IDENTIFICATION OF NOVEL ATP AND INTERLEUKIN-10 SOURCES IN THE SPINAL CORD



Eryn Louise Werry

A thesis submitted to the University of Sydney in fulfilment of the requirements for the degree of Doctor of Philosophy

(November, 2008)

DECLARATION

This thesis is submitted to the University of Sydney in fulfilment of the requirements for the degree of Doctor of Philosophy. The work presented in this thesis was undertaken by myself in the Bosch Institute and Brain and Mind Research Institute, University of Sydney apart from the electrophysiology experiments in section 3.3.1 and 3.3.4 which were carried out by Dr Guo Jun Liu. To the best of my knowledge the work presented in this thesis has not been published by another person, except where due reference is made in the text of the thesis. Parts of this thesis have been published in abstract form and in peer-reviewed journal form and they are listed on page vii.

Eryn L. Werry November 2008

ACKNOWLEDGEMENTS

Without the help of a number of people, this thesis would not have been possible. Firstly, thanks must be given to the financial bodies which funded this work. This work was supported by a NSW Spinal Cord Project Grant as well as an Australian Federation of University Women Fellowship. Personal financial support was provided by an Australian Postgraduate Award and a Major Patrick Hore-Ruthven Foundation Scholarship.

I would like to thank Professor Max Bennett for teaching me about neuroscience and what is involved in being a scientist. Your passion for neuroscience is infective and I have enjoyed discussing ideas and results with you over the course of my degree.

I am indebted to Dr Guo Jun Liu. I can't imagine how anyone could fulfil the role of supervisor in a manner superior to your efforts. You frequently put others needs above your own and were always available to answer questions and give assistance, frequently going beyond the call of duty. You are extremely skilled as an experimentalist and have a brilliant critical mind and I have learnt many skills in these areas through working with you. Furthermore, I am glad to have you as a friend and I look forward to staying in touch.

I am grateful for meeting the many members of the Neurobiology Laboratory during my candidature. The group made sure there was rarely a dull moment. In particular I would like to thank Vlado Blujan, Rajini Nagarajah, Lorraine Kerr and William Lin for the many entertaining chats about life and everything related. Furthermore, I would like to thank Rajini Nagarajah for teaching me RT-PCR. I would also like to thank Peter Tan and Roger Dampney for donation of a mouse anti-rat NeuN antibody. Additionally, the staff at Laboratory Animal Services, University of Sydney must be thanked for their assistance in animal maintenance.

My family has given me wonderful opportunities to be able to try my best at everything I do. Many times they have put giving me these opportunities over their own needs and I am eternally grateful for the sacrifices they have made. Their support throughout my life is very much appreciated. Thanks also to my dogs Jill and Bella who have kept me company whilst I spent the last few months sitting on my floor and typing.

My friends are invaluable. They keep me laughing, chip off my rough edges and expand my mind. It was priceless during my candidature to have friends I could spend time with, leaving behind the world of science. I would especially like to thank Nicole Predl and Kalyani Kaneshalingam for their company and friendship.

Finally, I would like to thank my Lord and Saviour for my purpose and hope in life. Nothing makes more sense to me than putting You first and everything seems pointless in comparison to this. The writer of Ecclesiastes expresses this better than me:

"Of making many books there is no end and much study wearies the body. Now all has been heard here is the conclusion of the matter: Fear God and keep his commandments, for this is the whole duty of man."

Ecclesiastes 12:12b-13.

SUMMARY

Chronic pain has a large individual and societal impact. Although effective treatments for acute pain exist, a specific and effective treatment for chronic pain is still to be discovered. Recent use of genetic techniques has revealed that spinal ATP plays a role in the generation of allodynia and hyperalgesia, two symptoms of chronic pain. Furthermore, it has been shown that the symptoms of chronic pain can be alleviated by increasing spinal levels of the anti-inflammatory cytokine interleukin-10 (IL-10). These observations suggest that decreasing the spinal release of ATP and increasing the spinal release of IL-10 may be novel approaches to chronic pain treatment. Yet knowledge of the endogenous sources of spinal ATP and IL-10 is limited. Therefore, the aim of this thesis was to identify novel sources of ATP and IL-10 in the spinal cord.

The first series of experiments (Chapter 3) investigated whether spinal cord astrocytes could be a source of ATP in response to the nociceptive neurotransmitters glutamate and substance P. Glutamate stimulated ATP release from cultured spinal cord astrocytes and this release was greatly potentiated by substance P, even though substance P alone did not elicit ATP release. Substance P also potentiated glutamate-induced inward currents without alone causing such currents. Glutamate, when applied on its own, acted exclusively through AMPA receptors to stimulate Ca²⁺ influx-dependent ATP release. However when substance P was co-applied with glutamate, ATP release could be elicited by activation of NMDA and metabotropic glutamate receptors. ATP release resulting from AMPA and kainate receptor stimulation was not affected by substance P. Activation of neurokinin receptor

subtypes, protein kinase C and phospholipase A_2 , C and D were required for substance P to bring about its effects.

The second series of experiments (Chapter 4) explored whether spinal cord astrocytes could be a source of IL-10 in response to glutamate and stimulation of the Toll-like receptor 4 (TLR4), which is known to occur in chronic pain. TLR4 stimulation increased IL-10 release from cultured spinal cord astrocytes. TLR4-stimulated IL-10 release was enhanced in the presence of glutamate. Glutamate potentiated TLR4-stimulated IL-10 release by upregulating IL-10 mRNA and worked synergistically through metabotropic glutamate receptor groups I, II and possibly III.

The third series of experiments (Chapter 5) examined whether spinal cord microglia could be a source of IL-10 in response to glutamate and stimulation of the TLR4. TLR4 stimulation enhanced both IL-10 transcription and translation, resulting in an increase in IL-10 release from cultured spinal cord microglia. Glutamate significantly increased TLR4-stimulated IL-10 release from microglia by binding NMDA, AMPA, metabotropic glutamate receptors and possibly kainate receptors and enhancing TLR4-induced IL-10 mRNA expression.

The results of this thesis suggest that astrocytes may be a major source of ATP in the spinal cord on activation of nerve fibres that co-release substance P and glutamate. The results of this thesis also suggest that when glutamate and TLR4 agonists are released into the spinal cord after noxious stimulation, astrocytes and microglia may be a major source of IL-10. These results implicate spinal cord glia in both pronociceptive and anti-nociceptive responses to pain-related substances.

CONTENTS

Publications		viii
List of Figures		X
List of Tables		xii
List of Abbreviations		xiii
Chapter 1	Literature review and aims	1
Chapter 2	Methods and materials	87
Chapter 3	Glutamate-stimulated ATP release from spinal cord astrocytes is potentiated by substance P	120
Chapter 4	Release of IL-10 from spinal cord astrocytes after activation of the TLR4 and the effect of concomitant mGluR activation	156
Chapter 5	Spinal microglial IL-10 release on activation of the TLR4 and the effect of concomitant glutamate receptor activation	on 190
Chapter 6	General discussion	220
Chapter 7	References	242
Chapter 8	Appendix	283

PUBLICATIONS

Refereed journal articles related to this thesis

Werry EL, Liu GJ and Bennett MR (2006) Glutamate-stimulated ATP release from spinal cord astrocytes is potentiated by substance P. *Journal of Neurochemistry* 99:924-36.

Werry EL, Liu GJ, Nagarajah R and Bennett MR (2008) Spinal cord astrocyte interleukin-10 release that occurs after activation of Toll-like receptor 4 is potentiated by glutamate *Submitted*.

Werry EL, Liu GJ, Nagarajah R, Lovelace MD, Hickie IB and Bennett MR (2008) Microglial interleukin-10 release on activation of Toll-like receptor 4 is potentiated by glutamate *Submitted*.

Other refereed journal articles

Liu GJ, Kalous A, **Werry EL** and Bennett MR (2006) Purine release from spinal cord microglia after elevation of calcium by glutamate *Molecular Pharmacology* 70:851-59.

Liu GJ, Werry EL and Bennett MR (2005) Secretion of ATP from Schwann cells in response to uridine triphosphate *European Journal of Neuroscience* 21:151-60.

Abdipronoto A, Liu GJ, Werry EL and Bennett MR (2003) Mechanisms of secretion of ATP from cortical astrocytes triggered by uridine triphosphate *Neuroreport* 14:2177-81.

Conference proceedings

Werry EL, Liu GJ and Bennett MR (2004) Substance P modulates glutamatestimulated adenosine triphosphate release from spinal cord astrocytes *Proceedings of the Australian Health and Medical Research Congress* 2:119.

Werry EL, Liu GJ and Bennett MR (2005) Substance P allows glutamate to stimulate ATP release through NMDA and metabotropic glutamate receptors on spinal cord astrocytes *Proceedings of the Australian Neuroscience Society* 16:42.

Werry EL, Liu GJ and Bennett MR (2005) Glutamate-stimulated ATP release from spinal cord astrocytes is potentiated by substance P Journal of Neurochemistry 94:101. Werry EL, Liu GJ and Bennett MR (2006) Substance P enhances glutamatestimulated ATP release from cultured rat spinal cord astrocytes *Society for Neuroscience* 733.10/H14.

Werry EL, Liu GJ and Bennett MR (2007) Glutamate enhances antiinflammatory interleukin-10 release from spinal cord microglia and astrocytes after Toll-like receptor 4 stimulation *Proceedings of the World Congress of Neuroscience* 7:163.

Werry EL, Liu GJ and Bennett MR (2007) Interleukin-10 release from spinal cord microglia and astrocytes after Toll-like receptor 4 stimulation is enhanced by glutamate *Proceedings of the 25th International Australasian Winter Conference on Brain Research* 25:In press.

LIST OF FIGURES

Chapter 1

Figure 1.1	Neural pathways involved in acute pain	4
Figure 1.2	The morphology of microglia	10
Figure 1.3	The morphology of astrocytes	11
Figure 1.4	ATP and its receptors	16
Figure 1.5	Pathways of involvement of spinal ATP in chronic pain	20
Figure 1.6	Structure of the IL-10 dimer and IL-10 receptor	28
Figure 1.7	The anti-nociceptive actions of interleukin-10 (IL-10)	34
Figure 1.8	Common intracellular signaling pathways linked to metabotro	pic
	receptors 4	1-42
Figure 1.9	The structure of metabotropic and ionotropic glutamate recep	tors
Figure 1 10	Mechanisms of glutamate-stimulated ATP release from cortic	al
Figure 1.10	astrocytes	55
Figure 1.11	Structure of substance P and its receptors	59
Figure 1.12	The structure of lipopolysaccharide (LPS) and its receptor	69
Figure 1.13	TLR4 signaling pathways in microglia and astrocytes	79
G1 1 0		
Chapter 2		
Figure 2.1	ATP standard curve	97
Figure 2.2	IL-10 standard curve	108
0	and the second se	
Chapter 3		
Figure 3.1	SP potentiates glutamate-stimulated ATP release from culture	d
- 8	spinal cord astrocytes but not cortical astrocytes	130
Figure 3.2	Glutamate-stimulated ATP release and the potentiation of this	by
U	SP is concentration-dependent	131
Figure 3.3	SP acts on neurokinin receptor subtypes to potentiate glutama	te-
	stimulated ATP release	133
Figure 3.4	Glutamate stimulates ATP release by activating AMPA recep	137
D' 35	European of AMBA NMDA and mGluBs in spinal cord	157
Figure 3.5	expression of AMPA, NMDA and months in spinar cord	138
Figure 3.6	SP allows glutamate to stimulate ATP release through NMDA	A and
Figure 5.0	mGluRs	139
Figure 3.7	The modulation of glutamate-stimulated ATP release by SP	
	requires Ca2+ influx and activation of multiple intracellular	
	signaling pathways	144
Figure 3.8	A model of the mechanisms of glutamate-stimulated ATP rel	ease
1992	from spinal cord astrocytes and its potentiation by SP	154

x

Chapter 4

Figure 4.1	The TLR4 is expressed on spinal cord astrocytes	167
Figure 4.2	LPS-stimulates IL-10 release from spinal cord astrocytes and thi	is
B	release is increased in the presence of glutamate	168
Figure 4.3	The modulation of LPS-stimulated IL-10 release by glutamate is	5
	dose-dependent	171
Figure 4.4	The modulation of LPS-stimulated IL-10 release by glutamate is	S
- Bart I	time-dependent	173
Figure 4.5	Transcription is involved in the potentiating effect of glutamate	
i igure ne	1	176
Figure 4.6	Glutamate potentiates LPS-stimulated IL-10 release by co-	
I igure no	activating mGluR groups	179
Figure 4.7	A model of the mechanisms of LPS-stimulated IL-10 release in	the
- guite in	absence and presence of glutamate from spinal cord astrocytes	188
Chapter 5		
Figure 5.1	The TLR4 is expressed on spinal cord microglia	200
Figure 5.2	Glutamate significantly potentiates LPS-stimulated IL-10	
U	release from spinal cord microglia	201
Figure 5.3	The modulation of LPS-stimulated IL-10 release by glutamate	
U	is dose-dependent	204
Figure 5.4	LPS-stimulated IL-10 release and the potentiating effect of	
0	glutamate is time-dependent	206
Figure 5.5	The potentiating effect of glutamate involves transcription	208
Figure 5.6	Glutamate potentiates LPS-stimulated IL-10 release by activating range of glutamate receptor subtypes	ng a 212
Figure 5.7	A model of the mechanisms of LPS-stimulated IL-10 release in	the
I iguite out	absence and presence of glutamate from spinal cord microglia	218
Chapter 6		
Figure 6.1	A three stage hypothesis of spinal processing after noxious	

igure 6.1A three stage hypothesis of spinal processing after noxious
stimulation in chronic pain228-9

LIST OF TABLES

Chapter 1

 Table 1.1
 Nomenclature of glutamate receptor subtypes

48

ABBREVIATIONS

AC,	Adenylyl cyclase
ADP	adenosine diphosphate
AMPA	α -amino-3-hydroxy-5-methyl-4-isoxazoleproprionate
AP-1	activation protein-1
AP5	D(-)-2-Amino-5-phosphonopentanoic acid
APDC	(2R,4R)-4-Aminopyrrolidine-2,4-dicarboxylate
ARL 67156	6-N,N-Diethyl-β-γ-dibromomethylene-D-adenosine-5- triphosphate
ATP	adenosine triphosphate
BDNF	brain-derived neurotrophic factor
BSA	bovine serum albumin
cAMP	cyclic adenosine monophosphate
CD14	cluster determinant 14
cDNA	complementary deoxyribonucleic acid
c/EBP	CCAAT/enhancer binding protein
CNQX	6-cyano-7-nitroquinoxaline-2,3-dione
CNS	central nervous system
CPCCOEt,	7-Hydroxyiminocyclopropan[b]chromen-1α-carboxylic acid ethyl
	ester
DAG	diacyl glycerol
DAPI	4',6-diamidino-2-phenylindole
DDA	2'5'dideoxyadenosine
DHPG	(S)-3,5-Dihydroxyphenylglycine hydrate
DMEM	Dulbecco's modified Eagle's medium
DMEM+	supplemented Dulbecco's modified Eagle's medium
dNTP	2'-deoxynucleoside 5'-triphosphate
dsRNA	double stranded ribonucleic acid
EGLU	(2S)-α-Ethylglutamic acid
EGTA	ethylene glycol-bis(2-aminoethylether)-N,N,N',N'-tetraacetic acid
ELISA	enzyme-linked immunosorbent assay
ERK	extracellular signal-related kinase
FAD ²⁺	flavin adenine dinucleotide
GABA	γ-aminobutyric acid
GAMS	γ-D-glutamylamino methylsulfonic acid
GFAP	glial fibrillary acidic protein
GLAST	glutamate and aspartate transporter
GLT	glutamate transporter
Glu	glutamate
GYKI52466	1-(4-Aminophenyl)-4-methyl-7,8-methylenedioxy-5H-2,3-
	benzodiazepine hydrochloride
HBSS	Hank's balanced salt solution
IkB	inhibitor of nuclear factor-kB
IKK	inhibitor of nuclear factor-kB kinase
IL-6	interleukin-6

IL-10	interleukin-10
IL-10R	interleukin-10 receptor
IL-1B	interleukin-1β
IP ₃	inositol triphosphate
IRAK	IL-1 receptor associated kinase
IRF3	interferon regulatory factor-3
ISRE	interferon-stimulated response element
JAK1	Janus Kinase 1
JNK	JUN N-terminal kinase
KA	kainate
L703606	cis-2-(Diphenylmethyl)-N-[(2iodophenyl)methyl]-
	lazabicyclo[2.2.2]octan-3-amine oxalate salt
L-AP4	L-(+)-2-Amino-4-phosphonobutyric acid
LBP	lipopolysaccharide binding protein
LDH	lactate dehydrogenase
LPS	lipopolysaccharide
MAL	MyD88-adaptor-like
MAPK	mitogen activated protein kinase
MCPG	(+)□-α-methyl-4-carboxyphenylglycine
MD2	myeloid differentiation Ag
mGlu	metabotropic glutamate
mGluR	metabotropic glutamate receptor
MK-801	(+)-MK-801 hydrogen maleate
MKK	mitogen-activated protein kinase kinase
MPEP	6-Methyl-2-(phenylethynyl)pyridine
MSPG	(\pm) - α -Methyl-(4-sulfonophenyl)glycine
MyD88	myeloid differention 88
NAD^+	nicotinamide adenine dinucleotide
NBMPR	S-(4-Nitrobenzyl)-6-thioinosine
NBQX	2,3-dihydroxy-6-nitro-7-sulphamoyl-benzo(1)quinoxaline
NF-kB	nuclear factor-kB
NK1	neurokinin-1
NK2	neurokinin-2
NK3	neurokinin-3
NMDA	N-methyl-D-aspartate
PBS	phosphate buffered saline
PIP ₂	phosphatidylinositol-4,5-bisphosphate
PKC	protein kinase C
PLA ₂	phospholipase A ₂
PLC	phospholipase C
PLD	phospholipase D
REST	relative expression software tool
R-LPS	rough lipopolysaccharide
RT-PCR	reverse transcription-polymerase chain reaction
siRNA	small interfering ribonucleic acid
S-LPS	smooth lipopolysaccharide

STAT	Signal Transducer and Activator of Transcription
Sp1	Simian virus 40 promoter factor 1
SP	substance P
tACPD	trans-(1S,3R)-1-amino-1,3-cyclopentanedicarboxylic acid
TAB2	transforming growth factor-β-activated kinase 1-binding protein 2
TAK1	transforming growth factor-β-activated kinase 1
TBOA	threo-β-hydroxyaspartic acid
TIR	Toll/interleukin-1 receptor
TLR	Toll-like receptor
TLR4	Toll-like receptor 4
TMB	tetramethylbenzidine
TNF-α	tumor necrosis factor-α
TRAF	tumor-necrosis-factor-receptor-associated factor
TRAM -1	TIR-containing adaptor molecule-1
TRPV1	transient receptor potential channel, vanilloid receptor 1
t-SNARE	target-soluble N-ethylmaleimide-sensitive factor attachment receptor
Tyk2	tyrosine kinase 2
UBC13	ubiquitin-conjugating enzyme 13
UEV1A	ubiquitin-conjugating enzyme E2 variant 1;
UTP	uridine triphosphate;
v-SNARE	vesicle-soluble N-ethylmaleimide-sensitive factor attachment receptor

Chapter 1

CHAPTER 1

LITERATURE REVIEW AND AIMS

Contents

1.1 AN INTRODUCTION TO THE CONCEPT OF PAIN	4
1.1.1 Acute pain	2
1.1.2 Chronic pain	3
1.2. A BRIEF INTRODUCTION TO THE CELL TYPES IMPLICATED IN CHRONIC	
PAIN	6
121 Neurons	6
1.2.1 Neurons	7
1.2. MOLECIUES IMPLICATED IN CHRONIC PAIN	12
1.3 MOLECULES INTERCATED IN CONCOLUCITATION	12
1.3.1 ATF	12
1.3.1.1 An introduction to A11	15
1.3.1.2 The role of the F2X-receptor in inflammatory and neuropathic hyperalgesia	18
1.3.1.5 The role of the P2X-receptor in inflammatory and neuropathic hyperalgesia	22
1.3.1.4 The fold of the 12A3 receptor in minimum of y and the provide of the second se	24
1.3.2 Intercurve of pro-inflammatory cytokines in chronic Dain	24
1.3.2.1 The fole of pro-initiality systems in comparison pairs	26
1.3.2.2 An introduction to the to the list and the state of $H_{-1}(0)$ in chronic pain states	30
1.3.2.3 Stutes unless implicated in chronic pain	32
1.4 POSSIBLE SOURCES OF ATP AND IL-10 IN THE SPINAL CORD	36
1.4 1 Describe sources of ATP in the sninal cord in response to pain-related substances	36
1.4.1 Possible sources of A11 in the spinal cord	36
1.4.1.1 Documented sources of ATP release from cortical astrocytes	38
1.4.1.2 Gutamate stimulated ATP release from	
1.4.1.2.1 Acceptors and increasing participation of the second se	39
1 4 1 2 2 Secretion mechanisms involved in glutamate-mediated ATP release from cortical astrocytes	49
1.4.1.2.3 Glutamate untake and ATP degradation: the final steps in glutamate-stimulated ATP release	
from cortical astrocytes	52
1.4.1.3 Potential of spinal cord astrocytes to release ATP in response to glutamate	54
1.4.1.4 The potential for SP to be involved in ATP release from spinal cord astrocytes	57
1.4.2 Possible sources of IL-10 in the spinal cord in response to pain-related substances	62
1.4.2.1 Documented sources of IL-10 in the brain and evaluation of candidates for sources of the spinal	
pool of IL-10 in chronic pain	62
1.4.2.2 TLR4-activated IL-10 release from brain microglia and astrocytes	60
1.4.2.2.1 The Toll-like receptor 4 and its agonists	0
1.4.2.2.2 Recognition of LPS by the Toll-like receptor 4 and subsequent signaling mechanisms	7
1.4.2.2.3 Mechanisms of IL-10 secretion	7
1.4.2.2.4 Removal of LPS and degredation of IL-10	0
1.4.2.3 Potential of spinal cord astrocytes and microglia to release IL-10 on ILR4 activation	0
1.4.2.4 The potential of glutamate to be involved in IL-10 release from spinal cord microgita and	8
astrocytes	8/
1.5 AIMS AND OBJECTIVES	01

1.1 AN INTRODUCTION TO THE CONCEPT OF PAIN

1.1.1 Acute pain

The International Association for the Study of Pain (1994) defines pain as "an unpleasant sensory and emotional experience associated with actual or potential tissue damage, or described in terms of such damage". Pain can be acute where the area of sensation is the same as the area of noxious stimulation. Acute pain is reversible, that is, it goes away (Borsook, 2003; Holden and Pizzi, 2003). Acute pain is elicited by noxious stimuli such as mechanical force, thermal energy or The receptors that respond to these stimuli are called chemicals like acid. nociceptors and they are found in the skin, muscles, joint capsules and a range of internal organs such as the heart and gut. Nociceptors are the peripheral endings of Aδ and C fibres, which are intermediate (12-30 m/s) and slow (0.5-2 m/s) conducting peripheral afferent fibres respectively. When these nociceptors are activated to a degree whereby their fibres are brought to threshold, action potentials travel down Aδ and C fibres and synapse in the spinal cord dorsal horn. The cell body-containing grey matter of the spinal cord is divided up into 10 layers or laminae (I – X). C fibres synapse in laminae I, II, VI and X whilst A δ fibres synapse mostly in laminae I, II and to a lesser extent IV. At these synapses, through release of neurotransmitters such as substance P (SP) and glutamate, action potentials are elicited in post-synaptic dorsal horn neurons (for a review see Millan (1999)). Some of these dorsal horn neurons are interneurons, which synapse on ventral horn neurons. The result of this is the propagation of action potentials toward muscles in the originally stimulated region and consequent reflex

withdrawal from the noxious stimulus. Other dorsal horn neurons send axons up ascending tracts to the limbic system, thalamus and areas of the cortex, such as the somatosensory cortex, resulting in perceptions of pain (Andersen et al., 2004; Holden and Pizzi, 2003; M illan, 1999). These pathways are summarized in Figure 1.1. Acute pain is useful, providing a warning of impending damage to tissues (Millan, 1999).

1.1.2 Chronic pain

Chronic pain is pain which persists beyond the normal time of healing and which is present everyday for three or more months (Access Economics Pty Ltd, 2007). Allodynia is pain perceived on application of a normally non-painful stimulus, such as pain felt when clothes brush on the skin (tactile allodynia), and hyperalgesia is a heightened response to a noxious stimulus (Merskey and Bogduk, 1994). Primary hyperalgesia occurs in the area of initial injury, whilst secondary hyperalgesia occurs in tissue surrounding the initial injury. Pain can also be spontaneous both in the area of injury and the surrounding tissues. Allodynia, hyperalgesia and spontaneous pain are the three key symptoms of chronic pain, although secondary hyperalgesia and allodynia are seen in severe acute pain and are classed as chronic pain symptoms when they persist longer than would be expected for the acute injury to heal (Holden and Pizzi, 2003). The enduring pain is referred to as neuropathic pain if it results from nerve damage and inflammatory pain if it results from inflammatory mediators. Conditions involving chronic pain include arthritis, chronic migraine and peripheral nerve damage (Borsook, 2003).

3



Figure 1.1 Neural pathways involved in acute pain. 1) A nociceptive stimulus activates nociceptors which project action potentials down A^{δ} or C peripheral afferent fibres into the dorsal horn of the spinal cord. 2) These neurons synapse on an interneuron then a motor neuron, leading to reflex withdrawal. 3) Peripheral afferent fibres also synapse on dorsal horn neurons which project to higher areas of the central nervous system such as the thalamus and limbic system (4) and the somatosensory cortex (5). Adapted from McAlexander (2007).

Chapter 1

Chapter 1

Chronic pain is a widespread condition with an estimated 3.2 million Australians having experienced it in 2007 (Access Economics Pty Ltd, 2007). In contrast to acute pain, chronic pain interferes with the everyday life and wellbeing of sufferers, with many sufferers experiencing reduced work performance and employment outcomes (Access Economics Pty Ltd, 2007). Additionally, a high correlation with psychopathology such as depressive disorders has been reported, although the nature of the relationship between chronic pain and psychopathology is yet to be fully elucidated (Dersh et al., 2002). Given the individual effects of chronic pain, it has a large economic impact on Australian society. In 2007, chronic pain cost Australians an estimated \$34.3 billion (Access Economics Pty Ltd, 2007).

The large individual and societal impact of chronic pain means that an effective treatment is much sought after. Although treatments such as non-steroidal antiinflammatories, opiods and anaesthetics allow effective management of acute pain, these treatments do not effectively diminish chronic pain (Nitu et al., 2003). Moreover, a specific and effective treatment for chronic pain is yet to be found. A deeper understanding of the cellular and molecular mechanisms of chronic pain is needed to develop treatments that specifically target the dysfunctioning pathways.

1.2 A BRIEF INTRODUCTION TO THE CELL TYPES IMPLICATED IN CHRONIC PAIN

Many cell types are present in the central nervous system (CNS). The most numerous group of cells are glial cells, which is a cell group comprised of microglia, astrocytes, oligodendrocytes and ependymoglia. Glial cells outnumber the other major class of cell, neurons, ten to one (Stevens, 2003). In addition to glia and neurons, the CNS also contains endothelial cells, which line CNS blood vessels, and cells such as leptomeninges, which comprise the pia mater, and arachnoid lining, which covers the CNS. Of all these cell types, evidence is strongest for the involvement of neurons, microglia and astrocytes in chronic pain.

1.2.1 Neurons

An increased activity of nociceptive peripheral afferent and dorsal horn neurons has been suggested as the causative factor behind hyperalgesia and allodynia. Supporting this hypothesis are observations that nociceptive peripheral afferent neurons can develop spontaneous action potential firing (Liu et al., 2000) and become hyperexcitable (Campbell and Meyer, 2006; Millan, 1999) when injured or inflamed. In part, this is thought to be the cellular basis of hyperalgesia. Additionally, in such circumstances neurons that would normally fire action potentials to non-noxious tactile stimuli start to express mRNA for transmitters released by pain neurons (Neumann et al., 1996; Woolf and Salter, 2000). This phenomenon is thought to underlie allodynia. Recently, evidence has emerged that the hyperalgesia and allodynia which characterise chronic pain may not be due solely to dysfunction of neurons but that glial cells, such as microglia and astrocytes, may contribute to the aforementioned maladaptive neuronal changes.

1.2.2 Microglia and astrocytes

Microglia comprise 5–10 % of the total amount of glial cells in the CNS (Vaughan and Peters, 1974). They are the immune surveillance cells of the CNS, recognising foreign, infected or dead cells and proteins. When such cells and proteins are detected, microglia become "activated" and destroy and phagocytose the foreign, infected or dead material. This function restricts the spread of disease and infection in the CNS (Streit, 1995). Due to this T-cell-like responsibility of microglia, their morphology varies with their functional state as demonstrated in Figure 1.2A. In a non-pathological environment, microglia are described as "resting" and have thin, spiny processes radiating from a central compartment. "Activated" microglia, which have been exposed to pathological tissue but have not become phagocytic, have thicker processes than resting microglia. Phagocytic microglia are spherical (Streit, 1995). When activated, microglia exhibit an upregulation of cell surface markers such as OX-42 and endothelin-1 (Taylor et al., 2003; Tsuda et al., 2003).

Astrocytes comprise the majority $(60-65 \ \%)$ of glial cells in the CNS (Peinado et al., 1998). Astrocytes have processes that surround both blood vessels and synapses (Peters et al., 1991; Ventura and Harris, 1999) and were originally thought to function only as homeostatic regulators, uptaking neurotransmitter, H⁺ and K⁺ ions to generate perfect conditions for neuronal action potential propagation

(Ballanyani, 1995; Deitmer, 1995). Over the last ten years, evidence has accumulated that astrocytes can also release transmitters, termed gliotransmitters, which can actively modulate neuronal transmission (Koizumi et al., 2003), can elicit waves of intracellular calcium increases in surrounding astrocytes (Cornell-Bell et al., 1990; Koizumi et al., 2002) and can cause release of substances from microglia (Bianco et al., 2005). Figure 1.3A shows that two basic types of astrocytes can be observed in vivo, based on morphological characteristics. Protoplasmic astrocytes are found in grey matter and have thin processes with extensive process branching, whilst fibrillary astrocytes are found in white matter and have thicker, less branched processes and a stellate shape (Duffy, 1983). Categorisation into protoplasmic and fibrillary astrocytes may be too simplistic since white matter astrocytes that resemble protoplasmic astrocytes have been described, as have astrocytes that have features of both protoplasmic and fibrillary types (Levison and Goldman, 1993). Similarly, in vitro, there are several types of astrocytes. For example, in 10-day-old spinal cord astrocyte cultures, there are three major types of astrocyte present - type A, which are flat, polygonal astrocytes; type G, which have a small cell body with lots of small processes; and type H, which are flat, disk shaped astrocytes (Black et al., 1993). These can be seen in Figure 1.3C.

The role of glia in chronic pain is evidenced with careful use of fluorocitrate. Fluorocitrate acts in the Krebs cycle to decrease the activity of aconitase, preventing the conversion of citrate to isocitrate. As neurons do not use extracellular citrate in the Krebs cycle, the direct effect of fluorocitrate is to selectively and reversibly inhibit glial metabolism, effectively inactivating glia. The need for care in use comes at high doses and on prolonged exposure to fluorocitrate where it leads to affect neuronal health which indirectly irreversible glial changes (Milligan et al., 2003; Paulsen et al., 1987; Wieseler-Frank et al., 2005). Hyperalgesia is blocked when fluorocitrate is added at an appropriate dose and incubation time to selectively effect glia prior to the induction of inflammatory pain (Meller et al., 1994; Watkins et al., 1997). Also, when minocycline, a microglial activation inhibitor, is applied before the induction of neuropathic and inflammatory decreased is hyperalgesia allodynia and the degree of pain, (Ledeboer et al., 2005; Raghavendra et al., 2003).

It should be noted that in the majority of experiments, application of glial inactivators does not completely stop pain. Given the wealth of evidence for the involvement of neuronal dysfunction in hyperalgesia and allodynia, it is obvious that a variety of spinal cord cell types interact to bring about hyperalgesia and allodynia and that one type of cell cannot be completely responsible for the changes. It is not completely understood how glia and neurons interact to result in the changes seen in chronic pain. However, part of this process has recently been revealed with emerging evidence of some key molecular pathways involved in both the genesis and alleviation of chronic pain.



Figure 1.2 The morphology of microglia . A) In situ morphology of i) resting, ii) activated and iii) phagocytic cortical microglia stained by tomato lectin immunohistochemistry (see section 2.5 for details on immunohistochemistry). From Basu et al (2005). No scale was reported. B) In vitro morphology of resting and lipopolysaccharide (LPS) -activated spinal cord microglia as viewed by i) light microscopy and ii) immunohistochemical staining for cell marker isolectin B_4 . The scale bar represents 20^{μ} m. Adapted from Liu et al (2006).



Chapter 1

Literature review and aims

Figure 1.3 The morphology of astrocytes. A) Morp hology of freshly isolated i) fibrous and ii) protoplasmic rat cortical astrocytes stained for glial fibrillary acidic protein (GFAP) by immunohistochemistry. The arrow in i) indicates the endfoot of an astrocytic process while the arrows in ii) indicate cell bodies (magnification is x1000). From Connor and Berkowitz (1985). B) *In vitro* morphology of rat spinal cord astrocytes as viewed by light microscopy. The arrow indicates a cultured astrocyte resembling an *in situ* fibrous astrocyte, while all other astrocytes resemble protoplasmic astrocytes. No scale was indicated. Taken from Rosewater and Sontheimer (1994)(Everett and Senogles, 2004; 1994). C) *In vitro* morphology of spinal cord astrocytes as viewed by immunocytochem ical staining for GFAP. Three types of cultured spinal cord astrocytes can be identified by immunocytochemical staining 10 days after initial culturing. These are type A (i), type G (ii) and type H (iii). x400. Taken from Black et al (1993).

Chapter 1

1.3 MOLECULES IMPLICATED IN CHRONIC PAIN

Recent use of genetic techniques has shed light on some of the key molecules which may be involved in both generating and alleviating the symptoms of chronic pain like hyperalgesia and allodynia. Techniques such as receptor knock-down and knock-out have revealed a role for adenosine triphosphate (ATP) in the pathogenesis of neuropathic allodynia, neuropathic and inflammatory hyperalgesia, and gene therapy have revealed a role for interleukin-10 (IL-10) in alleviating the symptoms of chronic pain.

1.3.1 ATP

1.3.1.1 An introduction to ATP

ATP is a ubiquitous molecule and as illustrated in Figure 1.4A it consists of a ribose sugar with the purine base adenine attached to the 1' ribose carbon (the adenosine portion) and three phosphate groups attached to the 5' ribose carbon (the triphosphate portion). ATP acts on a large family of ionotropic and metabotropic purinergic receptors. Ionotropic purinergic receptors are referred to as P2X receptors and they are formed from a combination of 3 subunits. When a receptor is comprised of three subunits, it is termed a trimer. Each subunit of the P2X trimer has an intracellular N- and C-terminus connected to 2 hydrophobic transmembrane domains which are linked by a large extracellular loop as shown in Figure 1.4B. There are seven different receptor subunit proteins termed P2X₁₋₇ found throughout the CNS (Burnstock, 2007b). Apart from P2X₆, all can form homotrimer receptors in which all the subunits of the receptor are identical, for example a

 $P2X_1$ homotrimer receptor is comprised of only $P2X_1$ subunits. Additionally, heteromeric forms of P2X receptors exist, where the receptor is comprised of several different subunits such as with $P2X_{2/3}$ receptors (Burnstock, 2007b). All P2X receptors are activated by ATP, leading to a flow of ions through the receptors. In addition to this when $P2X_7$ receptors are repeatedly activated, a pore forms allowing the flow of larger molecules such as a 629 Da nucleic acid stain (Surprenant et al., 1996).

Metabotropic receptors which respond to ATP are referred to as P2Y receptors and each receptor consists of just one subunit. Each subunit contains an extracellular Nterminus, seven transmembrane domains and an intracellular C-terminus as exemplified in Figure 1.4C. There are 8 types (P2Y₁, ₂, ₄, ₆, ₁₁, ₁₂, ₁₃, ₁₄) (Fields and Burnstock, 2006). ATP effectively activates P2Y₂ and P2Y₄, while metabolites of ATP are more effective activators of other P2Y receptors. The first step in ATP metabolism is the removal of a phosphate group resulting in the nucleotide adenosine diphosphate (ADP) which activates P2Y₁, ₁₂ and P2Y₁₃ receptors. Another nucleotide, uridine diphosphate, activates P2Y₆ and uridine diphosphate glucose activates P2Y₁₄ receptors (Burnstock, 2007b). P2Y receptors are coupled to intracellular signaling pathways, for example P2Y₂ and P2Y₄ receptors can activate phospholipase C (PLC)- β (see section 1.4.1.2.1 and Figure 1.8 for the PLC pathway).

Although within a cell ATP plays a key role in oxidative metabolism and without it cells would die, it is also released as an extracellular modulator by some cells. In the spinal cord, ATP has a number of roles as an extracellular modulator. The exercise pressor reflex is an increase in blood pressure and heart rate that occurs after static contraction of muscles. The first synapse in this reflex circuit is in the dorsal horn and ATP release from muscle afferent nerves at this synapse plays a key role in bringing about the exercise pressor reflex (Gao et al., 2005). ATP can also excite spinal motor neurons and elicit an initial increase then a delayed decrease in activity of the diaphragm-innervating phrenic motor nerves (Deng and Fyffe, 2004; Miles et al., 2002).

Additionally, ATP has been shown to excite neurons and glia in the first central pain synapse in the dorsal horn, suggesting a role for ATP in nociceptive pain transmission (Burnstock, 2007a; Fam et al., 2000). For example, ATP increases glutamate release from peripheral nociceptive afferents onto post-synaptic dorsal horn neurons, consequently increasing evoked excitatory post-synaptic currents in these dorsal horn neurons (Nakatsuka and Gu, 2001). Moreover, a role in not just acute pain transmission but chronic pain is suggested by studies showing application of 30 nmol ATP to the spinal cord elicits allodynia (Nakagawa et al., 2007). Additionally, one common method of administration of substances into the spinal cord is to inject them into the subarachnoid space. This is also referred to as an intrathecal injection. Intrathecal administration of a P2X agonist elicits hyperalgesia (Tsuda et al., 1999). Further evidence for the role of ATP in allodynia

and hyperalgesia has come with genetic and pharmacological targeting of the $P2X_4$, $P2X_7$ and $P2X_3$ subtypes of purinergic receptor.

1.3.1.2 The role of the P2X4 receptor in neuropathic allodynia

The binding of ATP to the P2X4 receptor on microglia appears to play a key role in the maintenance of allodynia after peripheral nerve injury. Allodynia can be completely reversed using an antagonist that blocks the P2X4 receptor if intrathecally administered 7 days after injury, whereas allodynia isn't relieved with a broad spectrum P2X antagonist that doesn't block the P2X4 receptor (Tsuda et al., Additionally, allodynia can be decreased by knockdown of the P2X4 2003). receptor through intraspinal administration of an antisense oligodeoxynucleotide (see Figure 1.5A). Antisense oligodeoxynucleotide is a small sequence of DNA complementary to a sequence of the P2X4 mRNA, which when bound to the P2X₄ mRNA results in the inability of the P2X₄ subunit protein to be synthesized from its mRNA. Furthermore, the P2X4 receptor is upregulated on dorsal horn microglia after injury but was not detected on neurons or astrocytes after injury (Tsuda et al., 2003). Although expression on oligodendrocytes was not examined, almost all P2X₄ staining was on OX-42 positive cells. As OX-42 is a cellular marker for microglia, this suggests that the site of action of ATP on P2X4 receptors is microglia. Further supporting this is the observation that injection of P2X₄stimulated microglia into the spinal cord results in allodynia (Tsuda et al., 2003).



Chapter

Literature review and aims



16

One mechanism whereby binding of ATP to microglial P2X₄ receptors maintains allodynia is through production of brain-derived neurotrophic factor (BDNF). ATP-stimulated microglia release BDNF, and BDNF application to the spinal cord causes the neurotransmitter γ -aminobutyric acid (GABA) to produce depolarizing currents in lamina I neurons. This is in stark contrast to the usual hyperpolarizing effect of GABA on these neurons. This change in direction of the effect of GABA is commonly seen after nerve injury and may contribute to the hyperexcitability of nociceptive neurons after injury. This change is also accompanied by allodynia (Coull et al., 2005). When P2X₄-stimulated microglia are injected into the spinal cord and microglia are subjected to small interfering RNA (siRNA) specific for BDNF which interferes with BDNF production, allodynia and the change in polarization of GABA-induced currents is reversed. Additionally, allodynia resulting from neuropathy and P2X₄-stimulated microglial injection is prevented by application of_an antibody to a receptor of BDNF (Coull et al., 2005).

The involvement of the $P2X_4$ receptor in chronic inflammatory pain and neuropathic hyperalgesia is unclear. Although one study has reported an increase in microglial expression of the $P2X_4$ receptor after formalin injection, one method of inducing inflammatory pain (Guo et al., 2005), results of other studies indicate the involvement of the $P2X_4$ receptor seems to be specific to neuropathic allodynia as no changes in $P2X_4$ expression are seen after injection of complete Freund's adjuvant, a diminished mycobacteria which induces inflammatory pain (Tsuda et al., 2003). Also, both inflammatory and neuropathic hyperalgesia can be

completely blocked by a $P2X_7$ gene deletion as discussed below (Chessell et al., 2005).

1.3.1.3 The role of the P2X₇ receptor in inflammatory and neuropathic hyperalgesia

It is unlikely that the P2X₄ receptor is the only purinergic receptor that ATP works on to contribute to chronic pain. Suramin is a poor antagonist at the P2X₄ receptor but a strong antagonist at most other P2X and P2Y receptors (excluding the P2X₆, P2Y₄ and P2Y₆ receptor) (Bo et al., 1995; Collo et al., 1996; Garcia-Guzman et al., 1997; Ralevic and Burnstock, 1998; Soto et al., 1996). Intrathecal application of suramin one day before induction of inflammatory hyperalgesia led to a decrease in hyperalgesia which lasted 4 weeks after cessation of suramin injection (Wu et al., 2004). Suramin is known to have inhibitory effects at non-purinergic receptors (Ralevic and Burnstock, 1998; Wu et al., 2004) and a portion of the observed inhibitory effect on hyperalgesia may be due to these non-purinergic actions, however a portion of the effects of suramin is likely due to effects on purinergic receptors since PPADS, an antagonist at all P2X receptors apart from P2X₄ and P2X₆ (Tsuda et al., 2003) had a similar effect on inflammatory hyperalgesia to suramin (Tsuda et al., 1999).

One non-P2X₄ microglial receptor that ATP may be working through in generating hyperalgesia is the P2X₇ receptor. Systemic and intraperitoneal application of specific P2X₇ receptor antagonists decreases the degree of tactile allodynia in

neuropathic pain (Honore et al., 2006; McGaraughty et al., 2007). Inflammatory and neuropathic hyperalgesia are compeletely removed in mice with a disrupted P2X7 receptor gene, as shown in Figure 1.5B (Chessell et al., 2005). Peripherally, the P2X7 receptor is found on cells such as monocytes, macrophages, lymphocytes, basophils, neutrophils, satellite cells and Schwann cells eosinophils, (Burnstock and Knight, 2004; Chessell et al., 2005). In the CNS, there is solid immunohistochemical evidence for the presence of microglial P2X7 receptors (Sim et al., 2004). The presence of functional P2X7 receptors on neurons and astrocytes, however, is controversial (Anderson and Nedergaard, 2006; Sperlagh et Antibodies used in previous immunohistochemical studies which al., 2006). identified P2X7 receptors on neurons and astrocytes have been shown to be nonspecific, positively staining these cells in P2X7 knockout mice (Sim et al., 2004). By and large, pharmacological efforts to demonstrate functional P2X7 receptors have been hampered by a lack of availability of specifc agonists and antagonists for the P2X7, as well as the potential of agonists to be metabolized to adenosine (Anderson and Nedergaard, 2006). These factors make it difficult to identify which cell types contribute to the effect of P2X7 gene disruption on allodynia and hyperalgesia and further work will be needed to accurately identify which cells have pain-related P2X7 receptors. Although it is likely that peripheral cells play a role (Martucci et al., 2007), there is evidence for at least a partial role for the P2X7 receptor on spinal microglia in the reported effects of gene deletion. After neuropathic hyperalgesia induction, PPADS administration in the spinal cord decreased levels of the chronic pain-inducing cytokine interleukin-1ß (IL-1ß; see





Figure 1.5 Pathways of involvement of spinal ATP in chronic pain. A) A mouse with allodynia will withdraw its paw when subjected to a smaller amount of mechanical force (measured in g) than it would when in a non-allodynic state. The amount of force required to cause paw withdrawal before nerve injury (BL) was greater than that required after nerve injury (MM). When mice were treated with anti-sense oligodeoxynucleotide to the P2X4 receptor (AS), nerve-injury induced allodynia was partially reversed. *p<0.05. Taken from Tsuda et al (2003). B) A mouse with hyperalgesia will remove its paw from a heat source faster than when in a non-hyperalgesic state. When mice were placed on a hotplate set at 50-55°C, male and female mice with functional P2X7 receptors (+/+) displayed hyperalgesia after an operation that induced neuropathic damage. Male and female mice without P2X7 receptors did not display this hyperalgesia when subjected to neuropathic damage. Initial recordings marked as post-operative day 0 were taken before neuropathy-inducing surgery. Taken from Coull et al (2005). C) Action potentials were electrically induced in A\delta fibres and the frequency of subsequent postsynaptic potentials was calculated. Responses of rats not subjected to further treatment (baseline) were set as 100%. A subpopulation of rats had either a vehicle liquid or a novel P2X3 antagonist, compound A, applied to the spinal cord and subsequent frequency of electrically induced action potentials were given as a ratio of baseline values. It can be seen compound A significantly reduced the activity of Aδ fibres. *p<0.05, **p<0.01, ***p<0.001. Taken from Sharp et al (2006). D) In the dorsal horn of the spinal cord, ATP from an unknown source can activate microglial P2X4 receptors resulting in release of brain-derived neurotrophic factor (BDNF) which can activate its receptor TrkB. ATP may also activate microglial P2X7 receptors (and possibly astrocytic and neuronal P2X7 receptors). Activation of microglial P2X7 receptors in conjunction with stimulation of microglial Toll-like receptor 4 (TLR4) results in interleukin-1ß (IL-1ß) release. ATP also activates neuronal P2X₃ receptors. The net result of this is increased nociceptive neuronal transmission. Pathways linked to neuropathic allodynia are indicated with red arrows and pathways linked to inflammatory and neuropathic hyperalgesia are indicated with green arrows.
section 1.3.2.1) and subsequently the degree of hyperalgesia displayed (Martucci et al., 2007). In neuropathic pain conditions, the Toll-like receptor 4 (TLR4, detailed in section 1.4.2.2.1) is activated (Tanga et al., 2005). To date, the only recorded spinal source of IL-1 β in response to ATP is by P2X₇ stimulation of microglia that have had their TLR4s activated (Chessell et al., 2005; Sanz and Di Virgilio, 2000). This suggests IL-1 β release from P2X₇-stimulated spinal microglia is important in the pathogenesis of hyperalgesia.

Recently, functional heterotrimeric $P2X_{4/7}$ receptors have been identified (Dubyak, 2007; Guo et al., 2007). It should be noted that ATP may not only bind to homotrimeric $P2X_4$ and $P2X_7$ receptors to bring about allodynia and hyperalgesia but also to these heteromeric receptors. It is yet to be determined if the evidence presented above results from ATP acting on the heterotrimeric or homotrimeric types of $P2X_4$ and $P2X_7$ receptors.

1.3.1.4 The role of the $P2X_3$ receptor in inflammatory and neuropathic hyperalgesia

A further P2X receptor subtype that may be involved in the pathogenesis of chronic pain is the P2X₃ receptor. The P2X₃ subtype is involved in the early stages of pain processing, being found both on peripheral and central (spinal cord) terminals of sensory neuronal afferents including nociceptive fibres (Chen et al., 1995; Jarvis et al., 2002; Lewis et al., 1995; Vulchanova et al., 1997). Additionally in the spinal cord, P2X₃ receptors are found in the dorsal commissure, which is a tract involved in a visceral pain pathway, and also in the lateral collateral pathway of Lissauer with P2X3 subunits found on cells projecting to visceral organs (Studeny et al., 2005). This subtype is also found in other areas of the CNS (Kukley et al., 2001; Seguela et al., 1996), however peripheral afferent and spinal cord P2X₃ receptors have been specifically linked to pain. Subcutaneous and intraplantar application of P2X3 receptor antagonists decrease inflammatory for peripheral $P2X_3$ receptors highlighting role hyperalgesia, a (McGaraughty et al., 2003; Wu et al., 2004). A clear role for spinal P2X₃ receptors in neuropathic hyperalgesia is seen in a study involving intrathecal administration of siRNA to the P2X₃ receptor. This resulted in a decreased P2X₃ receptor expression in the spinal cord dorsal horn and subsequently reduced levels of neuropathic hyperalgesia (Dorn et al., 2004). Similarly, spinal administration of P2X₃ receptor antagonists had specific spinal cord neuronal effects, namely decreased A&- and C-fibre-evoked responses after neuropathic injury (Sharp et al., 2006), as seen in Figure 1.5C. Since the P2X₃ subunit can form heterotrimers such as the P2X_{2/3} receptor and the antagonists used in the aforementioned studies block these heterotrimeric receptors, it is possible that both homotrimeric and heterotrimeric P2X3-containing receptors play a role in chronic pain (Sharp et al., 2006). The pathways of involvement of spinal ATP in chronic pain are summarised in Figure 1.5D.

1.3.2 Interleukin-10 (IL-10)

1.3.2.1 The role of pro-inflammatory cytokines in chronic pain

The body responds to injury or disease with a series of cellular and vascular changes that can limit damage. This is called the inflammatory response. Cytokines are a group of diverse, small, soluble proteins which act to mediate inflammatory responses (Benveniste, 1992). Cytokines can be divided into two groups, namely, pro-inflammatory cytokines and anti-inflammatory cytokines on the basis of their effect on the inflammatory response. Pro-inflammatory cytokines start and maintain an inflammatory response whereas anti-inflammatory cytokines decrease the response (Lin, 2003). Three such pro-inflammatory cytokines are tumor necrosis factor- α (TNF- α), IL-1 β and interleukin-6 (IL-6). In the CNS, these cytokines are released from a number of different cell types. The resident CNS immune cells microglia and astrocytes can release these cytokines. In response to substances involved in pain, both astrocytes and microglia are a source of TNF- α and IL-6 and microglia can also release biologically active IL-1ß (Chao et al., 1992; Chauvet et al., 2001; Grimaldi et al., 1994; Lee et al., 2000; Martin et al., 1993; Millan, 1999; Sanz and Di Virgilio, 2000; Sawada et al., 1999; Tanga et al., 2005). Macrophages may also be a source of pro-inflammatory cytokines in the CNS in painful situations. Macrophages are phagocytic cells which do not natively reside in the CNS. After a peripheral nervous system injury, blood-bourne macrophages can infiltrate into the spinal cord and differentiate into microglia (Zhang et al., 2007a). Although virtually all macrophages differentiate into microglia (Zhang et al., 2007a), there is a possibility that prior to differentiation these infiltrating macrophages can be a source of these cytokines in pain as they are known to release them (Benveniste, 1992). Finally, after peripheral inflammation, leptomeningeal cells which make up the meninges around the CNS, have been shown to release TNF- α (Wu et al., 2005).

The presence of spinal pro-inflammatory cytokines has been linked with chronic pain. Levels of TNF- α , IL-1 β and IL-6 are known to increase in the spinal cord after induction of hyperalgesia and allodynia (DeLeo et al., 1996; DeLeo et al., 1997; Sweitzer et al., 1999). These cytokines can increase the efficiency of transmission of neurotransmitters involved in spinal nociceptive processing such as SP, glutamate and ATP (Millan, 1999). Specifically, IL-1 β enhances the spinal release of SP (Malcangio et al., 1996), expression of the hyperalgesia-linked P2X₇ receptor (Narcisse et al., 2005) and increases N-methyl-D-aspartate (NMDA)mediated Ca²⁺-currents (Viviani et al., 2003). TNF- α can also make transmission via glutamate more efficient by increasing insertion of one of its receptors into neuronal membranes (Stellwagen et al., 2005). Additionally, these cytokines can further exacerbate pain states by increasing the release of each other, for example IL-1 β and TNF- α increase IL-6 release from astrocytes (Aloisi et al., 1992) and IL-6 increases the release of TNF- α from microglia (Wanaka et al., 1997).

Considering the effects of these cytokines on nociceptive neurotransmission, it is not surprising that the introduction of IL-1 β into the spinal cord is sufficient to produce hyperalgesia and allodynia (Reeve et al., 2000; Sung et al., 2004) and

introduction of IL-6 is sufficient to produce allodynia (DeLeo et al., 1996). Further, blocking spinal cytokine receptors with intrathecally administered antagonists during the induction of neuropathic hyperalgesia and allodynia (IL-1 β , TNF- α , IL-6) and inflammatory hyperalgesia (IL-1 β and not TNF- α) decreases the severity of these pain-related symptoms (Arruda et al., 2000; Milligan et al., 2003; Watkins et al., 1997). As such, one way to decrease the severity of chronic pain symptoms is to decrease the release of cytokines such as IL-1 β , IL-6 and TNF- α by glia in the spinal cord and to combat the deleterious effects of released spinal cytokines.

1.3.2.2 An introduction to IL-10 and its anti-inflammatory actions

Interleukin-10 (IL-10) is an anti-inflammatory cytokine, along with interleukin-4, interleukin-12, interleukin-13, interleukin-18 and interleukin-23 (Isomaki and Punnonen, 1997; Yamagata and Ichinose, 2006). IL-10 belongs to the helical structural group of cytokines, so named because the structure of cytokines in this class have right handed coils called α -helices collected in a bundle (Conklin, 2004). IL-10 is a helical cytokine which is encoded by a single gene and in the human, is made up of two polypeptide chains, each consisting of 160 amino acid residues. These chains are combined and intercalate in a dimer to produce a 36 kDa protein, as illustrated in Figure 1.6A and B (Donnelly et al., 1999; Zdanov, 2004). Rat IL-10 shares 73% of the human IL-10 amino acid sequence (Ball et al., 2001).

IL-10 exerts its cellular effects by binding the IL-10 receptor (IL-10R) which is found on cultured rat astrocyte and microglial cells in the CNS (Ledeboer et al., 2002). Additionally, although extensive studies identifying the presence of the IL-10R across a range of neuronal populations is yet to be carried out, the IL-10R has been identified on retinal ganglion cells where IL-10 increased the survival of retinal ganglion cells in culture (Boyd et al., 2003).

The IL-10R is a cell surface receptor consisting of 4 chains. The first two are highaffinity IL-10 binding helical α -chains that can recruit Janus Kinase 1 (JAK1) to the receptor complex. JAK1 is a type of protein tyrosine kinase, which are enzymes that can effect the function of a protein by phosphorylating it at a tyrosine residue. The second two chains in the IL-10R are accessory β -chains which can recruit tyrosine kinase 2 (Tyk2), another protein tyrosine kinase, to the receptor complex (Finbloom and Winestock, 1995; Ho et al., 1995; Moore et al., 2001). The first two chains are delineated IL-10R1 whilst the second two are commonly referred to as IL-10R2. Both chains are necessary for signal transduction (Kotenko et al., 1997). When IL-10 binds IL-10R1, a process demonstrated in Figure 1.6C, JAK1 and Tyk2 are phosphorylated. These proteins can then phosphorylate IL-10R1 which allow the transcription factors Signal Transducer and Activator of Transcription 1 and 3 (STAT1 and STAT3) to bind the receptor and be phosphorylated by JAK1 and Tyk2. In their phosphorylated form, the two STATs can translocate into the nucleus of cells and bind STAT binding elements of gene promoter sites, effecting the production of mRNA and protein



Figure 1.6 Structure of the IL -10 dimer and IL-10 receptor. A) Structure of one of the two polypeptide chains that compose the IL-10 dimer. The letters A to F indicate sections of the polypeptide chain that are classified as α -helices. Yellow segmen ts represent disulfide bonds. B) Structure of the IL -10 dimer showing how the two IL -10 polypeptide chains fit together. One subunit is in pink with the o ther in green. C) Representation of how the IL -10 dimer in green and pink interacts with the two IL-10R1 chains represented in red. N and C represent the amino and carboxy terminals of the indicated proteins respectively. Adapted from Zdanov (2004). Chapter

Chapter 1

from specific genes (Moore et al., 2001). In microglia, binding of the IL-10 receptor induces STAT1 but not STAT3 (Zocchia et al., 1997).

IL-10 was formerly known as cytokine synthesis inhibitory factor as typically the net outcome of its effect on mRNA and protein production is to decrease the release and action of pro-inflammatory cytokines. A TLR4 agonist induces the release of IL-6 and TNF-α from microglia and astrocytes (Benveniste et al., 1995; Heyen et al., 2000; Kremlev and Palmer, 2005; Ledeboer et al., 2000; Mizuno et al., 1994; Sawada et al., 1999) and additionally IL-1ß from microglia (Sawada et al., 1999) although this release can be quite inefficient compared to when a TLR4 agonist acts in conjuction with other transmitters (Ledeboer et al., 2002; Martin et al., 1993; Sanz and Di Virgilio, 2000; Sawada et al., 1999). The anti-inflammatory effect of IL-10 is exemplified by the finding that IL-10 greatly diminishes the amount of IL-1 β , IL-6 and TNF- α stimulated by the TLR4 agonist from microglia and astrocytes (Benveniste et al., 1995; Heyen et al., 2000; Kremlev and Palmer, 2005; Ledeboer et al., 2000; Mizuno et al., 1994; Sawada et al., 1999). As suggested by its links with STAT, there is evidence that IL-10 works at the level of transcription to produce these anti-inflammatory outcomes by decreasing the activation of transcription factors and the level of TLR4-stimulated pro-inflammatory cytokine mRNA (Heyen et al., 2000; Ledeboer et al., 2002; Zhou et al., 2007).

Furthermore, the anti-inflammatory action of IL-10 is not limited to decreasing the release of pro-inflammatory cytokines in inflammatory conditions, but it also acts to

Chapter 1

reduce the impact of pro-inflammatory cytokines that do get released. IL-10 exposure decreases the density of IL-6 receptors on microglia (Sawada et al., 1999) and decreases the ability of IL-1 β to activate the intracellular signaling pathways linked to its receptors in astrocytes (Pousset et al., 2000).

1.3.2.3 Studies demonstrating the effects of IL-10 in chronic pain states

Given that numerous spinal pro-inflammatory cytokines contribute to allodynia and hyperalgesia, and that IL-10 can decrease the release and impact of these cytokines in inflammatory conditions, it is not surprising that spinal introduction of IL-10 can reverse allodynia and hyperalgesia in chronic pain states. Initial studies of the effect of IL-10 on chronic pain symptoms utilized intrathecal injections of IL-10 protein 10 days after induction of chronic constriction injury. These injections were found to partially reverse neuropathic mechanical allodynia and thermal hyperalgesia_4 days after administration (Milligan et al., 2005a). However, the effects were only transient with chronic pain symptoms returning to pre-injection levels within 24 hours of injection, probably explained by the observation that injected IL-10 had a half life of 2 hours in the spinal cord (Milligan et al., 2005a).

To combat this limitation and prolong the time IL-10 could be administered in the spinal cord, subsequent studies took advantage of the technique of gene therapy where DNA encoding the desired gene is administered into relevant parts of the body, resulting in the production of the desired protein. Intrathecally administered adeno-associated virus expression vectors containing rat IL-10 complementary

DNA (cDNA) predominantly infected the leptomeningeal cells, leading to the production and release of IL-10 into the spinal cord. The result of this was partial reversal of neuropathy-induced mechanical allodynia and thermal hyperalgesia. The effect diminished two to three weeks after administration when symptoms returned to pre-administration levels (Milligan et al., 2005a; Milligan et al., 2005b). In an attempt to extend the duration and amplitude of impact of intrathecal gene therapy, naked plasmid DNA encoding IL-10 was administered into the spinal cord without a vector. This led to a prolonged (> 40 days) and complete reversal of drug)-induced allodynia cancer paclitaxel (a neuropathicand (Ledeboer et al., 2007; Milligan et al., 2006a). Due to inefficient uptake of plasmid, the drawback of these studies was that reversal of symptoms required numerous high doses of plasmid, which was estimated to equate to a dose of 290 mg for a 81 This limits the therapeutic usefulness of naked plasmid delivery kg person. (Milligan et al., 2006b).

The most successful example of the benefit of IL-10 in chronic pain states came from a study which used a synthetic polymer to deliver IL-10 DNA intrathecally. Complete reversal of neuropathic allodynia occurred for over 40 days with only two injections of much lower doses of DNA to those used in the naked DNA study. These effects are shown in Figure 1.7A (Milligan et al., 2006b).

Not only can exogenous IL-10 influence chronic pain symptoms, but there is evidence that endogenous IL-10 levels influence the experience of chronic pain.

Patients that experience a painful neuropathy have on average half the blood IL-10 mRNA levels than those with a painless neuropathy (Uceyler et al., 2007). Although this does not necessarily mean that endogenous spinal IL-10 levels are imperative, it does suggest endogenous IL-10 plays a role in abrogating the symptoms of chronic pain. The anti-inflammatory and anti-nociceptive actions of IL-10 are summarized in Figure 1.7B.

1.3.3 Other molecules implicated in chronic pain

In addition to IL-10, pro-inflammatory cytokines, glutamate, ATP and its downstream mediator BDNF, a number of other molecular mediators are currently being explored as potential targets of chronic pain therapies. Subcutaneous injection of capsazepine, an antagonist to the transient receptor potential channel, vanilloid receptor 1 (TRPV1), produces a maximum 44% reversal of inflammatory hyperalgesia and 80% reversal of mechanical hyperalgesia in guinea pigs (Walker et al., 2003). Infusion of Na⁺ channel blocker lidocaine results in a 45% reduction in mechanical allodynia in rats with neuropathic pain (Abdi et al., 1998). Gabapentin reduces neuropathic mechanical allodynia in rats by 27% through its inhibitory actions on voltage-gated Ca²⁺ channels (Abdi et al., 1998; Field et al., 2006). In addition, the synthetic cannabinoid WIN55,212-2 completely reversed neuropathic allodynia and hyperalgesia in rats for 20 min (Bridges et al., 2001).

When results of similar experiments targeting ATP and IL-10 are compared with the results of the aforementioned experiments targeting TRPV1, Na⁺ channels and Ca^{2+} channels, it can be seen that ATP and IL-10 may play a more central role in pain and analgesia than these other 3 potential molecular mediators. As highlighted in sections 1.3.1 and 1.3.2.3, interrupting the normal role of ATP in the spinal cord can completely prevent hyperalgesia (Coull et al., 2005) and introducing increased amounts of IL-10 in the spinal cord can completely reverse allodynia (Milligan et al., 2006b). Hence, the anti-chronic pain affects of targeting ATP and IL-10 in animals are much more powerful than targeting TRPV1, Na⁺ channels and Ca²⁺ channels. Furthermore, in contrast to these latter three channels, a more central role of ATP can be conceptualized on the basis of the multiple pathways through which it may contribute to chronic pain, namely through the P2X₃ receptor, P2X₄ receptor and the P2X₇ receptor. Furthermore, ATP has demonstrated a contribution to both allodynia and hyperalgesia, whilst at present, evidence supports a role for TRPV1 primarily in hyperalgesia, and for Na⁺ and Ca²⁺ channels primarily in allodynia.

Whilst targeting the endocannabinoid system can also completely reverse alldoynia and hyperalgesia in rats, numerous human clinical trials have shown that cannabis and synthetic cannabinoids either have no effect on the level of neuropathic pain experienced by patients or they are only as effective as codeine (Attal et al., 2004; Campbell et al., 2001). Furthermore these compounds produce unwanted side effects (Campbell et al., 2001). As such, the mediators ATP and IL-10 will be the selective focus of this thesis.



Figure 1.7 The anti-nociceptive actions of interleukin-10 (IL-10). A) The amount of force that could be applied to a hindpaw before paw withdrawal was measured in rats subjected to a range of conditions. Black rectangles denote animals subjected to neuropathy (induced by chronic constriction injury or CCI) and subsequently injected twice with a synthetic polymer which delivers IL-10 DNA into the spinal cord. Black diamonds denote animals subjected to neuropathy and subsequently injected once with a synthetic polymer which delivers IL-10 DNA into the spinal cord. Empty diamonds represent animals subjected to neuropathy and injected twice with a synthetic polymer which did not deliver IL-10 DNA into the spinal cord. Empty triangles represent animals subjected to sham surgery which did not result in neuropathy and two subsequent injections of a synthetic polymer which did not deliver IL-10 DNA into the spinal cord. It can be seen that delivery of IL-10 DNA completely reverses neuropathic allodynia. BL = baseline. *p<0.05. Taken from Milligan et al (2006b). B) In the dorsal horn of the spinal cord, microglia and astrocytes release the pro-inflammatory cytokines interleukin-1ß (IL-1 β), tumor necrosis factor- α (TNF- α) and interleukin-6 (IL-6) (black arrows). These cytokines cause increased nociceptive neuronal transmission (purple arrows). Interleukin-10 (IL-10) derived from an unknown source decreases the release of these substances from glia by acting through the IL-10 receptor (IL-10R) and consequently reverses the cytokine-induced changes in neuronal transmission (red arrows, dashes and cross). The orange circle represents the nuclei of glial cells.

1.4 POSSIBLE SOURCES OF ATP AND IL-10 IN THE SPINAL CORD

In consideration of the evidence that ATP can contribute to neuropathic allodynia through spinal P2X₄ receptors and hyperalgesia through spinal P2X₃ and potentially P2X₇ receptors, knowledge of the sources of spinal ATP before and during chronic pain is needed to gain a greater understanding of the pathomechanisms of chronic pain. Whilst two possible sources of spinal ATP in chronic pain have been identified, it is possible that more sources exist. Additionally, given the evidence that spinal IL-10 is important in alleviating hyperalgesia and allodynia, knowledge of the endogenous spinal sources of IL-10 is important in understanding the endogenous protective mechanisms of the CNS in chronic pain. To date, however, it has not been examined which spinal cells release IL-10 in response to nociceptive substances. This present research aims to identify novel sources of ATP and IL-10 in the spinal cord in response to pain-related substances.

1.4.1 Possible sources of ATP in the spinal cord in response to pain-related substances

1.4.1.1 Documented sources of ATP in the spinal cord

Knowledge of spinal sources of ATP in response to pain-specific stimuli is limited. Release from dorsal horn neurons is a possible source and there are two studies documenting release from spinal cord glia. When dorsal roots of spinal cord slices are stimulated, many of the post-synaptic currents set up in the post-synaptic laminae II dorsal horn neurons are blocked by antagonists to glutamate, GABA and glycine. However a subset of excitatory post-synaptic currents are blocked only by an ATP receptor antagonist (Bardoni et al., 1997), indicating that ATP is one transmitter released by afferents into the dorsal horn. It is not completely clear which neuronal fibre types are the source of this ATP and hence it is not completely clear whether this ATP release is in response to nociceptive stimuli. Lesion of C fibres does not effect depolarization-stimulated release of ATP in the dorsal horn, the source of ATP fibres are not nociceptive these indicating (Sawynok et al., 1993). Dorsal rizhotomy, which kills all afferent neurons, does not completely eliminate the release of ATP when synaptosomes are exposed to depolarizing K⁺ concentrations, indicating some ATP is released from interneurons originating in the dorsal horn (Sawynok et al., 1993). Further support for this comes with the observation that stimulation of cultured dorsal horn neurons results in a co-release of both GABA, a common interneuron transmitter, and ATP (Jo and Schlichter, 1999). Hence, at the very least it appears dorsal horn interneurons are a source of ATP in the spinal cord subsequent to peripheral The ability of the non-nociceptive $A\beta$ and the afferent action potentials. nociceptive A\delta fibres to release ATP can not be ruled out and is yet to be investigated. Hence, although neuronal ATP release is seen in the dorsal horn, the extent to which it occurs subsequent to painful stimulation is yet to be discovered.

There is evidence that glia may be a source of spinal ATP. In addition to the immune surveillance role outlined in section 1.2.2, microglia have recently been shown to respond to neurotransmitters. In particular, spinal microglia have been

shown to release ATP on exposure to glutamate (Liu et al., 2006). As glutamate release occurs in increased amounts in chronic pain conditions (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992), it is conceivable that microglia could be a source of ATP on induction of chronic pain. Furthermore, spinal cord astrocytes have been documented to release ATP on exposure to the P2Y₂ and P2Y₄ agonist uridine triphosphate (Abdipranoto et al., 2003). This suggests that when astrocytes are exposed to a source of ATP in the spinal cord they can release more ATP in response but does not shed light on the initial source of ATP in the spinal cord after nociceptive stimulation.

1.4.1.2 Glutamate-stimulated ATP release from cortical astrocytes

Apart from glutamate-stimulated ATP release from spinal cord microglia, UTPstimulated ATP release from spinal cord astrocytes and potentially release from peripheral afferent neurons, there are no other reported sources of ATP release in response to pain-related compounds from spinal cells in the literature. It is important, however, to identify other potential spinal sources to get a clear idea of what could be contributing to the pain-problematic spinal pool of ATP.

One place to start in this exploration may be to identify which cells are known to release ATP in response to pain-related transmitters in other areas of the CNS and examine if this release also occurs in the spinal cord. Queiroz et al (1997) demonstrated that application of 1 mM of the neurotransmitter glutamate to cultured

cortical astrocytes led to a nineteen-fold increase in ATP release above basal level from these astrocytes. ATP release was obtained using glutamate concentrations as low as 100μ M. The receptors, intracellular pathways and secretion mechanisms that were involved in this process were investigated.

1.4.1.2.1 Receptors and intracellular pathways involved in glutamatestimulated ATP release from cortical astrocytes

There are four known subtypes of glutamate receptors in the CNS, classified on the basis of their pharmacological, physiological and structural properties. The first subtype is the G-protein linked metabotropic glutamate receptor (mGluR). These exist as homodimers, which is unusual as G-protein linked receptors are commonly found as monomers (Abe et al., 1992; Aramori and Nakanishi, 1992; Pin et al., mGluRs are classified into three groups. Group I mGluRs activate 2006). phospholipase_C (PLC) (Pin and Duvoisin, 1995). Figure 1.8A shows this results in diaclyglycerol liberation from inositol triphosphate (DAG) (IP_3) and phosphatidylinositol-4.5-bisphosphate (PIP2) and consequently increases in the intracellular level of Ca²⁺ from the opening of IP₃ gated Ca²⁺ stores. Group II and III mGluRs inhibit adenylyl cyclase (AC) and consequently inhibit the production of cyclic adenosine monophosphate (cAMP), as shown in Figure 1.8B (Pin and Duvoisin, 1995). There are eight subunits of mGluRs, namely mGluR1 up to mGluR8 classified on the basis of their protein structure and pharmacology. An example of their structure is shown in Figure 1.9A. Group I mGluRs consist of

subtypes mGluR1 and 5, group II mGluRs consist of mGluR2 and 3 and finally group III mGluRs consist of mGluR 4,6,7 and 8 (Pin and Duvoisin, 1995).

The remaining three subtypes of glutamate receptors are ionotropic receptors, which can depolarise the cell through the influx of ions and the activation of voltagedependent Ca²⁺ channels, which open on depolarization, resulting in Ca²⁺-influx. This further depolarises the cell and can activate Ca²⁺-sensitive PLC (Haydon, 2001). The subunits of the ionotropic glutamate receptor subtypes share some features including subunits with an extracellular glutamate binding domain which binds glutamate in a manner akin to the motion of a venus fly trap, an intracellular C-terminal domain and a three-transmembrane segment domain which contains a fourth section that arches inside the membrane. This segment contributes to the formation of the ion channel (Madden, 2002). The general structure of the subunits of these ionotropic receptors is shown in Figure 1.9B.

One form of ionotropic glutamate receptor is the α -amino-3-hydroxy-5-methyl-4isoxazolepropionate (AMPA) receptor, encoded by 4 different genes named GluR1 through to GluR4. Ionotropic glutamate receptors encoded by GluR5 through to GluR7 and KA1 and KA2 are termed kainate (KA) receptors. Ion channels are comprised of subunits that have the same name as the gene that encodes them and subunits are approximately 100 kDa in molecular weight (Bettler and Mulle, 1995). Each receptor contains four or five subunits (Dingledine et al., 1999).



Figure 1.8 Common intracellular signaling pathways linked to metabotropic receptors. A) Phospholipase C intracellular signaling pathway. On binding of a G_q -protein coupled receptor, the G-protein cleaves and one of the products is $G_{\alpha q}$. This can then activate phospholipase C (PLC) which cleaves phosphatidylinositol-4,5-bisphosphate (PIP₂), resulting in diacylglycerol (DAG) and inositol triphosphate (IP₃). IP₃ can open intracellular Ca²⁺ stores and DAG can activate protein kinase C (PKC). B) Adenylyl cyclase intracellular signaling pathway. On binding of a G_s-protein coupled receptor, the G-protein cleaves and one of the products is G_{cs}. This can activate adenylyl cyclase (AC) which results in cyclic adenosine monophosphate (cAMP) production. This can then activate cAMP dependent protein-kinase.



Figure 1.8 continued. C) Phospholipase A_2 intracellular signaling pathway. On binding of a G_i-protein coupled receptor, the G-protein cleaves and one of the products is G_{αi}. This can then activate phospholipase A_2 (PLA₂) which cleaves phosphoinositide (PI), resulting in arachidonic acid (AA) release. D) Phospholipase D intracellular signaling pathway. On binding of a G-protein coupled receptor, a G-protein cleaves, which results in activation of phospholipase D (PLD) and subsequent cleavage of phosphatidylcholine (PC) to produce phosphatidic acid (PA). The nature of the G-protein that cleaves in the PLD pathway depends on the receptor and cell bound (for example Gayral et al (2006) and Everett and Senogles (2004)). This figure was adapted from Kandel and Schwartz (2000).

Eryn Louise Werry



Figure 1.9 The structure of metabotropic and ionotropic glutamate receptors. A) The structure of a metabot ropic glutamate receptor 1 (mGluR1) subunit as an example of the structure of mGluR subunits. Note the large extracellular and intracellular domains and the seven transmembrane segments. The portions of the structure marked as black indicate amino acid r esidues that are conserved between all mGluR subunits. Taken from Pin & Duv oisin (1995). B) The general structure of ionotropic glutamate receptor subunits. Note the large extracellular domains and the three transmembrane domain ns with a further fourth segment which arches inside the membrane. Glu indicates the proposed site for glutamate binding, G indicates a proposed site for glycosylation of the GluR6 subunit and P indicates a proposed site for NR1 phosphorylation. The blac k circles labeled IV, YC and QR indicate sites of possible RNA editing. Taken from Bettler and Mulle (1995).

As their name suggests, AMPA receptors have a high affinity to the agonist AMPA, whereas kainate receptors have a high affinity to the agonist kainate. When glutamate or synthetic agonists bind AMPA or kainate receptors, they become permeable to ions such as Na⁺ and K⁺. These receptors are also permeable to Ca²⁺ ions, except AMPA receptors that contain the subunit GluR2 (Bettler and Mulle, 1995; Hollmann et al., 1991). A feature of AMPA and kainate receptors is their quick de- and re-sensitisation (Zorumski and Thio, 1992).

Another ionotropic glutamate receptor subtype, the N-methyl-D-aspartate (NMDA) receptor, named because of its affinity for the agonist NMDA, is a tetramer composed of a combination four out of seven different subunits, which are named NR1, NR2A, NR2B, NR2C, NR2D, NR3A and NR3B (Cull-Candy et al., 2001; Schuler et al., 2008). NMDA receptors are usually comprised of NR1/NR2 subunits or NR1/NR3 subunits. NMDA receptors comprised of NR1 and NR2 subunits require both glutamate to be bound to the NR2 subunit and a co-agonist such as glycine or a neutral amino acid like D-serine or D-alanine to be bound to the NR1 subunit for the receptor to be fully activated (Brugger et al., 1990; Johnson and Ascher, 1987; Laube et al., 1997; White et al., 1989) whilst receptors built of NR1 and NR3 subunits are activated by glycine alone (Chatterton et al., 2002). In addition, NMDA receptors can be comprised of a mixture of NR1, NR2 and NR3 subunits and these receptors have a reduced response to glutamate as compared to receptors composed of only NR1 and NR2 subunits (Das et al., 1998).

The NMDA channel differs from the AMPA and KA ionotropic glutamate receptors in several important ways. Firstly, in neurons the NMDA receptor is both ligand and voltage-gated. In neurons, it is blocked by a Mg²⁺ ion which is only removed after partial depolarisation of the cell, leading to the voltage-gated nature of the channel. Subsequent to depolarisation, glutamate or a similar agonist is needed to open the ion channel, exemplifying the ligand-gated nature of the channel. Once opened, the NMDA ion channel has a much higher Ca²⁺ permeability and results in a longer lasting depolarisation than the AMPA or KA receptors. Additionally, the NMDA receptor has a higher affinity for glutamate than the other ionotropic glutamate receptors (Mori and Mishina, 1995). The nomenclature for the glutamate receptor subtypes discussed above is summarised in Table 1.1.

There is evidence for the presence of all four types of glutamate receptors on freshly isolated (*in situ*) brain astrocytes. AMPA and KA expression (Conti et al., 1994; Petralia et al., 1994) and activity (Clark and Mobbs, 1992; Jabs et al., 1994; Porter and McCarthy, 1995) is highly documented in astrocytes in brain regions such as the hippocampus, retina, cerebellum and cortex. The most common AMPA subunits localised to these astrocytes are the GluR1 and 4 subunits, with less than 10% of astrocytes expressing the GluR2 and 3 subunits. This suggests that most astrocytes with AMPA receptors are of the Ca²⁺ permeable variety (Conti et al., 1994). There is also immunohistochemical (Aoki et al., 1994; Van Bockstaele and Colago, 1996) and functional evidence (Lehmann, 1987; Porter and McCarthy, 1995; Ziak et al., 1998) for NMDA receptors on *in situ* astrocytes. The mGluR3

Chapter 1

and 5 subtypes of mGluR have been identified on astrocytes *in situ* (Cohen et al., 1968; Fotuhi et al., 1994; Romano et al., 1995), although there is evidence that the mGluR1, 2 and 4 subtypes are lacking in cortical astrocytes (Fotuhi et al., 1994; Martin et al., 1992; Shigemoto et al., 1992; Tanabe et al., 1993). To date, evidence for the presence of other *in situ* mGluR subtypes on brain astrocytes is lacking (Porter and McCarthy, 1997).

To isolate the glutamate receptors that were responsible for glutamate-stimulated ATP release, Queiroz et al (1997) applied 2,3-dihydroxy-6-nitro-7-sulphamoylbenzo(f)quinoxaline (NBQX), a claimed AMPA specific antagonist and found it abolished glutamate's ability to stimulate ATP release. This observation in isolation seemed to suggest that glutamate-mediated ATP release was through AMPA receptors. However, they also found that NBQX decreased the magnitude of ATP release elicited by NMDA and kainate receptor stimulation, suggesting that NBQX also antagonised these receptors. Additionally, they found that agonists of all four glutamate receptors produced ATP release from cortical astrocytes (Queiroz et al., 1997; Queiroz et al., 1999). These observations suggest that all four glutamate receptor subtypes may be involved the release of ATP from cultured cortical astrocytes.

It was found that at high concentrations of glutamate (1 mM), the influx of extracellular Ca^{2+} through voltage-dependant Ca^{2+} channels was necessary for ATP release. Conversely, the blockade of voltage-dependant Ca^{2+} channels and external

 Ca^{2+} availability at lower concentrations (300 µM) of glutamate did not reduce ATP release. This suggests the intracellular signaling pathway involved in glutamatemediated ATP release depends on the concentration of glutamate that is in the vicinity of astrocytes. A possible explanation of this concentration dependency could be that different receptors mediate glutamate-elicited ATP release at different concentrations of glutamate. Queiroz et al (1999) found that AMPA-stimulated ATP release is not disrupted by removing extracellular Ca²⁺ and voltage-dependant Ca²⁺ channels function, whereas NMDA and KA-stimulated ATP release subsequent to stimulation of the AMPA receptor is the predominant mechanism of ATP release, whereas at the higher concentration of 1 mM, the NMDA and KA subtypes are the predominant subtypes acted upon.

Queiroz et al (1999) found that LiCl significantly attenuated ATP release stimulated by glutamate, NMDA, AMPA and kainate. LiCl is an inositol monophosphatase inhibitor. Inositol monophosphatase dephosphorylates inositol monophosphate, resulting in the production of inositol. Levels of inositol are important for regenerating PIP₂, so in effect LiCl functions as a PIP₂-synthesis inhibitor and consequently reduces the amount of IP₃ and DAG generated prior to PLC activation (del Rio et al., 1998; Murray and Greenberg, 1997; Murray and Greenberg, 2000). Hence, this result suggests that ATP release from ionotropic glutamate receptor stimulation on cortical astrocytes involves activation of PLC.

 Table 1.1 Nomenclature of glutamate receptor subtypes.
 Information from Bettler

 and Mulle (1995), Mori and Mishina (1995), Pin and Duviosin (1995) and Schuler et al (2008).

Metabotropic glutamate receptors

Subtype	Group	Subunits in group	Intracellular signaling pathway name
mGluR	I	mGluR1, 5	+ PLC/IP ₃
	П	mGluR2, 3	- AC/cAMP
	III	mGluR4, 6, 7, 8	- AC/cAMP

Ionotropic glutamate receptors

Subtype	Subunit composition
AMPA	GluR1-4
Kainate	GluR5-7, KA1-2
NMDA	NR1, NR2A, NR2B, NR2C, NR2D, NR3A, NR3B

As some isoforms of PLC are Ca^{2+} -dependent (Haydon, 2001), this is congruent with the findings discussed above that NMDA and KA receptors bring about ATP release by creating an influx of Ca^{2+} through voltage-dependent Ca^{2+} channels, which would then activate PLC, resulting in ATP release. However, how PLC activation is evoked by AMPA, which was not found to use extracellular Ca^{2+} to bring about ATP release, is not clear. AMPA might bring about ATP release through activation of PLC by an unknown, Ca^{2+} -independent mechanism.

1.4.1.2.2 Secretion mechanisms involved in glutamate-mediated ATP release from cortical astrocytes

Work on cortical astrocytes suggests there are at least four methods of ATP release from these cells. These include release by cell lysis (Queiroz et al., 1999), exocytosis (Araque et al., 2001; Coco et al., 2003), transporter proteins and connexin hemichannels (Stout et al., 2002).

When the membrane of a cell becomes irreversibly permeable or lyses, molecules such as ATP can leave the cell. In this way, when an astrocyte lyses, ATP can be released (Queiroz et al., 1999).

Exocytosis is the Ca²⁺-dependent (Nicholls, 1998) process by which vesicle-soluble N-ethylmaleimide-sensitive factor attachment receptor (v-SNARE) proteins on neurotransmitter-containing vesicles interact with target (t)-SNARE proteins on the membrane of a cell, causing the vesicle to fuse with the membrane. Consequently the vesicle's contents are released into the extracellular space (Calakos and Scheller, 1996). A number of different types of vesicles have been found inside astrocytes which can undergo exocytosis – analogues of small synaptic vesicles which contain glutamate and other classical neurotransmitters (Calakos and Scheller, 1996), large dense core vesicles which contain ATP and other peptides (Coco et al., 2003) and lysosomes containing ATP (Zhang et al., 2007b). Astrocytic release of ATP by exocytosis of large dense core vesicles has been shown after stimulation with uridine triphosphate (UTP), a metabotropic ATP receptor agonist (Abdipranoto et al., 2003; Coco et al., 2003).

There are three types of transporters in astrocytes that are involved in allowing the passage of ATP from inside the cell to outside and all are members of the ATP binding cassette protein family (Queiroz et al., 1999). They are the cystic fibrosis transmembrane conductance regulator, which is also an ion channel, multidrug resistance-associated-proteins and P-glycoprotein. It has been shown that antagonising these ABC proteins decreases the amount of ATP released from astrocytes in response to mechanical stimulation and stimulation with UTP, the metabotropic purinergic receptor antagonist (Ballerini et al., 2002; Queiroz et al., 1999). However, it should be noted that the effects of the cystic fibrosis transmembrane conductance regulator on ATP efflux may only be indirect, as ATP does not appear to flow through the cystic fibrosis transmembrane conductance regulator, and the effects of the cystic fibrosis transmembrane conductance regulator on stimulation, despite the fact that antagonising the cystic fibrosis.

transmembrane conductance regulator decreases ATP release after stimulation (Wu et al., 2001).

Proteins known as connexin hemichannels, which can join together to form a gap junction, have been suggested to have a functional role as an ion channel in their unjoined form. Evidence is mounting that inhibition of these hemichannels with octanol and flufenamic acid leads to decreased release of ATP from astrocytes after mechanical stimulation (Cotrina et al., 1998; Stout et al., 2002). However, doubt has recently been cast on this evidence as these hemichannel blockers have also been shown to block P2X₇ receptors (Suadicani et al., 2006), hence further work with more specific antagonists is needed to conclude about the role of connexin hemichannels in ATP release.

Queiroz et al. (1999) examined the role that some of these possible secretion mechanisms played in glutamate-mediated ATP release. There was no lysis of cells on exposure to glutamate, thus cell lysis was not an explanation of ATP release. They also found that inhibiting the CFTR reduced glutamate-stimulated ATP release by 23%, which was not a significant difference from the uninhibited state. On the other hand, (Pangrsic et al., 2007) used a fluorescent dye to image vesicles in astrocytes that contained ATP co-localised with atrial naturetic peptide and observed these vesicles underwent exocytosis on exposure to glutamate. Presumably these vesicles were large dense core vesicles, given their peptidergic contents. In further support of an exocytotic release from astrocytes, Zhang et al (2007b) found that on exposure to glutamate, ATP-containing lysosomes underwent partial exocytosis and selective lysis of lysosomes abolished glutamate-stimulated ATP release from cultured cortical astrocytes. The effect of hemichannel disruption on glutamate-stimulated ATP release from cortical astrocytes has not been examined.

1.4.1.2.3 Glutamate uptake and ATP degradation: the final steps in glutamatestimulated ATP release from cortical astrocytes

The final steps in glutamate-stimulated ATP release are removal of extracellular glutamate and degradation of ATP. Excess glutamate in the extracellular space must be removed. Found on the surface of murine astrocytes are the glutamate and aspartate transporter (GLAST) and the glutamate transporter (GLT) (Rothstein et al., 1994). These proteins uptake glutamate into astrocytes across a Na⁺ gradient, with a stoichiometry of one glutamate molecule uptaken per three Na⁺ ions that enter the astrocyte (Anderson and Swanson, 2000). Astrocytes can also uptake manner through Na⁺-independent chloride-dependent glutamate in a glutamate/cystine antiporters, although this uptake only accounts for less than five percent of total glutamate uptake (Cho and Bannai, 1990; Tsai et al., 1996). When uptaken glutamate has entered the astrocyte, it is metabolised to the amino acid glutamine by glutamine synthetase, an enzyme found exclusively in astrocytes (Martinez-Hernandez et al., 1977). Neurons also express GLAST along with three other glutamate transporters, referred to as excitatory amino acid transporter 3, 4 and 5. The excitatory amino acid transporter 4 is primarily found on neurons in the cerebellum and the excitatory amino acid transporter 5 is primarily found on neurons in the retina (Furuta et al., 1997; Pow and Barnett, 2000; Rothstein et al., 1994). These transporters, however, only clear an insignificant fraction of the total glutamate cleared from the extracellular space (Anderson and Swanson, 2000). Furthermore, under pathological conditions such as ischaemia, the excitatory amino acid transporter 3 can reverse its normal operation and lead to release of glutamate into the extracellular space (Rossi et al., 2000)).

When ATP is released from an astrocyte, it is degraded to ATP by ecto-ATPase and then to AMP by ecto-ADPase and finally to adenosine by ecto-nucleotidase (Trapido-Rosenthal et al., 1990). Ecto-ATPases are found on the same areas of the cell surface from which ATP is released from astrocytes (Joseph et al., 2003), although not all released ATP is immediately decomposed, as some released ATP can diffuse to effect other cells (Fam et al., 2003; Koizumi et al., 2003; Scemes et al., 2000).

The work reviewed above and summarised in Figure 1.10 suggests that glutamatestimulated ATP release from cultured cortical astrocytes is possibly mediated by all four subtypes of glutamate receptor, leading to both extracellular Ca^{2+} influx through voltage-dependent Ca^{2+} channels and activation of the PLC/PIP₂ second messenger system. The result of this is ATP release through exocytosis of lysosomes and large-dense core vesicles. Glutamate is subsequently taken up by

GLAST and GLT and ATP is degraded by ecto-ATPase, ecto-ADPase and ectonucleotidase.

1.4.1.3 Potential of spinal cord astrocytes to release ATP in response to glutamate

Given that a focus of this present research is to identify novel sources of spinal ATP, it is appealing to hypothesise glutamate-stimulated ATP release from spinal cord astrocytes as a potential source of ATP in the spinal cord for numerous reasons. Firstly, glutamate is the most common excitatory neurotransmitter and is released in the dorsal horn in increased amounts during chronic pain (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992). Astrocytes are ideally positioned to be exposed to this synaptically released glutamate as their processes enwrap synapses (Ventura and Harris, 1999). In addition to this, there are more astrocytes in the CNS then there are neurons or other glial cell types suggesting that they could be a large contributor to the total pool of ATP if they were to release ATP to glutamate stimulation (Peinado et al., 1998; Stevens, 2003). Finally results from cortical astrocytes suggest astrocytes could be a powerful source of ATP compared to other cell types. Glutamate-stimulated ATP release from cortical astrocytes was a much more dominant source of ATP than glutamate-stimulated cortical neuronal ATP release as mixed astrocyte-neuronal cultures containing 10% of neurons did not differ in the amount of glutamate-stimulated ATP release than those with the same amount of astrocytes but further neurons added to increase the total percentage of



Figure 1.10 Mechanisms of glutamate-stimulated ATP release from cortical Glutamate (1 mM) brought about ATP release by binding to astrocytes. N-methyl-D-aspartate (NMDA), kainate, metabotropic (mGluR) and q-amino-3hydroxy-5-methyl-4-isoxazolepropionate (AMPA) receptors. On glutamate binding, cations entered the cell through NMDA and kainate receptors, and this subsequently activated voltage dependent Ca²⁺ channels, causing further Ca²⁺ influx. The Ca²⁺ ions then activated phospholipase C (PLC). PLC was also activated when glutamate bound mGluRs and AMPA receptors, although the mechanism by which AM PA receptor binding leads to PLC activation is The result of PLC activation was ATP release by exocytosis of unknown. lysosomes and large dense core vesicles (LDCVs). Extracellular glutamate was uptaken by the glutamate transporter (GLT) and the glutamate and aspartate transporter (GLAST).

neurons to 30% (Queiroz et al., 1999).

It cannot be assumed that because astrocytes from the cortex display glutamatestimulated ATP release, that the same will be true for spinal cord astrocytes because a regional heterogeneity amongst astrocytes has been reported. There are numerous examples of this. SP can stimulate phosphoinositide turnover and prostaglandin release in spinal cord astrocytes but not in cortical astrocytes (Cholewinski et al., 1988; Marriott et al., 1991). Conversely, the neuropeptides vasopressin and oxytocin can stimulate phosphoinositide turnover in cortical but not spinal cord astrocytes (Cholewinski et al., 1988). A β-adrenergic agonist can increase cAMP in spinal cord and cortical astrocytes, however the hormone somatostatin blocks this response from cortical astrocytes but does not block it from spinal cord astrocytes (Cholewinski and Wilkin, 1988). Finally, another hormone, thyrotropin releasing hormone, causes phosphoinositide turnover in spinal cord but not cortical astrocytes (McDermott et al., 1992). This raises the question of what evidence exists to suggest glutamate may be able to stimulate ATP release from spinal cord astrocytes. Spinal cord astrocytes have some of the cellular machinery needed for glutamatestimulated ATP release. For example immunocytochemical experiments have shown cultured spinal cord astrocytes express group I and II mGluRs (Silva et al., 1999) and in situ spinal cord astrocytes express NMDA receptors and the GluR1, 4 and 5 sub-units of the AMPA and kainate receptors (Aicher et al., 1997; Brand-Schieber et al., 2004). Additionally, as mentioned above, spinal cord astrocytes are able to release ATP to another transmitter, UTP (Abdipranoto et al., 2003).

Exposure of spinal cord astrocytes to glutamate results in an increase in intracellular Ca^{2+} and, as mentioned in section 1.4.1.2.1, intracellular Ca^{2+} elevations are involved in the process of ATP release (Ahmed et al., 1990; Queiroz et al., 1999). This demonstrated ability of spinal cord astrocytes to release ATP and to have receptors to respond to glutamate suggests that glutamate-stimulated ATP release may be a viable process in spinal cord astrocytes. Given the appealing potential of spinal cord astrocytes to release ATP to glutamate, *one objective of the current research is to explore whether glutamate can stimulate ATP release from spinal cord astrocytes and if release is found, to explore the mechanisms involved.*

1.4.1.4 The potential for SP to be involved in ATP release from spinal cord astrocytes

Glutamate is co-released with other neurotransmitters at many CNS synapses so that when glutamate is released by neurons and binds to astrocytes, these other transmitters will also be present in the extracellular environment surrounding astrocytes (Millan, 1999). This opens up the possibility for these simultaneously released transmitters to be potential modulators of glutamate-stimulated ATP release, if not releasing agents themselves. The neurotransmitter SP is found colocalised with glutamate in spinal cord terminals of nociceptive fibres (De Biasi and Rustioni, 1988) and although only a small amount of these fibres have been reported to contain proteins required for loading of glutamate into exocytotic vesicles and for the exocytosis of these vesicles, SP-containing fibre terminals are localized directly next to fibres that contain both glutamate and the proteins
required for its exocytosis (Morris et al., 2005). Consequently upon noxious stimulation both glutamate and SP are simultaneously released in laminae I, II, IV and V of the dorsal horn (Allen et al., 1999; Sakamoto and Atsumi, 1989; Sasaki et al., 1998; Womack et al., 1988). In the spinal cord SP and glutamate are also released together in the interomediolateral column and the ventral horn bringing about increases in blood pressure and motor neuron excitability respectively (Helke et al., 1982; Hokfelt et al., 2000; Kapoor et al., 1992; Nicholas et al., 1992). SP is a peptide containing 11 amino acid residues (Maggio, 1988) and once released SP is degraded by a range of enzymes including angiotensin converting enzyme and neutral endopeptidase 24.11 (Borson, 1991; Khawaja and Rogers, 1996; Skidgel et al., 1984) which cleave SP at the sites indicated in Figure 1.11A. Prior to deactivation, however, SP is known to bind at least three distinct receptors. These are the neurokinin-1 (NK1), neurokinin-2 (NK2) and neurokinin-3 (NK3) receptors (Saria, 1999). A fourth receptor was proposed in the human (Krause et al., 1997) however this receptor has since been shown to have a very similar pharmacology and amino acid composition to the human NK3 receptor and a search for a separate gene for this fourth receptor has failed, suggesting this fourth receptor is actually a variant of NK3 (Krause et al., 1997; Sarau et al., 2000). NK1, NK2 and NK3 receptors exist as monomers that contain seven transmembrane domains with an extracellular amino terminal and an intracellular carboxy terminal as illustrated in Figure 1.11B and differ in the residues indicated in the figure (Khawaja and Rogers, 1996; Meyer et al., 2006). Of these three neurokinin receptors, SP has the highest



Figure 1.11 Structure of substance P and its receptors. A) The amino acid composition of substance P. Blue arrows indicate the bonds which are cleaved by neutral endopeptidase 24.11 and the red arrows indicate the bonds which are cleaved by angiotensin converting enzyme. The amino acid compositions of neurokinin-A and neurokinin-B are also displayed, with amino acids common to these neuropeptides indicated in capitals. Adapted from Maggio (1988). B) The secondary structure of the neurokinin-1 receptor as an example of the structure of neurokinin receptors. The amino acid residues conserved between the three subtypes of neurokinin receptor are indicated by black circles. TM ₁₋₇ denotes the seven transmembrane segments of the receptor. Adapted from Khawaja and Rogers (1996).

relative affinity to the NK1 receptor and primarily exerts its biological effects through this receptor (Harrison and Geppetti, 2001) although it will act on all neurokinin receptor classes (Mussap et al., 1993; Regoli et al., 1994). Two peptides with similar amino acid sequences to SP (see Figure 1.11A) are neurokinin-A and neurokinin-B and these have a higher affinity than SP at the neurokinin-1 and neurokinin-2 receptors respectively (Cascieri et al., 1992). There is both immunohistochemical and functional evidence that astrocytes in the murine spinal cord express all three types of SP receptor (Palma et al., 1997; Wienrich and Kettenmann, 1989; Yashpal et al., 1990; Zaratin et al., 2000; Zerari et al., 1997; Zerari et al., 1998). Consequently, when SP is released at the same time as glutamate in the spinal cord, the receptor machinery is present for SP to bind to astrocytes.

All three neurokinin receptors are G-protein coupled receptors which are known to stimulate the PLC/IP₃ signaling pathway, the phospholipase A_2 (PLA₂)/arachidonic acid pathway, the cAMP/AC pathway and the phospholipase D (PLD) pathway (Garcia et al., 1994; Montuschi et al., 1996; Nakajima et al., 1992; Regoli et al., 1994; Yau et al., 1991). By acting through these pathways, which are summarised in Figure 1.8, SP has been shown to cause membrane potential depolarisation (Wienrich and Kettenmann, 1989), phosphoinositide turnover, prostaglandin release (Hartung et al., 1988; Marriott et al., 1991) and increases in intracellular Ca²⁺ levels (Heath et al., 1994) in astrocytes from the spinal cord. Some of these events, such as phosphoinositide turnover (Abdipranoto et al., 2003) and increases in

intracellular Ca²⁺ concentration (Queiroz et al., 1999), have been shown to increase ATP release from astrocytes. Hence, when SP and glutamate are co-released from neurons onto astrocytes, not only will glutamate bind to astrocytes to effect them, but SP can potentially bind to astrocytes and bring about changes linked with ATP release, allowing the possibility that the effects of SP on astrocytes may include either release of ATP itself or interference with the process of glutamate-stimulated ATP release.

SP has been postulated to be a modulatory peptide for a range of neurotransmitter systems. For example, Bourque and Robitaille (1998) found that SP on its own caused increases in intracellular Ca^{2+} in perisynaptic Schwann cells. Conversely, when combined with muscarinic acetycholine receptor agonists or ATP, SP reduced the percentage of cells that responded to these agonists. SP has also been shown to potentiate noradrenaline-induced cAMP activation in astrocytes (Rougon et al., 1983) and enhanced NMDA-induced currents in spinal cord dorsal horn neurons (Randic et al., 1990; Rusin et al., 1993).

Considering this evidence, it can be seen that SP is co-released with glutamate onto spinal astrocytes that have both SP and glutamate receptors, SP has been shown to bring about changes in astrocytes linked to ATP release and has been shown to modulate the response to neurotransmitters in glial cells and spinal cord cells. Put together, these suggest that SP is a promising neurotransmitter to examine in the search for either a substance to elicit ATP release from spinal astrocytes or a

potential endogenous modulator of glutamate-stimulated ATP release. Hence, a further objective of this present research will be to examine if SP elicits ATP release from spinal cord astrocytes or if it modulates any observed glutamatestimulated ATP release from spinal cord astrocytes.

1.4.2 Possible sources of IL-10 in the spinal cord in response to pain-related substances

1.4.2.1 Documented sources of IL-10 in the brain and evaluation of candidates for sources of the spinal pool of IL-10 in chronic pain

Apart from discovering novel sources of ATP in the spinal cord, the second main aim of this present research is to identify novel sources of IL-10 in the spinal cord in chronic pain. Although it has not been investigated which cells in the spinal cord release IL-10, release of IL-10 has been shown from a number of cell types in other areas of the CNS. One way to generate candidate spinal stimulating substances and cellular sources of IL-10 may be to review what is known about IL-10 release in these non-spinal areas of the CNS. The most studied cellular sources of IL-10 in the CNS are astrocytes and microglia. In the brain, they have been recorded to release IL-10 on exposure to a range of agents. One group of such agents are pathogens. Parasites, such as the abortion-causing Neospora caninum, elicit IL-10 release from astrocytes (Pinheiro et al., 2006), neonatal Borna disease virus, which causes chronic infection of neurons elicits IL-10 release from mixed glia (Ovanesov et al., 2006) and bacteria such as Borrelia burgdorferi and Neisseria meningitidis elicit IL-10 release from both microglia and astrocytes (Rasley et al., 2006).

Chapter 1

Athough the most common result of exposure to pathogens is an increase in IL-10 release from these glia, some pathogens such as Japanese encephalitis virus decrease IL-10 release levels from microglia (Swarup et al., 2007).

Pathogens have molecular structures on their surface which are invariant. These pathogen-associated molecular patterns are recognized by Toll-like receptors on cells of the mammalian innate immune system (Cristofaro and Opal, 2006). Thirteen different Toll-like receptors (TLR) have been described in mice and eleven in humans. Agonists to two different types of Toll-like receptors also elicit IL-10 release from glia. A TLR4 agonist stimulates the release of IL-10 from microglia and astrocytes of the brain (Ledeboer et al., 2002; Mizuno et al., 1994; Seo et al., 2004; Williams et al., 1996) and when stimulated with a TLR3 agonist such as double stranded RNA or double stranded messenger RNA, astrocytes taken from the subcortical white matter release IL-10 (Alexopoulou et al., 2001; Bsibsi et al., 2006; Kariko et al., 2004). Double stranded RNA can be released extracellularly on lysis of virus-infected cells (Majde et al., 1998).

Binding of infiltrating T-cells to microglia results in IL-10 release (Chabot et al., 1999). Infiltration of T-cells can occur in blood-brain barrier breakdown or in diseases such as multiple sclerosis (Chabot et al., 1999). Finally, cytokines such as TNF- α and IL-6 can elicit IL-10 release from human brain microglia (Sheng et al., 1995). Interestingly, on induction of peripheral inflammation, the level of IL-10 increases in the CNS near microglia and astrocytes

close to leptomeningeal cells (Wu et al., 2005). When exposed to peripheral inflammation leptomeningeal cells were shown to release TNF- α , possibly explaining why adjacent cells such as microglia showed an increased staining for IL-10 (Wu et al., 2005).

Other non-glial cells have been documented to release IL-10 in the brain. Cultured brain endothelial cells have been shown to spontaneously release IL-10 on their luminal side and hence could be a basal source of IL-10 in the brain (Verma et al., 2006). Little is known about how pain-related substances effect the secretion of IL-10 by brain endothelial cells, however an agonist of the TLR4 does not significantly increase brain endothelial cell-stimulated IL-10 release (Verma et al., 2006).

As mentioned in section 1.3.2.1, after the induction of peripheral neuropathic pain blood-borne macrophages can infiltrate into the spinal cord and differentiate into microglia. Peripheral macrophages secrete IL-10 in the blood both spontaneously and on exposure to a TLR4 agonist (Hagiwara et al., 1995; Kim et al., 2004). It is not known if macrophages infiltrating into the spinal cord secrete IL-10 in a similar manner before they differentiate into microglia.

If all the brain cellular sources described above were to be sources of IL-10 in the spinal cord, it is probable that microglia and astrocytes would play a more dominant role in chronic pain than endothelial cells and macrophages. Although endothelial

cells are in contact with perivascular microglia, endothelial cells are located quite a distance from the majority of the pro-inflammatory cytokine releasing microglia and hence any potential release of IL-10 may be limited in its effectiveness in preventing pro-inflammatory cytokine release. Also macrophages are only transiently in the spinal cord during chronic pain before they differentiate into microglia (Zhang et al., 2007a), temporally restricting the impact of any possible IL-10 release. On the other hand, if spinal microglia and astrocytes could be shown to release IL-10 this may have a large impact in chronic pain as it presents the opportunity for IL-10 to act in an autocrine manner to shut down the action of pro-inflammatory cytokines released by these cells. This makes it appealing to investigate if microglia and astrocytes may be a source of IL-10 in the spinal cord.

Of the factors documented to stimulate IL-10 from brain microglia and astrocytes, most pertinent to the situation of chronic pain are reports that an agonist for the TLR4 can increase the production of IL-10 from microglia and astrocytes. Pathogens, T-cells or extracellular dsRNA/mRNA in the spinal cord occur in infection and certain disease states and it would not be expected that these IL-10 inducers would be present in a person whose only pathology is chronic pain (Chabot et al., 1999; Majde et al., 1998). In contrast, the TLR4 is known to be activated in neuropathic pain (Tanga et al., 2005). It has not been investigated whether it is also activated in inflammatory pain. Additionally, although IL-6 and TNF- α are present in painful situations, at least in neuropathic hyperalgesia and allodynia their production appears to be subsequent to TLR4 stimulation as TLR4

Chapter 1

antisense oligodeoxynucleotide administration completely removes the increase in mRNA of these cytokines seen on induction of neuropathic pain (Tanga et al., 2005). Consequently, any effect on IL-10 levels that IL-6 and TNF- α may have in neuropathic pain will occur after effects of TLR4 stimulation.

Taking into consideration the above-mentioned possible autocrine antiinflammatory mechanism and the early TLR4 activation in neuropathic pain, it is appealing to hypothesise that TLR4-stimulated IL-10 release from microglia and astrocytes could be a source of IL-10 in the spinal cord in chronic pain. What is known about TLR4-stimulated IL-10 release is detailed below in order to explore whether spinal cord astrocytes and microglia possess any of the proteins known to be involved in the process of TLR4-stimulated IL-10 release from glia and other cells in other areas of the CNS and body.

1.4.2.2 TLR4-activated IL-10 release from brain microglia and astrocytes

The TLR4 agonist lipopolysaccharide (LPS) stimulates the release of IL-10 from human, mouse and rat brain microglia (Ledeboer et al., 2002; Mizuno et al., 1994; Seo et al., 2004; Williams et al., 1996) and from rat brain astrocytes (Ledeboer et al., 2002). In microglia, LPS concentrations of $0.1 - 1 \mu g/ml$ applied for 24 h elicits an accumulation of IL-10 in the extracellular medium that is 9 to 11 times higher than that which accumulates in the medium without stimulation (Ledeboer et al., 2002; Mizuno et al., 1994; Seo et al., 2004). The amount of accumulated IL-10 continues rising with an increased time of exposure to LPS and

even at 96 h of exposure the level of IL-10 continues to accumulate (Seo et al., 2004). The amount of accumulated IL-10 in the medium of cortical astrocytes stimulated with $0.1 \,\mu$ g/ml of LPS for 24 h rises 17-fold above the unstimulated amount. The timecourse of release from astrocytes has not been mapped (Ledeboer et al., 2002). Details of the TLR4 and intracellular mechanisms of TLR4-stimulated IL-10 release from immune cells have been investigated.

1.4.2.2.1 The Toll-like receptor 4 and its agonists

The TLR4 is a complex receptor with both an intracellular and extracellular domain and exists as a homodimer as can be seen in Figure 1.12A. As mentioned in section 1.4.2.1, Toll-like receptors respond to certain types of pathogen. The pathogen that the TLR4 responds to is gram-negative bacteria (Takeuchi et al., 1999). In addition, this receptor also responds to a range of other endogenous and Gangliosides are a gylocophospholipid-containing exogenous substances. component of mammalian cell-membranes. Gangliosides can be released into the extracellular space for example after exposure to oligomeric amyloid-ß protein (Matsuzaki and Horikiri, 1999; Michikawa et al., 2001). Gangliosides have been shown to activate glial TLR4s leading to an upregulation of cytokines such as IL-1β and TNF-a (Jou et al., 2006). Biglycan, a proteoglycan rich in leucine which accumulates in the culture medium of astrocytes also signals through the TLR4 (and TLR2) leading to the release of cytokines from macrophages (Koops et al., 1996; Schaefer et al., 2005). It is unknown what stimulates biglycan release from astrocytes. Saturated fatty acids also signal through the TLR4 (Hwang, 2001).

Although a number of other proteins have been suggested as TLR4 agonists one problem in the identification of TLR4 agonists is that recombinantly produced compounds are often applied to cells to test whether a substance acts on the TLR4. Recombinant products contain residual bacterial cell wall components such as LPS and these contaminants can produce false positive results, such as what was found for heat shock protein 60 and 70 and surfactant protein-A (for a review see Tsan and Gao, 2004).

Apart from these endogenous ligands, the most commonly used exogenous TLR4 ligand is LPS which is often used to experimentally mimic inflammatory conditions. LPS is a component of gram-negative bacterial walls which enables the immune system to respond to gram negative bacteria. LPS from wild type gram-negative bacteria such as *Escherichia Coli, Porphyromonas gingivalis* or *Salmonellae typhimurium* is referred to as smooth LPS (S-LPS) and is composed of three major regions. The first is a polysaccharide side chain region with antigens (Osborn et al., 1964). This region is often referred to as the O-antigen region and its structure can vary depending on the bacterial source of LPS (Dixon and Darveau, 2005). The next portion is a core polysaccharide linking the side chain portion to the final lipid component called lipid A, which is unusual compared to most lipids as it contains glucosamine instead of glycerol (Osborn et al., 1964). In gram negative bacteria, lipid A is embedded in the cell membrane of

Eryn Louise Werry

68



Figure 1.12 The structure of lipopolysaccharide (LPS) and its receptor. A) A schematic of the three different domains of LPS when it is embedded in the cell membrane of gram negative bacteria. The final extracellular segments of S-LPS and R-LPS are indicated. Adapted from Dixon and Darveau (2005). B) Schematic of the binding of LPS to the Toll-like receptor 4 (TLR4) complex. 1. A single LPS molecule is removed from an aggregate of LPS by LPS binding protein (LBP) and is presented to CD14. 2. CD14 then presents LPS to myeloid differentiation Ag (MD2). 3. The binding of LPS by MD2 brings the two Toll/interleukin-1 receptor (TIR) domains of the TLR4 homodimer together. Adapted from Jerala (2007).

the 2 polysaccharide components are extracellular (Dixon and Darveau, 2005). It is the lipid A portion of lipopolysaccharide that is the main biologically active component as administration of just this portion can elicit the same immune effects as LPS (Chaby, 2004). The structure of S-LPS can be seen in Figure 1.12A. Additionally, rough LPS (R-LPS) is produced by rough mutants of gram-negative bacteria. R-LPS does not contain the O-antigen region (Caroff and Karibian, 2003).

Considering the different origins of the endogenous and exogenous TLR4 ligands, it can be seen that this receptor is activated during infection with gram negative bacteria, after ganglioside-releasing cell damage and potentially on exposure to non-stimulated astrocytes. Additionally, as detailed in section 1.3.2.1, there is evidence that the TLR4 is activated during neuropathic pain although it is currently unknown which endogenous ligands contribute to its activation (Tanga et al., 2005).

1.4.2.2.2 Recognition of LPS by the Toll-like receptor 4 and subsequent signaling mechanisms

Numerous steps are required for LPS to activate intracellular signaling pathways through the TLR4. The TLR4 on its own is not responsive to LPS and requires a protein named myeloid differentiation Ag (MD2) to first bind LPS. The MD2/LPS complex can then bind to the extracellular portion of TLR4 (Kennedy et al., 2004; Shimazu et al., 1999). Hence, MD2 is in effect the ligand binding segment of the TLR4 receptor complex (Visintin et al., 2005).

Evidence suggests that, at least in part, LPS works at the transcriptional level to increase IL-10 release as all studies that have found a release of IL-10 from LPS-exposed microglia and astrocytes report a concomitant upregulation of IL-10 mRNA (Ledeboer et al., 2002; Mizuno et al., 1994; Seo et al., 2004; Williams et al., 1996). There are a number of different intracellular signaling pathways through which activation of the TLR4 can lead to changes in transcription. The intracellular segment of each TLR4 monomer contains a sequence of amino acids known as a Toll/interleukin-1 receptor (TIR) domain and when the TLR4 homodimer is bound by MD2/LPS, the TLR4 changes conformation bringing the TIR domains of each TLR4 monomer closer together. This enables one of two proteins that also contain TIR domains to attach to the TLR4. Depending on which TIR protein attaches to the TLR4, two distinct signaling pathways can be activated. One is a fast response pathway and the other is a delayed response pathway. The quick response myeloid differention 88 (MyD88)-dependent pathway is characterized by attachment of the MyD88-adaptor-like (M AL) protein to the TLR4 while the delayed response MyD88-independent pathway is characterized by the attachment of TIR-containing adaptor molecule-1 (TRAM -1) to the TLR4 (O'Neill and Bowie, 2007).

After attachment to the TLR4, MAL binds PIP_2 in the plasma membrane which recruits MyD88 (Fitzgerald and Chen, 2006). MyD88 recruits IL-1 receptor associated kinase 4 (IRAK4) which then recruits IRAK1. The target of IRAK1 is tumor-necrosis-factor-receptor-associated factor-6 (TRAF6). TRAF6 recruits a range of intracellular signaling molecules such as TGFβ-activated kinase 1 (TAK1),

TAK1-binding protein 2 (TAB2), ubiquitin-conjugating enzyme E2 variant 1 (UEV1A) and ubiquitin-conjugating enzyme 13 (UBC13) (O'Neill and Bowie, 2007). These enzymes activate 2 main pathways. The first is initiated by inhibitor of nuclear factor-kB (IkB) kinase complex (IKK). In its inactive form the transcription factor nuclear factor-kB (NF-kB) is bound to IkB. This interaction prevents NF-kB from interacting with DNA. IKK is a complex of several IKK subunits which acts to phosphorylate IkB, resulting in its degradation and detachment from NF-kB. This allows the translocation of NF-kB into the nucleus then subsequent attachment of it to NF-kB binding motifs on genes resulting in the production of a variety of pro-inflammatory cytokines.

NF-kB exists as a dimer made of a combination of p65 and p50 subunits. The most abundant form of NF-kB is the p50/p65 heterodimer. In a macrophage-like cell line, when the p50 homodimer form of NF-kb is stimulated by TLR4 activation IL-10 promoter activity is increased. No other form of NF-kB has IL-10 promoter activity (Cao et al., 2006). It is as yet unclear whether TLR4 stimulation in glia activates a p50 homodimer form of NF-kB and hence whether the TRAF-6/NF-kB pathway contributes to LPS-induced IL-10 release from glia.

Additionally, on activation of TAK1, TAB2, UEV1A and UBC13 by TRAF6, mitogen-activated protein kinase (MAPK) kinase (MKK) 3 and 7 are activated. MKK3 then activates the p38 MAPK whilst MKK7 activates the JUN N-terminal kinase (JNK) MAPK pathway (O'Neill and Bowie, 2007). In glia, LPS-induced p38

and JNK increase the activity of the transcription factor c/EBP β (CCAAT/enhancer binding protein β). Also, in macrophages it has been demonstrated that LPSstimulated p38 activation increases the activity of the transcription factor Simian virus 40 promoter factor 1 (Sp1) (Cao et al., 2006; Ejarque-Ortiz et al., 2007; Kelicen and Tindberg, 2004; Ma et al., 2001). Whether LPS activates Sp1 in glia has not been investigated although it has been shown that Sp1 is present in glia (Mao et al., 2007). The promoter site on the IL-10 gene has two areas responsive to c/EBP β and Sp1 respectively and distrupting both sites almost completely blocks LPS-activated IL-10 expression in macrophages (Chiang et al., 2006; Liu et al., 2003).

Apart from MAL, the other protein that can attach to the intracellular segment of LPS/MD2-bound TLR4 is TRAM -1. Once attached, TRAM -1 binds the protein TIR-domain -containing adaptor inducing IFN- β (TRIF) which activates two pathways. The first pathway activated by TRIF is via tumor-necrosis-factor-receptor-associated factor-3 (TRAF3) and leads to activation of the transcription factor interferon regulatory factory-3 (IRF-3), which when combined with the NF-kB subunit p65 can bind interferon-stimulated response elements (ISRE) on genes, leading to the production of the cytokine interferon- β (O'Neill and Bowie, 2007). The second pathway that TRIF activates is via TRAF6 and leads to activation of NF-kB (Fitzgerald et al., 2003). Activation of the MyD88/TRAF6/IKK pathway leads to the production of NF-kB in a faster manner than does the TRIF/TRAF6 pathway.

Prior to MD2 binding of LPS, LPS is presented to MD2 by either a serum or glycolphosphatidylinositol anchored membrane protein named cluster determinant 14 (CD14), which is necessary for some of the physiological effects of LPS (Tobias et al., 1995; Wright et al., 1990). S-LPS cannot induce MyD88-dependent signaling through the TLR4 without CD14, while R-LPS does not need CD14 to elicit the MyD88-dependent pathway (Huber et al., 2006; Jiang et al., 2005). Both R- and S-LPS require CD14 to signal through the TRIF pathway (Jiang et al., 2005).

Another protein that can increase the sensitivity of the response to LPS is LPS binding protein (LBP), although LPS can bind MD2 without LBP (Viriyakosol et al., 2001; Visintin et al., 2005; Visintin et al., 2001). LBP can either be present in the plasma or be associated with the cell membrane. In the blood LPS often exists in aggregates and usually the first step towards binding of the TLR4 by LPS involves binding of a single LPS molecule by LBP (Roes et al., 2006; Schumann et al., 1990). When it is present in low concentrations, the role of plasma LBP has been proposed as binding individual LPS monomers to separate them from aggregates (Jerala, 2007; Schumann et al., 1990). It is unclear if the role of membrane-associated LBP is the same as plasma LBP since membrane-associated LBP is the same as plasma LBP since membrane-associated LBP is the same as plasma LBP since membrane-associated LBP is the same as plasma LBP. Nonce bound, LBP then delivers LPS to CD14. The ease of transfer of LBP-bound LPS to CD14 depends on the bacterial source of the LPS (Cunningham et al., 1996). The role of CD14 and LBP is illustrated in Figure 1.12B.

Work has been carried out to investigate whether endogenous ligands bind to the TLR4 and bring about intracellular changes in a similar manner to LPS. It is known that gangliosides require CD14 to bind TLR4 and a subsequent intracellular event is NF-kB activation (Jou et al., 2006). Additionally, there is evidence that biglycan signals through the TLR4 mainly via the MyD88 pathway, resulting in activation of MAPKs and NF-kB. Also, saturated fatty acids activate NF-kB (Hwang, 2001; Schaefer et al., 2005). Studies providing a more in-depth knowledge of how endogenous ligands activate the TLR4 are needed if interventions in TLR4 signaling are to be made to control outcomes of endogenous stimulation of this receptor.

A summary of the literature reviewed above shows that after CD14-dependent binding of S-LPS (the type of LPS used in most experimental studies) to MD2, this complex can attach to the TLR4 homodimer stimulating a fast MyD88-dependent and a more delayed MyD88-independent signaling pathway. The result of these pathways is interferon- β release, pro-inflammatory cytokine release, from a quick or delayed activation of NF-kB potentially accompanied by NF-kB stimuated IL-10 release and activation of p38 and JNK MAPK, resulting in activation of c/EBP and consequent upregulation of IL-10 mRNA and IL-10 release. This pathway is summarized in Figure 13.

1.4.2.2.3 Mechanisms of IL-10 secretion

The mechanisms of IL-10 secretion from cells are unknown. It is proposed that another cytokine, IL-1 β may be released via secretory lysosomes, which are organelles that can be exocytosed, or microvesicles, which are small blebs of membrane (MacKenzie et al., 2001; Mambula and Calderwood, 2006). In some lysosomal storage disorders like Gaucher disease and Globoid cell leukodystrophy an upregulation of IL-10 is seen (Allen et al., 1997; Wu et al., 2000), perhaps providing circumstantial support for a lysosomal release of IL-10, however a colocalisation and co-secretion of IL-10 with lysosomal enzymes will have to be shown in further experiments to conclude a lysosomal mechanism. Further, other secretion mechanisms such as exocytosis of classic vesicles and microvesicles will also have to be explored.

1.4.2.2.4 Removal of LPS and degredation of IL-10

The final steps in TLR4-mediated IL-10 release are removal of the TLR4 agonist and released IL-10 from the extracellular space. There are two main documented methods of LPS deactivation. LPS can bind membrane-anchored CD14, be internalized, and then degraded, for example, by lysosomal acid phosphate, which is an enzyme identified in the CNS among other areas (Chaby, 2004; Gegner et al., 1995; Geier et al., 1992; Luchi and Munford, 1993). The second documented method of LPS deactivation utilises phospholipid transfer proteins and LBP. Phospholipid transfer proteins are plasma proteins with 24% sequence homology to LBP, which bind the lipid A portion of LPS and can transfer LPS from aggregates to high density lipoprotein particles, neutralizing its biological activity. These high density lipoprotein particles are then phagocytosed (Chaby, 2004; Hailman et al., 1996). Addition of phospholipid transfer proteins to blood prevents LPS-stimulated cytokine release (Hailman et al., 1996). As well as its ability to transfer LPS to CD14, LBP can also transfer LPS to high density lipoprotein particles (Wurfel et al., 1994). Phospholipid transfer proteins have been documented in the brain (Jiang and Zhou, 2006) but it is as yet unknown if LBP is present in the CNS.

Although cytokines can be degraded by bacterial proteases (Wilson et al., 1998) and in the periphery can be degraded by proteases released from mast cells (Zhao et al., 2005) there are currently no delineated endogenous spinal cord mechanisms for IL-10 degredation (Milligan et al., 2005a). Although a bolus intrathecal injection of IL-10 was found to have a half life of 2 hours in the spinal cord, the rate of IL-10 clearance fell within the range of cerebral spinal fluid production rates and hence this half life was attributed to bulk cerebral spinal fluid flow as opposed to breakdown or degredation (Milligan et al., 2005a).



Chapter 1

Figure 1.13 TLR4 signaling pathways in microglia and astrocytes. On binding of the MD2/TLR4 complex by LPS, two signaling pathways are activated. The first is a MAL/MyD88/TRAF pathway which leads to activation of the transcription factors NF-kB, p38 and JNK and results in upregulation of IL-10 and pro-inflammatory cytokine mRNA and protein levels. The second pathway is a TRAM/TRIF/TRAF pathway which leads to activation of IRF3 and NF-kB, resulting in upregulation of IFN-B, pro-inflammatory cytokine and IL-10 mRNA and protein. Question marks indicate processes which have been shown to occur in other cell types but have not been explored in glia. It is unclear whether MKK, p38, JNK and c/EBP activation occurs in the cytoplasm or nucleus. Abbreviations: c/EBP, CCAAT/enhancer binding protein; IFN-B, interferon-B; IKK, inhibitor of nuclear factor-kB kinase complex; IL-10, interleukin-10; IRF3, kinase: LPS, JNK. JUN N-terminal factory-3; regulatory interferon lipopolysaccharide; MAL, MyD88-adaptor-like; MD2, myeloid differentiation Ag; MKK, mitogen-activated protein kinase kinase; MyD88, myeloid differention 88: NF-kB, nuclear factor-kB; PIP₂, phosphatidylinositol bisphosphate; Sp1, Simian virus 40 promoter factor 1; TLR4, Toll-like receptor 4; TRAF, tumornecrosis-factor-receptor-associated factor; TRAM, TIR-containing adaptor molecule; TIR, Toll/interleukin-1 receptor; TRIF, TIR-domain containing adaptor inducing IFN-β. Adapted from Jerala (2007) and O'Neill and Bowie (2007).

1.4.2.3 Potential of spinal cord astrocytes and microglia to release IL-10 on TLR4 activation

As mentioned in section 1.4.1.3, a regional heterogeneity between spinal cord and cortical astrocytes has been described and it cannot be assumed that, because cortical astrocytes display TLR4-stimulated IL-10 release, spinal cord astrocytes will also display this. Additionally, there is a lack of studies comparing the responses of spinal cord and cortical microglial cells, hence it cannot be assumed that spinal cord microglia react to neurotransmitters in a similar fashion to cortical transmitters.

This raises the question of what evidence there is that spinal cord astrocytes and microglia can release IL-10 in reponse to LPS. Spinal cord astrocytes and microglia have some of the cellular machinery that is needed for TLR4-stimulated IL-10 release. There is evidence that spinal cord microglia and astrocytes express functional TLR4s. Cultured spinal cord astrocytes and microglia release substances in response to LPS, with astrocytes shown to release nerve growth factor (Pehar et al., 2004) and microglia shown to release TNF- α , IL-1 β and IL-6 (Ogata et al., 2003). Additionally, it is known that spinal cord astrocytes contain JNK and spinal cord microglia contain p38 (Svensson et al., 2003; Zhuang et al., 2006), which as detailed above are two key MAPKs involved in IL-10 transcription activation.

Given i) it has not been examined which cells release IL-10 in the spinal cord despite the possible anti-nociceptive impact of potential IL-10 release and ii) in consideration that there is evidence for the presence of some of the key proteins involved in TLR4-stimulated IL-10 release on spinal cord microglia and astrocytes, one objective of this thesis will be to examine if TLR4 stimulation can elicit IL-10 release from spinal cord microglia and astrocytes.

1.4.2.4 The potential of glutamate to be involved in IL-10 release from spinal cord microglia and astrocytes

As noted in section 1.3.1, when the TLR4 is being activated in chronic pain conditions, astrocytes and microglia are simultaneously exposed to many other neurotransmitters (Millan, 1999; Tanga et al., 2005). This suggests that to gain a full knowledge of the spinal sources of IL-10 it should be examined whether these substances elicit IL-10 release and/or affect the process of TLR4-stimulated IL-10 release if this process occurs. As time is restricted, the effects of every relevant substance can not be explored in the present research, however glutamate is an attractive starting point in the identification of further spinal sources of IL-10 and possible modulators of TLR4-stimulated IL-10 release. Glutamate shows potential for a number of reasons. Firstly, as discussed in section 1.3.1, glutamate is found in increased amounts in the spinal cord in chronic pain (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992). In conditions such as chronic pain where there is TLR4 activation and hence subsequent pro-inflammatory cytokine release, cytokines such

as TNF- α contribute to these increased levels of glutamate by stimulating glutamate release from astrocytes and microglia, and inhibiting astrocytic glutamate uptake (Bezzi et al., 2001; Takeuchi et al., 2006; Tilleux and Hermans, 2007). Secondly, spinal cord glia can respond to these increased levels of glutamate as they have glutamate receptors in their membranes. Spinal cord microglia have been shown to have NMDA receptors, AMPA receptors and mGluRs on their cell surface (Liu et al., 2006). Although the presence of kainate receptors has not been examined in spinal cord microglia, there is evidence for these receptors on cortical and hippocampal microglia (Christensen et al., 2006; Yamada et al., 2006). As discussed in section 1.4.1.3, spinal cord astrocytes express all subtypes of glutamate receptors on their cell surface. Hence when the TLR4 is being bound on these cells. glutamate receptors are also being bound, opening the possibility that glutamate may modulate any TLR4-stimulated IL-10 release. Furthermore, one outcome of the ligation of astrocytic and microglial glutamate receptors is the activation of key signaling molecules involved in IL-10 mRNA upregulation. NDMA elicits p38 activation in microglial cultures (Tikka and Koistinaho, 2001) and glutamate increases c/EBPß mRNA and protein levels in astrocytes (Yano et al., 1996). Both these proteins are involved in the ability of LPS to bring about IL-10 release (as discussed in section 1.4.2.2.2).

A final reason why glutamate is an attractive starting point in the search for possible IL-10 stimulating agents is that glutamate has a predominantly anti-inflammatory effect on astrocytes and activated microglia. Exposure of resting microglia to

glutamate results in pro-inflammatory outcomes such as TNF- α release and subsequent neurotoxicity (Noda et al., 2000; Taylor et al., 2005). In contrast, in activated microglia glutamate decreases microglial activation, as evidenced by a decrease in cell surface expression of the activation marker ED1. Additionally, activated microglia can be neurotoxic but the degree of neurotoxicity induced by these microglia is decreased if they are exposed to glutamate (Taylor et al., 2003). Further examples of the anti-inflammatory nature of glial glutamate exposure can be seen when astrocytes are exposed to glutamate. When neurons are exposed to high levels of an excitotoxin such as NMDA, they undergo apoptosis or programmed cell death. If these neurons are exposed to astrocytes that have been stimulated by glutamate, the amount of NMDA-elicited neuronal death is greatly decreased (Bruno et al., 1998; D'Onofrio et al., 2001). Similarly, astrocytes exposed to LPS cause neuronal death but this neuronal death can be greatly decreased if- astrocytes are co-exposed to glutamate (Zhou et al., 2006). Furthermore, one outcome of the exposure of astrocytes to cytokines is the induction of nitric oxide synthase, an enzyme that is involved in the production of the reactive oxygen species and inflammatory mediator nitric oxide. If glutamate is added to astrocytes simultaneous to cytokine exposure, the extent of nitric oxide synthase induction is reduced (Levy and Zochodne, 1998; Murphy et al., 1995). As the release of IL-10 is an anti-inflammatory action and glutamate often has an antiinflammatory effect on these glial cells, it suggests glutamate is a promising potential source of IL-10 or a modulator of TLR4-stimulated IL-10 release.

84

In summary, it can be seen that in chronic pain there is increased levels of glutamate and glutamate receptors are present on spinal cord glia to bind this glutamate whilst the TLR4 is being bound. Given this, and also that the ligation of glutamate receptors results in anti-inflammatory effects and activation of key molecules involved in the production of IL-10, it is conceivable that glutamate may elicit IL-10 release from spinal cord glia or may modulate any TLR4-stimulated IL-10 release. For these reasons, *the final objective of the present research will be to examine if glutamate can elicit IL-10 release and modulate any TLR4-stimulated IL-10 release from spinal cord astrocytes and microglia*.

1.5 AIMS AND OBJECTIVES

Given that ATP in the spinal cord has been shown to play a role in bringing about the symptoms of chronic pain and IL-10 in the spinal cord has been shown to play a role in bringing about the alleviation of these symptoms and considering that the spinal sources of these remain largely unexplored, this work aims:

i) to investigate whether glutamate can stimulate ATP release from spinal cord astrocytes in Chapter 3 and if it can,

ii) to investigate whether SP can either cause ATP release itself or modulate glutamate-stimulated ATP release from spinal cord astrocytes in Chapter 3;

iii) to investigate whether stimulation of the TLR4 on spinal astrocytes results in IL-10 release in Chapter 4;

iv) to investigate whether glutamate can either cause IL-10 release itself or modulate any TLR4-activated IL-10 release from spinal cord astrocytes in Chapter 4; -

v) to investiage whether stimulation of the TLR4 on spinal microglia results in IL-10 release in Chapter 5;

vi) to investigate whether glutamate can either cause IL-10 release itself or modulate any TLR4-activated IL-10 release from spinal cord microglia in Chapter 5.

If any of the above processes are found to occur, mechanisms involved in these will be explored.

Methods and materials

CHAPTER 2

METHODS AND MATERIALS

Contents

2.1 OVERVIEW	88
2.2 PREPARATION OF GLIAL CELL CULTURES	89
2.2.1 Initial culture of mixed spinal cord glial cells	89
2.2.2 Purification of spinal cord microglial cell cultures	90
2.2.3 Purification of spinal cord astrocyte cell cultures	91
2.2.4 Culture of cortical astrocytes	92
2.3 ONLINE BIOLUMINESCENCE	93
2.3.1 Preparation	93
2.3.2 ATP release measurement	94
2.3.3 Controls	95
2.3.4 Construction of a standard curve	95
2.3.5 Analysis of data	96
2.4 PATCH-CLAMP RECORDING	98
2.4.1 Identification of astrocytes	98
2.4.2 Patch-clamp recordings	99
2.4.3 Analysis	99
2.5 FLUORESCENCE IMMUNOCYTOCHEMISTRY	100
2.5.1 Fixation, blockade of non-specific binding and permeabilisaton	100
2.5.2 Antibody incubation and coverslip mounting	101
2.5.3 Fluorescence microscopy	102
2.5.4 Analysis	102
2.6 SANDWICH ENZYME-LINKED IMMUNOSORBENT ASSAY	104
2.6.1 Sample preparation	105
2.6.2 Assay	105
2.6.3 Analysis	107
2.7 LACTATE DEHYDROGENASE TOXICOLOGY ASSAY	109
2.7.1 Sample preparation	109
2.7.2 Assay	110
2.7.3 Analysis	110
2.8 REVERSE TRANSCRIPTION-POLYMERASE CHAIN REACTION	111
2.8.1 Extraction of total RNA	111
2.8.2 Reverse transcription of RNA	112
2.8.3 PCR	112
2.8.4 Analysis	113
2.9 PHARMACOLOGICAL MATERIALS	115
2.9.1 Agonists	115
2.9.2 Inhibitors	116
2.9.3 Primary antibodies	118
2.9.4 Secondary antibodies	119

2.1 OVERVIEW

A number of methods have been employed to explore and characterise the release of ATP and IL-10 from spinal cord microglia and astrocytes. First, tissue culture was used to isolate and purify spinal cord microglia and astrocytes, and also cortical astrocytes. ATP release was measured by online-bioluminescence. Membrane currents induced by agonist exposure were measured by electrophysiology. The presence of glutamate receptor subtypes, TLR4s and cell specific markers on cultured glia were probed with immunocytochemistry. Release of IL-10 was detected by enzyme-linked immunosorbent assay (ELISA). Cell membrane integrity was investigated with a lactate dehydrogenase assay and finally, levels of IL-10 mRNA were quantified with a reverse transcription-polymerase chain reaction (RT-PCR).

All experiments were approved by the University of Sydney Animal Ethics Committee and were conducted in accordance with the Australian Government National Health and Medical Research Council's Code of Practice for the Care and Use of Animals for Scientific Purposes. Chemicals were purchased from Sigma-Aldrich (St. Louis, MO, USA) unless otherwise specified and experiments were performed at room temperature $(22 - 24^{\circ}C)$ unless otherwise specified.

2.2 PREPARATION OF GLIAL CELL CULTURES

Tissue culture is the process of sterile dissection of tissue from a specimen, dissociation of that tissue, propagation of dissociated cells in an artificial growth environment and removal of unwanted cells (Phelan, 2006). Tissue culture is a method useful in examining the function of an isolated cell type which would otherwise be impossible *in situ* due to the presence of other cell types.

2.2.1 Initial culture of mixed spinal cord glial cells

Mixed spinal cord glial cells were cultured according to Cole and de Vellis (1997) with modifications. Whole spinal cords were removed from 0-2 day old Sprague-Meninges and peripheral nerves were removed and cords were Dawley rats. incubated in porcine trypsin (0.125%) dissolved in Hank's balanced salt solution (HBSS, in mM: 136.9 NaCl, 5.4 KCl, 4.2 NaHCO₃, 0.4 KH₂PO₄, 0.3 Na₂HPO₄, 5.6 D-glucose, pH 7.4) for 20 min at 37°C. Subsequently the cords were washed twice in 2 ml Dulbecco's modified Eagle's medium (DMEM) supplemented with 10% cosmic calf serum (Hyclone, Logan, UT, USA) and 1% penicillin/ streptomycin/glutamine (Invitrogen, Carlsbad, CA, USA; referred to as DMEM+) to inactivate the trypsin. Cells were then dissociated by mechanical trituration and were plated onto poly-D-lysine (20 µg/ml)-coated flasks at 37°C for 4 h. Media was aspirated to remove detached cells and flasks washed twice with 2 mlDMEM+. Glial cells were maintained at 37°C in 5% CO₂ in DMEM+ and culture medium was replaced with DMEM+ every three to four days.

2.2.2 Purification of spinal cord microglial cell cultures

Microglia were purified according to Liu et al (2006) with modifications. To isolate microglia from mixed spinal cord glial cultures, fourteen days after initial plating, flasks containing mixed glial cells were shaken at 400 rpm at 37°C for 40 min on a Unimax100 shaker and incubator (Heidolph, Schwabach, Germany). Microglia and oligodendrocytes do not attach to the base of the flask as strongly as astrocytes and consequently detach upon prolonged shaking (Cole and deVellis, 1997). Medium containing detached glial cells was removed and centrifuged at 1500 rpm for 5 min. The supernatant medium layer contained substances released by cultured glia and this glial conditioned medium supernatant was kept at 37°C. The cell pellet layer was washed twice with 1 ml DMEM+ and then plated at a density of 1 x 10⁵ cells per well in a plastic 24-well plate (Sarstedt, Nümbrecht, Germany), at a density of 1 x 10⁵ cells per coverslip on a 13 mm diameter circular glass coverslip (Menzel-Glaser, Braunschweig, Germany) or 1 x 10⁶ cells per dish on a 35 mm diameter circular plastic Petri dish. Cell-adherent surfaces of these were pre-coated with 20 µg/ml poly-D-lysine. After 15 min, any unattached cells (predominantly oligodendrocytes) were removed and the remaining plated cells were maintained in a 1:1 mixture of DMEM+ and glial conditioned medium at 37°C in 5% CO₂. The purity of cultures was verified with immunocytochemistry. Cells were used for experiments 1 day after initial plating.

2.2.3 Purification of spinal cord astrocyte cell cultures

After removal of detached cells in section 2.2.2, flasks containing adherent cells were maintained in DMEM+ at 37°C in 5% CO₂ for a further 2 days. After in vitro day 16 the numbers of remaining microglia and oligodendrocytes were further reduced by replacing DMEM + with sorbitol medium (DMEM without glucose, 10% horse serum (Trace Bioscientific, Castle Hill, Australia), 1% penicillin/ streptomycin/glutamine and 25 mM sorbitol. Oligodendrocytes and microglia do not have aldose reductase and sorbitol dehydrogenase and consequently cannot use sorbitol for energy. Accordingly if sorbitol is substituted for glucose in the culture medium, oligodendrocytes and microglia die off, leaving astrocytes which have the enzymes necessary to use sorbitol for energy (Wiesinger et al., 1991). Twenty days after initial plating, flasks were shaken by the Unimax100 shaker and incubator at 400 rpm and 37°C for 16 h. Twenty-one days after initial plating, the age at which there becomes only one predominant (> 90%) subtype of cultured spinal cord astrocytes (Black et al., 1993), remaining astrocytes were washed twice with 3 ml HBSS and removed from the flask after incubation for 5 min at 23°C in 0.125% porcine trypsin. Suspended cells were centrifuged at 1500 rpm for 5 min and washed twice with 1 ml DMEM+. Subsequently cells were plated either on 13 mm glass coverslips, the base of wells in a plastic 24-well plate or on the base of a plastic 35 mm diameter Petri dish. Cells were plated at a density of 1 x 10⁵ cells per well or coverslip, or 6.77 x 10⁵ cells per Petri dish. Cell-adherent surfaces were precoated with 20 µg/ml poly-D-lysine. The purity of cultures was verified by immunofluorescent staining for markers of neurons and other glial cell classes. The

heterogeneity of the astrocyte population was examined based on morphological characteristics and over 90% of astrocytes resembled spinal cord astrocyte subtype 1a as described by Black et al. (1993). It is unclear whether these astrocytes are grey matter or white matter astrocytes. Cells were used for experiments 1-2 days after initial plating.

2.2.4 Culture of cortical astrocytes

The cortices of 0–2 day old Sprague-Dawley rats were removed and meninges were separated away. Preparations of mixed cortical glial cells were then prepared in the same manner as for mixed spinal cord glial cells in section 2.2.1. Cortical astrocytes were purified and plated in the same manner as for spinal cord astrocytes in section 2.2.3 and were used for experiments 1 to 2 days after initial plating.

2.3 ONLINE BIOLUMINESCENCE

Online bioluminescence is a technique that allows the real-time measurement of ATP release from cells. Using a luciferin-luciferase enzymatic ATP assay mix (firefly luciferase, luciferin, MgSO₄, ethylenediaminetetraacetic acid, dithiothreitol and bovine serum albumin (BSA) in a Tricine buffer according to manufacturer's proportions, Sigma-Aldrich) developed by Taylor et al. (1998), for every molecule of ATP that is evolved in solution, a photon of light is produced according to the equations:



Consequently, the amount of ATP afforded by cells can be indirectly calculated by measuring in real-time the number of photons produced by cells incubated in the luciferin-luciferase assay with a photomultiplier.

2.3.1 Preparation

Coverslips prepared in section 2.2 were placed in HEPES buffered salt solution (in mM; 140 NaCl, 5 KCl, 10 HEPES, 1 CaCl₂, 1 MgCl₂, pH 7.4) for 5 min to remove DMEM+. Excess solution was removed from cell-free sides of the coverslip and it was placed on an 18 mm diameter round plastic photomultiplier stage constructed by cutting individual wells out of a plastic 24-well plate. Eighty µl of ATP assay mix (1 mg/ml; Sigma-Aldrich) dissolved in HEPES buffered salt solution was
immediately added to the coverslip. The photomultiplier stage was placed inside a light-impermeable chamber constructed by the University of Sydney Physics Workshop (Camperdown, Australia).

2.3.2 ATP release measurement

Photons were counted with a P30CWAD5-45 Photodetector Package (Electron Tubes, Coventry, UK) and displayed using an RS232 Photon Counting program (Electron Tubes). Photons were collected every second. Initial photon count was high as there was an imbalance between the basal amount of ATP produced by cells and the amount of ATP consumed by the ATP assay mix and the ecto-ATPases. Sixty to eighty min later the amount of ATP produced by cells and the amount of ATP consumed by the assay and ecto-ATPase reached equilibrium, illustrated by the photon count readings not varying by more than 10%. When this stable photon count extended for 10 min, the average photon count over this period of time was recorded as baseline. After the baseline was reached, recording of photon counts was stopped and agonists of interest were applied at 10% of the volume of liquid on the coverslip. Recording was immediately recommenced (within 20 s) after addition of the chemicals. Chemicals were not washed out during experiments. At the end of each experiment, 1% Triton X-100 (Bio-Rad, Hercules, CA, USA) was added to lyse cells, allowing an estimate of the number of cells on the coverslip. Each experimental condition was tested at least three times and in experiments using antagonists, all antagonists were applied 1 h prior to agonist addition.

2.3.3 Controls

To investigate the effects of chemicals and their solvents (HEPES buffered solution, DMSO, NaOH, HCl and acetonitrile + trifluoroacetic acid as per section 2.9) on the intrinsic luciferin-luciferase activity, chemicals or solvents were added to non-cell coated coverslips. To examine the effect of solvents on glutamate-stimulated ATP release, solvents were added to cell-coated coverslips after baseline ATP release had been reached. The effect of antagonists on the baseline ATP release by cells was investigated by addition of the antagonist and subsequent recording of photon release. The possibility that the mechanical sensitivity of cells caused any observed ATP release was also investigated by addition of HEPES without chemicals to cells. These treatments did not elicit ATP release.

2.3.4 Construction of a standard curve

Raw data was obtained showing the change in photon count every second. To enable the conversion of photon counts to moles of ATP, a standard curve was constructed. Eighty μ l of ATP assay mix were added to a glass coverslip without cells on it. The baseline photon emission was recorded from this ATP assay mix and then ten known amounts of ATP between 3.3×10^{-15} and 1×10^{-10} moles were added to the ATP assay mix in increasing concentrations. The baseline photon emission was subtracted from the peak photon count values obtained from each amount of ATP, and the points plotted on a log scale. The gradient was computed and the photon counts for all experiments were divided by the reciprocal of this gradient value to enable conversion of photon count readings to moles of ATP. A

standard curve was constructed for each batch, both at the start and towards the end of use of each batch of luciferase to identify the degree of assay activity loss over time. No loss over time was found. An example of a standard curve is displayed in Figure 2.1.

2.3.5 Analysis of data

ATP release values were analysed to find the peak increase in ATP release relative to baseline (referred to as 'relative peak ATP level'). Results were expressed as mean relative peak ATP level \pm SEM. Means were analysed for statistically significant differences with an unpaired Student's two-sample two-tailed *t*-test. Statistical significance was considered at p < 0.05.



Figure 2.1 ATP standard curve. A representative image of the variation of photon counts on application of known amounts of ATP.

2.4 PATCH-CLAMP RECORDING

The opening of ion channels and receptors on a cell can lead to the flow of ions into and out of a cell. Whole cell patch-clamp recording is a technique that can be used to measure cellular currents set up by such ion flows. It involves the placement a glass pipette with a very small diameter tip (often about $1 - 2 \mu m$) on the surface of a cell. Negative pressure is applied to the electrode interior resulting in the formation of a seal between the pipette and cell. This seal has a resistance in the order of magnitude of gigaohms. Exposure to the antibiotic nystatin perforates the patch of membrane encircled by the electrode tip. On perforation of this patch of membrane, ions in the intracellular fluid of the pipette can exchange with ions in the cytoplasmic fluid of the cell, allowing the pipette to sense changes in voltage in the patched cell (Jackson, 1997).

2.4.1 Identification of astrocytes

Astrocytes were identified according to the morphological features of type A astrocytes, illustrated in Figure 1.3C, such as their large flat cell body in comparison to the small, rounded cell bodies of microglia and oligodendrocytes. The morphology of these astrocytes corresponded with the morphology of cells that stained positive for glial fibrillary acidic protein (GFAP) in immunocytochemistry experiments (see section 2.5) and differed from the smaller CDllb positive microglia.

2.4.2 Patch-clamp recordings

Resistances of pipettes were $4 - 8 \text{ M } \Omega$ when the pipette was filled with internal pipette solution (in mM ; 145 CsCl, 1 MgCl₂, 10 EGTA, 10 HEPES and 250 µg/mL nystatin, pH 7.3). The external bath solution consisted, in mM , of 140 NaCl, 5 KCl, 1 CaCl₂, 1 MgCl₂, 10 HEPES with pH 7.4. Mg²⁺ was omitted from the external bath solution when NMDA was used as an agonist. Whole-cell currents in astrocytes were recorded using a Multiclamp 700A amplifier (Axon Instruments, Sunnyvale, CA, USA) under voltage-clamp mode at a holding membrane potential of -60 mV. The currents were sampled on-line using a Digidata 1322A interface and pClamp8 software (Axon Instruments). Glutamate, its analogs and SP were applied through pressure ejection via a picosprizer (General Valve, Fairfield, NJ, USA).

2.4.3 Analysis

The number of cells that displayed inward currents in response to agonists was identified with pClamp8. For those cells that displayed currents, the peak amplitude of the currents were recorded and averaged. Means were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant and values were expressed as mean \pm SEM.

2.5 FLUORESCENCE IMMUNOCYTOCHEMISTRY

Fluorescence immunocytochemistry allows the identification and localization of cellular proteins through the attachment of fluorescent molecules to proteins of interest. In indirect fluorescence immunocytochemistry, cells are fixed, non-specific binding of antibodies is blocked and cells are permeabilised. Cells are then incubated in an antibody constructed to bind to the protein of interest. This is the primary antibody. A secondary antibody conjugated to a fluorescent protein is bound to the primary antibody. When mounted on a microscope and exposed to a wavelength of light that excites the fluorescent protein, filters can be used to visualize light emitted by the fluorescent molecule. Different primary and conjugated secondary antibody pairs can be used in the one preparation allowing the simultaneous investigation of several proteins (Goldstein and Watkins, 2008).

2.5.1 Fixation, blockade of non-specific binding and permeabilisation

Cells grown on coverslips described in section 2.2 were washed twice for 5 min each wash with the HEPES buffered salt solution described in section 2.3.1. Cells on coverslips were then fixed with 2% paraformaldehyde solution (BDH chemicals, Kilsyth, Australia) in phosphate buffered saline (PBS in mM; 136.89 NaCl, 2.68 KCl, 10.14 Na₂HPO₄, 1.76 KH₂PO₄, pH 7.4) for 10 min. Coverslips were washed twice for 5 min each wash with PBS and then incubated in PBS containing 1% BSA and 0.05% saponin for 30 min to block non-specific binding and permeabilise cells.

2.5.2 Antibody incubation and coverslip mounting

Cells on coverslips were then incubated either overnight at 4°C (in Chapter 3) or for 2 h at room temperature (in Chapters 4 and 5) with primary antibodies diluted in PBS containing 1% BSA and 0.05% saponin. Primary antibodies were washed out three times for 10 min each wash with PBS and then cells were incubated in secondary antibodies diluted in PBS containing 1% BSA and 0.05% saponin for 1 h at room temperature. Secondary antibodies were washed out 3 times with PBS for 10 min each wash. Excess moisture was removed from the back of coverslips and they were mounted on rectangular glass coverslips using ProLong Gold anti-fade mounting media with 4',6-diamidino-2-phenylindole (DAPI) (Invitrogen). Slides were sealed with nail polish and subsequently stored at 4°C until viewing.

To verify the results obtained using the above method, for every primary antibody used, cells were stained with and without the primary antibody and no positive staining was seen when the primary antibody was omitted. In double-labeling experiments, cells were also incubated in each single antibody to verify there was no cross-reactivity between primary antibodies. Patterns of antibody staining identified when cells were incubated in each single antibody matched patterns of antibody staining seen in double-labeling experiments. Furthermore, for Chapter 3, to verify the results obtained using the above method, experiments probing NR1 and GluR4 subunits were repeated with one of the following changes per experiment: omission of DAPI from the mounting media; substitution of the step incubating cells in saponin and BSA with three steps incubating cells in 1% Triton X-100 which was washed out three times with PBS then a 1 hr incubation in 1% BSA; use of a different rabbit anti-rat NR1 antibody (Chemicon, Temecula, CA, USA; 1:50–1:200); incubation of cells in the rabbit anti-rat glutamate receptor antibodies prior to fixation of cells with paraformaldehyde solution. Results using these different methods were consistent.

2.5.3 Fluorescence microscopy

Fluorescence was visualised with an Axioplan2 upright microscope (Zeiss, Jena, Germany) for experiments described in Chapter 3 and an Axio200M inverted microscope (Zeiss) for experiments described in Chapters 4 and 5. Alexa Fluor 488 was excited at 495 nm and emission was collected at 594 nm. Alexa Fluor 594 was excited at 590 nm and emission was collected at 617 nm. DAPI was excited at 345 nm and emission was collected at 458 nm. Images were acquired with an AxioCam HR digital monochrome charge-coupled device camera (Zeiss) using Axiovision 3.1 software (Zeiss). Experiments with each antibody were performed on at least three different batches of cultures and were repeated at least twice on each of the batches. For each coverslip at least three fields of view were imaged. No autofluorescence of primary antibodies or non-specific binding of secondary antibodies was observed.

2.5.4 Analysis

For analysis of experiments described in Chapters 4 and 5, a constant exposure time was used to acquire images assessed for brightness of TLR4 staining. Using Image J (U.S. National Institutes of Health, Bethesda, MD, USA), an arbitrary brightness threshold was applied to these images and the number of cells with regions of brightness above this threshold as a percentage of total cells was counted per field of view. This percentage was averaged over a minimum of three different fields of view per condition and mean percentages were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Percentages were expressed as mean \pm SEM.

2.6 SANDWICH ENZYME-LINKED IMMUNOSORBENT ASSAY

The release of protein by cells can be quantified using an enzyme-linked immunosorbent assay (ELISA). In a sandwich ELISA, a multi-well plate is coated with an antibody specifically designed to bind to the protein of interest. This antibody is termed the capture antibody. Wells are incubated in samples containing unknown amounts of the protein of interest, which bind the capture antibody. Subsequently, a detection antibody is added to the plate, which is specifically designed to bind to a different segment of the protein of interest to that of which the capture antibody is bound. This detection antibody has a biotin molecule added to it, the purpose of which is to react strongly with streptavidin conjugated to horseradish peroxidase. Effectively, the protein of interest is now labeled with horseradish peroxidase. The amount of horseradish peroxidase can be indexed by exposing the wells to hydrogen peroxide and tetramethylbenzidine (TMB). Horseradish peroxidase degrades hydrogen peroxide resulting in the oxidation of TMB. Oxidised TMB is a dark blue colour. The production of oxidized TMB can be ceased by addition of sulfuric acid, which turns the colour of oxidized TMB yellow. A spectrophotometer can be used to quantify the intensity of yellow in the wells, which can be converted to an amount of protein by construction of a standard curve (Hornbeck et al., 2001).

2.6.1 Sample preparation

Unless otherwise stated, 1 day after plating in 24-well plates, as per section 2.2, media was removed from wells and cells were incubated in either fresh DMEM+, 1 μ g/ml LPS from *Escherecia Coli 055:B5* in DMEM+ or 1 mM glutamate in DMEM+ for 8 h. These solutions were then removed and cells were incubated in 500 μ l of agonists and/or antagonists dissolved in DMEM stock with 2% BSA (DMEM/BSA) for 16 h. BSA was added to block non-specific binding to IL-10 capture antibodies (described in section 2.6.2). Plates were sealed on three of four sides to minimise loss of solution but to still allow gas exchange. Medium was then collected and centrifuged.

2.6.2 Assay

Ninety-six-well clear polystyrene microplates (R&D Systems, Minneapolis, MN, USA) were coated in anti-rat IL-10 monoclonal antibody (1:250; BD Biosciences, San Jose, CA, USA) diluted in a coating buffer consisting (in mM; 83 Na₂HPO₄, 134 NaH₂PO₄, pH 6.5) at 4°C overnight. Wells were washed three times for 5 min each wash with 0.05% Tween® 20 in PBS and incubated for 1 h in 1 % BSA in PBS. Wells were then washed for three periods of 5 min with 0.05% Tween® 20 in PBS. Immediately after centrifugation, medium samples were added to wells along with six standard concentrations of IL-10 (BD Biosciences) diluted in DMEM/2% BSA. After a 2 h incubation at room temperature wells were washed five times for 5 min each wash with 0.05% Tween® 20 in PBS. Wells were incubated for 2 h in a biotinylated anti-rat IL-10 antibody (1:250 diluted in 1% BSA in PBS, BD

Biosciences) and wells were washed 5 times for five min each wash with 0.05 % Tween® 20 in PBS. A streptavidin-horseradish peroxidase conjugate (1:250 diluted in 1 % BSA in PBS, BD Biosciences) was added to wells for 20 min and wells were washed 5 times for 5 min each wash with 0.05% Tween® 20 in PBS. Wells were incubated in a 1:1 mix of hydrogen peroxide and TMB for 20 min and 1 M sulfuric acid was added to stop the reaction. Absorbance of plates at 450 nm was read within 20 min of sulfuric acid addition using a Fluostar Galaxy Multiplate reader (BMG Labtechnologies, Offenberg, Germany) and wavelength correction was performed at 550 nm. A standard curve was constructed for each plate and was used to convert absorbance measurements for each condition into IL-10 concentration values. An example of a standard curve can be seen in Figure 2.2. Each experimental condition was tested on at least three different culture batches and at least two repetitions of measurement were made per batch giving a minimum n = 6, apart from select conditions in the time- and dose-response curves where n = 4.

Inhibitors were added 30 min prior to addition of agonists and left in for the duration of agonist exposure, in experiments involving inhibitors. To examine the effect of solvents on agonist-stimulated IL-10 release, solvents were added 30 min prior to addition of agonists. Solvents (dimethyl sulfoxide and NaOH as per section 2.9) at the final concentration used did not effect agonist-stimulated IL-10 release.

2.6.3 Analysis

A standard curve was constructed for each plate based on the absorbance readings of six known concentrations of IL-10. Absorbance readings of unknown samples were converted to concentrations of IL-10 using the reciprocal of the gradient of the standard curve. An example of a standard curve can be seen in Figure 2.2. Values in time- and concentration-dependency experiments and agonist and antagonist experiments were normalized to the average level of LPS-stimulated IL-10 release using the LPS-stimulated IL-10 release value for that batch. In most cases, results were presented as mean concentration of IL-10 (pg/ml) \pm SEM. However, to enhance description of the time-dependency of the glutamate effect and the ability of glutamate receptor agonists to elicit IL-10 release, each repeat in these experiments was analysed according to the following statement:

Ratio of agonist effect = <u>IL-10 release stimulated by LPS + agonist</u>, IL-10 release stimulated by LPS

and results were reported as mean ratio \pm SEM. To enhance description of the ability of glutamate receptor antagonists to effect LPS + glutamate-stimulated IL-10 release, each repeat in these experiments was analysed according to the following statement:

Ratio of antagonist effect = $\underline{IL-10}$ release stimulated by LPS + Glu + antagonist , IL-10 release stimulated by LPS + antagonist

and results were presented as a mean ratio \pm SEM. Statistically significant differences between means were analysed with a one-tailed Student's two-sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant.

Chapter 2



Figure 2.2 IL-10 standard curve. A representative image of the variation of absorbance on application of known amounts of IL-10.

2.7 LACTATE DEHYDROGENASE TOXICOLOGY ASSAY

When exposed to a toxic substance, cell membrane integrity is compromised leading to the release of cytoplasmic contents into the extracellular space. Lactate dehydrogenase (LDH) is a cytoplasmic enzyme which catalyses the conversion of nicotinamide adenine dinucleotide (NAD⁺) to its reduced form, NADH. The quantity of LDH in the extracellular medium as a percentage of the quantity of LDH in the intracellular medium is reflective of cell membrane integrity and can be quantified by exposure of LDH to NAD⁺ (Decker and Lohmann-Matthes, 1988; Legrand et al., 1992). In the presence of LDH, NAD⁺ is reduced to NADH. In the presence of NADH, flavin adenine dinucleotide (FAD²⁺) is reduced to FADH₂. If a tetrazolium dye is present, FADH₂ can supply it with electrons and the reaction can be stopped with acid. The intensity of the colour produced can be measured spectrophotometrically.

2.7.1 Sample preparation

Cells were prepared and incubated as described in sections 2.2 and 2.6.1. When medium was collected for ELISA experiments, leftover medium was retained and stored at -80°C for up to 2 weeks. Additionally, to quantify total intracellular LDH levels for each batch, 1% Triton X-100 (Bio-Rad, Hercules, CA, USA) was added to the medium to lyse cells. After 5 min incubation, medium was collected and was stored at -80°C. Medium was then defrosted and used to quantify extracellular LDH for select incubation conditions of ELISA experiments.

2.7.2 Assay

Membrane integrity was assessed with a *In Vitro* Toxicology Assay Kit (Sigma-Aldrich) as per the manufacturer's protocol. Briefly, 75 μ l of medium per condition was added to 150 μ l of LDH Assay Mixture (Sigma-Aldrich) in separate wells of a 96-well plate. After 20 min incubation, 22.5 μ l of 1 N HCl was added to stop the reaction. Within 30 min, absorbance was measured at 490 nm with a Fluostar Galaxy Multiplate reader (BMG Labtechnologies) and wavelength correction was performed at 650 nm.

2.7.3 Analysis

The absorbance reading corresponding to each extracellular LDH sample was divided by the absorbance reading for the corresponding total intracellular LDH sample to obtain a percentage of total LDH released per condition in accordance with Dringen and Hamprecht (1996). Statistically significant differences between means were analysed with a one-tailed Student's two-sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant.

2.8 REVERSE TRANSCRIPTION-POLYMERASE CHAIN REACTION

DNA is transcribed into mRNA in the nucleus of cells. mRNA is transported into the cytoplasm where it is translated into protein. This process involves association of the mRNA with ribosomes (constructed of proteins and rRNA) as well as association with amino-acid bound tRNA. The level of expression of mRNA in a cell can be indexed with a reverse transcription-polymerase chain reaction (RT-PCR). Total RNA, consisting of mRNA, rRNA and tRNA is extracted from cells and is reverse transcribed to cDNA with the enzyme reverse transcriptase. A polymerase chain reaction (PCR) is then used to amplify the cDNA. The real time production of cDNA can be indexed using Sybr Green I, a dye which binds to double stranded DNA and when bound can be excited at 488 nm, consequently emitting at 522 nm (Nolan et al., 2006; Beverly, 2001)

2.8.1 Extraction of total RNA

Total RNA was extracted from cells plated onto Petri dishes in section 2.2.2 and 2.2.3 using a GenElute^M Mammalian Total RNA Miniprep Kit (Sigma-Aldrich), according to the manufacturer's protocol. Briefly, on ice, a 2-mercaptoethanol (1%)-containing lysis buffer was used to lyse cells. The lysate was filtered and 70% ethanol added to precipitate RNA. The lysate/ethanol mixture was added to a binding column to extract the RNA. The sample was treated with amplification grade DNase I (Invitrogen) for 15 min at room temperature to remove any contaminating DNA before purified RNA was eluted from the binding column. The

quantity of isolated RNA was determined spectroscopically using a NanoDrop ND-1000 Spectrophotometer (NanoDrop Technologies, Wilmington, DE, USA).

2.8.2 Reverse transcription of RNA

Extracted total RNA was reverse transcribed to cDNA using a SuperScript^M III First-Strand Synthesis System (Invitrogen) and oligo(dT)₂₀ primer (Invitrogen) according to the manufacturer's protocol. Briefly, a mixture of 1 µl oligo(dT)₂₀ (50 µM), 1 µl 2'-deoxynucleoside 5'-triphosphate (dNTP) mix (10 mM) and 40 ng astrocytic RNA (Chapter 4) or 3 ng microglial RNA (Chapter 5) extracted as per section 2.8.1, was made up to 10 µl with RNAse free water and was thermally denatured at 65°C for 5 min and placed on ice for at least 1 min. A reverse transcriptase master mix containing 4 µl 5x first strand buffer, 1 µl dithiothreitol (0.1 M), 2 µl MgCl₂ (25 mM), 1 µl RNaseOUT^M (40 U/µl) and 1 µl SuperScript^M III reverse transcriptase (200 U/ml) was made up. Residual DNA contamination was tested for by replacing the reverse transcriptase enzyme with RNAse free water. The master mix was added to the mixture on ice and the new mixture was heated at 50°C for 50 min to synthesise cDNA in a Hybaid Gradient PCR machine (Hybaid Limited, Hampshire, UK). The reaction was terminated by heating at 70°C for 15 min. The mixture was chilled on ice and then stored at -20°C until the PCR.

2.8.3 PCR

Aliquots of first strand cDNA (40 ng) were amplified in 10 µL of Platinum SYBR-Green qPCR SuperMix-UDG (Invitrogen) with 0.8 µL of forward primer and

0.8 μ L of reverse primer made up to a total volume of 20 μ L with diethylpyrocarbonate-treated H₂O. Primers were as follows:

- 18S forward primer: 5'-GCCGCTAGAGGTGAAATTCTTG-3';
- 18S reverse primer: 5'-AAAACATTCTTGGCAAATGCTTT-3';
- IL-10 forward primer: 5'-TGCCTTCAGTCAAGTGAAGAC-3';
- IL-10 reverse primer: 5'-AAACTCATTCATGGCCTTGTA-3'.

All were synthesized by Invitrogen.

PCR was carried out using the Rotor-Gene 3000 system (Corbett Life Science, Sydney, Australia). Incubation conditions were as follows: 2 min at 50°C, 2 min at 95°C, then thermo-cycling for 40 cycles of 20 s at 95°C, 20 s at 60°C then 20 s at 72°C. Melt curves were constructed at the end of the PCR run and any sample with a melt curve with multiple peaks suggesting contamination was discarded.

2.8.4 Analysis

The threshold cycle number, crossing point at threshold and reaction efficiencies of each condition in a reaction were identified with Rotor-Gene 3000 v3.1 software (Corbett Life Science). Relative quantification was carried out using the relative expression software tool (REST; Pfaffl et al., 2002) whereby IL-10 mRNA expression stimulated by the agonists glutamate, LPS or LPS + glutamate was standardised by the level of expression of the house-keeping gene 18S rRNA and compared to IL-10 mRNA expression of unstimulated cells according to the equation:

Relative mRNA expression induced by an agonist = $\frac{(E_{IL-10 \text{ agonist}})^{CP_{IL-10 \text{ unstimulated}} - CP_{IL-10 \text{ agonist}}}{(E_{18S \text{ agonist}})^{CP_{18S \text{ unstimulated}} - CP_{18S \text{ agonist}}}$

where $E_{IL-10 \text{ agonist}}$ is the efficiency of the PCR performed on IL-10 cDNA derived from cells stimulated by an agonist, $E_{18S \text{ agonist}}$ is the efficiency of the reaction performed on 18S cDNA taken from cells stimulated by an agonist; $CP_{IL-10 \text{ unstimulated}}$ is the crossing point or cycle threshold of the PCR performed on IL-10 cDNA derived from unstimulated cells; $CP_{IL-10 \text{ agonist}}$ is the crossing point or cycle threshold of the PCR performed on IL-10 cDNA derived from cells stimulated by an agonist, $CP_{18S \text{ unstimulated}}$ is the crossing point or cycle threshold of the PCR performed on 18S cDNA derived from unstimulated cells; and $CP_{18S \text{ agonist}}$ is the crossing point or cycle threshold of the PCR performed on 18S by an agonist.

At least three batches were examined per condition and mean relative mRNA levels were determined. M ean levels were compared using a one-tailed Student's twosample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Values are expressed as mean relative IL-10 mRNA expression ± SEM.

2.9 PHARMACOLOGICAL MATERIALS

2.9.1 Agonists

Agonists used in experiments were: neurotransmitters glutamate (1 µM - 10 mM ; Chapters 3,4 and 5) and SP (3 - 300 µM; Chapter 3); neurokinin-1 receptor agonist [Sar⁹, Met(O₂)¹¹]-substance P (100 µM; Chapter 3); neurokinin-2 receptor agonist [β-Ala⁸]-neurokinin A Fragment 4-10 (64 µM; Chapter 3); neurokinin-3 receptor agonist succinyl-[Asp⁶, N-Me-Phe⁸]-Substance P Fragment 6-11 (senktide: 100 µM; Chapter 3); glutamate receptor subtype agonist kainate (Tocris Bioscience, Bristol, UK; 100 µM - 5 mM ; Chapters 3, 4 and 5), glutamate receptor subtype agonist α -amino-3-hydroxy-5-methyl-4-isoxazoleproprionate (AMPA; 50 μ M -100 µM ; Chapters 3, 4 and 5), glutamate receptor subtype agonist N-methyl-Daspartate (NMDA; 500 µM - 5 mM ; Chapters 3, 4 and 5), mGluRI and II agonist trans-(1S,3R)-1-amino-1,3-cyclopentanedicarboxylic acid (tACPD; 10 µM - 1 mM ; Chapters 3, 4 and 5), mGluRI agonist (S)-3,5-Dihydroxyphenylglycine hydrate (DHPG; 500 µM; Chapters 4 and 5); mGluRII agonist (2R,4R)-4-Aminopyrrolidine-2,4-dicarboxylate (APDC; Tocris Bioscience; 5 µM; Chapters 4 and 5) and mGluRIII agonist L-(+)-2-Amino-4-phosphonobutyric acid (L-AP4; 1 mM; Chapters 4 and 5), Ca^{2+} ionophore A23187 (10 μ M; Chapter 3) and the TLR4 agonist lipopolysaccharides from Escherichia Coli 055:B5 (LPS; 1 µg/ml; Chapters 4 and 5). All agonists were dissolved in HEPES buffered salt solution in Chapter 3 or DMEM in Chapter 4 and 5, except A23187 which was dissolved in dimethyl sulfoxide (DMSO; maximum final concentration 0.17%).

2.9.2 Inhibitors

Inhibitors used in experiments in proceeding chapters were: NMDA receptor open channel antagonist (+)-MK-801 hydrogen maleate (MK-801; 40 µM; Chapters 3 and 5); kainate receptor antagonist y-D-glutamylamino methylsulfonic acid 30 μM; Chapter 3); mGluRI and II antagonist (+)-α-methyl-4-(GAMS: carboxyphenylglycine (MCPG; 500 µM; Chapter 3); AMPA receptor antagonist 1-(4-Aminophenyl)-4-methyl-7,8-methylenedioxy-5H-2,3-benzodiazepine hydrochloride (GYKI52466; 50 µM; Chapter 3); kainate and AMPA receptor antagonist 6-cyano-7-nitroquinoxaline-2,3-dione (CNQX; 20 µM; Chapters 3, 4 and 5); mGluR1 antagonist 7-Hydroxyiminocyclopropan[β]chromen-1αcarboxylic acid ethyl ester (CPCCOEt; 75 µM; Chapters 4 and 5); mGluR5 antagonist 6-Methyl-2-(phenylethynyl)pyridine (MPEP; 1 µM; Chapters 4 and 5); mGluRII antagonist (2S)-α-ethylglutamic acid (EGLU; 500 μM; Tocris Bioscience; antagonist (\pm) - α -Methyl-(4-III mGluRII and Chapters 4 and 5): sulfonophenyl)glycine (MSPG; 1 mM; Chapters 4 and 5) and NMDA receptor antagonist D(-)-2-Amino-5-phosphonopentanoic acid (AP5; 200 µM; Chapter 4); phosphatidylinositol-4,5-bisphosphate (PIP₂) synthesis inhibitor LiCl (1 mM ; Chapter 3); endoplasmic reticulum Ca2+-dependent ATPase inhibitor thapsigargin (1 µM; Chapter 3); the inactive form of thapsigargin, called thapsigargin epoxide (Alomone Labs, Jerusalem, Israel; 1 µM; Chapter 3); PKC inhibitor chelerythrine chloride (20 µM; Chapter 3); voltage-dependent Ca2+ channel inhibitor CdCl2 (1 mM : Chapter 3); Ca²⁺ chelator ethylene glycol-bis(2-aminoethylether)-N, N, N', N'-

tetraacetic acid (EGTA; 5 mM; Chapter 3); phospholipase D inhibitor 1-butanol (0.5%; Chapter 3); cell-permeable AC inhibitor 2'5'dideoxyadenosine (DDA; 50 µM; Chapter 3); neurokinin-1 and neurokinin-2 receptor antagonist spantide I (3 µM; Chapter 3); neurokinin-1 antagonist cis-2-(Diphenylmethyl)-N-[(2iodophenyl)methyl]-lazabicyclo[2.2.2]octan-3-amine oxalate salt (L703606; 10 μ M; Chapter 3); phospholipase A₂ (PLA₂) antagonist aristolochic acid (50 μ M; Chapter 3); the NMDA receptor glycine co-binding site antagonist 7-chlorokynurenic acid (5 μ M; Chapter 3), equilibrative nucleoside transporter 1 inhibitor S-(4-Nitrobenzyl)-6-thioinosine (NBMPR; 10 µM; Chapter 3), the adenosine kinase inhibitor iodotubercidin (EMD Biosciences, San Diego, CA, USA; 10 μ M; Chapter 3), the ecto-ATPase inhibitor 6-N,N-Diethyl- β - γ dibromomethylene-D-adenosine-5-triphosphate (ARL 67156; 100 µM; Chapter 3), the glutamate transporter antagonist with preferential activity at excitatory amino acid carrier † L-threo-B-benzyl-aspartate (TBOA; 10 µM; Chapter 3), transcription inhibitor actinomycin D (1 µg/ml; Chapters 4 and 5) and translation inhibitor cycloheximide (1 µg/ml; Chapters 4 and 5).

All inhibitors were dissolved in HEPES buffered salt solution in Chapter 3 or DMEM in Chapters 4 and 5, except thapsigargin, thapsigargin epoxide, aristolochic acid, DDA, 7-chlorokynurenic acid, CPCCOEt, NBMPR and iodotubercidin were dissolved in dimethyl sulfoxide (maximum final concentration 0.17%); MCPG and EGLU were dissolved in 0.1M NaOH (maximum final concentration 1 mM NaOH); GYKI52466 was dissolved in 2.5 mM HCl and [β-Ala⁸]-Neurokinin A Fragment

4-10 was dissolved in 20% acetonitrile and 0.01% trifluoroacetic acid. As mentioned in sections 2.3.3 and 2.6.2, solvents were examined for effects on ATP or IL-10 release and no effects were found. Additionally, at 1 mM NaOH did not affect solution pH.

2.9.3 Primary antibodies

Primary monoclonal antibodies used were: the astrocyte intermediate filament marker mouse anti-rat GFAP (1:800; Chapters 3, 4 and 5; raised against GFAP from pig spinal cord), the microglia- and macrophage marker mouse anti-rat CD11b (1:100; Chemicon; Chapters 3, 4 and 5; raised against T cell-enriched splenocytes from B10 mice) and the neuron-specific marker antibody mouse anti-rat NeuN (1:150, Chemicon; Chapter 3; raised against purified cell nuclei from mouse brain). Primary polyclonal antibodies used were the oligodendrocyte marker rabbit anti-rat galactocerebroside (1:50 - 1:200; Chemicon; Chapter 3; raised against galactocerebroside from bovine brain), rabbit anti-rat glutamate receptor 2/3 (1:50 -1:200; Chapter 3; immunogen information not available), rabbit anti-rat glutamate receptor 4 (1:50 - 1:200; Tocris, Bristol, UK, Chapter 3; synthetic peptide corresponding to amino acids 883 - 902 of rat precursor GluR4), rabbit anti-rat metabotropic glutamate receptor 2/3 (1:50 - 1:200; Chapter 3; synthetic peptide corresponding to the final 13 C-terminal amino acids of mGluR2/3), rabbit anti-rat mGluR1a and mGluR5 receptors (1:250; Chemicon; Chapter 3; no immunogen information available), rabbit anti-rat NR1 (1:50 - 1:200; Chapter 3; synthetic peptide corresponding to amino acids 918 - 938 of the NR1 protein), and rabbit

anti-rat Toll-like receptor 4 (1:50; Santa Cruz Biotechnology, Santa Cruz, CA, USA; synthetic peptide corresponding to amino acids 339 – 638 of the TLR4 protein).

2.9.4 Secondary antibodies

Secondary antibodies used were goat anti-mouse IgG conjugated to Alexa Fluor 488 (1:100) to detect primary monoclonal antibodies and goat anti-rabbit IgG conjugated to Alexa Fluor 594 (1:100) to detect primary polyclonal antibodies. Both secondary antibodies were from Invitrogen.

CHAPTER 3

GLUTAMATE-STIMULATED ATP RELEASE FROM SPINAL CORD ASTROCYTES IS POTENTIATED BY SP

Contents

3.1 INTRODUCTION	121
3.2 MATERIALS AND METHODS	123
3.2.1 Spinal cord and cortical astrocyte cell cultures	123
3.2.2 Online bioluminescence	123
3.2.3 Patch-clamp recording	124
3.2.4 Fluorescence immunocytochemistry	125
3.3 RESULTS	127
3.3.1 SP enhances glutamate-stimulated ATP release from spinal cord astrocytes	127
3.3.2 Glutamate-stimulated ATP release and potentiation of this by SP is concentration- and tin	ne-
dependent	132
3.3.3 Neurokinin receptors mediate the potentiation of glutamate-stimulated ATP release by SP	134
3.3.4 Glutamate activates AMPA receptors to bring about ATP release	136
3.3.5 SP selectively enables glutamate to stimulate ATP release by binding NMDA receptors and	d
mGluRs	140
3.3.6 Glutamate transporters are not involved in the potentiation by SP of glutamate-stimulated	ATP
release	141
3.3.7 Multiple intracellular signaling pathways are involved in the potentiation of glutamate-	
stimulated ATP release by SP	142
3.3.8 An influx of extracellular Ca ²⁺ is needed for glutamate-stimulated ATP release	143
3.4 DISCUSSION	146
3.4.1 Receptors involved in glutamate-stimulated ATP release and its potentiation by SP	146
3.4.1.1 Glutamate receptor subtypes involved in glutamate-stimulated ATP release	146
3.4.1.2 The effect of SP on glutamate receptor responses	147
3.4.1.3 Neurokinin receptors involved in the potentiating effects of SP	149
3.4.2 Ca ²⁺ influx and various intracellular signaling pathways are involved in glutamate-stimul	ated
ATP release and its potentiation by SP	150
3.4.2.1 Signaling mechanisms involved in glutamate-stimulated ATP release in the absence of SP	150
3.4.2.2 Signaling mechanisms involved in glutamate-stimulated ATP release in the presence of SP	151
5.4.5 Specificity of the potentiation to spinal cord distrocytes	152
5.4.4 A summary of the mechanisms of guitamate-stimulated ATP release from spinal cord	152
astrocytes in the absence and presence of Sr	155

3.1 INTRODUCTION

It has recently been discovered that allodynia induced by nerve-injury involves upregulation of ATP P2X₄ receptors on microglia as a required molecular mediator (Tsuda et al., 2003). Furthermore, inflammatory and neuropathic pain are completely removed in mice lacking the ATP P2X₇ receptor which is found on microglial cells amongst others (Chessell et al., 2005; Sim et al., 2004). Activation of P2X₃ receptors on nociceptor nerve terminals in the spinal cord enhances transmission of chronic inflammatory and neuropathic pain (Kennedy et al., 2003; Sharp et al., 2006). Furthermore, inflammatory and neuropathic hyperalgesia is significantly reduced in mice lacking the ATP P2X₃ receptor (Dorn et al., 2004; McGaraughty et al., 2003; Souslova et al., 2000; Wu et al., 2004) as it is following intrathecal administration of compounds that disrupt glial function, such as fluorocitrate (Watkins et al., 1997). It appears that the level of ATP in the vicinity of peripheral afferent synapses in the dorsal horn is likely to be an important regulator of inflammatory and neuropathic pain.

Few studies have examined potential spinal sources of ATP in inflammatory and neuropathic pain. Depolarisation of peripheral afferent fibres results in dorsal horn ATP release however it is unclear whether noxious stimulation can elicit this release (Bardoni et al., 1997; Sawynok et al., 1993). Stimulation of ATP receptors elicits ATP release from spinal cord astrocytes, but this does not give insight into the original source of ATP in chronic pain (Abdipranoto et al., 2003; Salter and Hicks, 1994). Glutamate, a transmitter released from spinal nociceptive terminals in increased amounts during chronic pain (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992), stimulates ATP release from spinal microglial cells (Liu et al., 2006). No other pathways of ATP release have been explored apart from these potential sources. Given the paucity of knowledge of spinal sources of ATP despite the important role of spinal ATP in chronic pain, further identification of spinal sources of ATP would contribute to the knowledge of the pathogenesis of chronic pain.

Spinal cord astrocytes are in intimate contact with nociceptor nerve terminals (Ventura and Harris, 1999; Zerari et al., 1998) and they possess receptors for two important nociceptive transmitters - glutamate (Agrawal and Fehlings, 1997) and SP (Marriott et al., 1991; Palma et al., 1997; Torrens et al., 1986; Wienrich and Kettenmann, 1989; Zerari et al., 1998). Binding of these receptors activates intracellular signaling processes which have been linked to ATP release, including Ca²⁺ liberation and phosphoinositide turnover (Abdipranoto et al., 2003; Ahmed et al., 1990; Hartung et al., 1988; Heath et al., 1994; Marriott et al., 1991; Queiroz et al., 1999). In addition, cortical astrocytes release ATP in response to glutamate and release much greater amounts of ATP than neurons, suggesting astrocytes could be a much more powerful source of ATP than other cell types (Queiroz et al., 1997; Queiroz et al., 1999). Considering this, it is natural to enquire whether spinal cord astrocytes, excited by the concomitant application of glutamate and SP, release ATP in relatively large amounts that could excite the range of ionotropic purinergic receptors, which have been so clearly implicated in neuropathic and inflammatory pain. We report here that glutamate can stimulate ATP release from spinal cord astrocytes and this release is greatly enhanced when SP is present. Mechanisms involved in this phenomenon are reported.

3.2 MATERIALS AND METHODS

3.2.1 Spinal cord and cortical astrocyte cell cultures

Detailed protocol for the preparation of astrocyte cultures is given in section 2.2. Briefly, spinal cords and cortices from 0-2 day old Sprague-Dawley rats were dissected out, and the meninges and, where appropriate, peripheral nerves were The remaining spinal cord or cortical tissue was enzymatically and removed. mechanically dissociated then washed with DMEM+. Dissociated cells were plated on a poly-D-lysine (20 µg/mL) coated culture flask for 4 h in a 5% CO2 incubator at 37°C, were washed and then maintained at 5% CO2 and 37°C in DMEM +, replaced every 3-4 days. To purify astrocytes, 14 days after initial plating, flasks were shaken for 40 min at 400 rpm and 37°C and this was repeated 20 days after initial plating for 16 h. Additionally, 16 days after initial plating DMEM+ was replaced with DMEM supplemented by 10% horse serum, 1% penicillin/streptomycin/ glutamine and 25 mM sorbitol. After 21 days, the age at which there remains only one predominant (> 90%) subtype of spinal cord astrocyte (Black et al., 1993), cells were transferred to 13 mm diameter round glass coverslips coated in 20 µg/ml polyfor ATP release, electrophysiology and Cells were used D-lysine. immunocytochemistry experiments 1 to 2 days after transfer. The purity of cultures was verified by immunocytochemistry.

3.2.2 Online bioluminescence

Detailed protocol for the measurement of ATP by online-bioluminescence is given in section 2.3. Briefly, coverslips were placed inside a light-impermeable chamber with ATP assay mix added (1 mg/ml) and the number of photons released were measured

every second by a P30CWAD5-45 Photodetector Package. When photon count readings did not varying by more than 10% over 10 minutes, the average photon count over this period of time was recorded as the baseline level. After the baseline was reached, recording of photon counts was stopped and agonists of interest were applied at 10% of the volume of liquid on the coverslip. Recording was immediately Chemicals were not washed out during recommenced (within 20 s). experimentation. At the end of each experiment, 1% Triton X-100 was added to lyse cells, allowing an approximation of the number of cells on the coverslip. For each batch of ATP assay mix used for experiments, a standard curve was computed using known concentrations of ATP allowing conversion of the photon count data to ATP concentration values. ATP release data was analysed to find the peak increase in ATP release relative to baseline (referred to as 'relative peak ATP level'). Experiments were repeated at least three times and results were expressed as mean relative peak ATP level \pm SEM. Means were analysed for statistically significant differences with an unpaired Student's two-sample two-tailed *t*-test. p < 0.05 was considered statistically significant. All inhibitors were applied 1 h prior to addition of agonists.

3.2.3 Patch-clamp recording

Detailed protocol for the measurement of whole-cell currents by patch-clamp recording is given in section 2.4. Briefly, pipettes used had resistances between $4 - 8M \Omega$ when a patch pipette was filled with pipette solution detailed in section 2.4. Mg²⁺ was omitted from the bath solution when NMDA was used as an agonist. Whole-cell currents were recorded under voltage-clamp mode at a holding membrane potential of -60 mV. Currents were recorded with a Multiclamp 700A amplifier and

they were sampled on-line using a Digidata 1322A interface and pClamp8 program. Glutamate, its analogs and SP were applied through pressure ejection via a picosprizer.

3.2.4 Fluorescence immunocytochemistry

Detailed protocol for the detection of cellular proteins using fluorescent antibodies is given in section 2.5. Briefly, coverslips described in section 3.2.1 taken from the same culture batches used in bioluminescence experiments were washed twice with HEPES buffered salt solution, were fixed with 2% paraformaldehyde solution for 10 min, washed twice for 5 min each with PBS and then incubated in PBS containing 1% BSA and 0.05% saponin for 30 min. Cells on coverslips were then incubated overnight at 4°C with primary antibodies diluted in PBS containing 1% BSA and 0.05% saponin. Primary antibodies were washed out three times for 10 min each with PBS and then cells were incubated in secondary antibodies diluted in PBS containing 1% BSA and 0.05% saponin for 1 h at room temperature. Secondary antibodies were washed out three times with PBS for 10 min each wash. Excess moisture was removed from the back of coverslips and they were mounted on rectangular glass coverslips using ProLong Gold anti-fade mounting media with DAPI. Slides were sealed with nail polish and subsequently stored at 4°C until viewing.

Fluorescence was visualised with an Axioplan2 upright microscope and images were acquired with an AxioCam HR digital monochrome CCD camera using Axiovision 3.1 software. Experiments with each antibody were performed on at least three different batches of cultures and were repeated at least twice on each of the batches. For each coverslip at least three fields of view were imaged. As a control for each antibody, experiments were also run without a primary antibody and no non-specific binding of secondary antibodies was observed. As a further control, experiments were run without secondary antibodies and no autofluorescence of primary antibodies was observed.

To verify the results obtained using the method above, experiments probing the NR1 and GluR4 subunits were repeated with one of the following changes per experiment: omission of DAPI from the mounting media; substitution of the step incubating cells in saponin and BSA with three steps incubating cells in 1% Triton X-100 which was washed out three times with PBS then a 1 h incubation in 1% BSA; use of a different rabbit anti-rat NMDA receptor 1 antibody; incubation of cells in the rabbit anti-rat glutamate receptor antibodies prior to fixation of cells with formaldehyde solution. Results using these different methods were consistent.

3.3 RESULTS

3.3.1 SP enhances glutamate-stimulated ATP release from spinal cord astrocytes

We initially determined the ability of SP and glutamate to stimulate ATP release from spinal cord astrocytes when applied separately and together using online bioluminescence. Glutamate (1 mM) elicited approximately a two-fold increase in ATP release levels at the peak of its release compared to the baseline level (2.19 \pm 0.11; n = 67; Figure 3.1A, B). This measure is referred to as relative peak ATP level. Peak ATP release was reached 5 - 7 min after application of glutamate and ATP release levels returned to baseline within 30 min after application (Figure 3.1B). When applied on its own at concentrations of 100 µM or less, SP did not change ATP release levels from the baseline level (1 µM, n = 3; 10 µM, n = 12; 30 µM, n = 3; 100 µM, n = 15; Figure 3.1A, B). However, when 100 µM SP and 1 mM glutamate were co-applied, ATP release levels significantly increased to approximately seven-fold above the baseline level (7.35 \pm 0.68; n = 38; p < 0.001; Figure 3.1A, B). This equated to a 336% potentiation of glutamate-stimulated ATP release by SP compared to the effect of glutamate on its own. SP did not alter the time course of glutamate-stimulated ATP release (Figure 3.1B).

To verify the identity of the cells being acted upon by SP and glutamate to bring about the modulation of ATP release, immunocytochemistry was carried out. Over 90% of cells were positive for the astrocyte-specific intermediate filament marker GFAP, whilst less than 10% were positive for microglial and macrophage marker CD11b. Anti-galactocerebroside non-specifically labelled all cells. Consequently oligodendrocyte presence was identified on the basis of morphological characteristics and oligodendrocytes were found to comprise < 1% of total cells. There were no neurons present in cultures as evidenced by a lack of NeuN staining. To examine a possible contribution of microglia and oligodendrocytes to the modulation of glutamate-stimulated ATP release by SP, the effect of SP on glutamate-stimulated ATP release from coverslips with total microglia and oligodendrocyte populations of \leq 4% was compared to coverslips with the same number of astrocytes but with microglia and oligodendrocytes added until the total microglial and oligodendrocyte population reached 30%. In cells with \leq 4% microglia and oligodendrocytes, SP (100 µM) increased glutamate (1mM)-stimulated ATP release by 412% with a glutamate-stimulated ATP release of 1.64 ± 0.16 (n = 3) fold above basal ATP release values and a SP and glutamate-stimulated ATP release of 6.78 ± 1.32 (n = 4) Similarly, on cells with 30% microglia and fold above basal levels. oligodendrocytes, SP (100 µM) increased glutamate (1mM)-stimulated ATP release by 379% with an ATP release of 1.61 ± 0.15 (n = 3) fold above basal ATP release values being elicited by glutamate only and of 6.10 ± 2.98 (n = 3) fold by coapplication of SP and glutamate. There was no significant difference in values between the \leq 4% contaminant cultures and the 30% contaminant cultures. Furthermore, glutamate (1 mM) stimulated an ATP release of 2.51 ± 0.30 (n = 5) fold above basal ATP release values from pure microglial cultures and substance P (100 μ M) did not significantly increase this release (3.26 ± 1.67; n = 5). These results suggest microglia and oligodendrocytes do not contribute to the interaction of SP with glutamate-stimulated ATP release and that the potentiation results from SP and glutamate acting on astrocytes.

It was possible that ATP measured on application of agonists could be a consequence of adenosine release through the equilibrative nucleoside transporter 1 and subsequent anabolism by adenosine kinase then adenylate kinase to ATP. Neither the equilibrative nucleoside transporter 1 antagonist NBMPR (10 μ M; 7.06 \pm 1.78, n = 6) nor the adenosine kinase antagonist iodotubercidin (10 μ M; 7.92 \pm 1.01; n = 3) significantly changed SP and glutamate-stimulated ATP release. Similarly the ecto-ATPase inhibitor ARL 67156 did not have a significant effect on SP and glutamatestimulated ATP release (100 μ M; 6.27 \pm 1.10; n = 4).

To examine if SP also potentiated glutamate-induced currents in spinal cord astrocytes, glutamate-induced whole cell currents were recorded and the effects of SP on them were determined. In 17 out of 19 cells tested, glutamate (1 mM) caused slow inward currents which took approximately 10 s to peak. The average peak amplitude of the inward currents was 49 ± 9 pA. SP (100 µM) did not cause inward currents in any cells examined (n = 4). However, on co-application of glutamate and SP, inward currents were significantly larger than that recorded from astrocytes exposed to glutamate alone (150 ± 29 pA; n = 3; p < 0.05; Figure 3.1C).

As cortical astrocytes express both glutamate (Janssens and Lesage, 2001) and SP receptors (Beaujouan et al., 1990), it was determined whether SP could also potentiate glutamate-stimulated ATP release from cortical astrocytes. When applied by itself, 100 μ M SP did not cause ATP release (n = 5). Glutamate (1 mM) elicited ATP release from cortical astrocytes (1.80 \pm 0.14; n = 14), but this relative ATP release level was not significantly enhanced by SP (2.17 \pm 0.24; n = 12;).


Figure 3.1 SP potentiates glutamate-stimulated ATP release from cultured spinal cord astrocytes. A) The effect of application of 100 μ M SP (SP; n = 15), 1 mM glutamate (Glu; n = 67) and co-application of 1 mM Glu and 100 μ M SP (Glu + SP; n = 38) on ATP release relative to the basal release level from cultured spinal cord astrocytes. In this figure and histograms in following figures, relative peak ATP level is defined as the peak ATP release divided by the baseline ATP value and is expressed as mean ± SEM in this and subsequent figures involving relative peak ATP level data. B) Representative time courses of ATP release induced by these agonists from spinal cord astrocytes. The arrow represents time of addition of these agonists to the bath. C) A representative trace of a glutamate (1 mM)-induced inward current from a spinal cord astrocyte, the amplitude of which was dramatically enhanced by 100 μ M SP, while SP itself did not induce currents. The black bars above the current traces represent the duration of agonist application. ***p < 0.001.



Figure 3.2 Glutamate-stimulated ATP release and the potentiation of this by SP is concentration-dependent. A) Glutamate dose-dependently stimulated ATP release (lower line/black circles; $EC_{50} = 234 \,\mu\text{M}$; $E_{max} = 2.11$) and the degree of modulation by SP (100 μ M) increased with increasing concentrations of glutamate (top line/white circles; $EC_{50} = 347 \,\mu\text{M}$; $E_{max} = 7.91$). B) SP dose-dependently increased glutamate-stimulated ATP release when it was co-applied with a constant 1 mM glutamate ($EC_{50} = 34 \,\mu\text{M}$; $E_{max} = 7.60$). For each concentration, n is at least 3.

3.3.2 Glutamate-stimulated ATP release and potentiation of this by SP is concentration- and time-dependent

To examine if the interaction between SP and glutamate was dependent on the concentration of glutamate, ATP release was measured on co-application of increasing concentrations of glutamate with a constant 100 µM SP, and compared with the level of ATP release stimulated by the corresponding concentration of glutamate on its own. In the absence of SP, 30 µM glutamate elicited ATP release $(1.51 \pm 0.25; n = 4)$ and increasing concentrations augmented ATP release in a dosedependent manner (100 μ M, 1.66 \pm 0.24, n = 4; 300 μ M, 1.78 \pm 0.29, n = 3; 1 mM, 2.19 ± 0.11 , n = 67). Concentrations above 1 mM (3 mM, 1.99 \pm 0.51, n = 4; 10 mM, 2.15 ± 0.25 , n = 3) did not further increase ATP release. The EC₅₀ was 234 μ M and E_{max} was 2.11 when glutamate was applied alone (Figure 3.2A). When 100 µM SP was co-administered with glutamate, potentiation of the level of glutamate-stimulated ATP release was seen even at the lowest concentrations of glutamate examined (30 μ M, 2.11 \pm 0.43, n = 3; 100 μ M, 2.39 \pm 0.66, n = 4). Although ATP release stimulated by co-application of SP with 30 µM and 100 µM glutamate was not significantly larger than the ATP release by corresponding concentrations of glutamate in the absence of SP, their relative values were comparable with the E_{max} of glutamate alone. Statistically significant increases in levels of ATP release were seen at concentrations of 300 μ M glutamate (3.41 ± 0.78, n = 6, p < 0.05) and above (1 mM, 7.35 \pm 0.68, n = 38, p < 0.001; 3 mM, 7.74 \pm 1.40, n = 4, p < 0.01; 10 mM, 7.96 \pm 1.51, n = 3, p < 0.05). The EC₅₀ was 347 μ M which was comparable with that obtained on application of glutamate in the absence of SP,

however the E_{max} was 7.91 which was appreciably higher than that obtained with glutamate alone (Figure 3.2A).

To examine if the interaction between SP and glutamate was dependent on the concentration of SP, ATP release was measured on co-application of varied concentrations of SP with a constant 1 mM glutamate and compared with that elicited by 1 mM glutamate on its own. No significant increases above the glutamate-only level were seen with co-application of 1 μ M (1.65 ± 0.16, n = 3) or 10 μ M (1.63 ± 0.13, n = 3) SP. Significant increases were seen with higher concentrations of SP (30 μ M, 3.47 ± 0.53, n = 4, p < 0.05; 100 μ M, 7.35 ± 0.68, n = 38, p < 0.001; 300 μ M, 7.52 ± 0.92, n = 7, p < 0.001). The EC₅₀ of SP was 34 μ M and the E_{max} was 7.60 (Figure 3.2B). Consequently, SP was used at a concentration of 100 μ M and glutamate at 1 mM for the remaining studies.

The time-dependence of the potentiation of glutamate-stimulated ATP release by SP was investigated by applying glutamate at different intervals after SP application. SP potentiated glutamate-stimulated ATP release when the two agonists were co-applied (0 min, 7.35 ± 0.68 , n = 38). However when glutamate was added 30 secs (2.10 ± 0.21 , n = 3), 2 min (2.23 ± 0.29 , n = 3), 5 min (2.13 ± 0.39 , n = 3), 10 min (2.27 ± 0.35 , n = 3) and 15 min (1.76 ± 0.29 , n = 3) after SP, no potentiation was observed. Therefore SP and glutamate were co-applied in further experiments. An examination of the degree of potentiation when SP was applied at different intervals after glutamate application could not be performed. This was because levels of ATP release were increasing after application of glutamate, and this prevented comparison of potentiation between experiments. It was not examined whether the time-

dependency of the potentiation was inherent in the mechanisms leading to the modulation or whether it was attributable to the breakdown of SP by endopeptidases, such as neutral endopeptidase 24.11, known to be expressed on spinal cord astrocytes (Horsthemke et al., 1984; Lentzen and Palenker, 1983; Marcel et al., 1990; Vincent et al., 1994).

3.3.3 Neurokinin receptors mediate the potentiation of glutamatestimulated ATP release by SP

It was investigated what subtypes of neurokinin receptor were acted on by SP to potentiate glutamate-stimulated ATP release. The selective NK1 agonist [Sar9, Met(O₂)¹¹]-substance P (100 μ M; 2.06 \pm 0.45; n = 4), the selective NK2 agonist [β -Ala⁸]-Neurokinin A Fragment 4-10 (64 μ M; 1.74 \pm 0.51; n = 3) and the selective NK3 agonist senktide (100 μ M; 1.83 \pm 0.21; n = 3) did not significantly potentiate glutamate-stimulated ATP release (Figure 3.3A). However when [Sar⁹, Met(O₂)¹¹]substance P and [B-Ala⁸]-neurokinin A Fragment 4-10 were co-applied with glutamate, a significant potentiation of glutamate-stimulated ATP release was observed $(3.51 \pm 0.38; n = 3; p < 0.05;$ Figure 3.3A). Furthermore when all three neurokinin agonists were co-applied with glutamate, greater potentiation of glutamate-stimulated ATP release was recorded (4.85 \pm 0.94; n = 6; p < 0.05; Figure 3.3A). The NK1 and NK2 preferring antagonist spantide I $(3 \mu M)$ significantly reduced SP and glutamate-stimulated relative ATP release to levels seen when glutamate was applied alone $(2.03 \pm 0.31; n = 4; p < 0.001;$ Figure 3.3B). Similarly, the selective neurokinin-1 antagonist L703606 (10 µM) abolished the potentiating effects of SP $(1.90 \pm 0.25; n = 4; p < 0.001;$ Figure 3.3B, C).



Figure 3.3. SP acts on neurokinin receptor subtypes to potentiate glutamatestimulated ATP release. A) Summary histogram of the effect of neurokinin receptor agonists [Sar⁹, Met(O₂)¹¹]-substance P (SarMet; NK1 receptor agonist; 100 μ M; n = 4), [β-Ala⁸]-Neurokinin A Fragment 4-10 (β-Ala; NK2 receptor agonist; 64 μ M; n = 3) and senktide (Senk; NK3 receptor agonist; 100 μ M; n = 3) on glutamatestimulated ATP release. Significant potentiation of glutamate-stimulated ATP release was seen only on co-application of neurokinin receptor subtype agonists. B) Preincubation in Spantide I (Spantide; NK1 and NK2 receptor antagonist; 3 μ M; n = 4) and L703606 (NK1 receptor antagonist; 10 μ M; n = 4) abolished the potentiation of glutamate-stimulated ATP release by SP. C) Representative time courses of ATP release stimulated by addition of glutamate, glutamate and SP and glutamate and SP in the presence of L703606. * p <0.05, *** p < 0.001.

3.3.4 Glutamate activates AMPA receptors to bring about ATP release

The contribution of NMDA, AMPA, kainate and metabotropic glutamate receptors to glutamate-stimulated ATP release in the absence of SP was examined. Application of 50 μ M AMPA stimulated ATP release (2.41 ± 0.34; n = 9) whereas 5 mM kainate (n = 3), 1 mM tACPD (n = 4), an mGluR group I and II agonist, and 5 mM NMDA (n = 3) did not lead to ATP release (Figure 3.4A, B). In agreement with these results, glutamate-stimulated ATP release was significantly reduced by 50 μ M GYKI52466 hydrochloride (AMPA receptor antagonist; 1.16 ± 0.16; n = 5; p < 0.001) and 20 μ M CNQX (AMPA and kainate antagonist; 1.2 ± 0.07; n = 4; p < 0.001; Figure 3.4C, D). In contrast, it was not significantly affected by 30 μ M GAMS, a kainate receptor antagonist (1.78 ± 0.22; n = 4), 500 μ M MCPG, an mGluR group I and II antagonist (1.85 ± 0.21; n = 4) and 40 μ M MK-801, an NMDA receptor antagonist (2.39 ± 0.44; n = 5; Figure 3.4C). Additionally, AMPA (1 mM) induced slow inward currents which were similar to those induced by glutamate in all spinal cord astrocytes tested (50 – 75 pA; n = 3; Figure 3.4E). NMDA (1 mM; n = 4) and tACPD (1 mM; n = 5) did not elicit inward currents in spinal cord astrocytes (Figure 3.4E).

To examine whether glutamate receptor subtypes were available on the membrane of spinal cord astrocytes to be bound by glutamate, immunocytochemistry was conducted. The AMPA GluR2/3 subunit was found in fine clusters throughout the cytoplasm and membrane of all GFAP-positive spinal cord astrocytes imaged (n = 35 astrocytes; Figure 3.5A). In contrast, the AMPA GluR4 subunit was localized specifically to an area overlapping and immediately surrounding the DAPI-stained nuclei of spinal cord astrocytes (n = 36 astrocytes; Figure 3.5B). An antibody



Figure 3.4 Glutamate stimulates ATP release by activating AMPA receptors. A) AMPA stimulated ATP release (50 μ M; n = 7), however no ATP release was seen on application of kainate (KA; 5 mM; n = 3), tACPD (mGluRI and II agonist; 1 mM; n = 4) or NMDA (5 mM; n = 3). B) Time course of the effect of AMPA and kainate on ATP release. C) Glutamate-stimulated ATP release was significantly reduced by application of GYKI52466 (GYKI; AMPA receptor antagonist; 50 μ M; n = 5) and CNQX (AMPA and kainate receptor antagonist; 20 μ M; n = 4), but was not significantly reduced by application of GAMS (kainate receptor antagonist; 30 μ M; n = 4), MCPG (mGluRI and II antagonist; 500 μ M; n = 4) and MK-801 (MK801; NMDA receptor antagonist; 40 μ M; n = 5). D) Time course of the effect of CNQX on glutamate-stimulated ATP release. E) Representative traces of patch clamp recordings from spinal cord astrocytes on application of AMPA (1 mM), NMDA (1 mM) and tACPD (1 mM). Inward membrane currents were induced by AMPA but not by NMDA or tACPD. The black bars above the current traces indicate the duration of addition of the agonists. ***p<0.001.



Figure 3.5 Expression of AMPA, NMDA and mGluRs in spinal cord astrocytes. Representative examples of AMPA receptor subunits GluR2 and GluR3 staining of spinal cord astrocytes (A), AMPA receptor subunit GluR4 staining (B), mGluR1a and mGluR5 staining (C) and NMDA receptor subunit NR1 staining (D). Images on the left display glutamate receptor subtype staining in red, the middle images display astrocyte intermediate filament marker GFAP staining in green and the right images are the previous two images merged with additional DAPI staining of the nucleus in blue. Co-localisation of glutamate receptor and GFAP staining appears yellow while co-localisation of glutamate receptor and nucleus staining appears violet. The white bar in the bottom right box represents 20 µm and holds for all images in the figure.



Figure 3.6 SP allows glutamate to stimulate ATP release through NMDA and mGluRs. A) SP (100 μ M) did not affect the amount of ATP stimulated by glutamate receptor subtype agonists AMPA (50 μ M; n = 9 without SP and n = 5 with SP) and kainate (KA; 5 mM; n = 3 without SP and n = 3 with SP). tACPD (1 mM; n = 4) and NMDA (500 μ M, n = 3) did not lead to ATP release on their own but, when coapplied with SP, ATP release was seen (n = 7 for tACPD plus SP and n = 9 for NMDA plus SP). When NMDA and tACPD were co-applied with SP (n = 6), relative ATP release levels approximated that obtained by summating release levels from each agonist plus SP separately. B) Time course of the effect of SP on NMDA- and C) NMDA plus tACPD-stimulated ATP release. D) The potentiation of glutamate-stimulated ATP release was significantly reduced by application of MCPG (500 μ M; n = 5) and MK-801 (40 μ M; n = 6). Co-application of MCPG and MK-801 completely removed the effect of SP (n = 4). ***p<0.001.

recognising the mGluR1a (a splice variant of the mGluR1 receptor) and the mGluR5 receptor stained positively on 8.8% of imaged GFAP-positive astrocytes (n = 29 of 329 astrocytes) and was distributed in thick clusters throughout the membrane and cytoplasm of spinal cord astrocytes (Figure 3.5C). The NMDA receptor subunit NR1 was found localised to an area overlapping the DAPI-stained nuclei of spinal cord astrocytes in 64% of cells imaged (n = 74 of 115 astrocytes) whilst distribution of NR1 also extended throughout the membrane and cytoplasm in the remaining cells imaged (n = 41 of 115 astrocytes; Figure 3.5D). The co-localisation of GluR4 and NR1 with astrocytic nuclei persisted despite the use of a variety of immunocytochemical methods and antibodies.

3.3.5 SP selectively enables glutamate to stimulate ATP release by binding NMDA receptors and mGluRs

SP is known to enhance the activity of some neuronal glutamate receptor subtypes such as the NMDA receptor (Mjellem-Joly et al., 1991; Randic et al., 1990; Rusin et al., 1993). It was therefore determined whether SP changes the ability of glutamate receptor agonists to stimulate ATP release. Co-application of SP and 500 μ M NMDA led to ATP release (2.29 ± 0.26; n = 9; Figure 3.6A, B), as did co-application of SP and 1 mM tACPD (2.15 ± 0.23; n = 7; Figure 3.6A). Application of 500 μ M NMDA, 1 mM tACPD and 100 μ M SP led to further increases in ATP release (3.34 ± 1.02; n = 6; Figure 3.6A, C). No ATP release was seen on co-application of SP and kainate (5 mM; n = 3; Figure 3.6A). Co-application of SP and AMPA (50 μ M; 2.46 ± 0.49; n = 5) led to an ATP release which was not significantly different from application of AMPA alone (Figure 3.6A). The potentiation by SP of glutamate-stimulated ATP release was significantly reduced by 500 μ M MCPG (3.48 ± 0.50; n = 5; p ≤ 0.001; Figure 3.6D) and 40 μ M MK-801 (3.57 ± 0.64; n = 6; p < 0.001; Figure 3.6D). Furthermore, SP and glutamate co-stimulated relative ATP release levels were reduced to glutamate-only levels when pre-incubated with both 500 μ M MCPG and 40 μ M MK-801 (2.13 ± 0.41; n = 4; p < 0.001; Figure 3.6D).

The possibility that SP may have brought about the modulation of NMDA receptors by potentiating the function of the glycine co-binding site of the NMDA receptor or by increasing extracellular levels of NMDA receptor co-agonists glycine, D-serine or D-alanine was examined. The potentiation of NMDA (500 μ M)-induced ATP release by SP was not significantly changed by pre-incubation in the glycine cobinding site antagonist 7-chlorokynurenic acid (5 μ M; 1.76 \pm 0.27; n = 6). Another possible mechanism by which SP may have modulated the NMDA receptor was by depolarizing the cell and removing the Mg²⁺ block, however because of the Mg²⁺ dependence of the activity of the bioluminescence enzymes (Wulff, 1983), it was not possible to conduct experiments in which the concentration of Mg²⁺ was varied.

3.3.6 Glutamate transporters are not involved in the potentiation by SP of glutamate-stimulated ATP release

SP is known to be able to inhibit glutamate transporters (Johnson and Johnson, 1993). It is possible, then, that inhibition of glutamate transporters may have led to increased extracellular glutamate concentrations, which may have contributed to the potentiation seen on co-application of SP with glutamate. However glutamate-

stimulated ATP release was not significantly affected by the glutamate transporter antagonist TBOA (10 μ M; 2.30 ± 0.27; n = 3).

3.3.7 Multiple intracellular signaling pathways are involved in the potentiation of glutamate-stimulated ATP release by SP

As neurokinin receptors are G-protein coupled receptors linked to the phospholipase C, PLA₂, PLD and AC (Garcia et al., 1994; Hartung et al., 1988; Nakajima et al., 1992; Torrens et al., 1998) signaling pathways, potential roles of each of these cell The PKC inhibitor signaling pathways in the effects of SP were explored. chelerythrine chloride (20 µM) abolished the effects of SP on glutamate-stimulated ATP release (2.09 \pm 0.53; n = 6; p < 0.001; Figure 3.7A), as did PLA₂ inhibitor aristolochic acid (50 μ M; 2.40 \pm 0.16; n = 4; p < 0.001; Figure 3.7A, B). The potentiation of glutamate-stimulated ATP release by SP was also significantly reduced by 0.5% 1-butanol (PLD antagonist; 3.44 ± 0.87 ; n = 5; p < 0.01) and 1 mM LiCl (inositol monophosphatase inhibitor; 3.93 ± 0.73 ; n = 8; p < 0.01), however these antagonists did not fully abolish the effects of SP (Figure 3.7A). Thapsigargin (1 μ M), an endoplasmic reticulum Ca²⁺ pump activator inhibitor, also decreased the potentiation (5.43 \pm 0.56; n = 6; p < 0.05). However thapsigargin reduced the potentiation to the same levels of ATP release as the inactive form of thapsigargin, thapsigargin epoxide (1 μ M; 5.55 ± 1.16; n =4). The cell-permeable AC antagonist DDA (50 μ M; 8.63 \pm 1.66; n = 5) did not significantly affect SP and glutamatestimulated ATP release (Figure 3.7A). It should be noted that none of these significantly decreased glutamate-stimulated ATP release inhibitors (see section 3.3.8). Furthermore, the potentiation of NMDA-stimulated ATP release by SP was significantly decreased by 20 μ M chelerythrine chloride (1.12 ± 0.33; n = 4; p < 0.05; Figure 3.7C), but was not significantly affected by 0.5% 1-butanol (2.54 ± 0.68; n = 4; Figure 3.7C).

As NMDA receptors are Ca²⁺ permeable, the role of extracellular Ca²⁺ influx on NMDA and SP-stimulated ATP release was examined. The potentiation of NMDAstimulated ATP release by SP was significantly decreased in Ca²⁺-free bath solution containing 5 mM EGTA compared with the relative peak ATP release level seen when NMDA and SP were added to cells in bath solution containing 500 μ M Ca²⁺ (1.10 ± 0.14; n = 4; p < 0.001; Figure 3.7D). Influx of extracellular Ca²⁺ was shown to be sufficient to induce ATP release as the Ca²⁺ ionophore A23187 triggered ATP release (10 μ M; 2.38 ± 0.64; n = 3; Figure 3.7D).

3.3.8 An influx of extracellular Ca²⁺ is needed for glutamate-

As some AMPA receptors are Ca^{2+} permeable, the role of extracellular Ca^{2+} influx in glutamate-stimulated ATP release was examined. Astrocytes incubated in Ca^{2+} -free bath solution containing the Ca^{2+} chelator EGTA (5 mM) had significantly reduced glutamate-stimulated ATP release levels (1.30 ± 0.07; n = 5; p < 0.001; Figure 3.7D) compared with glutamate-stimulated ATP release from astrocytes in bath solution which contained 500 μ M Ca^{2+} , the concentration used throughout the studies described previously. A possible non-receptor source of Ca^{2+} influx on application of glutamate may have been voltage-dependent Ca^{2+} channels which are known to be expressed on astrocytes (D'Ascenzo et al., 2004). However a voltage-dependent Ca^{2+} channel blocker, CdCl₃, did not significantly affect glutamate-stimulated ATP release



Figure 3.7 The modulation of glutamate-stimulated ATP release by SP requires Ca²⁺ influx and activation of multiple intracellular signaling pathways. A) In the presence of chelerythrine chloride (CheCl; PKC inhibitor; $20 \,\mu$ M; n = 6) or aristolochic acid (Aristol; PLA₂ inhibitor; 50 μ M; n = 4) the potentiating effect of SP on glutamate-stimulated ATP release was blocked. In the presence of 1-butanol (PLD antagonist; 0.5%; n = 5) and lithium chloride (LiCl; inositol monophosphatase inhibitor; 1 mM; n = 8) the modulatory effects of SP were also significantly decreased. The potentiation was not significantly altered by DDA (cell-permeable AC inhibitor; 50 μ M; n = 5). B) A typical time course of the effect of Aristol on the potentiation of glutamate-stimulated ATP release by SP. C) The potentiation of NMDA-stimulated ATP release by SP was significantly decreased by CheCl (n = 4), but not by 1-butanol (n = 4). D) In Ca²⁺ free bath with 5 mM EGTA (zero Ca²⁺; n = 5), glutamate-stimulated ATP release was significantly reduced. Similarly, NMDA plus SP-stimulated ATP release was significantly reduced in these Ca2+-free conditions (zero Ca^{2+} ; n = 4). The Ca^{2+} ionophore A23187 (10 μ M; n = 3) stimulated ATP release. *p<0.05, **p<0.01, ***p<0.001.

(1 mM; 2.40 \pm 0.54; n = 3). Additionally, inhibitors to other intracellular signaling molecules known to be associated with glutamate receptors, such as the inhibitors chelerythrine chloride (20 μ M; 1.84 \pm 0.16; n = 4), aristolochic acid (50 μ M; 2.38 \pm 0.32; n = 4), 1-butanol (0.5%; 2.31 \pm 0.28; n = 5), LiCl (1 mM; 1.62 \pm 0.22; n = 4), thapsigargin (1 μ M; 1.84 \pm 0.25; n = 4) and DDA (50 μ M; 1.75 \pm 0.11; n = 3), did not significantly alter glutamate-stimulated ATP release from spinal cord astrocytes.

3.4 DISCUSSION

In this chapter, the ability of glutamate and SP to stimulate ATP release from spinal cord astrocytes was investigated. It was found that glutamate elicited ATP release in a time- and concentration-dependent manner, as is seen with cortical astrocytes (Queiroz et al., 1997). On its own, SP did not elicit ATP release, but greatly potentiated glutamate-stimulated ATP release.

3.4.1 Receptors involved in glutamate-stimulated ATP release and its potentiation by SP

3.4.1.1 Glutamate receptor subtypes involved in glutamate-stimulated ATP release

The glutamate receptors involved in generating glutamate-invoked inward currents and ATP release primarily consist of the AMPA subtype as the effects of glutamate on ATP release were blocked by an AMPA receptor antagonist, but were not significantly blocked by antagonists to kainate, NMDA or mGluRs. Furthermore, only AMPA evoked inward currents and stimulated ATP release. In agreement with what has been reported on *in situ* spinal cord astrocytes (Brand-Schieber et al., 2004; Brand-Schieber and Werner, 2003), immunocytochemistry showed the existence of AMPA GluR2 and GluR3 subunits on the surface of these cultured spinal cord astrocytes. The GluR4 subunit of the AMPA receptor was not found on spinal cord astrocyte membranes. GluR4 subunits have been selectively identified on the end feet of astrocytes surrounding blood vessels (Brand-Schieber et al., 2004), which were absent from cultures in the present study.

Chapter 3

Glutamate-stimulated ATP release is potentiated by SP

Surprisingly ATP release was not elicited by kainate, which is known to activate AMPA receptors (Rodriguez et al., 2005). Kainate has been reported to cause small membrane potential changes in spinal cord astrocytes, although it has not been investigated whether these responses are due to the action on kainate or AMPA receptors (Ziak et al., 1998). Consequently, we hypothesise that in the present case, either kainate does not activate AMPA receptors on spinal cord astrocytes to the extent needed to produce ATP release or AMPA receptors on spinal cord astrocytes are insensitive to kainate, possibly due to the presence of a variant form of kainate receptor that has an altered primary structure at the key amino acid sites that regulate the kainate sensitivity of AMPA receptors (Rodriguez et al., 2005).

3.4.1.2 The effect of SP on glutamate receptor responses

Although NMDA and mGlu receptor agonists did not elicit ATP release when applied alone, when they were each co-applied with SP a release of ATP was observed. SP had no effect on the ATP release responses of spinal cord astrocytes to AMPA and kainate. These results suggest a model whereby glutamate stimulates ATP release through AMPA receptors, but in the presence of SP it can also stimulate ATP release through NMDA and mGlu receptors, leading to the observed potentiation of glutamate-stimulated ATP release. Further supporting this conclusion, simultaneous antagonism of NMDA and mGlu receptors completely abolished the potentiation of glutamate-stimulated ATP release by SP. Also, with regards to the concentration studies carried out, the E_{max} value of glutamatestimulated ATP release in the presence of SP was much higher than in the absence of SP. This suggests that SP enhanced the number of ATP release-linked glutamate receptors in the membrane. Immunocytochemistry identified NR1 subunits and mGluR1a and mGluR5 receptors expressed on the surface of a portion of spinal cord astrocytes, in accordance with previous reports (Aicher et al., 1997; Silva et al., 1999; Ziak et al., 1998), yet the fact that glutamate on its own does not stimulate ATP release through these receptors suggests that they are not functionally linked to the ATP release pathway. How SP may link these receptors to the ATP release pathway remains unclear. PKC is known to enhance the activity of NMDA receptors (Liao et al., 2001; Skeberdis et al., 2001) and arachidonic acid downstream of PLA₂ activation is known to enhance the activity of mGluRs (Collins et al., 1995; Sistiaga and Sanchez-Prieto, 2000). As both these signaling molecules are known to be activated on NK1 binding (Monastyrskaya et al., 2005; Rayner and Van Helden, 1997; Tanabe et al., 1996) and are central to the effects of SP in this study, SP may possibly potentiate the activity of these receptors to a degree where they can activate enough signaling molecules to release ATP. In addition, SP exposure may increase the density of NMDA receptors and mGluRs in spinal cord astrocyte membranes.

Although the precise mechanisms of action remain unknown, the interaction between SP and glutamate receptor subtypes described in this current research is similar to the interaction of SP with glutamate receptors on freshly isolated rat dorsal horn neurons. Exposure of these preparations to SP results in a potentiation of NMDA-mediated currents. This effect is specific for the NMDA receptor as AMPA and kainate mediated currents are not affected by SP (Randic et al., 1990). The effect of SP on neuronal mGluRs was not examined.

Glutamate-stimulated ATP release is potentiated by SP

Surprisingly, the SP and NMDA co-stimulated responses occurred despite the presence of Mg^{2+} at a concentration known to inhibit neuronal NMDA receptormediated responses (Ziak et al., 1998). There are several examples of NMDAmediated responses in glial cells that occur at a physiological level of Mg^{2+} , such as NMDA-stimulated ATP release from cortical astrocytes (Queiroz et al., 1997) and NMDA-induced inward currents in spinal cord astrocytes from rats younger than 14 days (Ziak et al., 1998). As suggested by Ziak et al (1998) and Kutsuwada et al. (1992) perhaps glial NMDA receptors contain Mg^{2+} resistant subunits, such as the NR2C (formerly ε 3) subunit.

3.4.1.3 Neurokinin receptors involved in the potentiating effects of SP

To bring about potentiation of glutamate-stimulated ATP release, it is likely that SP acted synergistically on multiple neurokinin receptor subtypes. A NK1 receptor preferring antagonist abolished the effects of SP suggesting that action on the NK1 subtype was necessary for the effects of SP. However a selective NK1 receptor agonist was not sufficient to potentiate glutamate-stimulated ATP release. Similarly, selective stimulation of NK2 or NK3 was not sufficient to potentiate glutamate-stimulated ATP release. It was only when agonists to the neurokinin receptor subtypes were applied in combination that potentiation was observed. This suggests the NK1 receptor is necessary but not sufficient for the actions of SP on glutamate-stimulated ATP release and synergistic action on NK2 and/or NK3 is also necessary for the potentiation. This is consistent with reports of SP acting as an agonist at all three neurokinin receptor subtypes (Drew et al., 2005; Regoli et al., 1994) and with reports of functional NK1, NK2 and NK3 receptors on spinal cord astrocytes (Wienrich and Kettenmann, 1989; Zerari et al., 1998).

It was unlikely that the ATP found in solution after addition of agonists was due to release of adenosine and subsequent actions of adenosine kinase and adenylate kinase to produce ATP. Supporting this statement were the findings that antagonists to an adenosine transporter and adenosine kinase did not affect SP and glutamate-stimulated ATP release. Potential mechanisms of direct ATP release were not explored in this study, however previous studies have reported direct glutamate-stimulated ATP release from cortical astrocytes through exocytosis (Pangrsic et al., 2007; Zhang et al., 2007b).

3.4.2 Ca²⁺ influx and various intracellular signaling pathways participate in glutamate-stimulated ATP release and its potentiation by SP

3.4.2.1 Signaling mechanisms involved in glutamate-stimulated ATP release in the absence of SP

The PLC, PLA₂, AC and PLD signaling pathways did not play a role in glutamatestimulated ATP release when SP was not present, since antagonists to these pathways had no significant affect. However, influx of Ca^{2+} was centrally involved in glutamate-stimulated ATP release since Ca^{2+} influx was shown to be sufficient to elicit ATP release and removing extracellular Ca^{2+} greatly attenuated glutamatestimulated ATP release. This Ca^{2+} influx could be directly through non-GluR2 containing AMPA receptors (Hollmann et al., 1991; Verdoorn et al., 1991).

3.4.2.2 Signaling mechanisms involved in glutamate-stimulated ATP release in the presence of SP

Significant decreases in the glutamate-evoked release of ATP in the presence of SP were obtained on blocking PKC, PLA₂, PLD, IP₃ and DAG, but not AC. It is known that activation of NK1 receptors activates PKC in rheumatoid synovial cells (Tanabe et al., 1996) and HEK293 cells (Monastyrskaya et al., 2005) and PLA2 is endothelial cells of NK1 receptors on activation elevated on also (Rayner and Van Helden, 1997). In murine astrocytes, activation of PKC leads to phosphorylation of PLA2, necessary for its subsequent action in releasing arachidonic acid (Xu et al., 2002) and PLA2 and arachidonic acid can activate and translocate PKC (Mau and Vilhardt, 1997; O'Flaherty et al., 2001), possibly explaining the result that SP's potentiating effects are completely removed when blocking PKC but leaving PLA2 active and vice versa. It is known that SP elevates PLD, PLC and phosphatidylinositol hydrolysis in Chinese hamster cells and astrocytoma cells (Nakajima et al., 1992; Torrens et al., 1998), but that it does not elevate cAMP in cortical astrocytes (Rougon et al., 1983).

ATP release evoked by activation of NMDA glutamate receptors in the presence of SP is greatly reduced in Ca^{2+} -free external solution, indicating that an influx of Ca^{2+} is necessary for this effect. It is unlikely that this influx of Ca^{2+} ions is due to activation of neurokinin receptors as we were unable to detect any inward current in spinal cord astrocytes in response to applied SP. It is known that SP can directly enhance PLC and phophatidylinositol hydrolysis in membrane preparations independent of Ca^{2+} (Nakajima et al., 1992). We suggest that the dependence of the

effect of SP on NMDA receptor-activated release of ATP involves an influx of Ca²⁺ ions through the NMDA receptor channel.

3.4.3 Specificity of the potentiation to spinal cord astrocytes

Interestingly, the potentiation of glutamate-stimulated ATP release by SP was not replicated with cortical astrocytes. The density of NK1 receptors on spinal cord astrocytes has been shown to be twelve times higher than that on cortical astrocytes in rat cultures (Beaujouan et al., 1990) and six times higher in human cultures (Palma et al., 1997). Consequently, in cortical astrocytes SP will be able to activate less NK1-mediated signaling molecules, which are necessary for the interaction to occur, resulting in no interaction. For example, the activation of PLC by NK1 receptors is involved in the potentiation of SP on glutamate-stimulated ATP release, as shown in this study. Phosophoinositide turnover, a consequence of PLC activation, occurs when spinal cord astrocytes are exposed to SP but not when cortical astrocytes are exposed (Cholewinski et al., 1988; Marriott et al., 1991).

Although modulation of glutamate-stimulated ATP release by SP did not occur in cortical astrocytes, it may be of interest to examine whether the modulation occurs with astrocytes from other areas of the CNS that have dense tachykinin receptor expression, such as the basal ganglia and amygdala (Dam et al., 1990; Humpel and Saria, 1993; McLean et al., 1985). Since the actions of SP in the basal ganglia are emerging as an effective target in Parkinson's disease (for a review see Chen et al. (2004)) and its actions in the amygdala are emerging as an effective target in depression treatment (for a review see Rupniak (2002a; 2002b)), the actions of SP on all cell types in these regions should be investigated.

3.4.4 A summary of the mechanisms of glutamate-stimulated ATP release from spinal cord astrocytes in the absence and presence of SP

These observations, of the receptors and signaling molecules important in the interaction between SP and glutamate, suggest that glutamate can elicit ATP release from spinal cord astrocytes via two pathways. The first is the action of glutamate on AMPA receptors resulting in influx of Ca^{2+} . An additional pathway is recruited in the presence of SP through its joint actions on NK1, NK2 and NK3 receptors. Binding of these receptors leads to activation of PKC, PLA₂, PLD and the formation of IP₃ and DAG. The result of this activation is the additional ability of glutamate to bring about ATP release by binding NMDA receptors and mGluRs. These pathways are summarised in Figure 3.8.





Figure 3.8 A model of the mechanisms of glutamate-stimulated ATP release from spinal cord astrocytes and its potentiation by SP. A) In isolation, glutamate acts on the AMPA receptor to bring about an influx of Ca^{2+} . B) When SP is also present, it synergistically acts on the NK1, NK2 and NK3 receptors to activate the NMDA receptor and mGluR. ATP release occurs in a phospholipase C (PLC) and phospholipase A₂ (PLA₂)-dependent manner. The result of this is an increase in the level of glutamate-stimulated ATP release in the presence of SP. The mechanism of ATP release in these two situations is unknown.

CHAPTER 4

RELEASE OF IL-10 FROM SPINAL CORD ASTROCYTES AFTER ACTIVATION OF THE TLR4 AND THE EFFECT OF CONCOMITANT mGLUR ACTIVATION

Contents

4.1 INTRODUCTION	157
4.2 MATERIALS AND METHODS	160
4.2.1 Spinal cord astrocyte cell cultures	160
4.2.2 Sandwich ELISA	161
4.2.3 Lactate Dehydrogenase Toxicology Assay	162
4.2.4 Reverse Transcription-Polymerase Chain Reaction	163
4.2.5 Fluorescence Immunocytochemistry	164
4.3 RESULTS	166
4.3.1 The TLR4 is present on spinal cord astrocytes	166
4.3.2 LPS stimulates IL-10 release from spinal cord astrocytes and this release is enhanced	on
co-exposure to glutamate	166
4.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of	
glutamate _	170
4.3.4 LPS-stimulated IL-10 release from spinal cord astrocytes and enhancement in the	
presence of glutamate is time-dependent	172
4.3.5 Transcription is key to the potentiating effect of glutamate exposure	174
4.3.6 Glutamate acts on multiple mGluR groups to bring about IL-10 release in the presence	e of
LPS	177
4.4 DISCUSSION	180
4.4.1 The interaction between glutamate and LPS involves upregulation of IL-10 mRNA	180
4.4.2 Transcriptional activation by LPS does not contribute to LPS-stimulated IL-10 levels	182
4.4.3 Co-ligation of group I and II mGluRs is necessary for the increase in LPS-stimulated	IL-
10 release in the presence of glutamate	183
4.4.4 Spinal cord astrocytes express TLR4 and this expression is increased by LPS exposure	e185
4.4.5 Summary of mechanisms	186

4.1 INTRODUCTION

IL-10 is a member of the four α -helix bundle family of cytokines (Zdanov, 2004). It has an anti-inflammatory effect in the CNS, for example it has been reported to decrease the morphological signs of glial activation (Wirjatijasa et al., 2002; Woiciechowsky et al., 2004). Associated with this, IL-10 can decrease the production of pro-inflammatory cytokines such as TNF-α, IL-1β, and IL-6 from activated glia (Benveniste et al., 1995; Heyen et al., 2000; Kremlev and Palmer, 2005; Ledeboer et al., 2000; Ledeboer et al., 2002; Mizuno et al., 1994; Sawada et al., 1999). Glial activation and high levels of spinal pro-inflammatory cytokines contribute to numerous CNS pathologies, including inflammatory and neuropathic pain (Arruda et al., 2000; DeLeo et al., 1996; Milligan et al., 2003; Reeve et al., 2000; Sung et al., 2004; Watkins et al., 1997; Wieseler-Frank et al., 2005). Spinal application of IL-10 can be beneficial in such conditions, for example intrathecal IL-10 gene therapy abolishes neuropathic pain (Ledeboer et al., 2007; Milligan et al., 2006a; Milligan et al., 2006b). Moreover, patients with a painless neuropathy have on average double the blood IL-10 mRNA levels than those with a painful neuropathy (Uceyler et al., 2007).

Possible sources of IL-10 in the spinal cord during chronic pain have not been examined. Given that spinal cord IL-10 has important anti-nociceptive outcomes, it is of interest to identify possible sources of IL-10. Astrocytes are a source of IL-10 in the brain. Cultured rat and human astrocytes from the brain express an innate immune pattern recognition receptor called the TLR4 which is often experimentally

stimulated with the agonist LPS (Bowman et al., 2003; Bsibsi et al., 2002; Jou et al., 2006). Unstimulated cultured rat brain astrocytes express very low IL-10 mRNA levels and do not release IL-10. When these astrocytes are exposed to LPS, IL-10 mRNA levels are upregulated and IL-10 protein release is increased (Ledeboer et al., 2002). There is a functional regional heterogeneity amongst astrocytes from different regions of the brain, for example, vasopressin and oxytocin elicit phosphoinositide turnover from cortical but not spinal cord astrocytes (Cholewinski et al., 1988). Hence, even though cortical astrocytes show TLR4-stimulated IL-10 release it can not be assumed that spinal cord astrocytes will behave similarly. It is appealing, however, to hypothesise that TLR4 stimulation of astrocytes may be a source of IL-10 in the spinal cord astrocytes express functional TLR4s (Pehar et al., 2004) and also express a key protein involved in TLR4-stimulated IL-10 release, JNK (Zhuang et al., 2006).

In addition to TLR4-stimulated IL-10 release from spinal cord astrocytes, it is appealing to suggest the neurotransmitter glutamate may be involved in IL-10 release from astrocytes in the spinal cord. Glutamate is a neurotransmitter released in increased amounts in the dorsal horn of the spinal cord in chronic pain (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992). Astrocytes are exposed to glutamate released from neurons, and possibly neighbouring astrocytes and activated microglia (Chen et al., 2005; Lalo et al., 2006; Takeuchi et al., 2006). Moreover, spinal cord

astrocytes express all four subtypes of glutamate receptor, namely AMPA, KA, NMDA and mGluRs (see the results of section 3.3.4, and Aicher et al. (1997), Silva et al. (1999) and Brand-Schieber et al. (2004)). Exposure of astrocytes to glutamate often results in anti-inflammatory outcomes (Bruno et al., 1998; Murphy et al., 1995; Zhou et al., 2006) and also the activation of c/EBP β , a key transcription factor involved in IL-10 release (Liu et al., 2003; Yano et al., 1996).

Considering no spinal sources of IL-10 have been identified despite the demonstration that spinal IL-10 levels are anti-nociceptive, the question will be addressed as to whether spinal cord astrocytic TLR4 and glutamate receptor stimulation can result in IL-10 release. We report here that TLR4-stimulated IL-10 release occurs from spinal cord astrocytes and glutamate co-application significantly increases TLR4-stimulated IL-10 release.

4.2 MATERIALS AND METHODS

4.2.1 Spinal cord astrocyte cell cultures

Detailed protocol for the preparation of astrocyte cultures is given in section 2.2. Briefly, spinal cords from 0-2 day old Sprague-Dawley rats were dissected out and the meninges and peripheral nerves were removed. The remaining spinal cord tissue was enzymatically and mechanically dissociated then washed with DMEM+. Dissociated cells were plated on a poly-D-lysine (20 µg/mL) coated culture flask for 4 hours in a 5% CO2 incubator at 37°C, were washed and then maintained at 5% CO2 and 37°C in DMEM+, replaced every 3 - 4 days. To purify astrocytes, 14 days after initial plating, flasks were shaken for 40 min at 400 rpm and 37°C and this was repeated 20 days after initial plating for 16 h. Additionally, 16 days after initial plating DMEM+ was replaced with DMEM supplemented by 10% horse serum, 1% penicillin/streptomycin/glutamine and 25 mM sorbitol. After 21 days cells were transferred and plated onto plastic 24-well plates for ELISA and LDH assay experiments or 13 mm diameter round glass coverslips for immunocytochemistry experiments at a density of 1 x 10⁵ cells per well or coverslip. For RT-PCR experiments cells were plated on 35 mm diameter plastic Petri dishes at a density of 6.77×10^5 cells per dish. Cells were used for ELISA, immunocytochemistry or LDH assay experiments 1 to 2 days after transfer. The purity of cultures was verified by immunocytochemistry.

4.2.2 Sandwich ELISA

Detailed protocol for the measurement of IL-10 by ELISA is given in section 2.6. Briefly, after a 16 h incubation of astrocytes in agonists or antagonists and DMEM/2% BSA, medium was collected and centrifuged. Ninety-six well clear polystyrene microplates were coated in anti-rat IL-10 monoclonal antibody (1:250) at 4°C overnight. Wells were washed three times for 5 min each wash with 0.05 % Tween® 20 in PBS and incubated for 1 h in 1 % BSA in PBS. Wells were then washed three times for 5 min a wash with 0.05 % Tween® 20 in PBS. Immediately after centrifugation, medium samples were added to wells along with standard concentrations of IL-10 diluted in DMEM/2% BSA. After a 2 h incubation at room temperature wells were washed five times for 5 min each wash with 0.05 % Tween® 20 in PBS. Wells were incubated for 2 h in a biotinylated anti-rat IL-10 antibody (1:250) and wells were washed five times for 5 min each wash with 0.05 % Tween® 20 in PBS. A streptavidin-horseradish peroxidase conjugate (1:250) was added to wells for 20 min and wells were washed five times for 5 min each wash with 0.05 % Tween 20 in PBS. Wells were incubated in a 1:1 mix of hydrogen peroxide and TMB for 20 min and 1 M sulfuric acid was added to stop the reaction. Absorbance of plates at 450 nm was read within 20 min of sulfuric acid addition using a Fluostar Galaxy Multiplate reader and wavelength correction was performed at 550 nm. A standard curve was constructed for each plate and was used to convert absorbance measurements for each condition into IL-10 concentration values.

Each experimental condition was tested on at least three different culture batches and at least two repetitions of measurement per batch giving a minimum n = 6, apart from select conditions in the time- and dose-response curves where n = 4. Values in timeand concentration-experiments, agonist and antagonist experiments were compared to the average level of LPS-stimulated IL-10 release. Results were presented as mean \pm SEM . Statistically significant differences between means were analysed with a onetailed Student's two-sample t-test assuming unequal variance, with p < 0.05considered as statistically significant.

4.2.3 Lactate Dehydrogenase Toxicology Assay

Detailed protocol for the assessment of membrane integrity with an LDH assay is given in section 2.7. Briefly, cell toxicity was assessed with a Lactate Dehydrogenase (LDH) based *In Vitro* Toxicology Assay Kit as per the manufacturer's protocol. Absorbance was measured at 490 nm with a Fluostar Galaxy Multiplate reader and wavelength correction was performed at 650 nm. The absorbance reading corresponding to each extracellular LDH level was divided by the absorbance reading for each total intracellular LDH level to obtain a percentage of total LDH released per condition in accordance with Dringen and Hamprecht (1996). Statistically significant differences between means were analysed with a one-tailed Student's two-sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant.

4.2.4 Reverse Transcription-Polymerase Chain Reaction

Detailed protocol for RT-PCR is given in section 2.8. Briefly, total RNA was extracted from astrocytes using a GenElute[™] Mammalian Total RNA Miniprep Kit, according to the manufacturer's protocol. The quantity of isolated RNA was determined spectroscopically using a NanoDrop ND-1000 Spectrophotometer. Extracted total RNA was reverse transcribed to cDNA using a SuperScript[™] III First-Strand Synthesis System and oligo(dT)20 primer according to the manufacturer's protocol. The potential for residual DNA contamination was examined by replacing the reverse transcriptase enzyme with RNAse free water. The cDNA was stored at -20°C until the polymerase chain reaction (PCR). Aliquots of first strand cDNA (40 ng) were amplified in 10 µL of Platinum SYBR-Green qPCR SuperMix-UDG with 0.8 µL of forward primer, 0.8 µL of reverse primer, made up to a total volume of 20 µL with diethylpyrocarbonate-treated H2O. PCR was carried out using the Rotor-Gene 3000 system. Incubation conditions were as follows: 2 min at 50°C, 2 min at 95°C, then thermo-cycling for 40 cycles of 20 s at 95°C, 20 s at 60°C then 20 s at 72°C. The threshold cycle number and reaction efficiencies of each condition were identified with Rotor-Gene 3000 v3.1 Software. Relative quantification was carried out using the relative expression software tool (Pfaffl et al., 2002) whereby IL-10 mRNA expression was standardized by the house-keeping gene 18S rRNA expression. Mean relative mRNA levels were determined with at least three separate reactions per condition and mean levels were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Values are expressed as mean relative IL-10 mRNA expression ± SEM.

4.2.5 Fluorescence Immunocytochemistry

Detailed protocol for the detection of cellular proteins using fluorescent antibodies is given in section 2.5. Briefly, 24 h after transfer of coverslips described in section 4.2.1, a number of coverslips were incubated in 1 µg/ml LPS for 24 h, while others were left in DMEM/BSA. Subsequently coverslips were washed twice with HEPES buffered salt solution, fixed with 2% formaldehyde solution in PBS for 10 min, washed twice with PBS and then incubated in PBS containing 1% BSA and 0.05% saponin for 30 min. Cells on coverslips were then incubated for 2 h at room temperature with primary antibodies diluted in PBS containing 1% BSA and 0.05% saponin. Primary antibodies were washed out 3 times with PBS and cells were then incubated in secondary antibodies diluted in PBS containing 1% BSA and 0.05% saponin for 1 hour at room temperature. Secondary antibodies were washed out three times with PBS and coverslips were then mounted on rectangular glass coverslips using ProLong Gold anti-fade mounting media with DAPI. Slides were sealed with nail polish and subsequently stored at 4°C until viewing.

Fluorescence was visualised with an inverted Axiovert 200M inverted microscope and images were acquired with an AxioCam HR digital monochrome charge-coupled device camera using Axiovision 3.1 software. A constant exposure time was used to acquire images assessed for brightness of TLR4 staining. Using Image J a constant arbitrary brightness threshold was applied to all these images and the number of cells with regions of brightness above this threshold as a percentage of total cells was counted per field of view. This percentage was averaged over a minimum of three different fields of view per condition and mean percentages were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Percentages were expressed as mean \pm SEM.
4.3 RESULTS

4.3.1 The TLR4 is present on spinal cord astrocytes

Although there is functional evidence for the presence of the TLR4 on spinal cord astrocytes (Pehar et al., 2004), to date immunocytochemical studies have not been carried out to verify the presence of the TLR4. Immunocytochemistry revealed that cultured spinal cord astrocytes positively expressed the TLR4 (Figure 4.1). In unstimulated astrocytes, $88 \pm 9\%$ of cells (n = 169 cells; 3 batches) stained brightly for TLR4 (as determined by use of an arbitrary brightness threshold detailed in section 2.5). Exposure of astrocytes to LPS (1 µg/ml, 24 h) significantly increased to 100% the percentage of spinal cord astrocytes that were classified as bright (n = 139 cells; 3 batches; p < 0.05).

4.3.2 LPS stimulates IL-10 release from spinal cord astrocytes and this release is enhanced on co-exposure to glutamate

Determination was next made as to whether glutamate or the TLR4 agonist LPS stimulated IL-10 release from spinal cord astrocytes. Astrocytes were pre-exposed to 1 µg/ml LPS for 8 h. A time period of 8 h was chosen as this duration of LPS exposure did not cause significant IL-10 release from brain astrocytes (Ledeboer et al., 2002). After an 8 h pre-incubation in LPS, a further 16 h exposure to LPS stimulated 561 ± 45 pg/ml IL-10 release (n = 65). After an 8 h pre-incubation in LPS, a further 16 h exposure to LPS + 1 mM glutamate stimulated significantly more IL-10 release than that stimulated by LPS only (872 ± 62 pg/ml; n = 65; p < 0.001;



Figure 4.1 The TLR4 is expressed on spinal cord astrocytes. Representative images of A) unstimulated spina 1 cord astrocytes and B) spinal cord astrocytes exposed to 24 h LPS (1 μ g/ml). The left column represents TLR4 staining in red. The middle column represents staining of the astrocyte intermediate filament marker glial fibrillary acidic protein (GFAP) in green. The right column displays a merged image of the left and middle columns. In the merged image, blue represents DAPI-staining of cell nuclei. The scale bars represent 20^µ m and hold for all images in the row.

Chapter 4



Figure 4.2 LPS stimulates IL-10 release from spinal cord astrocytes and this release is increased in the presence of glutamate. Treatments of astrocytes were as follows: basal – no addition of agonists (n = 57); Glu – no pre-incubation in LPS but a 16 h exposure to 1 mM glutamate (n = 10); LPS – 8 h pre-incubation in LPS (1 µg/ml) after which the media was removed and replaced with LPS for 16 h (n = 65); LPS + Glu – 8 h pre-incubation in LPS after which the media was removed and replaced with LPS + 1 mM glutamate for 16 h (n = 65). ***p<0.001.

Figure 4.2), whilst unstimulated astrocytes released 3.1 ± 3.3 pg/ml IL-10 (n = 57) and this value was not enhanced by the application of 1 mM glutamate for 16 h (3.4 ± 0.9 pg/ml; n = 10; Figure 4.2). The average percentage increase of LPS-stimulated IL-10 release in the presence of glutamate was $166 \pm 6 \%$ (n = 65).

To explore if an 8 h pre-incubation in LPS was necessary for LPS to stimulate IL-10 release or for the potentiating effect of glutamate co-application, agonists were applied with different pre-incubation conditions. Without a pre-incubation in LPS, it was found a 16 h application of 1 µg/ml LPS stimulated 211 ± 12 pg/ml IL-10 release (n = 8). When 1 mM glutamate was co-applied with LPS, this release significantly increased to 240 ± 8 pg/ml (n = 8; p < 0.05). If cells were subjected to an 8 h pre-incubation in 1 mM glutamate, the amount of IL-10 released from a subsequent 16 h exposure to LPS + glutamate was significantly higher than that seen if there was no pre-incubation (291 ± 18 pg/ml, n = 6, p < 0.01). As the highest levels of IL-10 release were seen with an 8 h pre-incubation in LPS, subsequent experiments were performed with an 8 h pre-incubation.

To verify the effect of LPS and glutamate on IL-10 release did not reflect compromised membrane integrity, an LDH assay was performed. The amount of LDH found in the extracellular medium as a percentage of total intracellular LDH did not significantly differ in unstimulated astrocytes over 16 h (3.78 ± 0.86 %; n = 4) as compared to astrocytes which were stimulated with LPS (1 µg/ml) and glutamate (1 mM) for 16 h after an 8 h LPS pre-incubation (4.42 ± 0.29 %; n = 4). In addition,

immunocytochemical analysis of spinal cord astrocyte cultures revealed only a small percentage of cells (8%, n = 284) were found to be positive for CD11b, an adhesion molecule found on microglia, and morphologically there appeared to be no oligodendrocytes present in the culture, suggesting that the IL-10 release observed was from astrocytes. The issue of culture purity is discussed further in section 6.1.7.

4.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate

Glutamate potentiated LPS-stimulated IL-10 release in a dose-dependent manner with an EC₅₀ of 400 μ M (Figure 4.3). After pre-incubation in LPS, when glutamate was applied for 16 h with 1 μ g/ml LPS, 1 μ M glutamate elicited 554 \pm 14 pg/ml IL-10 release (n = 4), 10 μ M glutamate elicited 563 \pm 7 pg/ml (n = 4) and 100 μ M glutamate elicited 601 \pm 18 pg/ml (n = 4). Compared to levels elicited without glutamate (561 \pm 45 pg/ml; n = 65), 500 μ M glutamate significantly increased IL-10 release to 782 \pm 13 pg/ml (n = 4; p < 0.001), as did 1 mM glutamate (872 \pm 62 pg/ml; n = 65) and 10 mM glutamate (928 \pm 38 pg/ml; n = 4; p < 0.001). To ensure a concentration above EC₅₀ was used, all subsequent experiments involving glutamate were carried out with a concentration of 1 mM. LPS was used at a concentration of 1 μ g/ml in subsequent experiments as this concentration is commonly used in the previous literature (for example Beck et al. (2008), Noga et al. (2008) and Seo et al. (2004)).



Figure 4.3 The modulation of LPS-stimulated IL-10 release by glutamate is dose-dependent. Astrocytes were pre-incubated in 1 μ g/ml LPS for 8 h. Media was removed and LPS with various concentrations of glutamate (Glu) were added. EC₅₀ was 400 μ M. For each concentration of glutamate, n was at least 4.

4.3.4 LPS-stimulated IL-10 release from spinal cord astrocytes and enhancement in the presence of glutamate is time-dependent

To examine if LPS-stimulated IL-10 release was dependent on the duration of LPS exposure, LPS was added for various times after a constant 8 h pre-incubation in LPS. A subsequent 2 h exposure to LPS produced IL-10 levels of 147 ± 9 pg/ml (n = 4), a 4 h exposure produced 210 ± 24 pg/ml (n = 6), an 8 h exposure produced 323 ± 41 pg/ml (n = 6), a 16 h exposure produced 561 ± 45 pg/ml (n = 65), a 24 h exposure produced 1094 ± 124 pg/ml (n = 6) and a 48 h exposure produced 2551 ± 353 pg/ml IL-10 (n = 6; Figure 4.4A).

To examine if the increase in IL-10 release on co-exposure to glutamate and LPS was also dependent on the duration of agonist exposure, LPS and glutamate were added for various times after a constant 8 h pre-incubation in LPS. A further 2 h co-exposure to LPS and glutamate produced IL-10 levels of 169 ± 3 pg/ml (n = 4), a 4 h co-exposure produced 244 ± 21 pg/ml (n = 6), an 8 h co-exposure produced 391 ± 45 pg/ml (n = 6), a 16 h co-exposure produced 872 ± 62 pg/ml (n = 65), a 24 h co-exposure produced 1466 ± 120 pg/ml (n = 6) and a 48 h co-exposure produced 2921 ± 225 pg/ml IL-10 (n = 6; Figure 4.4A).

The level of IL-10 release measured on exposure to glutamate and LPS was compared to the level stimulated on exposure to LPS alone for the same amount of time



Figure 4.4 The modulation of LPS-stimulated IL-10 release by glutamate is time-dependent. Astrocytes were pre-incubated in 1 μ g/ml LPS for 8 h. Media was removed and either LPS (1 μ g/ml) or LPS + glutamate (1 mM) was added for various durations. A) Summary histogram of the amount of LPS (1 μ g/ml; black column) and LPS+Glu (1 mM; white column)-stimulated IL-10 release over the differing time periods. B) Summary histogram of how the ratio of LPS+Glu to LPS-stimulated IL-10 release varied over the differing time periods. Values are expressed as a mean ratio of LPS + Glu release to LPS release ± SEM. The peak modulatory effect occurred when glutamate was applied with LPS for 16 h. For each condition, n was at least 4.

(Figure 4.4B). The relative increase of LPS + glutamate-stimulated IL-10 release beyond that stimulated by LPS alone was largest when agonists were added for 16 h $(1.66 \pm 0.06 \text{ fold higher}; n = 65)$. A 2 h exposure to LPS + glutamate elicited a 1.16 ± 0.08 fold higher IL-10 release than a 2 h exposure to LPS alone (n = 4). A 4 h exposure to agonists elicited a 1.21 ± 0.13 fold higher release (n = 6). An 8 h exposure elicited a 1.22 ± 0.03 fold higher release (n = 6). A 24 h exposure to agonists elicited a 1.36 ± 0.04 fold higher release (n = 6) and a 48 h exposure elicited a 1.23 ± 0.14 fold higher response (n = 6). Consequently, agonists were applied for 16 h in subsequent experiments to ensure maximal interaction between glutamate and LPS.

4.3.5 Transcription is key to the potentiating effect of glutamate exposure

In spinal cord astrocytes, glutamate is known to activate c/EBP β which is a transcription factor that can stimulate increased IL-10 production (Liu et al., 2003; Yano et al., 1996). Additionally, the 16 h optimal time of exposure to glutamate described above falls within IL-10 mRNA production time lengths (Ledeboer et al., 2002). An investigation was therefore made into whether transcription and subsequent translation were involved in the potentiating effect of glutamate presence.

Blocking mRNA transcription with actinomycin D (1 μ g/ml) did not significantly decrease LPS-stimulated IL-10 release (483 ± 53 pg/ml; n = 14) but significantly decreased LPS + glutamate-stimulated IL-10 release (428 ± 37 pg/ml; n = 14;

p < 0.001). Figure 4.5A illustrates that glutamate had no potentiating effect in the presence of actinomycin D. Blocking translation with cycloheximide (1 µg/ml) ablated both LPS-stimulated IL-10 release (13 ± 5 pg/ml; n = 10; p < 0.001) and LPS + glutamate-stimulated IL-10 release (11 ± 7 pg/ml; n = 10; p < 0.001; Figure 4.5A).

Incubation in actinomycin D and cycloheximide caused astrocytic processes to hypertrophy. An LDH assay was performed to verify that this did not reflect compromised membrane integrity or cell death. The amount of LDH found in the extracellular medium as a percentage of total intracellular LDH did not significantly differ in astrocytes incubated in actinomycin D with LPS + glutamate (4.51 ± 0.33 %; n = 4) compared to the percentage found in the medium of unstimulated astrocytes or astrocytes incubated in LPS + glutamate (as reported in section 4.3.2). Similarly, incubation in cycloheximide with LPS + glutamate did not generate significantly different amounts of released LDH (4.35 ± 0.06 %; n = 4) compared to unstimulated or LPS + glutamate-stimulated astrocytes.

The effect of glutamate on IL-10 mRNA expression in spinal cord astrocytes was investigated using RT-PCR to further explore the role of transcription. Glutamate on its own stimulated a 1.3 ± 1.6 (n = 3) fold increase in mRNA levels over those found in unstimulated astrocytes. LPS stimulated a 6.2 ± 2.7 (n = 3) fold increase whilst LPS + glutamate significantly increased expression levels beyond this LPS-stimulated level to 30.3 ± 7.9 fold above basal expression (n = 3; p < 0.05; Figure 4.5B).



Figure 4.5 Transcription is involved in the potentiating effect of glutamate. A) The effect of incubation in actinomycin D (Actino; transcription inhibitor; 1 µg/ml) and cycloheximide (Cyclo; translation inhibitor; 1 µg/ml) on LPS (1 µg/ml) and LPS plus glutamate (Glu; 1 mM)-stimulated IL-10 release. Whilst actinomycin D did not significantly effect LPS-stimulated IL-10 release (n = 14), cycloheximide ablated it (n = 10). Conversely, both actinomycin D (n = 14) and cycloheximide (n = 10) removed the potentiating effect of glutamate. B) IL-10 mRNA expression after a 24 h incubation in Glu (n = 3), LPS (n = 3) or LPS + Glu (n = 3) normalised to 18S rRNA expression and relative to basal IL-10 expression *p<0.05, ***p<0.001.

4.3.6 Glutamate acts on multiple mGluR groups to bring about IL-10 release in the presence of LPS

Agonists and antagonists to glutamate receptor subtypes were used to examine the role of AMPA, kainate, NMDA and mGluRs in the potentiation of IL-10 release seen in the presence of glutamate and LPS. tACPD, an mGluRI and II receptor agonist, increased LPS-stimulated IL-10 release across a range of concentrations. The ratio of IL-10 release in the presence of 10 μ M tACPD + LPS compared to LPS alone was 1.25 \pm 0.08 (n = 6). This ratio for 100 μ M tACPD was 1.58 \pm 0.06 (n = 6; Figure 5.6A) and for 1 mM tACPD was 1.84 \pm 0.03 (n = 6). Surprisingly, application of agonists for individual mGluR groups showed no potentiation of LPS-stimulated IL-10 release. The ratio of IL-10 release in the presence of 500 μ M DHPG (mGluRI agonist) + LPS compared to LPS alone was 1.05 \pm 0.02 (n = 6; Figure 5.6A). The ratio for the mGluRII agonist L-AP4 was 1.13 \pm 0.03 (1 mM; n = 6; Figure 5.6A). Kainate (100 μ M; 1.06 \pm 0.01; n = 6), AMPA (100 μ M; 1.10 \pm 0.03; n = 6) and NMDA (500 μ M, 1.10 \pm 0.02; n = 6) also had little impact on LPS-stimulated IL-10 release (Figure 5.6A).

The ratio of IL-10 release in the presence of LPS + glutamate + 1 μ M MPEP (mGluR5 antagonist) to IL-10 release in the presence of LPS + 1 μ M MPEP was 1.64 \pm 0.06 (n = 8; Figure 5.6B). This ratio for 75 μ M CPCCOEt (mGluR1 antagonist) was 1.85 \pm 0.09 (n = 13; Figure 5.6B) and for 500 μ M EGLU (mGluRII antagonist) was 1.63 \pm 0.09 (n = 9; Figure 5.6B). None of these ratios were

significantly different from the ratio of IL-10 release stimulated by LPS + glutamate to that stimulated by LPS alone $(1.66 \pm 0.06; n = 65;$ Figure 5.6B). However, mGluR antagonists that concurrently block multiple receptor groups significantly decreased the potentiation of glutamate on LPS-stimulated IL-10 release levels. The ratio of IL-10 release in the presence of LPS + glutamate + 1 mM MSPG (mGluRII and III antagonist) to IL-10 release in the presence of LPS + 1 mM MSPG was 1.40 ± 0.09 (n = 11; p <0.01; Figure 5.6B). The potentiation was further decreased by coexposure to 1 μ M MPEP and 500 μ M EGLU, which produced a ratio of 1.19 \pm 0.13 (n = 6; p < 0.01; Figure 5.6B) and co-exposure to 1 μ M MPEP, 500 μ M EGLU and 75 μ M CPCCOEt which produced a ratio of 1.13 \pm 0.13 (n = 6, p < 0.01; Figure 5.6B).

Ionotropic glutamate receptor antagonists had no significant effect on the increase in IL-10 release seen on co-application of glutamate and LPS. The ratio of IL-10 release stimulated by LPS + glutamate + 200 μ M AP5 (NMDA receptor antagonist) to IL-10 release stimulated by LPS + AP5 was 1.51 ± 0.09 (n = 6; Figure 5.6B) and the ratio was 1.72 ± 0.03 (n = 6; Figure 5.6B) for the AMPA and kainate receptor antagonist 20 μ M CNQX.



Glutamate potentiates LPS-stimulated IL-10 release by co-Figure 4.6 activating mGluR groups. A) The effect of glutamate receptor agonists glutamate (Glu; 1 mM; n = 65), tACPD (mGluRI and II agonist, 100µM; n = 6), DHPG (mGluRI agonist, 500 μ M; n = 6), APDC (mGluRII agonist, 5 μ M; n = 6), L-AP4 (mGluRIII agonist, 1 mM; n = 6), KA (100 μ M; n = 6), AMPA (100 μ M; n = 6) and NMDA (500 μ M; n = 6) on LPS (1 μ g/ml)-stimulated IL-10 release. Values are expressed as mean ratio of LPS + agonist-stimulated release to LPSstimulated release \pm SEM. B) The effect of glutamate receptor antagonists MPEP (mGluR5 antagonist, 1 μ M; n = 8), CPCCOEt (mGluR1 antagonist, 75 μ M; n = 13), EGLU (mGluRII antagonist, 500 μ M; n = 9), MSPG (mGluRII and III antagonist, 1 mM; n = 11), MPEP (1 μ M) + EGLU (500 μ M; n = 6), MPEP $(1 \mu M)$ + EGLU (500 μM) + CPCCOEt (75 μM ; n = 6), AP5 (NMDA receptor antagonist, 200 µM; n = 6) and CNQX (AMPA and kainate receptor antagonist, 20 µM; n = 6) on glutamate + LPS-stimulated IL-10 release. Values are expressed as mean ratio of LPS + Glu + antagonist to LPS + antagonist ± SEM. **p<0.01.

4.4 DISCUSSION

Unstimulated rat spinal cord astrocytes release very little IL-10 but increase the amount of IL10 release on exposure to LPS, reflecting what has been reported with rat forebrain astrocytes (Ledeboer et al., 2002). LPS-stimulated IL-10 release is significantly increased in the presence of glutamate in a dose and time-dependent manner, whilst glutamate itself lacked an effect on IL-10 release levels.

4.4.1 The interaction between glutamate and LPS involves upregulation of IL-10 mRNA

The increase in release of IL-10 in the presence of LPS and glutamate compared to that in the sole presence of LPS appears to require upregulation of IL-10 mRNA expression. In comparison to the level of IL-10 mRNA stimulated by LPS, co-exposure to glutamate greatly increased IL-10 mRNA expression. Additionally, maximum potentiation of IL-10 release occurred after 16 h, which is within the time required for IL-10 mRNA production (Ledeboer et al., 2002). Glutamate had no effect on LPS-stimulated IL-10 release when transcription was prevented.

It is intriguing that glutamate alone does not substantially increase IL-10 mRNA levels, despite the large enhancement seen on co-exposure with LPS. It has recently been discovered that in order for transcription factors to effectively access IL-10 promoter sites, there must be covalent modification of histones at the IL-10 locus (Lucas et al., 2005). In macrophages, LPS alone does not stimulate such modifications and so only produces moderate IL-10 mRNA and protein increases

despite activating a range of transcription factors (Lucas et al., 2005). Extracellular signal-related kinase (ERK) is a MAPK, the activation of which brings about the needed covalent modifications to render the IL-10 promoter accessible to relevant transcription factors. Activation of ERK during LPS exposure greatly increases the ability of LPS to enhance IL-10 mRNA and protein levels (Lucas et al., 2005). Moreover, it has been shown that activation of ERK on its own does not increase IL-10 release, indicating that ERK activation is necessary but not sufficient to potentiate IL-10 release (Lucas et al., 2005). In astrocytes, stimulation of mGluRI and II receptors results in phosphorylation and increased activity of ERK (D'Onofrio et al., 2001; Peavy et al., 2001; Schinkmann et al., 2000). Hence, although the exact mechanism of IL-10 mRNA increase is still to be determined, these factors suggest the hypothesis that glutamate may increases in IL-10 levels by itself but allowing LPS-induced transcription factors to more readily access IL-10 promoter sites. ERK inhibitors could be used to investigate this hypothesis.

It should be noted, however, that without such experiments, it can not be ruled out that TLR4 stimulation may be facilitating the ability of glutamate to bring about IL-10 release. As mentioned, glutamate is known to elicit c/EBP β activation in astrocytes (Peavy et al., 2001) and activation of this transcription factor can result in increased IL-10 production (Liu et al., 2003). An alternative hypothesis to explain the data may be that there is a threshold as to the level of c/EBP β activation required to affect IL-10 production and glutamate on its own does not increase activation of

c/EBPβ above this threshold. When glutamate is co-applied with LPS, however, the amount of c/EBPβ stimulated by LPS on its own in combination with that elicited by glutamate is enough to result in the observed enhancement of IL-10 mRNA and protein levels.

4.4.2 Transcriptional activation by LPS does not contribute to LPSstimulated IL-10 levels

IL-10 release resulting from TLR4 stimulation of spinal cord astrocytes does not require the process of transcription. Spinal cord astrocytes display low constitutive IL-10 mRNA levels which are upregulated by LPS, in correspondence with previous RT-PCR studies on rat and mouse cortical astrocytes (Ledeboer et al., 2002; Mizuno et al., 1994). Even though LPS upregulates IL-10 mRNA levels, this upregulation did not contribute to the increased IL-10 release as inhibition of transcription did not significantly reduce LPS-stimulated IL-10 release. Rather, release was ablated by blockade of translation. This result suggests that the LPS-induced upregulation of IL-10 mRNA was not large enough to result in a measurable difference in IL-10 release and that LPS must exert its enhancing effects on IL-10 release in spinal cord astrocytes by a post-transcriptional mechanism. Post-transcriptional modifications in the process of IL-10 release have been documented. The 3' end of IL-10 mRNA contains an untranslated region with a number of mRNA destabilising sequences. IL-10 mRNA constructed to lack this region has a much longer half life than that which contains it (Nemeth et al., 2005). In macrophages, adenosine potentiates LPSstimulated IL-10 release by inducing cytoplasmic proteins to bind to the destabilising

regions of the 3' untranslated end of IL-10 mRNA, relieving the associated transcriptional repression (Nemeth et al., 2005). The possibility that LPS may have affected the 3' untranslated end of IL-10 mRNA to result in the increased IL-10 release observed in the present work could be tested in further studies by using luciferase constructs with the sequence of the 3' untranslated end of IL-10 mRNA, as in the study by Nemeth et al. (2005).

4.4.3 Co-ligation of group I and II mGluRs is necessary for the increase in LPS-stimulated IL-10 release in the presence of glutamate

The increase in LPS-stimulated IL-10 release in the presence of glutamate exclusively required group I, II and possibly III mGluRs. Ionotropic glutamate receptor agonists did not enhance LPS-stimulated IL-10 release and antagonists to ionotropic receptors did not decrease the potentiating effect of glutamate. Large increases in LPS-stimulated IL-10 release were only seen when both mGluRI and II were stimulated together and the potentiating effect of glutamate presence was only reversed when antagonists to mGluR5, 2 and 3 were co-applied. When agonists and antagonists to mGluR I or II receptors were added separately, no significant effects were found suggesting co-activation of mGluR2, 3 and 5 is necessary. A small contribution of mGluRIII cannot be ruled out due to the small increase in IL-10 release elicited by an mGluRIII agonist and residual IL-10 release remaining after LPS and glutamate-stimulated IL-10 release was antagonized with mGluR2, 3 and 5 antagonists. This is congruent with reports of the expression of mGluRI subunits (mGluR 1 and 5) and

mGluRII subunits (mGluR 2 and 3) on spinal cord astrocytes by Silva et al (1999) and in this thesis in section 3.3.4. Although no reports exist of the mGluRIII on unperturbed spinal cord astrocytes, expression has been identified in astrocytes in pathological conditions such as in multiple sclerosis (Geurts et al., 2005). It is unknown if exposure to a chronic pain-inducing pathology results in upregulation of the mGluRIII.

It remains a matter of speculation why co-activation of both mGluRI and II are necessary for the effects of glutamate on LPS-stimulated IL-10 release. Stimulation of the mGluRII class of metabotropic receptor has been shown to potentiate mGluRIstimulated IP₃ accumulation in hippocampal and cerebral slices without itself affecting IP₃ accumulation (Mistry et al., 1998). This may be relevant to the current study if ERK activation is involved in the potentiation as postulated above. In astrocytes, one pathway of phosphorylation of ERK after stimulation of mGluR5 and mGluRII involves stimulation of intracellular Ca²⁺ increases and PKC activation subsequent to PLC- β activation (Schinkmann et al., 2000) and presumably IP₃ production. Perhaps stimulation of either mGluR class on its own does not enhance ERK activation via IP₃ generation sufficiently and the reported synergy in mGluR5 and mGluR II-mediated IP₃ production is necessary to bring about the required levels of ERK activation to cause increased IL-10 release.

Alternatively, it could be conceptualised that co-activation of both mGluRI and II may be necessary if the astrocytic densities of each of these subtypes were low.

Although the relative densities of mGluRI and II on spinal cord astrocytes are unknown, it may be possible that activation of each individual subtype could produce a response below the threshold of IL-10 detection if their densities were low. Furthermore, if co-stimulated, this may be enough to generate IL-10 release levels above the detectable threshold level.

Although the reason remains unclear as to why co-activation of mGluRI and II is necessary in this present study, another report of this phenomenon exists in the literature. Co-activation of mGluRI and mGluRII is necessary for the induction of long term depression in the perirhinal cortex at resting potentials (Cho et al., 2002). This present study provides further evidence for a functional interaction between these two groups of mGluR.

4.4.4 Spinal cord astrocytes express TLR4 and this expression is increased by LPS exposure

The TLR4 has been identified on the surface of astrocytes in the brain (Bowman et al., 2003; Bsibsi et al., 2002; Jou et al., 2006). In the existing literature there is functional evidence for the presence of the TLR4 on spinal cord astrocytes whereby it was demonstrated that LPS could stimulate nerve growth factor release from these cells (Pehar et al., 2004). In the current work the presence of the TLR4 on cultured spinal cord astrocytes was further confirmed immunocytochemically.

Exposure to LPS led to a significant increase in the number of highly expressing-TLR4 spinal cord astrocytes. In other areas of the nervous system such as the cortex and brain, exposure of astrocytes to TLR4 agonists has been reported to result in a down-regulation of TLR4 mRNA and protein in some studies (Bowman et al., 2003; Jou et al., 2006) and has been reported to not affect TLR4 mRNA in other studies (Laflamme and Rivest, 2001; Lehnardt et al., 2002). It is unclear why TLR4s on spinal cord astrocytes respond to LPS differently to astrocytes from other areas of the brain, however reports exist of a regional heterogeneity in the response of spinal cord and cortical astrocytes to transmitters (Cholewinski et al., 1988; Marriott et al., 1991), as detailed more fully in section 1.4.1.3. In the present study, a higher level of surface TLR4 after LPS pre-incubation may be one reason why pre-incubation in LPS produced a higher level of LPS-stimulated IL-10 release to that seen without LPS preincubation.

4.4.5 Summary of mechanisms

In summary, it can be seen that TLR4 stimulation of spinal cord astrocytes results in an increase in IL-10 mRNA expression and IL-10 release. The increase in IL-10 release is not dependent on the increase in IL-10 mRNA expression, presumably suggesting TLR4 stimulation brings about increased IL-10 release via posttranscriptional mechanisms. This increase in IL-10 release on LPS application is greatly enhanced in the presence of glutamate, whilst glutamate itself does not affect IL-10 protein levels. This potentiation in the presence of glutamate requires coactivation of mGluRI and II (and possibly III) receptors, resulting in a synergistic

Chapter 4

increase in the expression of IL-10 mRNA in the presence of LPS and glutamate and subsequently the observed increase in IL-10 release. These pathways are summarised in Figure 4.7.





Figure 4.7 A model of the mechanisms of LPS-stimulated IL-10 release in the absence and presence of glutamate from spinal cord astrocytes. A) Binding of the TLR4 by LPS results in an increase in IL-10 release, presumably by a post-transcriptional mechanism. TLR4 activation also results in activation of transcription but this does not contribute to the increase in IL-10 release. B) Co-activation of mGluRI and II and possibly III receptors at the same time as TLR4 activation results in an increase in IL-10 mRNA levels. One explanation may be that this occurs through mGluR-mediated enhancement of LPS-stimulated IL-10 transcription. The result is an increase in IL-10 release. Glutamate itself does not directly affect IL-10 mRNA levels. IL-10 release mechanisms are not clear.

Chapter 5

CHAPTER 5

SPINAL MICROGLIAL IL-10 RELEASE ON ACTIVATION OF THE TLR4 AND THE EFFECT OF CONCOMITANT GLUTAMATE RECEPTOR ACTIVATION

Contents

5.2 MATERIALS AND METHODS1935.2.1 Spinal cord microglial cell cultures1935.2.2 Sandwich ELISA1945.2.3 Lactate Dehydrogenase Toxicology Assay1955.2.4 Reverse Transcription-Polymerase Chain Reaction1965.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.2.1 Spinal cord microglial cell cultures1935.2.2 Sandwich ELISA1945.2.3 Lactate Dehydrogenase Toxicology Assay1955.2.4 Reverse Transcription-Polymerase Chain Reaction1965.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co-1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of1935.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the2035.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.2.2 Sandwich ELISA1945.2.3 Lactate Dehydrogenase Toxicology Assay1955.2.4 Reverse Transcription-Polymerase Chain Reaction1965.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co-1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.2.3 Lactate Dehydrogenase Toxicology Assay1955.2.4 Reverse Transcription-Polymerase Chain Reaction1965.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.2.4 Reverse Transcription-Polymerase Chain Reaction1965.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.2.5 Fluorescence immunocytochemistry1975.3 RESULTS1995.3.1 The TLR4 is present on spinal cord microglia1995.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
5.3 RESULTS 199 5.3.1 The TLR4 is present on spinal cord microglia 199 5.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate 199 5.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate 203 5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 205 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves 205
5.3.1 The TLR4 is present on spinal cord microglia 199 5.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co- exposure to glutamate 199 5.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate 203 5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 205 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves 205
5.3.1 The table of place of on spinal cord microglia which is enhanced on co- exposure to glutamate1995.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate2035.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent2055.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves205
exposure to glutamate 199 5.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate 203 5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 205 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
5.3.3 The enhancement of IL-10 release by glutamate is dependent on the concentration of glutamate 203 5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 205 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
203 glutamate 5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
presence of glutamate is time-dependent 5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves
transcription 207
5.3.6 Glutamate acts on multiple glutamate receptor subtypes to bring about increased IL-10
release in the presence of LPS 210
5.4 DISCUSSION 213
5.4 1 The interaction between LPS and glutamate involves IL-10 mRNA upregulation 213
5.4.2 Chutamate activates AMPA, NMDA, mGlu and possibly KA receptors to interact with LPS
and enhance II -10 release 215
5 A 3 Presence of the TI R4 on spinal cord microglia 216
5.4.4 Summary of the mechanisms of IL-10 release from spinal cord microglia on exposure to
IPS and alutamate 216

5.1 INTRODUCTION

As detailed in sections 1.3.2 and 4.1, IL-10 is an anti-inflammatory cytokine with important anti-nociceptive effects. To date, the demonstration in Chapter 4 that stimulation of spinal cord astrocyte TLR4s results in release of IL-10 is the only documented report of spinal IL-10 release. It is of interest to identify further possible sources of IL-10 to get a more complete picture of processes that occur during pain and to identify other novel targets for intervention in chronic pain treatment.

It is possible that TLR4-stimulated IL-10 release described from spinal cord astrocytes may also occur from spinal cord microglia. Cultured spinal cord microglia have functional TLR4s (Ogata et al., 2003) and also p38, a key MAPK involved in IL-10 transcription activation (Svensson et al., 2003). In addition to TLR4-stimulated IL-10 release from spinal cord microglia, it is appealing to suggest that the neurotransmitter glutamate may be involved in IL-10 release from microglia in the spinal cord, as it is with astrocytes, as described in Chapter 4. In inflammatory conditions such as chronic pain, microglia are exposed to increased levels of glutamate (Juranek and Lembeck, 1997; Kawamata and Omote, 1996; Sasaki et al., 1998; Somers and Clemente, 2002; Sorkin et al., 1992) and spinal cord microglia express AMPA, NMDA and mGlu receptors (Liu et al., 2006), allowing them to respond to these increased levels of glutamate. When glutamate binds receptors on activated microglia anti-inflammatory effects such as a decrease in microglial activation and a decrease in microglial-induced neurotoxicity have been described (Taylor et al., 2003). In addition to these anti-inflammatory effects, exposure of

spinal cord microglia to glutamate results in activation of p38 (Ejarque-Ortiz et al., 2007; Tikka and Koistinaho, 2001), suggesting a possible effect of glutamate on IL-10 levels.

Considering this evidence, this present study will examine if spinal cord microglial TLR4 and glutamate receptor stimulation produce IL-10 release. Results show that glutamate significantly increased TLR4-stimulated IL-10 release from microglia, dependent upon an upregulation of IL-10 mRNA.

5.2 MATERIALS AND METHODS

5.2.1 Spinal cord microglial cell cultures

Detailed protocol for the preparation of spinal cord microglial cultures is given in section 2.2. Briefly, spinal cords from 0-2 day old Sprague-Dawley rats were dissected out and the meninges and peripheral nerves were removed. The remaining spinal cord tissue was enzymatically and mechanically dissociated then washed with Dissociated cells were plated on a poly-D-lysine (20 µg/mL) coated DMEM+. culture flask for 4 h in a 5% CO2 incubator at 37°C, were washed and then maintained at 5% CO₂ and 37°C in DMEM+, replaced every 3-4 days. To purify microglia, 14 days after initial plating, flasks were shaken for 40 min at 400 rpm and 37°C. Media containing detached glial cells were removed and centrifuged at 1500 rpm for 5 min. The glial conditioned medium supernatant was kept at 37°C and the pellet washed-twice with 1 ml DMEM+ and then plated onto plastic 24-well plates for ELISA and LDH assay experiments, 13 mm diameter round glass coverslips for immunocytochemistry experiments or 25 mm diameter plastic Petri dishes. Cells were plated at a density of 1×10^5 cells per well or coverslip or 1×10^6 cells per dish. All cell-adherent surfaces were pre-coated in 20 µg/ml poly-D-lysine. After 15 min, cells were washed and a 1:1 mix of DMEM+ and glial conditioned medium was added. Cells were used 1 to 2 days after this for experiments. The purity of cultures was verified by immunocytochemistry.

5.2.2 Sandwich ELISA

Detailed protocol for the measurement of IL-10 with ELISA is given in section 2.6. Briefly, after a 16 h incubation of microglia in agonists or antagonists and DMEM/2% BSA, medium was collected and centrifuged. Ninety-six well clear polystyrene microplates were coated in anti-rat IL-10 monoclonal antibody (1:250) at 4°C overnight. Wells were washed three times for 5 min each wash with 0.05 % Tween® 20 in PBS and incubated for 1 h in 1 % BSA in PBS. Wells were then washed three times for 5 min a wash with 0.05 % Tween® 20 in PBS. Immediately after centrifugation, medium samples were added to wells along with standard concentrations of IL-10 diluted in DMEM/2% BSA. After a 2 h incubation at room temperature wells were washed five times for 5 min each wash with 0.05 % Tween® 20 in PBS. Wells were incubated for 2 h in a biotinylated anti-rat IL-10 antibody (1:250) and wells were washed 5 times for 5 min each wash with 0.05 % Tween® 20 in PBS. A streptavidin-horseradish peroxidase conjugate (1:250) was added to wells for 20 min and wells were washed 5 times for 5 min each wash with 0.05 % Tween 20 in PBS. Wells were incubated in a 1:1 mix of hydrogen peroxide and tetramethylbenzidine for 20 min and 1 M sulfuric acid was added to stop the reaction. Absorbance of plates at 450 nm was read within 20 min of sulfuric acid addition using a Fluostar Galaxy Multiplate reader and wavelength correction was performed at 550 nm. A standard curve was constructed for each plate and was used to convert absorbance measurements for each condition into IL-10 concentration values.

Each experimental condition was tested on at least three different culture batches and at least two repetitions of measurement per batch giving a minimum n = 6, apart from select conditions in the time- and dose-response curves where n = 4. Values in time- and concentration-experiments, agonist and antagonist experiments were normalised to the average level of LPS-stimulated IL-10 release. Results were presented as mean \pm SEM . Statistically significant differences between means were analysed with a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant.

5.2.3 Lactate Dehydrogenase Toxicology Assay

Detailed protocol for the assessment of membrane integrity with an LDH assay is given in section 2.7. Briefly, cell toxicity was assessed with a Lactate Dehydrogenase (LDH) based *In Vitro* Toxicology Assay Kit as per the manufacturer's protocol. Absorbance was measured at 490 nm with a Fluostar Galaxy Multiplate reader and wavelength correction was performed at 650 nm. The absorbance reading corresponding to each extracellular LDH level was divided by the absorbance reading for each total intracellular LDH level to obtain a percentage of total LDH released per condition in accordance with Dringen and Hamprecht (1996). Statistically significant differences between means were analysed with a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant.

5.2.4 Reverse Transcription-Polymerase Chain Reaction

Detailed protocol for RT-PCR is given in section 2.8. Briefly, total RNA was extracted from microglia using a GenElute[™] Mammalian Total RNA Miniprep Kit, according to the manufacturer's protocol. The quantity of isolated RNA was determined spectroscopically using a NanoDrop ND-1000 Spectrophotometer. Extracted total RNA was reverse transcribed to cDNA using a SuperScript[™] III First-Strand Synthesis System and oligo(dT)20 primer according to the manufacturer's protocol. The possibility of residual DNA contamination was tested for by replacing the reverse transcriptase enzyme with RNAse free water. The cDNA was stored at -20°C until the polymerase chain reaction (PCR). Aliquots of first strand cDNA (40 ng) were amplified in 10 µL of Platinum SYBR-Green qPCR SuperMix-UDG with 0.8 µL of forward primer, 0.8 µL of reverse primer, made up to a total volume of 20 µL with diethylpyrocarbonate-treated H2O. PCR was carried out using the Rotor-Gene 3000 system. Incubation conditions were as follows: 2 min at 50°C, 2 min at 95°C, then thermo-cycling for 40 cycles of 20 s at 95°C, 20 s at 60°C then 20 s at 72°C. The threshold cycle number and reaction efficiencies of each condition were identified with Rotor-Gene 3000 v3.1 software. Relative quantification was carried out using the relative expression software tool (Pfaffl et al., 2002) whereby IL-10 mRNA expression was standardized by the house-keeping gene 18S rRNA expression. Mean relative mRNA levels were determined with at least 3 separate reactions per condition and mean levels were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Values are expressed as mean relative IL-10 mRNA expression ± SEM.

5.2.5 Fluorescence immunocytochemistry

Detailed protocol for the detection of cellular proteins using fluorescent antibodies is Briefly, 24 h after transfer of coverslips prepared in given in section 2.5. section 4.2.1, a number of coverslips were incubated in 1 µg/ml LPS for 24 h, while others were left in DMEM/BSA. Subsequently coverslips were washed twice for 5 min each wash HEPES buffered salt solution. Cells on coverslips were then fixed with 2% paraformaldehyde solution in PBS for 10 min. Coverslips were washed twice for 5 min each with PBS and then incubated in PBS containing 1% BSA and 0.05% saponin for 30 min. Cells on coverslips were then incubated for 2 h at room temperature with primary antibodies diluted in PBS containing 1% BSA and 0.05% saponin. Primary antibodies were washed out three times for 10 min each with PBS and then cells were incubated in secondary antibodies diluted in PBS containing 1% BSA and 0.05% saponin for 1 hour at room temperature. Secondary antibodies were washed out three times with PBS for 10 min each wash. Excess moisture was removed from the back of coverslips and they were mounted on rectangular glass coverslips using ProLong Gold anti-fade mounting media with DAPI. Slides were sealed with nail polish and subsequently stored at 4°C until viewing. Fluorescence was visualised with an inverted Axiovert 200M inverted microscope and images were acquired with an AxioCam HR digital monochrome charge-coupled device camera at a constant exposure time using Axiovision 3.1 software. Using Image J a constant arbitrary brightness threshold was applied to all these images and the number of cells with regions of brightness above this threshold as a percentage of total cells was counted per field of view. This percentage was averaged over a minimum of three

different fields of view per condition and mean percentages were compared using a one-tailed Student's two sample t-test assuming unequal variance, with p < 0.05 considered as statistically significant. Percentages were expressed as mean \pm SEM.

5.3 RESULTS

5.3.1 The TLR4 is present on spinal cord microglia

There is functional evidence for the presence of the TLR4 on spinal cord microglia (Ogata et al., 2003). However the presence of the TLR4 protein on the surface of spinal cord microglia is yet to be demonstrated by immunocytochemistry. In the present study, $85 \pm 3\%$ of microglial cells examined (n = 157 cells; 3 batches) stained brightly for TLR4 (as determined by use of an arbitrary brightness threshold; Figure 5.1). Pre-incubation in 1 µg/ml LPS for 24 h did not significantly alter the percentage of brightly stained cells ($83 \pm 5\%$; n = 158 cells; 3 batches).

5.3.2 LPS stimulates IL-10 release from spinal cord microglia which is enhanced on co-exposure to glutamate

The ability of glutamate and the TLR4 agonist LPS to stimulate IL-10 release from activated spinal cord microglia was explored. Microglia were pre-incubated in 1 µg/ml LPS for 8 h as it takes at least 4 h for microglia to display an activated morphology on exposure to LPS (Sugaya et al., 1998). After this pre-incubation, a further 16 h exposure to LPS stimulated release of 174 ± 17 pg/ml IL-10 (n = 59) while a 16 h exposure to LPS + 1 mM glutamate significantly enhanced IL-10 release as compared to release elicited by LPS alone (266 ± 22 pg/ml; n = 59; p < 0.001; Figure 5.2). Conversely, unstimulated microglia released 31 ± 9 pg/ml IL-10 over 16 h (n = 53) and this value was not enhanced by the application of 1 mM glutamate for 16 h (26 ± 8 ; n = 7; Figure 5.2). The average percentage increase of IL-10 release



Figure 5.1 The TLR4 is expressed on spinal cord microglia. Representative images of unstimulated spinal cord microglia. The left panel displays TLR4 staining in red. The middle panel represents staining in green of CD11b, an adhesion molecule found on microglia. The right column displays a merged image of the left and middle columns. In the merge d image blue represents DAPI-staining of cell nuclei. The scale bar represents 20^{4} m and holds for all images in the row.

Chapter 5



Figure 5.2 Glutamate significantly potentiates LPS-stimulated IL-10 release from spinal cord microglia. Treatments of microglia were as follows: basal – no addition of agonists (n = 53); Glu – a 16 h exposure to 1 mM glutamate (n = 7); LPS – 8 h pre-incubation in LPS (1 µg/ml) after which the media was removed and replaced with LPS for 16 h (n = 59); LPS + Glu – 8 h pre-incubation in LPS after which the media was removed and replaced with LPS + 1 mM glutamate for 16 h (n = 59). ***p<0.001.
in the presence of glutamate was 172 ± 7 % (n = 59). Immunocytochemistry revealed there were no astrocytes present in cultures displaying enhanced IL-10 release (n = 234) and no cells displayed the morphology of oligodendrocytes. All cells examined were CD11b positive (n = 234), and although this does not rule out the possibility that blood-born macrophages may have been in cultures, the morphology of cells resembled that of microglia as opposed to blood-born macrophages. Additionally, it has been reported that > 99 % of mononuclear cells isolated from murine central nervous system tissue are microglia (Giulian et al., 1995), strongly suggesting that the cell type which displayed LPS and glutamate-stimulated IL-10 release in these experiments was microglia.

Agonists were applied with different pre-incubation conditions in order to explore if an 8 h pre-incubation in LPS was necessary for LPS-stimulated IL-10 release and the potentiating effect of glutamate presence. Without prior exposure to LPS, a 16 h application of 1 µg/ml LPS stimulated 12 \pm 7 pg/ml IL-10 release (n = 4). This release significantly increased to 48 \pm 2 pg/ml (n = 4; p < 0.01) when 1 mM glutamate was co-applied with LPS. If cells were exposed to an 8 h pre-incubation in 1 mM glutamate, then the amount of IL-10 released from a subsequent 16 h exposure to LPS + glutamate was significantly higher than that seen if there was no preincubation (39 \pm 5 pg/ml, n = 6, p < 0.01). Ensuing experiments were performed with an 8 h pre-incubation in LPS as the highest levels of IL-10 release were seen with this time of pre-incubation.

To verify that release of IL-10 on exposure to LPS and glutamate was not a result of an increase in cell death, an LDH assay was conducted. The amount of LDH accumulated in the extracellular medium of unstimulated cells over 16 h as a percentage of the total intracellular level of LDH (13.30 \pm 2.52 %; n = 6) did not significantly differ from the percentage for cells exposed to LPS (1 µg/ml) and glutamate (1 mM) for 16 h (7.85 \pm 1.77 %; n = 6).

5.3.3 The enhancement of IL-10 release by glutamate is dependent

on the concentration of glutamate

LPS-stimulated IL-10 release was potentiated by glutamate in a dose-dependent manner with an EC₅₀ of 103 μ M (Figure 5.3). After pre-incubation in LPS, when glutamate was applied for 16 h with 1 μ g/ml LPS, 1 μ M glutamate elicited 189 ± 12 pg/ml IL-10 release (n = 4), 10 μ M elicited 154 ± 13 pg/ml (n = 4), 100 μ M elicited 217 ± 22 pg/ml (n = 4) and 500 μ M elicited 245 ± 39 pg/ml IL-10 release (n = 4). Glutamate at 1 mM enhanced LPS-stimulated release of IL-10 to 266 ± 22 pg/ml (n = 59) which is significantly greater than IL-10 release stimulated by LPS alone (174 ± 17 pg/ml; p < 0.05) and at 10 mM glutamate enhanced release to 264 ± 26 pg/ml (n = 4; p < 0.05). To ensure a concentration above EC₅₀ was used, all proceeding experiments involving glutamate were carried out with a glutamate concentration of 1 mM. As mentioned in section 4.3.3, LPS was used at a concentration of 1 μ g/ml in subsequent experiments as this concentration is commonly used in the previous literature (for example Beck et al. (2008), Noga et al. (2008) and Seo et al. (2004)).



Figure 5.3 The modulation of LPS-stimulated IL-10 release by glutamate is dose-dependent. Microglia were pre-incubated in 1 μ g/ml LPS for 8 h. Media was removed and LPS with various concentrations of glutamate (Glu) was added. EC₅₀ was 103 μ M. For all concentrations, n was at least 4.

5.3.4 LPS-stimulated IL-10 release from spinal cord microglia and enhancement in the presence of glutamate is time-dependent

In section 4.3.4 it was seen that LPS-stimulated IL-10 release from spinal cord astrocytes was time-dependent. It was examined whether LPS-stimulated IL-10 release from spinal cord microglia was also dependent on the duration of LPS application. After an 8 h pre-incubation in LPS, a subsequent 2 h exposure to LPS produced IL-10

levels of 48 ± 7 pg/ml (n = 6), a 4 h exposure produced 92 ± 19 pg/ml (n = 6), an 8 h exposure produced 132 ± 21 pg/ml (n = 6), a 16 h exposure produced 174 ± 17 pg/ml (n = 65), a 24 h exposure produced 387 ± 30 pg/ml (n = 6) and a 48 h exposure produced 690 ± 16 pg/ml IL-10 (n = 6; Figure 5.4A).

To determine if the enhancement of IL-10 levels in the presence of glutamate and LPS were similarly time-dependent, LPS and glutamate were added for various times after an 8 h pre-incubation in LPS. Further co-exposure to LPS and glutamate for 2 h produced IL-10 levels of 46 \pm 7 pg/ml (n = 6), a 4 h co-exposure produced 132 \pm 28 pg/ml, an 8 h co-exposure produced 167 \pm 32 pg/ml (n = 6), a 16 h co-exposure produced 266 \pm 22 pg/ml (n = 65), a 24 h co-exposure produced 540 \pm 85 pg/ml (n = 6) and a 48 h co-exposure produced 733 \pm 88 pg/ml IL-10 (n = 6; Figure 5.4A).

The level of IL-10 release elicited by co-exposure to LPS and glutamate was compared to the level stimulated on exposure to LPS alone for the same amount of





Figure 5.4 LPS-stimulated IL-10 release and the potentiating effect of glutamate is time-dependent. Microglia were pre-incubated in 1 μ g/ml LPS for 8 h. Media was removed and either LPS (1 μ g/ml) or LPS + glutamate (1 mM) were added for various durations. A) Summary histogram of the effect of various exposure times to LPS alone (1 μ g/ml; black columns) or LPS + Glu (1 mM; white columns). B) Summary histogram of the ratio of LPS-stimulated IL-10 release in the absence and presence of glutamate. The peak modulatory effect occurred when glutamate was applied with LPS for 16 h. Values are expressed as a mean ratio of LPS + Glu release to LPS release ± SEM. For all time-points, n was at least 4.

time (Figure 5.4B). The relative increase of LPS + glutamate-stimulated IL-10 release beyond that stimulated by LPS alone was largest when agonists were added for 16 h (1.72 ± 0.07 fold higher; n = 59; Figure 5.4B). A 2 h exposure to LPS + glutamate elicited a 0.98 \pm 0.10 fold change in IL-10 release compared to a 2 h exposure to only LPS (n = 6). A 4 h exposure to agonists elicited a 1.46 \pm 0.07 fold higher release (n = 6). An 8 h exposure elicited a 1.33 \pm 0.19 fold higher release (n = 6). A 24 h exposure to agonists elicited a 1.39 \pm 0.20 fold higher release (n = 6) and a 48 h exposure elicited a 1.08 \pm 0.14 fold higher response (n = 6). Consequently, agonists were applied for 16 h in subsequent experiments to ensure maximal IL-10 release.

5.3.5 LPS-stimulated IL-10 release and the potentiating effect of glutamate exposure involves transcription

In spinal cord microglia glutamate is known to activate p38, a MAPK involved in IL-10 mRNA production (Ejarque-Ortiz et al., 2007; Tikka and Koistinaho, 2001). Furthermore, the 16 h optimal time of exposure to glutamate reported in section 5.3.4 is within the time documented for IL-10 mRNA production (Ledeboer et al., 2002). Given this, it was investigated if transcription and translation were involved in the potentiating effect of glutamate.

LPS-stimulated IL-10 release ($72 \pm 4 \text{ pg/ml}$; n = 8; p < 0.001) and LPS + glutamatestimulated IL-10 release (62 ± 5 ; n = 8; p < 0.001) were significantly decreased by preventing mRNA transcription with actinomycin D (1 µg/ml). Glutamate had no



Figure 5.5 The potentiating effect of glutamate involves transcription. A) The effect of incubation in actinomycin D (Actino; transcription inhibitor; 1 µg/ml) and cycloheximide (Cyclo; translation inhibitor; 1 µg/ml) on LPS (1 µg/ml) and LPS plus glutamate (Glu; 1 mM)-stimulated IL-10 release. Both actinomycin D (n = 8) and cycloheximide (n = 8) significantly decreased LPS-stimulated IL-10 release. Similarly, both actinomycin D (n = 8) and cycloheximide (n = 8) significantly decreased the effects of glutamate. B) IL-10 mRNA expression after a 24 h incubation in Glu (n = 4), LPS (n = 3) or LPS + Glu (n = 3), normalised to 18S rRNA expression relative to unstimulated IL-10 mRNA expression. *p<0.05, ***p<0.001.

potentiating effect in the presence of actinomycin D, as shown in Figure 5.5A. Also, both LPS-stimulated IL-10 release ($6.62 \pm 1.63 \text{ pg/ml}$; n = 8; p < 0.001) and LPS + glutamate-stimulated IL-10 release ($7.06 \pm 2.38 \text{ pg/ml}$; n = 8; p < 0.001; Figure 5.5A) were ablated by preventing translation with cycloheximide (1 µg/ml).

Incubation in actinomycin D and cycloheximide caused the number of coverslipattached microglial cells to decrease. To ensure incubation in these inhibitors did not lead to increased cell death, an LDH assay was performed. The amount of LDH found in the extracellular medium as a percentage of total intracellular LDH was not significantly larger in microglia incubated in actinomycin D with LPS + glutamate $(13.91 \pm 1.96 \%; n = 4)$ from the percentage found in medium of unstimulated microglia $(13.30 \pm 2.52 \%; n = 6)$. Similarly, incubation in cycloheximide with LPS + glutamate did not generate significantly larger amounts of released LDH $(8.36 \pm 0.91 \%; n = 4)$ compared to unstimulated microglia.

RT-PCR was used to investigate the influence of glutamate on IL-10 mRNA expression in spinal cord microglia. Glutamate elicited a 2.42 ± 0.21 (n = 4) fold increase in mRNA levels over those found in unstimulated microglia. An 11.06 ± 5.21 (n = 3) fold increase was elicited by LPS. LPS + glutamate significantly increased expression levels beyond this LPS-stimulated level to 31.09 ± 7.26 fold above basal expression (n = 3; p < 0.05; Figure 5.5B).

5.3.6 Glutamate acts on multiple glutamate receptor subtypes to bring about increased IL-10 release in the presence of LPS

Agonists and antagonists were used to examine the role of AMPA, kainate, NMDA and mGluRs in the increase of IL-10 release seen on co-exposure to glutamate and LPS. A range of agonists to glutamate receptor subtypes potentiated LPS-stimulated IL-10 release. tACPD, a mGluRI and II agonist, increased LPS-stimulated IL-10 release across a variety of concentrations. The ratio of IL-10 release in the presence of 10 μ M tACPD + LPS compared to LPS alone was 1.36 ± 0.07 (n = 11). This ratio for 100 μ M tACPD was 1.61 ± 0.07 (n = 8; Figure 5.6A) and for 1 mM tACPD was 1.70 ± 0.10 (n = 6). The ratio of IL-10 release in the presence of 500 μ M DHPG (mGluRI agonist) + LPS compared to LPS alone was 1.50 ± 0.11 (n = 6; Figure 5.6A). The ratio for the mGluRII agonist APDC was 1.59 ± 0.24 (5 μ M; n = 6; Figure 5A). Contrastingly, for the mGluRIII agonist L-AP4 this ratio was 0.89 ± 0.08 (1 mM; n = 11; Figure 5.6A). Additionally, kainate (100 μ M; 1.22 ± 0.05 ; n = 11), AMPA (100 μ M; 1.28 ± 0.05 ; n = 7) and NMDA (500 μ M, 1.71 ± 0.17 ; n = 7) also showed increased levels of IL-10 release compared to the level seen on sole application of LPS (Figure 5.6A).

The potentiation of IL-10 release in the presence of LPS and glutamate was significantly decreased by mGluR antagonists. The ratio of IL-10 release in the presence of LPS + glutamate + 1 mM MSPG (mGluRII and III antagonist) to IL-10 release in the presence of LPS + 1 mM MSPG was 1.27 ± 0.10 (n = 11; p <0.001 compared to LPS + glutamate/LPS; Figure 5.6B). The potentiation was further

decreased by co-exposure to 1 μ M MPEP (mGluR 5 antagonist) and 75 μ M CPCCOEt (mGlu receptor 1 antagonist), producing a ratio of 1.06 \pm 0.16 (n = 6; p < 0.01; Figure 5.6B) and by co-exposure to 1 μ M MPEP, 75 μ M CPCCOEt and 500 μ M EGLU (mGluR group II antagonist), producing a ratio of 1.13 \pm 0.09 (n = 6; p < 0.001; Figure 5.6B).

lonotropic glutamate receptor antagonists also significantly decreased the potentiation by glutamate. The ratio of IL-10 release stimulated by LPS + glutamate + 40 μ M MK-801 (NMDA receptor antagonist) to IL-10 release stimulated by LPS + MK-801 was 1.31 ± 0.11 (n = 11; p < 0.01; Figure 5.6B) and this ratio was 1.11 ± 0.06 (n = 11; p < 0.001; Figure 5.6B) for the AMPA and kainate receptor antagonist 20 μ M CNQX.



Figure 5.6 Glutamate potentiates LPS-stimulated IL-10 release by activating a range of glutamate receptor subtypes. A) The effect of glutamate receptor agonists glutamate (Glu; 1 mM ; n = 59), tACPD (mGluRI and II agonist; 100µM ; n = 8), DHPG (mGluRI agonist; 500 μ M; n = 6), APDC (mGluRII agonist; 5 μ M; n = 6), L-AP4 (mGluRIII agonist; 1 mM; n = 11), KA (100 μ M; n = 11), AMPA (100 μ M ; n = 7) and NMDA (500 μ M ; n = 7) on LPS (1 μ g/ml)-stimulated IL-10 release. Values are expressed as mean ratio of LPS + agonist release to LPS release ± SEM. B) The effect of glutamate receptor antagonists M SPG (mGluRII and III antagonist; 1 mM ; n = 11), MPEP (mGluR5 antagonist; 1 μ M) + CPCCOEt (mGluR1 antagonist; 75 μ M ; n = 6), MPEP + CPCCOEt + EGLU (mGluRII antagonist, 500 µM ; n = 6), MK-801 (NMDA receptor antagonist, 40 μ M ; n = 11) and CNQX (AMPA and kainate receptor antagonist, 20 μ M ; n = 11) on glutamate + LPS-stimulated IL-10 release. Values are expressed as mean ratio of LPS + Glu + antagonist to LPS + antagonist ± SEM. **p<0.01, ***p<0.001.

5.4 DISCUSSION

In accordance with previous research on mouse, human and rat microglia from the brain, the present study shows unstimulated cultured rat spinal cord microglia release a small amount of IL-10 (Ledeboer et al., 2002; Mizuno et al., 1994; Williams et al., 1996). LPS stimulates IL-10 release from spinal cord microglia, as has been demonstrated with brain microglia (Ledeboer et al., 2002; Mizuno et al., 1994; Williams et al., 1994). In a dose- and time-dependent manner, the presence of glutamate significantly enhances the release of IL-10 on LPS exposure, without itself stimulating IL-10 release.

5.4.1 The interaction between LPS and glutamate involves IL-10 mRNA upregulation

Unstimulated spinal cord microglia displayed low constitutive IL-10 mRNA levels which were upregulated by LPS as previously reported for brain microglia (Ledeboer et al., 2002; Mizuno et al., 1994). LPS increased IL-10 release by working at a transcriptional level as the prevention of transcription greatly decreased the ability of LPS to stimulate IL-10 release. Prevention of transcription, however, did not completely ablate LPS-stimulated IL-10 release. This only occurred on blockade of translation, suggesting LPS also has post-transcriptional effects as discussed for astrocytes in section 4.4.2.

The interaction between glutamate and LPS resulting in enhanced IL-10 release appears to require the upregulation of LPS-induced IL-10 mRNA. IL-10 mRNA

levels were vastly increased by glutamate in the presence of LPS and glutamate had no effect on LPS-stimulated IL-10 release when transcription was prevented. Maximal potentiation in this present study occurred at 16 h after exposure to glutamate, within IL-10 mRNA production times (Ledeboer et al., 2002).

Despite the evidence that glutamate exposure led to increased IL-10 release via an enhancement of transcription, glutamate did not have a large impact on IL-10 mRNA levels on its own. This situation was similar to that described in section 4.4.1 with spinal cord astrocytes. As mentioned in section 4.4.1, ERK can enhance LPS-stimulated IL-10 mRNA levels and release, without alone increasing IL-10 release through covalent modification of histones at the IL-10 promoter locus (Lucas et al., 2005). It has not been examined whether glutamate activates ERK in spinal cord astrocytes (D'Onofrio et al., 2001; Peavy et al., 2001) and increases LPS-stimulated IL-10 mRNA without itself substantially increasing IL-10 mRNA levels in these cells, we hypothesise that glutamate also brings about an increase in LPS-stimulated IL-10 mRNA and release by activation of ERK. This could be tested in further experiments through use of ERK inhibitors.

Alternatively, the possibility exists that TLR4 stimulation may be enhancing the ability of glutamate to bring about IL-10 release. As mentioned in sections 1.4.2.4 and 5.1, glutamate activates p38, a MAPK which can lead to increased IL-10 production (Ejarque-Ortiz et al., 2007; Tikka and Koistinaho, 2001). Similar to what

was stated in section 4.4.1, perhaps there is a threshold as to the level of p38 activation required to affect IL-10 production and glutamate on its own does not increase activation of p38 above this threshold. When glutamate is co-applied with LPS, however, the amount of p38 stimulated by LPS on its own in combination with that elicited by glutamate may be enough to result in the observed increase of IL-10.

5.4.2 Glutamate activates AMPA, NMDA, mGlu and possibly KA receptors to interact with LPS and enhance IL-10 release

All glutamate receptor subtypes (apart from mGluRIII and possibly KA) contributed to the interaction with LPS. Antagonists to all subtypes significantly blocked the interaction and agonists to all subtypes (apart from mGluRIII) elicited IL-10 release in the presence of LPS. Spinal cord microglia have been shown to express AMPA, NMDA and mGlu receptors and although the presence of kainate receptors on spinal cord microglia has not been examined, microglia from the cortex and hippocampus have been shown to express kainate receptors (Christensen et al., 2006; Liu et al., 2006; Yamada et al., 2006). The functional significance of these kainate receptors is uncertain, however, as kainate-induced currents are completely ablated by a specific AMPA receptor blocker in cortical microglia after cell quiescence (Hagino et al., 2004; Yamada et al., 2006). In light of this, in the present study it can not be ruled out that the increase in IL-10 release seen on co-addition of kainate with LPS may be a result of kainate acting on AMPA receptors.

Although the effect of glutamate on microglia is not extensively investigated in the existing literature, there is another report of more than one subtype of glutamate receptor being involved in an effect of glutamate on microglia. All four receptor subtypes have been linked to the activation of the transcription factors c-fos and c-jun in microglia by glutamate (Eun et al., 2004). This present study is another example of the pantheic action of glutamate on its receptor subtypes in microglia.

5.4.3 Presence of the TLR4 on spinal cord microglia

The immunocytochemical identification of TLR4s on the surface of cultured spinal cord microglia in this study adds to the functional evidence that the TLR4 is present on the surface of spinal cord microglia (Ogata et al., 2003). In addition to the surface localization of the TLR4, a cytoplasmic localization was also apparent. LPS exposure could not redirect cytoplasmic TLR4 to the membrane, nor could it increase intensity of TLR4 staining. This is consistent with demonstrations that LPS exposure does not alter TLR4 mRNA expression in cortical microglia (Laflamme and Rivest, 2001; Lehnardt et al., 2002).

5.4.4 Summary of the mechanisms of IL-10 release from spinal cord microglia on exposure to LPS and glutamate

In conclusion, activation of TLR4s on spinal cord microglia results in an increase in IL-10 mRNA expression and subsequent increase in IL-10 release. This increase in IL-10 mRNA levels and IL-10 protein levels on LPS application is greatly enhanced in the presence of glutamate, whilst glutamate itself does not substantially affect

IL-10 mRNA expression and IL-10 protein levels. Glutamate acts on NMDA, AMPA and mGlu receptors to contribute to this increase, with a possible action on kainate receptors. These steps are summarized in Figure 5.7.

Chapter 5





Figure 5.7 A model of the mechanisms of LPS-stimulated IL-10 release in the absence and presence of glutamate from spinal cord microglia. A) TLR4 activation enhances IL-10 transcription and translation, resulting in IL-10 release. B) Co-activation of AMPA, NMDA, mGlu and possibly KA receptors at the same time as TLR4 activation results in an increase in IL-10 mRNA levels. One explanation may be that this occurs through glutamate receptor-mediated enhancement of LPS-stimulated IL-10 transcription. The result is an increase in IL-10 release. IL-10 release. Glutamate itself does not directly have a substantial effect on IL-10 mRNA levels. IL-10 release mechanisms are not clear.

Chapter 6

CHAPTER 6

GENERAL DISCUSSION

Contents

21
21
24
26
30
31
32
:34
:35
:35
238
:39
241

6.1 IMPLICATIONS OF FINDINGS

In Chapter 3 it is shown that spinal cord astrocytes release ATP in response to glutamate. This release is greatly potentiated in the presence of SP. In Chapters 4 and 5 it is shown that spinal cord microglia and astrocytes release IL-10 in response to TLR4 stimulation and this release is potentiated in the presence of glutamate. These findings may be applied to extend the current knowledge of spinal chronic pain mechanisms, general spinal processing and glial biology.

6.1.1 Potential implications of glutamate and SP-stimulated ATP release for the understanding of chronic pain mechanisms

By binding spinal P2X₄, P2X₃ and potentially P2X₇ receptors, ATP contributes to the production of the chronic pain symptoms hyperalgesia and allodynia (Chessell et al., 2005; Dorn et al., 2004; Tsuda et al., 2003). To date, however, the only confirmed source of ATP in the spinal cord in response to non-purinergic nociceptive substances was the report that microglia release ATP in response to glutamate (Liu et al., 2006). Results from the current research have identified further possible sources of spinal ATP in pain. SP and glutamate are released from nociceptive nerve terminals in the dorsal horn by relatively high frequencies of nerve stimulation (Allen et al., 1999; Honore et al., 2000; Sasaki et al., 1998), and their concomitant release enhances pain transmission in the dorsal horn (Ishizaki et al., 1999; Mjellem-Joly et al., 1991). Results from Chapter 3 suggest that one major source of spinal ATP release in pain may be from spinal cord astrocytes in response to the concomitant release of SP and glutamate during heightened activity of nociceptive peripheral afferent terminals.

Furthermore, the potentiation of glutamate-stimulated ATP release from spinal cord astrocytes may allow ATP levels in the spinal cord to be dependent on neuronal firing patterns. Generally, classical neurotransmitters such as glutamate are packaged into small synaptic vesicles for exocytotic release from neurons whilst peptides are released from large dense core vesicles (Morris et al., 2005). Whilst small synaptic vesicles can be released from neurons after a short burst of low frequency stimulation, large dense core vesicles containing peptides such as SP require either a prolonged low frequency stimulation or a short burst of high frequency stimulation (Whim and Lloyd, 1989). This allows a dissociation between the release of classical transmitters and peptides, for example, when Aplysia motor neurons are briefly stimulated at a low frequency they release only acetylcholine, whereas when they are stimulated at a high frequency they release both acetycholine and peptides (Whim and Lloyd, 1989). As such, peripheral afferent nociceptive fibres do not show a significantly measurable increase in SP release when stimulated with a short electrical pulse of 1 or 10 Hz (Marvizon et al., 1997). Rather, release of SP from these fibres is seen with a prolonged (8 min) stimulation at 1 Hz or on brief high frequency stimulation (Lever et al., 2001). In inflammatory hyperalgesia, peripheral afferents demonstrate a continuous low frequency firing of about 1 Hz. Prior to the induction of chronic pain these fibres do not display continuous discharges (Schaible and Schmidt, 1988). A similar pattern of low frequency continuous firing is seen in lesioned peripheral nerves. In addition, these fibres show a more prolonged after-discharge following noxious stimulation compared to that displayed in non-lesioned controls (Keller et al., 2007). Hence, it could be predicted that large neuroactive amounts of SP are only

released into the dorsal horn during a severe or prolonged noxious stimulation, such as in cases of inflammatory or neuropathic pain and not during acute pain. This hypothesis is supported by a study demonstrating that a low concentration of the C fibre activator capsaicin only caused a measurable release of SP if it was applied to neurons after inflammatory pain induction (Warsame Afrah et al., 2004). Consequently, it is likely in acute pain that astrocytes surrounding nociceptive nerve terminals may only be exposed to glutamate, whereas in neuropathic or inflammatory pain they may be exposed to both glutamate and substance P, leading to a much larger amount of ATP released in neuropathic and inflammatory pain as opposed to acute pain. This hypothesis of a 'switch like' mechanism of ATP release, in combination with other mechanisms (Herrero et al., 2000), may help explain the different outcomes of acute pain-producing versus chronic pain-producing stimuli.

The identification of SP and glutamate-mediated ATP release from spinal cord astrocytes as a potential major source of spinal ATP suggests this process may be a possible target in the quest for novel chronic pain treatments. Perhaps one way this process might be beneficially perturbed is by upregulating ecto-ATPases on spinal cord astrocyte membranes or intrathecally injecting ATP metabolizing compounds. This may be beneficial on two levels. Firstly the released ATP may be broken down before it can reach $P2X_4$ and $P2X_7$ receptors on microglia and $P2X_3$ receptors on neurons. Additionally, increasing the breakdown of ATP will increase the levels of the anti-nociceptive adenosine in the spinal cord (for a review of the anti-nociceptive actions of adenosine, see Li and Perl (1994)).

6.1.2 Potential implications of TLR4 and glutamate-stimulated IL-10 release for the understanding of chronic pain mechanisms

Chronic pain symptoms are abrogated by introduction of IL-10 into the spinal cord and high levels of IL-10 are common in people with a painless neuropathy whereas low levels are common in people with a painful neuropathy (Milligan et al., 2006a; Milligan et al., 2006b; Uceyler et al., 2007). Despite this, to date, endogenous sources of spinal IL-10 have not been identified. In neuropathic pain the TLR4 is activated. Results from Chapters 4 and 5 suggest that TLR4 activation of spinal cord astrocytes and microglia are potential sources of IL-10 in the spinal cord. Moreover, when TLR4 activation occurs in conjunction with glutamate receptor activation on these cells, the amount of IL-10 release is significantly increased. This suggests that the excitatory nervous system and immune system of the spinal cord can synergistically interact to form an endogenous protection mechanism against the deleterious inflammatory events that occur in the spinal cord during chronic pain. The present results also suggest that enhancing the interaction between these two systems may be a potential novel target in the search for a chronic pain treatment.

This demonstration of the anti-nociceptive effects of TLR4 stimulation should be considered alongside the pro-nociceptive effects of glial TLR4 stimulation. TLR4 stimulation elicits TNF- α , IL-1 β and IL-6 release from spinal cord glia (Benveniste et al., 1995; Heyen et al., 2000; Kremlev and Palmer, 2005; Ledeboer et al., 2000; Mizuno et al., 1994; Sawada et al., 1999). Release of such pro-inflammatory cytokines occurs soon after TLR4 stimulation. For example, TNF- α mRNA expression increases significantly within 30 min of microglial TLR4 activation (Kremlev and Palmer, 2005) and significant increases in IL-1 β release are seen within 1 h of microglial TLR4 activation (Sanz and Di Virgilio, 2000). Moreover, in animal models of pain increases in IL-1 β and its pro-inflammatory effects in the spinal cord are seen within 1 h of inflammatory pain induction (Sweitzer et al., 1999; Watkins et al., 1997) and the inflammatory effects of TNF- α in the spinal cord can be observed within 1 h of neuropathic pain induction (Milligan et al., 2003). Additionally, IL-6 release and IL-1 β release subsequent to TLR4 stimulation on microglia peaks at 6 h after TLR4 stimulation (Ledeboer et al., 2002; Sanz and Di Virgilio, 2000).

The timecourse of IL-10 release after TLR4 activation (and concomitant glutamate receptor stimulation) from glial cells is quite different to the timecourse of proinflammatory cytokine release. In the present study, it was not until 4 h and 8 h after TLR4 stimulation that substantial (> 150 pg/ml) amounts of IL-10 were released from astrocytes and microglia respectively. Levels of IL-10 continued to exponentially rise over the 48 h timecourse examined. This is in accordance with the timecourse of TLR4-mediated IL-10 release from cortical microglia (Ledeboer et al., 2002).

Considering the differences in timecourse between the pro- and anti-nociceptive actions of TLR4 stimulation, it could be hypothesised that after the induction of chronic pain TLR4 activation initially results in pro-nociceptive effects through the release of pro-inflammatory cytokines. As IL-10 release starts to increase in a delayed timeframe, it is hypothesised that the anti-inflammatory and anti-nociceptive

effects of TLR4 stimulation start to hinder the deleterious effects of the proinflammatory cytokines. In support of this, the peak in IL-1 β and IL-6 release at 6 h after TLR4 stimulation occurs at the time when IL-10 release is starting to become appreciable as demonstrated by the present study and that by Ledeboer et al. (2002), opening the possibility that the decrease in pro-inflammatory release could be a result of the inhibitory actions of IL-10. This hypothesis could be subsequently tested by measuring the timecourse of IL-6 and IL-1 β release in the presence of an anti-IL-10 antibody or IL-10R antagonist.

6.1.3 A proposed model of spinal events following noxious stimulation

Considering the discussions presented in sections 6.1.1 and 6.1.2 and what is known in the existing literature, a three-phase sequence of events in the spinal cord during chronic pain can be hypothesised. The release of glutamate and SP into the dorsal horn occurs immediately at the induction of chronic pain, when peripheral afferents are initially stimulated (Millan, 1999). This study suggests that immediately on exposure to glutamate and SP, spinal cord astrocytes release ATP. The same immediate timeframe is seen with release of ATP by spinal cord microglia on exposure to glutamate as reported by Liu et al (2006). This immediate release of ATP both from astrocytes in this study and microglia in the study by Liu et al (2006) peaks rapidly, within 15 min. Once released, ATP may then be able to bind P2X₄ receptors on microglia, P2X₃ receptors on neurons and possibly P2X₇ receptors on microglia to contribute to allodynia and hyperalgesia. These events are hypothesized to be the

first phase in spinal processing after noxious stimulation. Glutamate and SP are elevated in the dorsal horn weeks after injury (Allen et al., 1999; Kawamata and Omote, 1996). As the present study only examined the affects of acute exposure to SP and glutamate, it is not known if chronic exposure to these transmitters results in prolonged elevation of ATP.

It is not clear whether TLR4 agonists are present in the spinal cord as soon after noxious damage as SP and glutamate are. It is known that the TLR4 is activated one day after damage as anti-sense TLR4 oligodeoxynucleotide injection into the spinal cord one day after neuropathic damage effects pain behaviours (Tanga et al., 2005) but the effect of interfering with TLR4 expression less than 24 h after the induction of pain has not been examined. That being said, the activation of the TLR4 on spinal cord microglia and astrocytes within one day of induction of chronic pain is predicted to result in pro-inflammatory cytokine release (phase 2) followed by IL-10 release (phase 3), as discussed in section 6.1.2. In this way, it is predicted astrocytes and microglia are involved in bringing about both pro-nociceptive and delayed antinociceptive effects in response to noxious stimulation. Furthermore, it is predicted glutamate has both pro-nociceptive and delayed anti-nociceptive effects in the spinal cord in response to noxious stimulation. This hypothesis is illustrated in Figure 6.1. Chapter 6





Chapter 6



A three stage hypothesis of spinal processing after noxious Figure 6.1 stimulation in chronic pain. A) Phase 1 i) Substance P (SP) and glutamate (Glu) are released from dorsal horn neurons immediately after noxious stimulation. ii) On activation of NK receptors, mGluRs, AMPA receptors and NMDA receptors, ATP is released from astrocytes and microglia. iii) Spinal ATP can act on P2X4 and P2X₃ receptors to increase nociceptive transmission for example by stimulating release of brain-derived neurotrophic factor (BDNF) and its actions on the receptor TrkB. The result is allodynia and hyperalgesia. Although not on this diagram, if TLR4 activation occurs at the same time as ATP release, ATP may also act on P2X7 receptors on microglia to bring about IL-1ß release. Lines with circular endings denote receptor potentiation. B) Phase 2 i) Within a day of noxious stimulation, activation of TLR4s by an unknown agonist results in pro-inflammatory cytokine release from astrocytes and microglia, ii) which can increase nociceptive neuronal processing. C) Phase 3. i) After an initial pro-inflammatory cytokine release, TLR4 stimulation coupled with glutamate receptor stimulation produces a delayed release of IL-10 from microglia and astrocytes which can inhibit pro-inflammatory cytokine release, ii) decreasing nociceptive neuronal processing.

6.1.4 Potential implications of glutamate and SP-stimulated ATP release for the understanding of non-pain related spinal processing

SP and glutamate-containing nerve terminals are found in descending bulbospinal neurons in the ventral horn (Hokfelt et al., 2000) and in the intermediolateral column (Helke et al., 1982; Kapoor et al., 1992) of the spinal cord. The results of this present study suggest a consequence of SP and glutamate release in these areas may be an increased release of ATP from astrocytes. As these areas are rich in purinergic receptors (Deng and Fyffe, 2004; Deuchars et al., 2001; Miles et al., 2002) this may lead to an increase in purinergic transmission in these areas, with possible functional consequences such as increased phrenic motor innervation of the diaphragm (Miles et al., 2002) and increased motor neuron firing (Deng and Fyffe, 2004).

In addition, glutamate-stimulated ATP release from hippocampal astrocytes has been shown to be involved in a negative feedback effect on neuronal propagation (Zhang et al., 2003) and the spreading of intracellular Ca^{2+} concentration increases from astrocyte to astrocyte (Cornell-Bell et al., 1990; Guthrie et al., 1999; Wang et al., 2000). Termed an intercellular Ca^{2+} wave, this propagation has been implicated in the spreading of astrocytic-ATP mediated neuronal inhibition from the original site of stimulation (Koizumi et al., 2003), modulating changes in cerebral blood flow (Filosa and Blanco, 2007; Gordon et al., 2007) and uptake of glucose from the blood by astrocytes in response to neuronal activity (Bernardinelli et al., 2004). As glutamate is found in most synapses and astrocytes enwrap these synapses, the

demonstration of glutamate-stimulated ATP release from spinal cord astrocytes opens the possibility that astrocytes surrounding glutamatergic synapses in the spinal cord may affect neuronal propagation, astrocytic intercellular Ca²⁺ wave propagation, cerebral blood flow and uptake of glucose in response to neuronal activity.

6.1.5 Potential implications of TLR4 and glutamate-stimulated ATP

release for the understanding of non-pain related spinal processing

Glutamate is likely to potentiate IL-10 release induced by TLR4 activation from astrocytes and microglia in inflammatory conditions other than neuropathic pain, such as gram-negative bacterial infection and inflammatory neuronal damage. There are a number of reasons for this. Firstly, in such conditions TLR4 agonists are available. Gram-negative bacteria is a TLR4 agonist (Beutler, 2000) and neuronal damage and toxicity can cause membrane disruption and subsequent release of gangliosides, the TLR4-stimulating components of mammalian cell membranes (Jou et al., 2006). Secondly, levels of extracellular glutamate are elevated in such inflammatory conditions (Benveniste et al., 1984; Bezzi et al., 2001; Ko et al., 2003; Rego et al., 1996; Takeuchi et al., 2006) allowing the possibility that glutamate may rise to concentrations that potentiate LPS-induced IL-10 release from microglia and astrocytes.

In conditions of increased glutamate release and TLR4 activation, exposure to IL-10 has been shown to be beneficial. For example, dopaminergic neuronal degeneration resulting from exposure to gram-negative bacteria and pre-natal white matter damage

from maternal Escherecia Coli exposure can be decreased when infectious agents are co-applied with IL-10 (Mesples et al., 2003; Qian et al., 2006a; Qian et al., 2006b). IL-10 application at concentrations similar to that found released in the present study can decrease the amount of neuronal death when neuronal cultures are exposed to high levels of NMDA (Bachis et al., 2001). This suggests that one potential role of glutamate in such inflammatory conditions may be to restrict damage via potentiation of IL-10 levels. Indeed, the amount of neuronal death when neurons are exposed to LPS-activated astrocytes is decreased when these neurons are exposed to astrocytes which have had their mGluRs activated (Zhou et al., 2006). Similar neuroprotective effects are seen after microglial glutamate stimulation, for example, the level of neurotoxicity induced by LPS-activated microglia is greatly reduced when these microglia are stimulated with an mGluR agonist (Taylor et al., 2003). Results from this thesis may help explain part of the neuroprotective actions of glutamate in these studies. Consequently, artificial enhancement of the ability of glutamate to potentiate LPS-stimulated IL-10 release may be a therapeutic target in conditions such as infection and inflammatory neuronal damage.

6.1.6 A regional heterogeneity amongst astroctyes from the spinal cord and cortex

There is evidence in the literature for a functional difference between spinal cord and cortical astrocytes. For example SP can stimulate phosphoinositide turnover and prostaglandin release in spinal cord astrocytes but not in cortical astrocytes (Cholewinski et al., 1988; Marriott et al., 1991). Conversely, the neuropeptides

vasopressin and oxytocin can stimulate phosphoinositide turnover in cortical but not spinal cord astrocytes, even though both neuropeptides are found in the spinal cord and cortex and receptors to these neuropeptides have been identified on spinal cord astrocytes (Cholewinski et al., 1988; Condes-Lara et al., 2003; Hashimoto et al., 1985; Hosli and Hosli, 1992; Hosli et al., 1991). Additionally, a β-adrenergic agonist can increase cAMP in spinal cord and cortical astrocytes, however the hormone somatostatin blocks this response from cortical astrocytes but does not block it from spinal cord astrocytes (Cholewinski and Wilkin, 1988). It is unclear whether results presented in Chapter 4 support this idea of regional heterogeneity. In Chapter 4 it was found that LPS elicits IL-10 release from spinal cord astrocytes by acting through post-transcriptional mechanisms and not through transcription. In cortical astrocytes, LPS induces an increase in IL-10 mRNA expression and an activation of transcription factors in cortical astrocytes suggesting an action of LPS through transcription (Cao et al., 2006; Ejarque-Ortiz et al., 2007; Kelicen and Tindberg, 2004; Ledeboer et al., 2002; Ma et al., 2001; Mizuno et al., 1994; Seo et al., 2004; Williams et al., 1996). Nevertheless results in Chapter 4 show an increased mRNA expression does not necessitate increased IL-10 release, thereby clouding conclusions about the similarities of mechanism between cortical and spinal cord astrocytes.

On the other hand, the identification of AMPA as the sole receptor through which glutamate works to release ATP from spinal cord astrocytes described in Chapter 3 is in contrast with work done on cortical astrocytes by Queiroz et al (1997; 1999) as detailed in section 1.4.1.2. They found evidence that all glutamate receptor subtypes

were involved in glutamate-stimulated ATP release from cortical astrocytes. Additionally, AMPA-stimulated ATP release from spinal cord astrocytes required an influx of extracellular Ca^{2+} and was PLC-independent whereas AMPA-stimulated ATP release from cortical astrocytes was independent of Ca^{2+} influx and involved PLC (Queiroz et al., 1999). Furthermore, as explored in section 3.4.3, SP potentiates glutamate-stimulated ATP release from spinal cord but not cortical astroctyes. Hence this thesis provides support and further examples of the regional heterogeneity between spinal cord and cortical astroctyes.

6.1.7 A mechanistic heterogeneity amongst spinal cord astrocytes and microglia

In addition to the evidence for heterogeneity between spinal cord and cortical astrocytes, this thesis provides evidence for a heterogeneity in mechanisms between spinal cord astrocytes and microglia. Although spinal cord microglia and astrocytes both demonstrate glutamate-stimulated ATP release, LPS-stimulated IL-10 release, and show an enhancement of this release in the presence of glutamate, the mechanisms of each of these processes are different. Whereas spinal cord astrocytes require an influx of extracellular Ca²⁺ for glutamate-stimulated ATP release this is not involved in glutamate-stimulated ATP release from spinal cord microglia. Rather, this process requires a rise in intracellular Ca²⁺ levels and activation of PKC (Liu et al., 2006). LPS works at a transcriptional level to increase IL-10 release in spinal cord microglia but exerts its action post-transcriptionally in spinal cord astrocytes. Additionally, whereas glutamate works on multiple subtypes of microglial

glutamate receptors to enhance LPS-stimulated IL-10 release, only stimulation of mGluRs results in the effect in astrocytes.

Two points can be derived from this heterogeneity. Firstly, it provides a reassurance that observed protein release in astrocyte cultures was not due to contamination with microglia. For example, spinal cord astrocyte cultures were found to contain a small amount of microglia. The question may arise as to whether LPS-stimulated IL-10 release seen in astrocyte cultures was actually due to release from microglia. As microglial LPS-stimulated IL-10 release involved an enhancement of transcription but release from astrocyte cultures did not, this suggests that contaminating microglia did not contribute to observed IL-10 release from astrocyte cultures. Secondly, this heterogeneity suggests that it may be difficult to use the same treatment to alter the processes of glutamate-stimulated ATP release and glutamate and LPS-stimulated IL-10 release in both microglia and astrocytes. This needs to be kept in mind if these processes are to be novel targets in chronic pain treatment.

6.2 FURTHER QUESTIONS AND RESEARCH

6.2.1 Relevance to intact spinal cord function

Several hypotheses about the physiological and pathophysiological relevance of the results of this thesis have been raised in section 6.1 of this discussion. It must be remembered, when hypothesising about *in vivo* situations from the *in vitro* data of this research, that the properties and environment of cells in culture at room temperature may be different from those in the intact CNS. Amongst other differences, the

concentration of agonists in vivo may not be the same as those used in these It has been estimated that the concentration of synaptically released experiments. glutamate that reaches synapse-surrounding glia in vivo in the cerebellum is 190 µM (Dzubay and Jahr, 1999). In the present research, the EC₅₀ of glutamate-stimulated ATP release from spinal cord astrocytes was 240 μ M, which is in the vicinity of that reported by Dzubay and Jahr (1999). The EC₅₀ of glutamate in its interaction with LPS-stimulated IL-10 release from spinal cord microglia was about 100 µM, well below the level reported by Dzubay and Jahr (1999), although microglia do not wrap synapses as tightly as cerebellar glia and hence may not be exposed to as high a level of synaptic glutamate. The EC₅₀ of glutamate in its interaction with SP and with LPS on astrocytes was 347 μ M and 400 μ M respectively, which are values higher than the concentration reaching cerebellar glia on synaptic stimulation. In chronic pain and conditions of inflammation, extracellular levels of glutamate are increased. Additionally, synaptic glutamate is not the only source of endogenous extracellular glutamate with observations of release from astrocytes and microglia also reported (Chen et al., 2005; Takeuchi et al., 2006). Whether pooled levels of extracellular glutamate in pain and inflammation increase to those that allow potentiation of the effects of SP and LPS in vivo will need further investigation. Additionally, the concentration of SP and TLR4 agonists to reach glial membranes in vivo is unknown, so whether ATP and IL-10 releasing effects occur at in vivo concentrations will need further investigation.

Other possible differences between the *in vivo* and *in vitro* situations may be differences in density and expression of receptor subtypes. The expression of the TLR4, glutamate receptor subtypes, and neurokinin receptor subtypes, have been confirmed *in vitro* in this thesis and in work by Liu et al. (2005), Palma et al. (1997), Silva et al. (1999), Wienrich and Kettenmann (1989), Yashpal et al. (1990), Zaratin et al. (2000) and Zerari et al. (1997; 1998). Correspondingly, expression of glutamate receptor subtypes and the NK2 receptor on spinal cord astrocytes has been confirmed *in situ* (Aicher et al., 1997; Brand-Schieber and Werner, 2003; Zerari et al., 1998). On the other hand, it has not been investigated whether NK1 and NK3 receptors and the TLR4 are expressed on *in vivo* spinal cord astrocytes, and if glutamate receptor subtypes and the TLR4 are expressed on *in vivo* spinal cord microglia. Although beyond the scope of the present research, the expression of these receptors could be investigated in further studies by use of *in situ* immunohistochemistry.

As the similarity between the *in vitro* conditions of this present work and *in vivo* conditions is not known, the next step based on this research would be to investigate whether the processes shown in the current research occur in more physiological conditions. An exploration of whether SP potentiates glutamate-stimulated ATP release could be conducted *in vivo* by genetically engineering an animal to produce luciferase on the surface of its astrocytes, then focally injecting the CNS area of interest with luciferin. The fluorescence of the luciferase acting on ATP in the presence of luciferin could then be measured on stimulation of neurons that co-release SP and glutamate. Similarly, to explore whether the activation of TLR4s and
glutamate receptors on *in vivo* astrocytes and microglia results in IL-10 release, the cerebral spinal fluid IL-10 content of animals subject to intrathecal LPS and glutamate administration may be analysed with ELISA, as may the fluid of animals in conditions where there is increased glutamate and TLR4 activation such as neuropathic pain.

6.2.2 Effects of stimulation by and release of other pain-related substances

In chronic pain, glutamate is co-released with many different neurotransmitters, such as ATP, neurokinin-A, neurokinin-B and calcitonin gene related peptide (Millan, 1999; Sulzer et al., 1998). It would be interesting to investigate whether these neurotransmitters also modulate glutamate-stimulated ATP release and LPSstimulated IL-10 release, as the possibility exists that a number of different neurotransmitters can synergistically interact to contribute to the spinal pools of ATP and IL-10, providing further possible targets for chronic pain intervention. To investigate this, the experiments presented in this thesis could be repeated with these substances.

Chapters 4 and 5 of this thesis focused on the anti-inflammatory effect of glutamate on glia in pain. Whilst glutamate is well known as a CNS excitotoxin at high concentrations (Matute et al., 2006), has been reported to be able to stimulate IL-1 β release from astrocytes (Aronica et al., 2005) and has pro-inflammatory actions on resting microglia (Noda et al., 2000; Taylor et al., 2005), the exposure of astrocytes

and activated microglia to glutamate commonly results in neuroprotective or antiinflammatory outcomes (Bruno et al., 1998; Taylor et al., 2003; Venero et al., 2002; Zhou et al., 2006), as backed up by the results of Chapters 4 and 5 of this thesis. A further exploration of the anti-inflammatory capacities of glutamate could be conducted by exploring whether glutamate itself or in combination with LPS decreases pro-inflammatory cytokine secretion from astrocytes and activated microglia. Again, to investigate this, the experiments presented in Chapters 4 and 5 of this thesis could be repeated with measurement of pro-inflammatory cytokines.

6.2.3 Extension of findings to other cell types

It would be useful to see if the interaction between SP and glutamate occurred in other cell types, especially Schwann cells of the peripheral nervous system. Schwann cells have been shown to release ATP when stimulated with glutamate. This release may possibly affect the excitability of neurons (Liu and Bennett, 2003). When the periphery is subject to a noxious stimulus, glutamate and SP are released (Millan, 1999) and this release is in excess in chronic pain, allowing the possibility that if the potentiation of glutamate-stimulated ATP release by SP occurs in Schwann cells, excess glutamate and substance P released in chronic pain may lead to vast increases in ATP release from Schwann cells. As Schwann cells express $P2X_7$ receptors, exposure of these receptors to an excess of ATP may contribute to $P2X_7$ -mediated hyperalgesia (Chessell et al., 2005).

Additionally, it would be useful to see if the interaction between glutamate and the TLR4 occurred in other cell types. Again, one key cell type to investigate may be Schwann cells. Recently, functional TLR4s have been demonstrated on cultured Schwann cells which elicit TNF- α release on activation (Cheng et al., 2007). TNF- α has multiple roles after peripheral nerve injury. For example, it assists in macrophage recruitment, a key process in removing myelin debris after injury (Liefner et al., 2000) but also decreases Schwann cell viability in cultures (Myers et al., 2003). Although TNF-a may induce useful processes after axonal injury, the observation that TNF- α knockout mice show a higher number of preserved axons after the first 6 days of injury suggests that overall, TNF- α , has deleterious consequences after nerve injury within the first week after injury. Using immunohistochemistry to index the number of positive cells, it has been demonstrated that after a crush injury of the sciatic nerve there is an initial increase in cells expressing TNF- α which is reversed between 3 and 7 days after injury, the exact time where there is an increase in the number of cells expressing IL-10 (Sawada et al., 2007). If it can be shown that glutamate increases LPS-stimulated IL-10 release from Schwann cells, this may have implications for the process of axonal degeneration, repair and regeneration after peripheral nerve injury.

6.3 CONCLUSION

This thesis has provided evidence that spinal cord glia release both pro-nociceptive and anti-nociceptive compounds. Evidence was found that spinal cord astrocytes may contribute to the pool of the pro-nociceptive spinal ATP. By acting on AMPA receptors, glutamate was shown to stimulate a two-fold increase in ATP release from spinal cord astrocytes which was greatly potentiated in the presence of SP. SP brought about this potentiation by acting on all subtypes of neurokinin receptor, resulting in the ability of NMDA and mGlu receptor activation to result in ATP release. Additionally, this thesis has revealed that spinal cord microglia and astrocytes release the anti-nociceptive IL-10 on TLR4 activation and this release is enhanced in the presence of glutamate through an upregulation of IL-10 mRNA. These results can be integrated into a three-phase model of spinal chronic pain mechanisms which can be used as a basis for future research.

CHAPTER 7

REFERENCES

Abdi S, Lee DH and Chung JM (1998) The anti-allodynic effects of amitriptyline, gabapentin, and lidocaine in a rat model of neuropathic pain. *Anesthesia and analgesia* **87**(6):1360-1366.

Abdipranoto A, Liu GJ, Werry EL and Bennett MR (2003) Mechanisms of secretion of ATP from cortical astrocytes triggered by uridine triphosphate. *Neuroreport* **14**(17):2177-2181.

Abe T, Sugihara H, Nawa H, Shigemoto R, Mizuno N and Nakanishi S (1992) Molecular characterization of a novel metabotropic glutamate receptor mGluR5 coupled to inositol phosphate/Ca2+ signal transduction. *J Biol Chem* **267**(19):13361-13368.

Access Economics Pty Ltd (2007) The high price of pain: the economic impact of persistent pain in Australia, Report for MBF Foundation and the The University of Sydney Pain Management Research Institute, Canberra.

Agrawal SK and Fehlings MG (1997) Role of NMDA and non-NMDA ionotropic glutamate receptors in traumatic spinal cord axonal injury. *J Neurosci* 17(3):1055-1063.

Ahmed Z, Lewis CA and Faber DS (1990) Glutamate stimulates release of Ca2+ from internal stores in astroglia. *Brain Res* **516**(1):165-169.

Aicher SA, Sharma S, Cheng PY and Pickel VM (1997) The N-methyl-D-aspartate (NMDA) receptor is postsynaptic to substance P-containing axon terminals in the rat superficial dorsal horn. *Brain Res* 772(1-2):71-81.

Alberts B, Johnson A, Lewis J, Raff M, Roberts K and Walter P (eds) (2002) Cell Chemistry and Biosynthesis. Garland Science, New York.

Alexopoulou L, Holt AC, Medzhitov R and Flavell RA (2001) Recognition of double-stranded RNA and activation of NF-kappaB by Toll-like receptor 3. *Nature* **413**(6857):732-738.

Allen BJ, Li J, Menning PM, Rogers SD, Ghilardi J, Mantyh PW and Simone DA (1999) Primary afferent fibers that contribute to increased substance P receptor internalization in the spinal cord after injury. *J Neurophysiol* **81**(3):1379-1390.

Allen MJ, Myer BJ, Khokher AM, Rushton N and Cox TM (1997) Pro-inflammatory cytokines and the pathogenesis of Gaucher's disease: increased release of interleukin-6 and interleukin-10. Qjm **90**(1):19-25.

Aloisi F, Care A, Borsellino G, Gallo P, Rosa S, Bassani A, Cabibbo A, Testa U, Levi G and Peschle C (1992) Production of hemolymphopoietic cytokines (IL-6, IL-8, colony-stimulating factors) by normal human astrocytes in response to IL-1 beta and tumor necrosis factor-alpha. *J Immunol* **149**(7):2358-2366.

Andersen OK, Finnerup NB, Spaich EG, Jensen TS and Arendt-Nielsen L (2004) Expansion of nociceptive withdrawal reflex receptive fields in spinal cord injured humans. *Clin Neurophysiol* **115**(12):2798-2810.

Anderson CM and Nedergaard M (2006) Emerging challenges of assigning P2X7 receptor function and immunoreactivity in neurons. *Trends Neurosci* 29(5):257-262.

Anderson CM and Swanson RA (2000) Astrocyte glutamate transport: review of properties, regulation, and physiological functions. *Glia* **32**(1):1-14.

Aoki C, Venkatesan C, Go CG, Mong JA and Dawson TM (1994) Cellular and subcellular localization of NMDA-R1 subunit immunoreactivity in the visual cortex of adult and neonatal rats. *J Neurosci* 14(9):5202-5222.

Aramori I and Nakanishi S (1992) Signal transduction and pharmacological characteristics of a metabotropic glutamate receptor, mGluR1, in transfected CHO cells. *Neuron* **8**(4):757-765.

Araque A, Carmignoto G and Haydon PG (2001) Dynamic signaling between astrocytes and neurons. Annu Rev Physiol 63:795-813.

Aronica E, Gorter JA, Rozemuller AJ, Yankaya B and Troost D (2005) Activation of metabotropic glutamate receptor 3 enhances interleukin (IL)-1beta-stimulated release of IL-6 in cultured human astrocytes. *Neuroscience* **130**(4):927-933.

Arruda JL, Sweitzer S, Rutkowski MD and DeLeo JA (2000) Intrathecal anti-IL-6 antibody and IgG attenuates peripheral nerve injury-induced mechanical allodynia in the rat: possible immune modulation in neuropathic pain. *Brain Res* 879(1-2):216-225.

Attal N, Brasseur L, Guirimand D, Clermond-Gnamien S, Atlami S and Bouhassira D (2004) Are oral cannabinoids safe and effective in refractory neuropathic pain? *Eur J Pain* **8**(2):173-177.

Bachis A, Colangelo AM, Vicini S, Doe PP, De Bernardi MA, Brooker G and Mocchetti I (2001) Interleukin-10 prevents glutamate-mediated cerebellar granule cell death by blocking caspase-3-like activity. *J Neurosci* **21**(9):3104-3112.

Ball C, Vigues S, Gee CK, Poole S and Bristow AF (2001) Rat interleukin-10: production and characterisation of biologically active protein in a recombinant bacterial expression system. *Eur Cytokine Netw* **12**(1):187-193.

Ballanyani K (1995) Modulation of glial potassium, sodium, and chloride activities by the extracellular milieu in *Neuroglia* (Kettenmann H and Ransom B eds) pp 289-298, Oxford University Press, New York.

Ballerini P, Di Iorio P, Ciccarelli R, Nargi E, D'Alimonte I, Traversa U, Rathbone MP and Caciagli F (2002) Glial cells express multiple ATP binding cassette proteins which are involved in ATP release. *Neuroreport* **13**(14):1789-1792.

Bardoni R, Goldstein PA, Lee CJ, Gu JG and MacDermott AB (1997) ATP P2X receptors mediate fast synaptic transmission in the dorsal horn of the rat spinal cord. *J Neurosci* **17**(14):5297-5304.

Basu A, Lazovic J, Krady JK, Mauger DT, Rothstein RP, Smith MB and Levison SW (2005) Interleukin-1 and the interleukin-1 type 1 receptor are essential for the progressive neurodegeneration that ensues subsequent to a mild hypoxic/ischemic injury. *J Cereb Blood Flow Metab* **25**(1):17-29.

Beaujouan JC, Daguet de Montety MC, Torrens Y, Saffroy M, Dietl M and Glowinski J (1990) Marked regional heterogeneity of 125I-Bolton Hunter substance P binding and substance P-induced activation of phospholipase C in astrocyte cultures from the embryonic or newborn rat. *J Neurochem* **54**(2):669-675.

Benveniste EN (1992) Inflammatory cytokines within the central nervous system: sources, function, and mechanism of action. *Am J Physiol* **263**(1 Pt 1):C1-16.

Benveniste EN, Tang LP and Law RM (1995) Differential regulation of astrocyte TNF-alpha expression by the cytokines TGF-beta, IL-6 and IL-10. *Int J Dev Neurosci* **13**(3-4):341-349.

Benveniste H, Drejer J, Schousboe A and Diemer NH (1984) Elevation of the extracellular concentrations of glutamate and aspartate in rat hippocampus during transient cerebral ischemia monitored by intracerebral microdialysis. *J Neurochem* **43**(5):1369-1374.

Bernardinelli Y, Magistretti PJ and Chatton JY (2004) Astrocytes generate Na+mediated metabolic waves. *Proc Natl Acad Sci US A* **101**(41):14937-14942.

Bettler B and Mulle C (1995) Review: neurotransmitter receptors. II. AMPA and kainate receptors. *Neuropharmacology* **34**(2):123-139.

Beutler B (2000) Endotoxin, toll-like receptor 4, and the afferent limb of innate immunity. *Curr Opin Microbiol* **3**(1):23-28.

Bezzi P, Domercq M, Brambilla L, Galli R, Schols D, De Clercq E, Vescovi A, Bagetta G, Kollias G, Meldolesi J and Volterra A (2001) CXCR4-activated astrocyte glutamate release via TNFalpha: amplification by microglia triggers neurotoxicity. *Nat Neurosci* 4(7):702-710.

Bianco F, Pravettoni E, Colombo A, Schenk U, Moller T, Matteoli M and Verderio C (2005) Astrocyte-derived ATP induces vesicle shedding and IL-1 beta release from microglia. *J Immunol* **174**(11):7268-7277.

Black JA, Sontheimer H and Waxman SG (1993) Spinal cord astrocytes in vitro: phenotypic diversity and sodium channel immunoreactivity. *Glia* 7(4):272-285.

Bo X, Zhang Y, Nassar M, Burnstock G and Schoepfer R (1995) A P2X purinoceptor cDNA conferring a novel pharmacological profile. *FEBS Lett* **375**(1-2):129-133.

Borson DB (1991) Roles of neutral endopeptidase in airways. Am J Physiol 260(4 Pt 1):L212-225.

Borsook D (2003) Pain: the past, present and future. Adv Drug Deliv Rev 55(8):931-934.

Bourque MJ and Robitaille R (1998) Endogenous peptidergic modulation of perisynaptic Schwann cells at the frog neuromuscular junction. *J Physiol* 512 (Pt 1):197-209.

Bowman CC, Rasley A, Tranguch SL and Marriott I (2003) Cultured astrocytes express toll-like receptors for bacterial products. *Glia* **43**(3):281-291.

Boyd ZS, Kriatchko A, Yang J, Agarwal N, Wax MB and Patil RV (2003) Interleukin-10 receptor signaling through STAT-3 regulates the apoptosis of retinal ganglion cells in response to stress. *Invest Ophthalmol Vis Sci* **44**(12):5206-5211.

Brand-Schieber E, Lowery SL and Werner P (2004) Select ionotropic glutamate AMPA/kainate receptors are expressed at the astrocyte-vessel interface. *Brain Res* **1007**(1-2):178-182.

Brand-Schieber E and Werner P (2003) (+/-)-Alpha-amino-3-hydroxy-5methylisoxazole-4-propionic acid and kainate receptor subunit expression in mouse versus rat spinal cord white matter: similarities in astrocytes but differences in oligodendrocytes. *Neurosci Lett* **345**(2):126-130.

Bridges D, Ahmad K and Rice AS (2001) The synthetic cannabinoid WIN55,212-2 attenuates hyperalgesia and allodynia in a rat model of neuropathic pain. Br J Pharmacol 133(4):586-594.

Brugger F, Wicki U, Nassenstein-Elton D, Fagg GE, Olpe HR and Pozza MF (1990) Modulation of the NMDA receptor by D-serine in the cortex and the spinal cord, in vitro. *Eur J Pharmacol* **191**(1):29-38.

Bruno V, Battaglia G, Casabona G, Copani A, Caciagli F and Nicoletti F (1998) Neuroprotection by glial metabotropic glutamate receptors is mediated by transforming growth factor-beta. *J Neurosci* **18**(23):9594-9600.

Bsibsi M, Persoon-Deen C, Verwer RW, Meeuwsen S, Ravid R and Van Noort JM (2006) Toll-like receptor 3 on adult human astrocytes triggers production of neuroprotective mediators. *Glia* 53(7):688-695.

Bsibsi M, Ravid R, Gveric D and van Noort JM (2002) Broad expression of Toll-like receptors in the human central nervous system. *J Neuropathol Exp Neurol* **61**(11):1013-1021.

Burnstock G (2007a) Physiology and pathophysiology of purinergic neurotransmission. *Physiol Rev* 87(2):659-797.

Burnstock G (2007b) Purine and pyrimidine receptors. *Cell Mol Life Sci* 64(12):1471-1483.

Burnstock G and Knight GE (2004) Cellular distribution and functions of P2 receptor subtypes in different systems. *Int Rev Cytol* **240**:31-304.

Calakos N and Scheller RH (1996) Synaptic vesicle biogenesis, docking, and fusion: a molecular description. *Physiol Rev* **76**(1):1-29.

Campbell FA, Tramer MR, Carroll D, Reynolds DJ, Moore RA and McQuay HJ (2001) Are cannabinoids an effective and safe treatment option in the management of pain? A qualitative systematic review. *BMJ (Clinical research ed* **323**(7303):13-16.

Campbell JN and Meyer RA (2006) Mechanisms of neuropathic pain. Neuron 52(1):77-92.

Cao S, Zhang X, Edwards JP and Mosser DM (2006) NF-kappaB1 (p50) homodimers differentially regulate pro- and anti-inflammatory cytokines in macrophages. *J Biol Chem* **281**(36):26041-26050.

Caroff M and Karibian D (2003) Structure of bacterial lipopolysaccharides. Carbohydr Res 338(23):2431-2447.

Cascieri MA, Huang RR, Fong TM, Cheung AH, Sadowski S, Ber E and Strader CD (1992) Determination of the amino acid residues in substance P conferring selectivity and specificity for the rat neurokinin receptors. *Mol Pharmacol* **41**(6):1096-1099.

Chabot S, Williams G, Hamilton M, Sutherland G and Yong VW (1999) Mechanisms of IL-10 production in human microglia-T cell interaction. *J Immunol* **162**(11):6819-6828.

Chaby R (2004) Lipopolysaccharide-binding molecules: transporters, blockers and sensors. *Cell Mol Life Sci* 61(14):1697-1713.

Chao CC, Hu S, Close K, Choi CS, Molitor TW, Novick WJ and Peterson PK (1992) Cytokine release from microglia: differential inhibition by pentoxifylline and dexamethasone. *J Infect Dis* **166**(4):847-853.

Chatterton JE, Awobuluyi M, Premkumar LS, Takahashi H, Talantova M, Shin Y, Cui J, Tu S, Sevarino KA, Nakanishi N, Tong G, Lipton SA and Zhang D (2002) Excitatory glycine receptors containing the NR3 family of NMDA receptor subunits. *Nature* **415**(6873):793-798.

Chauvet N, Palin K, Verrier D, Poole S, Dantzer R and Lestage J (2001) Rat microglial cells secrete predominantly the precursor of interleukin-1beta in response to lipopolysaccharide. *Eur J Neurosci* 14(4):609-617.

Chen CC, Akopian AN, Sivilotti L, Colquhoun D, Burnstock G and Wood JN (1995) A P2X purinoceptor expressed by a subset of sensory neurons. *Nature* 377(6548):428-431.

Chen LW, Yung KK and Chan YS (2004) Neurokinin peptides and neurokinin receptors as potential therapeutic intervention targets of basal ganglia in the prevention and treatment of Parkinson's disease. *Curr Drug Targets* 5(2):197-206.

Chen X, Wang L, Zhou Y, Zheng LH and Zhou Z (2005) "Kiss-and-run" glutamate secretion in cultured and freshly isolated rat hippocampal astrocytes. *J Neurosci* **25**(40):9236-9243.

Cheng C, Qin Y, Shao X, Wang H, Gao Y, Cheng M and Shen A (2007) Induction of TNF-alpha by LPS in Schwann cell is regulated by MAPK activation signals. *Cell Mol Neurobiol* **27**(7):909-921.

Chessell IP, Hatcher JP, Bountra C, Michel AD, Hughes JP, Green P, Egerton J, Murfin M, Richardson J, Peck WL, Grahames CB, Casula MA, Yiangou Y, Birch R, Anand P and Buell GN (2005) Disruption of the P2X7 purinoceptor gene abolishes chronic inflammatory and neuropathic pain. *Pain* **114**(3):386-396.

Chiang BT, Liu YW, Chen BK, Wang JM and Chang WC (2006) Direct interaction of C/EBPdelta and Sp1 at the GC-enriched promoter region synergizes the IL-10 gene transcription in mouse macrophage. *J Biomed Sci* **13**(5):621-635.

Cho K, Brown MW and Bashir ZI (2002) Mechanisms and physiological role of enhancement of mGlu5 receptor function by group II mGlu receptor activation in rat perirhinal cortex. *J Physiol* **540**(Pt 3):895-906.

Cho Y and Bannai S (1990) Uptake of glutamate and cysteine in C-6 glioma cells and in cultured astrocytes. *J Neurochem* **55**(6):2091-2097.

Cholewinski AJ, Hanley MR and Wilkin GP (1988) A phosphoinositide-linked peptide response in astrocytes: evidence for regional heterogeneity. *Neurochem Res* **13**(4):389-394.

Cholewinski AJ and Wilkin GP (1988) Astrocytes from forebrain, cerebellum, and spinal cord differ in their responses to vasoactive intestinal peptide. *J Neurochem* **51**(5):1626-1633.

Christensen RN, Ha BK, Sun F, Bresnahan JC and Beattie MS (2006) Kainate induces rapid redistribution of the actin cytoskeleton in ameboid microglia. J *Neurosci Res* **84**(1):170-181.

Clark B and Mobbs P (1992) Transmitter-operated channels in rabbit retinal astrocytes studied in situ by whole-cell patch clamping. *J Neurosci* **12**(2):664-673.

Coco S, Calegari F, Pravettoni E, Pozzi D, Taverna E, Rosa P, Matteoli M and Verderio C (2003) Storage and release of ATP from astrocytes in culture. *J Biol Chem* 278(2):1354-1362.

Cohen MW, Gerschenfeld HM and Kuffler SW (1968) Ionic environment of neurones and glial cells in the brain of an amphibian. *J Physiol* **197**(2):363-380.

Cole R and deVellis J (1997) Astrocyte and oligodendrocyte cultures, in *Protocols for Neural Cell Culture* (Richardson SFaA ed) pp 117-130, Humana Press Inc., New Jersey.

Collins DR, Smith RC and Davies SN (1995) Interactions between arachidonic acid and metabotropic glutamate receptors in the induction of synaptic potentiation in the rat hippocampal slice. *Eur J Pharmacol* **294**(1):147-154.

Collo G, North RA, Kawashima E, Merlo-Pich E, Neidhart S, Surprenant A and Buell G (1996) Cloning OF P2X5 and P2X6 receptors and the distribution and properties of an extended family of ATP-gated ion channels. *J Neurosci* **16**(8):2495-2507.

Condes-Lara M, Gonzalez NM, Martinez-Lorenzana G, Delgado OL and Freund-Mercier MJ (2003) Actions of oxytocin and interactions with glutamate on spontaneous and evoked dorsal spinal cord neuronal activities. *Brain Res* 976(1):75-81.

Conklin D (2004) Recognition of the helical cytokine fold. J Comput Biol 11(6):1189-1200.

Connor JR and Berkowitz EM (1985) A demonstration of glial filament distribution in astrocytes isolated from rat cerebral cortex. *Neuroscience* **16**(1):33-44.

Conti F, Minelli A and Brecha NC (1994) Cellular localization and laminar distribution of AMPA glutamate receptor subunits mRNAs and proteins in the rat cerebral cortex. *J Comp Neurol* **350**(2):241-259.

Cornell-Bell AH, Finkbeiner SM, Cooper MS and Smith SJ (1990) Glutamate induces calcium waves in cultured astrocytes: long-range glial signaling. *Science* 247(4941):470-473.

Cotrina ML, Lin JH, Alves-Rodrigues A, Liu S, Li J, Azmi-Ghadimi H, Kang J, Naus CC and Nedergaard M (1998) Connexins regulate calcium signaling by controlling ATP release. *Proc Natl Acad Sci U S A* **95**(26):15735-15740.

Coull JA, Beggs S, Boudreau D, Boivin D, Tsuda M, Inoue K, Gravel C, Salter MW and De Koninck Y (2005) BDNF from microglia causes the shift in neuronal anion gradient underlying neuropathic pain. *Nature* **438**(7070):1017-1021.

Cristofaro P and Opal SM (2006) Role of Toll-like receptors in infection and immunity: clinical implications. Drugs 66(1):15-29.

Cull-Candy S, Brickley S and Farrant M (2001) NMDA receptor subunits: diversity, development and disease. *Curr Opin Neurobiol* **11**(3):327-335.

Cunningham MD, Seachord C, Ratcliffe K, Bainbridge B, Aruffo A and Darveau RP (1996) Helicobacter pylori and Porphyromonas gingivalis lipopolysaccharides are poorly transferred to recombinant soluble CD14. *Infect Immun* **64**(9):3601-3608.

D'Ascenzo M, Vairano M, Andreassi C, Navarra P, Azzena GB and Grassi C (2004) Electrophysiological and molecular evidence of L-(Cav1), N- (Cav2.2), and R- (Cav2.3) type Ca2+ channels in rat cortical astrocytes. *Glia* **45**(4):354-363.

D'Onofrio M, Cuomo L, Battaglia G, Ngomba RT, Storto M, Kingston AE, Orzi F, De Blasi A, Di Iorio P, Nicoletti F and Bruno V (2001) Neuroprotection mediated by glial group-II metabotropic glutamate receptors requires the activation of the MAP kinase and the phosphatidylinositol-3-kinase pathways. *J Neurochem* **78**(3):435-445.

Dam TV, Martinelli B and Quirion R (1990) Autoradiographic distribution of brain neurokinin-1/substance P receptors using a highly selective ligand [3H]-[Sar9,Met(O2)11]-substance P. *Brain Res* 531(1-2):333-337.

Das S, Sasaki YF, Rothe T, Premkumar LS, Takasu M, Crandall JE, Dikkes P, Conner DA, Rayudu PV, Cheung W, Chen HS, Lipton SA and Nakanishi N (1998) Increased NMDA current and spine density in mice lacking the NMDA receptor subunit NR3A. *Nature* **393**(6683):377-381.

De Biasi S and Rustioni A (1988) Glutamate and substance P coexist in primary afferent terminals in the superficial laminae of spinal cord. *Proc Natl Acad Sci USA* **85**:7820-7824.

Decker T and Lohmann-Matthes ML (1988) A quick and simple method for the quantitation of lactate dehydrogenase release in measurements of cellular cytotoxicity and tumor necrosis factor (TNF) activity. *J Immunol Methods* **115**(1):61-69.

Deitmer J (1995) pH Regulation, in *Neuroglia* (Kettenmann H and Ransom B eds), Oxford University Press, New York.

del Rio E, Shinomura T, van der Kaay J, Nicholls DG and Downes CP (1998) Disruption by lithium of phosphoinositide signalling in cerebellar granule cells in primary culture. *J Neurochem* **70**(4):1662-1669.

DeLeo JA, Colburn RW, Nichols M and Malhotra A (1996) Interleukin-6-mediated hyperalgesia/allodynia and increased spinal IL-6 expression in a rat mononeuropathy model. *J Interferon Cytokine Res* **16**(9):695-700.

DeLeo JA, Colburn RW and Rickman AJ (1997) Cytokine and growth factor immunohistochemical spinal profiles in two animal models of mononeuropathy. *Brain Res* **759**(1):50-57.

Deng Z and Fyffe RE (2004) Expression of P2X7 receptor immunoreactivity in distinct subsets of synaptic terminals in the ventral horn of rat lumbar spinal cord. Brain Res 1020(1-2):53-61.

Dersh J, Polatin PB and Gatchel RJ (2002) Chronic pain and psychopathology: research findings and theoretical considerations. *Psychosom Med* 64(5):773-786.

Deuchars SA, Atkinson L, Brooke RE, Musa H, Milligan CJ, Batten TF, Buckley NJ, Parson SH and Deuchars J (2001) Neuronal P2X7 receptors are targeted to presynaptic terminals in the central and peripheral nervous systems. *J Neurosci* **21**(18):7143-7152.

Dingledine R, Borges K, Bowie D and Traynelis SF (1999) The glutamate receptor ion channels. *Pharmacol Rev* **51**(1):7-61.

Dixon DR and Darveau RP (2005) Lipopolysaccharide heterogeneity: innate host responses to bacterial modification of lipid a structure. *J Dent Res* 84(7):584-595.

Donnelly RP, Dickensheets H and Finbloom DS (1999) The interleukin-10 signal transduction pathway and regulation of gene expression in mononuclear phagocytes. *J Interferon Cytokine Res* **19**(6):563-573.

Dorn G, Patel S, Wotherspoon G, Hemmings-Mieszczak M, Barclay J, Natt FJ, Martin P, Bevan S, Fox A, Ganju P, Wishart W and Hall J (2004) siRNA relieves chronic neuropathic pain. *Nucleic Acids Res* **32**(5):e49.

Drew GM, Mitchell VA and Vaughan CW (2005) Postsynaptic actions of substance P on rat periaqueductal grey neurons in vitro. *Neuropharmacology* **49**(5):587-595.

Dringen R and Hamprecht B (1996) Glutathione content as an indicator for the presence of metabolic pathways of amino acids in astroglial cultures. J Neurochem 67(4):1375-1382.

Dubyak GR (2007) Go it alone no more--P2X7 joins the society of heteromeric ATP-gated receptor channels. *Mol Pharmacol* 72(6):1402-1405.

Duffy PE (1983) Astrocytes : normal, reactive, and neoplastic. Raven Press, New York.

Dzubay JA and Jahr CE (1999) The concentration of synaptically released glutamate outside of the climbing fiber-Purkinje cell synaptic cleft. *J Neurosci* **19**(13):5265-5274.

Ejarque-Ortiz A, Medina MG, Tusell JM, Perez-Gonzalez AP, Serratosa J and Saura J (2007) Upregulation of CCAAT/enhancer binding protein beta in activated astrocytes and microglia. *Glia* 55(2):178-188.

Eun SY, Hong YH, Kim EH, Jeon H, Suh YH, Lee JE, Jo C, Jo SA and Kim J (2004) Glutamate receptor-mediated regulation of c-fos expression in cultured microglia. *Biochem Biophys Res Commun* **325**(1):320-327.

Everett PB and Senogles SE (2004) D3 dopamine receptor activates phospholipase D through a pertussis toxin-insensitive pathway. *Neurosci Lett* **371**(1):34-39.

Fam SR, Gallagher CJ, Kalia LV and Salter MW (2003) Differential frequency dependence of P2Y1- and P2Y2- mediated Ca 2+ signaling in astrocytes. *J Neurosci* **23**(11):4437-4444.

Fam SR, Gallagher CJ and Salter MW (2000) P2Y(1) purinoceptor-mediated Ca(2+) signaling and Ca(2+) wave propagation in dorsal spinal cord astrocytes. *J Neurosci* **20**(8):2800-2808.

Field MJ, Cox PJ, Stott E, Melrose H, Offord J, Su TZ, Bramwell S, Corradini L, England S, Winks J, Kinloch RA, Hendrich J, Dolphin AC, Webb T and Williams D (2006) Identification of the alpha2-delta-1 subunit of voltage-dependent calcium channels as a molecular target for pain mediating the analgesic actions of pregabalin. *Proc Natl Acad Sci U S A* **103**(46):17537-17542.

Fields RD and Burnstock G (2006) Purinergic signalling in neuron-glia interactions. Nat Rev Neurosci 7(6):423-436. Filosa JA and Blanco VM (2007) Neurovascular coupling in the mammalian brain. *Exp Physiol* **92**(4):641-646.

Finbloom DS and Winestock KD (1995) IL-10 induces the tyrosine phosphorylation of tyk2 and Jak1 and the differential assembly of STAT1 alpha and STAT3 complexes in human T cells and monocytes. *J Immunol* **155**(3):1079-1090.

Fitzgerald KA and Chen ZJ (2006) Sorting out Toll signals. Cell 125(5):834-836.

Fitzgerald KA, Rowe DC, Barnes BJ, Caffrey DR, Visintin A, Latz E, Monks B, Pitha PM and Golenbock DT (2003) LPS-TLR4 signaling to IRF-3/7 and NF-kappaB involves the toll adapters TRAM and TRIF. *J Exp Med* **198**(7):1043-1055.

Fotuhi M, Standaert DG, Testa CM, Penney JB, Jr. and Young AB (1994) Differential expression of metabotropic glutamate receptors in the hippocampus and entorhinal cortex of the rat. *Brain Res Mol Brain Res* **21**(3-4):283-292.

Furuta A, Rothstein JD and Martin LJ (1997) Glutamate transporter protein subtypes are expressed differentially during rat CNS development. *J Neurosci* 17(21):8363-8375.

Gao Z, Kehoe V, Sinoway LI and Li J (2005) Spinal P2X receptor modulates reflex pressor response to activation of muscle afferents. *Am J Physiol Heart Circ Physiol* **288**(5):H2238-2243.

Garcia-Guzman M, Soto F, Gomez-Hernandez JM, Lund PE and Stuhmer W (1997) Characterization of recombinant human P2X4 receptor reveals pharmacological differences to the rat homologue. *Mol Pharmacol* **51**(1):109-118.

Garcia M, Sakamoto K, Shigekawa M, Nakanishi S and Ito S (1994) Multiple mechanisms of arachidonic acid release in Chinese hamster ovary cells transfected with cDNA of substance P receptor. *Biochem Pharmacol* **48**(9):1735-1741.

Gayral S, Deleris P, Laulagnier K, Laffargue M, Salles JP, Perret B, Record M and Breton-Douillon M (2006) Selective activation of nuclear phospholipase D-1 by g protein-coupled receptor agonists in vascular smooth muscle cells. *Circ Res* **99**(2):132-139.

Gegner JA, Ulevitch RJ and Tobias PS (1995) Lipopolysaccharide (LPS) signal transduction and clearance. Dual roles for LPS binding protein and membrane CD14. *J Biol Chem* **270**(10):5320-5325.

Geier C, Kreysing J, Boettcher H, Pohlmann R and von Figura K (1992) Localization of lysosomal acid phosphatase mRNA in mouse tissues. *J Histochem Cytochem* **40**(9):1275-1282.

Geurts JJ, Wolswijk G, Bo L, Redeker S, Ramkema M, Troost D and Aronica E (2005) Expression patterns of Group III metabotropic glutamate receptors mGluR4 and mGluR8 in multiple sclerosis lesions. *J Neuroimmunol* **158**(1-2):182-190.

Giulian D, Li J, Bartel S, Broker J, Li X and Kirkpatrick JB (1995) Cell surface morphology identifies microglia as a distinct class of mononuclear phagocyte. *J* Neurosci **15**(11):7712-7726.

Goldstein M and Watkins S (2008) Immunohistochemistry. *Curr Protoc Mol Biol* Chapter 14:Unit 14 16.

Gordon GR, Mulligan SJ and MacVicar BA (2007) Astrocyte control of the cerebrovasculature. *Glia* 55(12):1214-1221.

Grimaldi M, Pozzoli G, Navarra P, Preziosi P and Schettini G (1994) Vasoactive intestinal peptide and forskolin stimulate interleukin 6 production by rat cortical astrocytes in culture via a cyclic AMP-dependent, prostaglandin-independent mechanism. *J Neurochem* **63**(1):344-350.

Guo C, Masin M, Qureshi OS and Murrell-Lagnado RD (2007) Evidence for functional P2X4/P2X7 heteromeric receptors. *Mol Pharmacol* 72(6):1447-1456.

Guo LH, Trautmann K and Schluesener HJ (2005) Expression of P2X4 receptor by lesional activated microglia during formalin-induced inflammatory pain. J Neuroimmunol 163(1-2):120-127.

Guthrie PB, Knappenberger J, Segal M, Bennett MV, Charles AC and Kater SB (1999) ATP released from astrocytes mediates glial calcium waves. *J Neurosci* **19**(2):520-528.

Hagino Y, Kariura Y, Manago Y, Amano T, Wang B, Sekiguchi M, Nishikawa K, Aoki S, Wada K and Noda M (2004) Heterogeneity and potentiation of AMPA type of glutamate receptors in rat cultured microglia. *Glia* 47(1):68-77.

Hagiwara E, Abbasi F, Mor G, Ishigatsubo Y and Klinman DM (1995) Phenotype and frequency of cells secreting IL-2, IL-4, IL-6, IL-10, IFN and TNF-alpha in human peripheral blood. *Cytokine* 7(8):815-822.

Hailman E, Albers JJ, Wolfbauer G, Tu AY and Wright SD (1996) Neutralization and transfer of lipopolysaccharide by phospholipid transfer protein. *J Biol Chem* **271**(21):12172-12178.

Harrison S and Geppetti P (2001) Substance p. Int J Biochem Cell Biol 33(6):555-576.

Hartung HP, Heininger K, Schafer B and Toyka KV (1988) Substance P stimulates release of arachidonic acid cyclooxygenation products from primary culture rat astrocytes. *Ann N Y Acad Sci* **540**:427-429.

Hashimoto H, Fukui K, Noto T, Nakajima T and Kato N (1985) Distribution of vasopressin and oxytocin in rat brain. *Endocrinologia japonica* **32**(1):89-97.

Haydon PG (2001) GLIA: listening and talking to the synapse. Nat Rev Neurosci 2(3):185-193.

Heath MJ, Womack MD and MacDermott AB (1994) Substance P elevates intracellular calcium in both neurons and glial cells from the dorsal horn of the spinal cord. *J Neurophysiol* 72(3):1192-1198.

Helke CJ, Neil JJ, Massari VJ and Loewy AD (1982) Substance P neurons project from the ventral medulla to the intermediolateral cell column and ventral horn in the rat. *Brain Res* 243(1):147-152.

Herrero JF, Laird JM and Lopez-Garcia JA (2000) Wind-up of spinal cord neurones and pain sensation: much ado about something? *Prog Neurobiol* **61**(2):169-203.

Heyen JR, Ye S, Finck BN and Johnson RW (2000) Interleukin (IL)-10 inhibits IL-6 production in microglia by preventing activation of NF-kappaB. *Brain Res Mol Brain Res* 77(1):138-147.

Ho AS, Wei SH, Mui AL, Miyajima A and Moore KW (1995) Functional regions of the mouse interleukin-10 receptor cytoplasmic domain. *Molecular and cellular biology* **15**(9):5043-5053.

Hokfelt T, Arvidsson U, Cullheim S, Millhorn D, Nicholas AP, Pieribone V, Seroogy K and Ulfhake B (2000) Multiple messengers in descending serotonin neurons: localization and functional implications. *J Chem Neuroanat* **18**(1-2):75-86.

Holden JE and Pizzi JA (2003) The challenge of chronic pain. Adv Drug Deliv Rev 55(8):935-948.

Hollmann M, Hartley M and Heinemann S (1991) Ca2+ permeability of KA-AMPA-gated glutamate receptor channels depends on subunit composition. *Science* **252**(5007):851-853.

Honore P, Donnelly-Roberts D, Namovic MT, Hsieh G, Zhu CZ, Mikusa JP, Hernandez G, Zhong C, Gauvin DM, Chandran P, Harris R, Medrano AP, Carroll W, Marsh K, Sullivan JP, Faltynek CR and Jarvis MF (2006) A-740003 [N-(1-{[(cyanoimino)(5-quinolinylamino) methyl]amino}-2,2-dimethylpropyl)-2-(3,4-dimethoxyphenyl)acetamide], a novel and selective P2X7 receptor antagonist, dose-dependently reduces neuropathic pain in the rat. *J Pharmacol Exp Ther* **319**(3):1376-1385.

Honore P, Rogers SD, Schwei MJ, Salak-Johnson JL, Luger NM, Sabino MC, Clohisy DR and Mantyh PW (2000) Murine models of inflammatory, neuropathic and cancer pain each generates a unique set of neurochemical changes in the spinal cord and sensory neurons. *Neuroscience* **98**(3):585-598.

Hornbeck P, Winston SE and Fuller SA (2001) Enzyme-linked immunosorbent assays (ELISA). *Curr Protoc Mol Biol* Chapter 11:Unit11 12.

Horsthemke B, Schulz M and Bauer K (1984) Degradation of substance P by neurones and glial cells. *Biochem Biophys Res Commun* **125**(2):728-733.

Hosli E and Hosli L (1992) Autoradiographic localization of binding sites for arginine vasopressin and atrial natriuretic peptide on astrocytes and neurons of cultured rat central nervous system. *Neuroscience* 51(1):159-166.

Hosli L, Hosli E, Lefkovits M and Wagner S (1991) Electrophysiological evidence for the existence of receptors for endothelin and vasopressin on cultured astrocytes of rat spinal cord and brainstem. *Neurosci Lett* **131**(2):193-195.

Huber M, Kalis C, Keck S, Jiang Z, Georgel P, Du X, Shamel L, Sovath S, Mudd S, Beutler B, Galanos C and Freudenberg MA (2006) R-form LPS, the master key to the activation of TLR4/MD-2-positive cells. *Eur J Immunol* **36**(3):701-711.

Humpel C and Saria A (1993) Characterization of neurokinin binding sites in rat brain membranes using highly selective ligands. *Neuropeptides* **25**(1):65-71.

Hwang D (2001) Modulation of the expression of cyclooxygenase-2 by fatty acids mediated through toll-like receptor 4-derived signaling pathways. Faseb J 15(14):2556-2564.

Ishizaki K, Sasaki M, Karasawa S, Obata H, Nara T and Goto F (1999) Intrathecal co-administration of NMDA antagonist and NK-1 antagonist reduces MAC of isoflurane in rats. *Acta Anaesthesiology Scandanavia* **43**:753-759.

Isomaki P and Punnonen J (1997) Pro- and anti-inflammatory cytokines in rheumatoid arthritis. Ann Med 29(6):499-507.

Jabs R, Kirchhoff F, Kettenmann H and Steinhauser C (1994) Kainate activates Ca(2+)-permeable glutamate receptors and blocks voltage-gated K+ currents in glial cells of mouse hippocampal slices. *Pflugers Arch* **426**(3-4):310-319.

Jackson MB (1997) Whole-cell voltage clamp recording. Curr Protoc Mol Biol Chapter 6:Unit6 1.

Janssens N and Lesage AS (2001) Glutamate receptor subunit expression in primary neuronal and secondary glial cultures. *J Neurochem* 77(6):1457-1474.

Jarvis MF, Burgard EC, McGaraughty S, Honore P, Lynch K, Brennan TJ, Subieta A, Van Biesen T, Cartmell J, Bianchi B, Niforatos W, Kage K, Yu H, Mikusa J, Wismer CT, Zhu CZ, Chu K, Lee CH, Stewart AO, Polakowski J, Cox BF, Kowaluk E, Williams M, Sullivan J and Faltynek C (2002) A-317491, a novel potent and selective non-nucleotide antagonist of P2X3 and P2X2/3 receptors, reduces chronic inflammatory and neuropathic pain in the rat. *Proc Natl Acad Sci U S A* **99**(26):17179-17184.

Jerala R (2007) Structural biology of the LPS recognition. Int J Med Microbiol 297(5):353-363.

Jiang XC and Zhou HW (2006) Plasma lipid transfer proteins. Curr Opin Lipidol 17(3):302-308.

Jiang Z, Georgel P, Du X, Shamel L, Sovath S, Mudd S, Huber M, Kalis C, Keck S, Galanos C, Freudenberg M and Beutler B (2005) CD14 is required for MyD88independent LPS signaling. *Nat Immunol* **6**(6):565-570.

Jo YH and Schlichter R (1999) Synaptic corelease of ATP and GABA in cultured spinal neurons. *Nat Neurosci* 2(3):241-245.

Johnson CL and Johnson CG (1993) Substance P regulation of glutamate and cystine transport in human astrocytoma cells. *Receptors Channels* 1(1):53-59.

Johnson JW and Ascher P (1987) Glycine potentiates the NMDA response in cultured mouse brain neurons. *Nature* **325**(6104):529-531.

Joseph SM, Buchakjian MR and Dubyak GR (2003) Colocalization of ATP release sites and ecto-ATPase activity at the extracellular surface of human astrocytes. *J Biol Chem* **278**(26):23331-23342.

Jou I, Lee JH, Park SY, Yoon HJ, Joe EH and Park EJ (2006) Gangliosides trigger inflammatory responses via TLR4 in brain glia. *Am J Pathol* **168**(5):1619-1630.

Juranek I and Lembeck F (1997) Afferent C-fibres release substance P and glutamate. *Can J Physiol Pharmacol* **75**(6):661-664.

Kandel ER and Schwartz JH (2000) Principles of neural science. McGraw-Hill, New York.

Kapoor V, Minson J and Chalmers J (1992) Ventral medulla stimulation increases blood pressure and spinal cord amino acid release. *Neuroreport* **3**(1):55-58.

Kariko K, Ni H, Capodici J, Lamphier M and Weissman D (2004) mRNA is an endogenous ligand for Toll-like receptor 3. *J Biol Chem* **279**(13):12542-12550.

Kawamata M and Omote K (1996) Involvement of increased excitatory amino acids and intracellular Ca2+ concentration in the spinal dorsal horn in an animal model of neuropathic pain. *Pain* **68**(1):85-96.

Kelicen P and Tindberg N (2004) Lipopolysaccharide induces CYP2E1 in astrocytes through MAP kinase kinase-3 and C/EBPbeta and -delta. *J Biol Chem* **279**(16):15734-15742.

Keller AF, Beggs S, Salter MW and De Koninck Y (2007) Transformation of the output of spinal lamina I neurons after nerve injury and microglia stimulation underlying neuropathic pain. *Mol Pain* **3**:27.

Kennedy C, Assis TS, Currie AJ and Rowan EG (2003) Crossing the pain barrier: P2 receptors as targets for novel analgesics. *J Physiol* **553**(Pt 3):683-694.

Kennedy MN, Mullen GE, Leifer CA, Lee C, Mazzoni A, Dileepan KN and Segal DM (2004) A complex of soluble MD-2 and lipopolysaccharide serves as an activating ligand for Toll-like receptor 4. *J Biol Chem* **279**(33):34698-34704.

Khawaja AM and Rogers DF (1996) Tachykinins: receptor to effector. Int J Biochem Cell Biol 28(7):721-738.

Kim HJ, Hart J, Knatz N, Hall MW and Wewers MD (2004) Janus kinase 3 downregulates lipopolysaccharide-induced IL-1 beta-converting enzyme activation by autocrine IL-10. *J Immunol* **172**(8):4948-4955.

Ko SY, Shim JW, Kim SS, Kim MJ, Chang YS, Park WS, Shin SM and Lee MH (2003) Effects of MK-801 (dizocilpine) on brain cell membrane function and energy metabolism in experimental Escherichia coli meningitis in the newborn piglet. *J Korean Med Sci* 18(2):236-241.

Koizumi S, Fujishita K, Tsuda M, Shigemoto-Mogami Y and Inoue K (2003) Dynamic inhibition of excitatory synaptic transmission by astrocyte-derived ATP in hippocampal cultures. *Proc Natl Acad Sci U S A* **100**(19):11023-11028.

Koizumi S, Saito Y, Nakazawa K, Nakajima K, Sawada JI, Kohsaka S, Illes P and Inoue K (2002) Spatial and temporal aspects of Ca2+ signaling mediated by P2Y receptors in cultured rat hippocampal astrocytes. *Life Sci* **72**(4-5):431-442.

Koops A, Kappler J, Junghans U, Kuhn G, Kresse H and Muller HW (1996) Cultured astrocytes express biglycan, a chondroitin/dermatan sulfate proteoglycan supporting the survival of neocortical neurons. *Brain Res Mol Brain Res* **41**(1-2):65-73.

Kotenko SV, Krause CD, Izotova LS, Pollack BP, Wu W and Pestka S (1997) Identification and functional characterization of a second chain of the interleukin-10 receptor complex. *Embo J* **16**(19):5894-5903.

Krause JE, Staveteig PT, Mentzer JN, Schmidt SK, Tucker JB, Brodbeck RM, Bu JY and Karpitskiy VV (1997) Functional expression of a novel human neurokinin-3 receptor homolog that binds [3H]senktide and [125I-MePhe7]neurokinin B, and is responsive to tachykinin peptide agonists. *Proc Natl Acad Sci U S A* 94(1):310-315.

Kremlev SG and Palmer C (2005) Interleukin-10 inhibits endotoxin-induced proinflammatory cytokines in microglial cell cultures. *J Neuroimmunol* **162**(1-2):71-80.

Kukley M, Barden JA, Steinhauser C and Jabs R (2001) Distribution of P2X receptors on astrocytes in juvenile rat hippocampus. *Glia* **36**(1):11-21.

Kutsuwada T, Kashiwabuchi N, Mori H, Sakimura K, Kushiya E, Araki K, Meguro H, Masaki H, Kumanishi T, Arakawa M and et al. (1992) Molecular diversity of the NMDA receptor channel. *Nature* **358**(6381):36-41.

Laflamme N and Rivest S (2001) Toll-like receptor 4: the missing link of the cerebral innate immune response triggered by circulating gram-negative bacterial cell wall components. *Faseb J* **15**(1):155-163.

Lalo U, Pankratov Y, Kirchhoff F, North RA and Verkhratsky A (2006) NMDA receptors mediate neuron-to-glia signaling in mouse cortical astrocytes. *J Neurosci* **26**(10):2673-2683.

Laube B, Hirai H, Sturgess M, Betz H and Kuhse J (1997) Molecular determinants of agonist discrimination by NMDA receptor subunits: analysis of the glutamate binding site on the NR2B subunit. *Neuron* **18**(3):493-503.

Ledeboer A, Breve JJ, Poole S, Tilders FJ and Van Dam AM (2000) Interleukin-10, interleukin-4, and transforming growth factor-beta differentially regulate lipopolysaccharide-induced production of pro-inflammatory cytokines and nitric oxide in co-cultures of rat astroglial and microglial cells. *Glia* **30**(2):134-142.

Ledeboer A, Breve JJ, Wierinckx A, van der Jagt S, Bristow AF, Leysen JE, Tilders FJ and Van Dam AM (2002) Expression and regulation of interleukin-10 and interleukin-10 receptor in rat astroglial and microglial cells. *Eur J Neurosci* **16**(7):1175-1185.

Ledeboer A, Jekich BM, Sloane EM, Mahoney JH, Langer SJ, Milligan ED, Martin D, Maier SF, Johnson KW, Leinwand LA, Chavez RA and Watkins LR (2007) Intrathecal interleukin-10 gene therapy attenuates paclitaxel-induced mechanical allodynia and proinflammatory cytokine expression in dorsal root ganglia in rats. *Brain Behav Immun* **21**(5):686-698.

Ledeboer A, Sloane EM, Milligan ED, Frank MG, Mahony JH, Maier SF and Watkins LR (2005) Minocycline attenuates mechanical allodynia and proinflammatory cytokine expression in rat models of pain facilitation. *Pain* **115**(1-2):71-83.

Lee YB, Schrader JW and Kim SU (2000) p38 map kinase regulates TNF-alpha production in human astrocytes and microglia by multiple mechanisms. *Cytokine* **12**(7):874-880.

Legrand C, Bour JM, Jacob C, Capiaumont J, Martial A, Marc A, Wudtke M, Kretzmer G, Demangel C, Duval D and et al. (1992) Lactate dehydrogenase (LDH) activity of the cultured eukaryotic cells as marker of the number of dead cells in the medium [corrected]. *J Biotechnol* **25**(3):231-243.

Lehmann A (1987) Evidence for a direct action of N-methylaspartate on nonneuronal cells. *Brain Res* **411**(1):95-101.

Lehnardt S, Lachance C, Patrizi S, Lefebvre S, Follett PL, Jensen FE, Rosenberg PA, Volpe JJ and Vartanian T (2002) The toll-like receptor TLR4 is necessary for

lipopolysaccharide-induced oligodendrocyte injury in the CNS. J Neurosci 22(7):2478-2486.

Lentzen H and Palenker J (1983) Localization of the thiorphan-sensitive endopeptidase, termed enkephalinase A, on glial cells. *FEBS Lett* **153**(1):93-97.

Lever IJ, Bradbury EJ, Cunningham JR, Adelson DW, Jones MG, McMahon SB, Marvizon JC and Malcangio M (2001) Brain-derived neurotrophic factor is released in the dorsal horn by distinctive patterns of afferent fiber stimulation. *J Neurosci* **21**(12):4469-4477.

Levison SW and Goldman JE (1993) Astrocyte Origins, in Astrocytes: Pharmacology and Function (Murphy S ed), Academic Press, Inc., San Diego.

Levy D and Zochodne DW (1998) Local nitric oxide synthase activity in a model of neuropathic pain. *Eur J Neurosci* **10**(5):1846-1855.

Lewis C, Neidhart S, Holy C, North RA, Buell G and Surprenant A (1995) Coexpression of P2X2 and P2X3 receptor subunits can account for ATP-gated currents in sensory neurons. *Nature* 377(6548):432-435.

Li J and Perl ER (1994) Adenosine inhibition of synaptic transmission in the substantia gelatinosa. *J Neurophysiol* 72(4):1611-1621.

Liao GY, Wagner DA, Hsu MH and Leonard JP (2001) Evidence for direct protein kinase-C mediated modulation of N-methyl-D-aspartate receptor current. *Mol Pharmacol* **59**(5):960-964.

Liefner M, Siebert H, Sachse T, Michel U, Kollias G and Bruck W (2000) The role of TNF-alpha during Wallerian degeneration. *J Neuroimmunol* **108**(1-2):147-152.

Lin V (ed) (2003) Mechanisms and Natural History of Spinal Cord Injury. Demos Medical Publishing, Inc New York.

Liu CN, Wall PD, Ben-Dor E, Michaelis M, Amir R and Devor M (2000) Tactile allodynia in the absence of C-fiber activation: altered firing properties of DRG neurons following spinal nerve injury. *Pain* **85**(3):503-521.

Liu GJ, Kalous A, Werry EL and Bennett MR (2006) Purine release from spinal cord microglia after elevation of calcium by glutamate. *Mol Pharmacol* **70**(3):851-859.

Liu YW, Tseng HP, Chen LC, Chen BK and Chang WC (2003) Functional cooperation of simian virus 40 promoter factor 1 and CCAAT/enhancer-binding

protein beta and delta in lipopolysaccharide-induced gene activation of IL-10 in mouse macrophages. J Immunol 171(2):821-828.

Lucas M, Zhang X, Prasanna V and Mosser DM (2005) ERK activation following macrophage FcgammaR ligation leads to chromatin modifications at the IL-10 locus. *J Immunol* **175**(1):469-477.

Luchi M and Munford RS (1993) Binding, internalization, and deacylation of bacterial lipopolysaccharide by human neutrophils. *J Immunol* **151**(2):959-969.

Ma W, Lim W, Gee K, Aucoin S, Nandan D, Kozlowski M, Diaz-Mitoma F and Kumar A (2001) The p38 mitogen-activated kinase pathway regulates the human interleukin-10 promoter via the activation of Sp1 transcription factor in lipopolysaccharide-stimulated human macrophages. J Biol Chem 276(17):13664-13674.

MacKenzie A, Wilson HL, Kiss-Toth E, Dower SK, North RA and Surprenant A (2001) Rapid secretion of interleukin-1beta by microvesicle shedding. *Immunity* **15**(5):825-835.

Madden DR (2002) The structure and function of glutamate receptor ion channels. *Nat Rev Neurosci* 3(2):91-101.

Maggio JE (1988) Tachykinins. Annu Rev Neurosci 11:13-28.

Majde JA, Guha-Thakurta N, Chen Z, Bredow S and Krueger JM (1998) Spontaneous release of stable viral double-stranded RNA into the extracellular medium by influenza virus-infected MDCK epithelial cells: implications for the viral acute phase response. *Arch Virol* 143(12):2371-2380.

Malcangio M, Bowery NG, Flower RJ and Perretti M (1996) Effect of interleukin-1 beta on the release of substance P from rat isolated spinal cord. *Eur J Pharmacol* **299**(1-3):113-118.

Mambula SS and Calderwood SK (2006) Heat shock protein 70 is secreted from tumor cells by a nonclassical pathway involving lysosomal endosomes. *J Immunol* **177**(11):7849-7857.

Mao X, Yang SH, Simpkins JW and Barger SW (2007) Glutamate receptor activation evokes calpain-mediated degradation of Sp3 and Sp4, the prominent Sp-family transcription factors in neurons. *J Neurochem* **100**(5):1300-1314.

Marcel D, Pollard H, Verroust P, Schwartz JC and Beaudet A (1990) Electron microscopic localization of immunoreactive enkephalinase (EC 3.4.24.11) in the neostriatum of the rat. *J Neurosci* **10**(8):2804-2817.

Marriott DR, Wilkin GP and Wood JN (1991) Substance P-induced release of prostaglandins from astrocytes: regional specialisation and correlation with phosphoinositol metabolism. *J Neurochem* **56**(1):259-265.

Martin FC, Anton PA, Gornbein JA, Shanahan F and Merrill JE (1993) Production of interleukin-1 by microglia in response to substance P: role for a non-classical NK-1 receptor. *J Neuroimmunol* **42**(1):53-60.

Martin LJ, Blackstone CD, Huganir RL and Price DL (1992) Cellular localization of a metabotropic glutamate receptor in rat brain. *Neuron* 9(2):259-270.

Martinez-Hernandez A, Bell KP and Norenberg MD (1977) Glutamine synthetase: glial localization in brain. *Science* **195**(4284):1356-1358.

Martucci C, Trovato AE, Costa B, Borsani E, Franchi S, Magnaghi V, Panerai AE, Rodella LF, Valsecchi AE, Sacerdote P and Colleoni M (2007) The purinergic antagonist PPADS reduces pain related behaviours and interleukin-1beta, interleukin-6, iNOS and nNOS overproduction in central and peripheral nervous system after peripheral neuropathy in mice. *Pain*.

Marvizon JC, Martinez V, Grady EF, Bunnett NW and Mayer EA (1997) Neurokinin 1 receptor internalization in spinal cord slices induced by dorsal root stimulation is mediated by NMDA receptors. *J Neurosci* **17**(21):8129-8136.

Matsuzaki K and Horikiri C (1999) Interactions of amyloid beta-peptide (1-40) with ganglioside-containing membranes. *Biochemistry* **38**(13):4137-4142.

Matute C, Domercq M and Sanchez-Gomez MV (2006) Glutamate-mediated glial injury: mechanisms and clinical importance. *Glia* **53**(2):212-224.

Mau SE and Vilhardt H (1997) Cross talk between substance P and melittin-activated cellular signaling pathways in rat lactotroph-enriched cell cultures. *J Neurochem* **69**(2):762-772.

McAlexander J (2007) Neurologic transmission of pain stimuli, Wild Iris Medical Education, Comptche.

McDermott AM, Dickinson SL and Wilkin GP (1992) Thyrotropin releasing hormone (TRH) and a degradation stabilized analogue (RX77368) stimulate phosphoinositide

turnover in cultured astrocytes in a regionally specific manner. Neurochem Int 20(3):307-313.

McGaraughty S, Chu KL, Namovic MT, Donnelly-Roberts DL, Harris RR, Zhang XF, Shieh CC, Wismer CT, Zhu CZ, Gauvin DM, Fabiyi AC, Honore P, Gregg RJ, Kort ME, Nelson DW, Carroll WA, Marsh K, Faltynek CR and Jarvis MF (2007) P2X7-related modulation of pathological nociception in rats. *Neuroscience* **146**(4):1817-1828.

McGaraughty S, Wismer CT, Zhu CZ, Mikusa J, Honore P, Chu KL, Lee CH, Faltynek CR and Jarvis MF (2003) Effects of A-317491, a novel and selective P2X3/P2X2/3 receptor antagonist, on neuropathic, inflammatory and chemogenic nociception following intrathecal and intraplantar administration. *Br J Pharmacol* **140**(8):1381-1388.

McLean S, Skirboll LR and Pert CB (1985) Comparison of substance P and enkephalin distribution in rat brain: an overview using radioimmunocytochemistry. *Neuroscience* 14(3):837-852.

Meller ST, Dykstra C, Grzybycki D, Murphy S and Gebhart GF (1994) The possible role of glia in nociceptive processing and hyperalgesia in the spinal cord of the rat. *Neuropharmacology* **33**(11):1471-1478.

Merskey H and Bogduk N (eds) (1994) Pain Terms, A Current List with Definitions and Notes on Usage. IASP Press, Seattle.

Mesples B, Plaisant F and Gressens P (2003) Effects of interleukin-10 on neonatal excitotoxic brain lesions in mice. *Brain Res Dev Brain Res* 141(1-2):25-32.

Meyer BH, Segura JM, Martinez KL, Hovius R, George N, Johnsson K and Vogel H (2006) FRET imaging reveals that functional neurokinin-1 receptors are monomeric and reside in membrane microdomains of live cells. *Proc Natl Acad Sci U S A* **103**(7):2138-2143.

Michikawa M, Gong JS, Fan QW, Sawamura N and Yanagisawa K (2001) A novel action of alzheimer's amyloid beta-protein (Abeta): oligomeric Abeta promotes lipid release. *J Neurosci* **21**(18):7226-7235.

Miles GB, Parkis MA, Lipski J and Funk GD (2002) Modulation of phrenic motoneuron excitability by ATP: consequences for respiratory-related output in vitro. *J Appl Physiol* **92**(5):1899-1910.

Millan MJ (1999) The induction of pain: an integrative review. *Prog Neurobiol* 57(1):1-164.

Milligan ED, Langer SJ, Sloane EM, He L, Wieseler-Frank J, O'Connor K, Martin D, Forsayeth JR, Maier SF, Johnson K, Chavez RA, Leinwand LA and Watkins LR (2005a) Controlling pathological pain by adenovirally driven spinal production of the anti-inflammatory cytokine, interleukin-10. *Eur J Neurosci* **21**(8):2136-2148.

Milligan ED, Sloane EM, Langer SJ, Cruz PE, Chacur M, Spataro L, Wieseler-Frank J, Hammack SE, Maier SF, Flotte TR, Forsayeth JR, Leinwand LA, Chavez R and Watkins LR (2005b) Controlling neuropathic pain by adeno-associated virus driven production of the anti-inflammatory cytokine, interleukin-10. *Mol Pain* **1**:9.

Milligan ED, Sloane EM, Langer SJ, Hughes TS, Jekich BM, Frank MG, Mahoney JH, Levkoff LH, Maier SF, Cruz PE, Flotte TR, Johnson KW, Mahoney MM, Chavez RA, Leinwand LA and Watkins LR (2006a) Repeated intrathecal injections of plasmid DNA encoding interleukin-10 produce prolonged reversal of neuropathic pain. *Pain* **126**(1-3):294-308.

Milligan ED, Soderquist RG, Malone SM, Mahoney JH, Hughes TS, Langer SJ, Sloane EM, Maier SF, Leinwand LA, Watkins LR and Mahoney MJ (2006b) Intrathecal polymer-based interleukin-10 gene delivery for neuropathic pain. *Neuron Glia Biol* **2**(4):293-308.

Milligan ED, Twining C, Chacur M, Biedenkapp J, O'Connor K, Poole S, Tracey K, Martin D, Maier SF and Watkins LR (2003) Spinal glia and proinflammatory cytokines mediate mirror-image neuropathic pain in rats. *J Neurosci* 23(3):1026-1040.

Mistry R, Golding N and Challiss RA (1998) Regulation of phosphoinositide turnover in neonatal rat cerebral cortex by group I- and II- selective metabotropic glutamate receptor agonists. *Br J Pharmacol* **123**(3):581-589.

Mizuno T, Sawada M, Marunouchi T and Suzumura A (1994) Production of interleukin-10 by mouse glial cells in culture. *Biochem Biophys Res Commun* **205**(3):1907-1915.

Mjellem-Joly N, Lund A, Berge OG and Hole K (1991) Potentiation of a behavioural response in mice by spinal coadministration of substance P and excitatory amino acid agonists. *Neurosci Lett* **133**(1):121-124.

Monastyrskaya K, Hostettler A, Buergi S and Draeger A (2005) The NK1 receptor localizes to the plasma membrane microdomains, and its activation is dependent on lipid raft integrity. *J Biol Chem* **280**(8):7135-7146.

Montuschi P, Curro D and Ciabattoni G (1996) Tachykinin NK2 receptor antagonists decrease eicosanoid release in lung anaphylaxis. *Eur J Pharmacol* **313**(1-2):R1-3.

Moore KW, de Waal Malefyt R, Coffman RL and O'Garra A (2001) Interleukin-10 and the interleukin-10 receptor. *Annu Rev Immunol* **19**:683-765.

Mori H and Mishina M (1995) Structure and function of the NMDA receptor channel. *Neuropharmacology* **34**(10):1219-1237.

Morris JL, Konig P, Shimizu T, Jobling P and Gibbins IL (2005) Most peptidecontaining sensory neurons lack proteins for exocytotic release and vesicular transport of glutamate. *J Comp Neurol* **483**(1):1-16.

Murphy S, Lin HL and Park SK (1995) Cytokine-induced expression of type II nitric oxide synthase in astrocytes is downregulated by ATP and glutamate. *Glia* **15**(1):77-82.

Murray M and Greenberg ML (1997) Regulation of inositol monophosphatase in Saccharomyces cerevisiae. *Mol Microbiol* **25**(3):541-546.

Murray M and Greenberg ML (2000) Expression of yeast INM1 encoding inositol monophosphatase is regulated by inositol, carbon source and growth stage and is decreased by lithium and valproate. *Mol Microbiol* **36**(3):651-661.

Mussap CJ, Geraghty DP and Burcher E (1993) Tachykinin receptors: a radioligand binding perspective. J Neurochem 60(6):1987-2009.

Myers RR, Sekiguchi Y, Kikuchi S, Scott B, Medicherla S, Protter A and Campana WM (2003) Inhibition of p38 MAP kinase activity enhances axonal regeneration. *Exp Neurol* **184**(2):606-614.

Nakagawa T, Wakamatsu K, Zhang N, Maeda S, Minami M, Satoh M and Kaneko S (2007) Intrathecal administration of ATP produces long-lasting allodynia in rats: differential mechanisms in the phase of the induction and maintenance. *Neuroscience* **147**(2):445-455.

Nakajima Y, Tsuchida K, Negishi M, Ito S and Nakanishi S (1992) Direct linkage of three tachykinin receptors to stimulation of both phosphatidylinositol hydrolysis and cyclic AMP cascades in transfected Chinese hamster ovary cells. *J Biol Chem* **267**(4):2437-2442.

Nakatsuka T and Gu JG (2001) ATP P2X receptor-mediated enhancement of glutamate release and evoked EPSCs in dorsal horn neurons of the rat spinal cord. J *Neurosci* **21**(17):6522-6531.

Narcisse L, Scemes E, Zhao Y, Lee SC and Brosnan CF (2005) The cytokine ILlbeta transiently enhances P2X7 receptor expression and function in human astrocytes. *Glia* **49**(2):245-258.

Nemeth ZH, Lutz CS, Csoka B, Deitch EA, Leibovich SJ, Gause WC, Tone M, Pacher P, Vizi ES and Hasko G (2005) Adenosine augments IL-10 production by macrophages through an A2B receptor-mediated posttranscriptional mechanism. *J Immunol* **175**(12):8260-8270.

Neumann S, Doubell TP, Leslie T and Woolf CJ (1996) Inflammatory pain hypersensitivity mediated by phenotypic switch in myelinated primary sensory neurons. *Nature* **384**(6607):360-364.

Nicholas AP, Pieribone VA, Arvidsson U and Hokfelt T (1992) Serotonin-, substance P- and glutamate/aspartate-like immunoreactivities in medullo-spinal pathways of rat and primate. *Neuroscience* **48**(3):545-559.

Nicholls DG (1998) Presynaptic modulation of glutamate release. Prog Brain Res 116:15-22.

Nitu AN, Wallihan R, Skljarevski V and Ramadan NM (2003) Emerging trends in the pharmacotherapy of chronic pain. *Expert Opin Investig Drugs* **12**(4):545-559.

Noda M, Nakanishi H, Nabekura J and Akaike N (2000) AMPA-kainate subtypes of glutamate receptor in rat cerebral microglia. *J Neurosci* 20(1):251-258.

O'Flaherty JT, Chadwell BA, Kearns MW, Sergeant S and Daniel LW (2001) Protein kinases C translocation responses to low concentrations of arachidonic acid. *J Biol Chem* **276**(27):24743-24750.

O'Neill LA and Bowie AG (2007) The family of five: TIR-domain-containing adaptors in Toll-like receptor signalling. *Nat Rev Immunol* 7(5):353-364.

Ogata T, Chuai M, Morino T, Yamamoto H, Nakamura Y and Schubert P (2003) Adenosine triphosphate inhibits cytokine release from lipopolysaccharide-activated microglia via P2y receptors. *Brain Res* **981**(1-2):174-183.

Osborn MJ, Rosen SM, Rothfield L, Zeleznick LD and Horecker BL (1964) Lipopolysaccharide of the Gram-Negative Cell Wall. *Science* 145:783-789.

Ovanesov MV, Sauder C, Rubin SA, Richt J, Nath A, Carbone KM and Pletnikov MV (2006) Activation of microglia by borna disease virus infection: in vitro study. *J Virol* 80(24):12141-12148.

Palma C, Minghetti L, Astolfi M, Ambrosini E, Silberstein FC, Manzini S, Levi G and Aloisi F (1997) Functional characterization of substance P receptors on cultured human spinal cord astrocytes: synergism of substance P with cytokines in inducing interleukin-6 and prostaglandin E2 production. *Glia* **21**(2):183-193.

Pangrsic T, Potokar M, Stenovec M, Kreft M, Fabbretti E, Nistri A, Pryazhnikov E, Khiroug L, Giniatullin R and Zorec R (2007) Exocytotic release of ATP from cultured astrocytes. *J Biol Chem* **282**(39):28749-28758.

Paulsen RE, Contestabile A, Villani L and Fonnum F (1987) An in vivo model for studying function of brain tissue temporarily devoid of glial cell metabolism: the use of fluorocitrate. *J Neurochem* **48**(5):1377-1385.

Peavy RD, Chang MS, Sanders-Bush E and Conn PJ (2001) Metabotropic glutamate receptor 5-induced phosphorylation of extracellular signal-regulated kinase in astrocytes depends on transactivation of the epidermal growth factor receptor. J Neurosci **21**(24):9619-9628.

Pehar M, Cassina P, Vargas MR, Castellanos R, Viera L, Beckman JS, Estevez AG and Barbeito L (2004) Astrocytic production of nerve growth factor in motor neuron apoptosis: implications for amyotrophic lateral sclerosis. *J Neurochem* **89**(2):464-473.

Peinado MA, Quesada A, Pedrosa JA, Torres MI, Martinez M, Esteban FJ, Del Moral ML, Hernandez R, Rodrigo J and Peinado JM (1998) Quantitative and ultrastructural changes in glia and pericytes in the parietal cortex of the aging rat. *Microsc Res Tech* **43**(1):34-42.

Peters A, Palay SL and Webster Hd (1991) The fine structure of the nervous system : neurons and their supporting cells. Oxford University Press, New York.

Petralia RS, Wang YX and Wenthold RJ (1994) Histological and ultrastructural localization of the kainate receptor subunits, KA2 and GluR6/7, in the rat nervous system using selective antipeptide antibodies. *J Comp Neurol* **349**(1):85-110.

Pfaffl MW, Horgan GW and Dempfle L (2002) Relative expression software tool (REST) for group-wise comparison and statistical analysis of relative expression results in real-time PCR. *Nucleic Acids Res* **30**(9):e36.

Phelan MC (2006) Techniques for mammalian cell tissue culture. *Curr Protoc Mol Biol* Appendix 3:Appendix 3F.

Pin J, Goudet C, Kniazeff J, Hlavackova V, Brock C, Binet V, Maurel D, Rondard P, Blahos J and Prezeau L (2006) Dimeric GPCRs: what did we learn from the metabotropic glutamate receptors?, in *Insights into Receptor Function and New Drug Development Targets* pp 105-115, Springer Berlin Heidelberg.

Pin JP and Duvoisin R (1995) The metabotropic glutamate receptors: structure and functions. *Neuropharmacology* **34**(1):1-26.

Pinheiro AM, Costa SL, Freire SM, Meyer R, Almeida MA, Tardy M, El Bacha R and Costa MF (2006) Neospora caninum: infection induced IL-10 overexpression in rat astrocytes in vitro. *Exp Parasitol* **112**(3):193-197.

Porter JT and McCarthy KD (1995) GFAP-positive hippocampal astrocytes in situ respond to glutamatergic neuroligands with increases in [Ca2+]i. *Glia* 13(2):101-112.

Porter JT and McCarthy KD (1997) Astrocytic neurotransmitter receptors in situ and in vivo. *Prog Neurobiol* **51**(4):439-455.

Pousset F, Dantzer R, Kelley KW and Parnet P (2000) Interleukin-1 signaling in mouse astrocytes involves Akt: a study with interleukin-4 and IL-10. *Eur Cytokine Netw* **11**(3):427-434.

Pow DV and Barnett NL (2000) Developmental expression of excitatory amino acid transporter 5: a photoreceptor and bipolar cell glutamate transporter in rat retina. *Neurosci Lett* **280**(1):21-24.

Qian L, Block ML, Wei SJ, Lin CF, Reece J, Pang H, Wilson B, Hong JS and Flood PM (2006a) Interleukin-10 protects lipopolysaccharide-induced neurotoxicity in primary midbrain cultures by inhibiting the function of NADPH oxidase. J Pharmacol Exp Ther **319**(1):44-52.

Qian L, Hong JS and Flood PM (2006b) Role of microglia in inflammation-mediated degeneration of dopaminergic neurons: neuroprotective effect of interleukin 10. J Neural Transm Suppl(70):367-371.

Queiroz G, Gebicke-Haerter PJ, Schobert A, Starke K and von Kugelgen I (1997) Release of ATP from cultured rat astrocytes elicited by glutamate receptor activation. *Neuroscience* **78**(4):1203-1208.

Queiroz G, Meyer D, Meyer A, Starke K and Kugelgen Iv (1999) A study of the mechanism of the release of ATP from rat cortical astroglial cells evoked by activation of glutamate receptors. *Neuroscience* 91(3):1171-1181.

Raghavendra V, Tanga F and DeLeo JA (2003) Inhibition of microglial activation attenuates the development but not existing hypersensitivity in a rat model of neuropathy. *J Pharmacol Exp Ther* **306**(2):624-630.

Ralevic V and Burnstock G (1998) Receptors for purines and pyrimidines. *Pharmacol Rev* 50(3):413-492.

Randic M, Hecimovic H and Ryu PD (1990) Substance P modulates glutamateinduced currents in acutely isolated rat spinal dorsal horn neurones. *Neurosci Lett* **117**(1-2):74-80.

Rasley A, Tranguch SL, Rati DM and Marriott I (2006) Murine glia express the immunosuppressive cytokine, interleukin-10, following exposure to Borrelia burgdorferi or Neisseria meningitidis. *Glia* 53(6):583-592.

Rayner SE and Van Helden DF (1997) Evidence that the substance P-induced enhancement of pacemaking in lymphatics of the guinea-pig mesentery occurs through endothelial release of thromboxane A2. *Br J Pharmacol* **121**(8):1589-1596.

Reeve AJ, Patel S, Fox A, Walker K and Urban L (2000) Intrathecally administered endotoxin or cytokines produce allodynia, hyperalgesia and changes in spinal cord neuronal responses to nociceptive stimuli in the rat. *Eur J Pain* 4(3):247-257.

Rego AC, Santos MS and Oliveira CR (1996) Oxidative stress, hypoxia, and ischemia-like conditions increase the release of endogenous amino acids by distinct mechanisms in cultured retinal cells. *J Neurochem* **66**(6):2506-2516.

Regoli D, Boudon A and Fauchere JL (1994) Receptors and antagonists for substance P and related peptides. *Pharmacol Rev* **46**(4):551-599.

Rodriguez J, Carcache L and Rein KS (2005) Low-mode docking search in iGluR homology models implicates three residues in the control of ligand selectivity. *J Mol Recognit* **18**(2):183-189.

Roes S, Mumm F, Seydel U and Gutsmann T (2006) Localization of the lipopolysaccharide-binding protein in phospholipid membranes by atomic force microscopy. *J Biol Chem* **281**(5):2757-2763.

Romano C, Sesma MA, McDonald CT, O'Malley K, Van den Pol AN and Olney JW (1995) Distribution of metabotropic glutamate receptor mGluR5 immunoreactivity in rat brain. *J Comp Neurol* **355**(3):455-469.

Rosewater K and Sontheimer H (1994) Fibrous and protoplasmic astrocytes express GABAA receptors that differ in benzodiazepine pharmacology. *Brain Res* **636**(1):73-80.

Rossi DJ, Oshima T and Attwell D (2000) Glutamate release in severe brain ischaemia is mainly by reversed uptake. *Nature* **403**(6767):316-321.

Rothstein JD, Martin L, Levey AI, Dykes-Hoberg M, Jin L, Wu D, Nash N and Kuncl RW (1994) Localization of neuronal and glial glutamate transporters. *Neuron* **13**(3):713-725.

Rougon G, Noble M and Mudge AW (1983) Neuropeptides modulate the betaadrenergic response of purified astrocytes in vitro. *Nature* **305**(5936):715-717.

Rusin KI, Bleakman D, Chard PS, Randic M and Miller RJ (1993) Tachykinins potentiate N-methyl-D-aspartate responses in acutely isolated neurons from the dorsal horn. *J Neurochem* **60**(3):952-960.

Sakamoto H and Atsumi S (1989) Substance P-immunoreactive elements in laminae I and II of the chicken spinal cord: a light- and electron-microscopic study. *Neurosci Res* 7(2):103-116.

Salter MW and Hicks JL (1994) ATP-evoked increases in intracellular calcium in neurons and glia from the dorsal spinal cord. *J Neurosci* 14(3 Pt 2):1563-1575.

Sanz JM and Di Virgilio F (2000) Kinetics and mechanism of ATP-dependent IL-1 beta release from microglial cells. *J Immunol* **164**(9):4893-4898.

Sarau HM, Mooney JL, Schmidt DB, Foley JJ, Buckley PT, Giardina GA, Wang DY, Lee JA and Hay DW (2000) Evidence that the proposed novel human "neurokinin-4" receptor is pharmacologically similar to the human neurokinin-3 receptor but is not of human origin. *Mol Pharmacol* **58**(3):552-559.

Saria A (1999) The tachykinin NK1 receptor in the brain: pharmacology and putative functions. *Eur J Pharmacol* **375**(1-3):51-60.

Sasaki M, Tohda C and Kuraishi Y (1998) Region-specific increase in glutamate release from dorsal horn of rats with adjuvant inflammation. *Neuroreport* 9(14):3219-3222.

Sawada M, Suzumura A, Hosoya H, Marunouchi T and Nagatsu T (1999) Interleukin-10 inhibits both production of cytokines and expression of cytokine receptors in microglia. *J Neurochem* **72**(4):1466-1471.

Sawada T, Sano M, Omura T, Omura K, Hasegawa T, Funahashi S and Nagano A (2007) Spatiotemporal quantification of tumor necrosis factor-alpha and interleukin-10 after crush injury in rat sciatic nerve utilizing immunohistochemistry. *Neurosci Lett* **41**7(1):55-60.

Sawynok J, Downie JW, Reid AR, Cahill CM and White TD (1993) ATP release from dorsal spinal cord synaptosomes: characterization and neuronal origin. *Brain Res* **610**(1):32-38.

Scemes E, Suadicani SO and Spray DC (2000) Intercellular communication in spinal cord astrocytes: fine tuning between gap junctions and P2 nucleotide receptors in calcium wave propagation. *J Neurosci* **20**(4):1435-1445.

Schaefer L, Babelova A, Kiss E, Hausser HJ, Baliova M, Krzyzankova M, Marsche G, Young MF, Mihalik D, Gotte M, Malle E, Schaefer RM and Grone HJ (2005) The matrix component biglycan is proinflammatory and signals through Toll-like receptors 4 and 2 in macrophages. *J Clin Invest* **115**(8):2223-2233.

Schaible HG and Schmidt RF (1988) Time course of mechanosensitivity changes in articular afferents during a developing experimental arthritis. J Neurophysiol **60**(6):2180-2195.

Schinkmann KA, Kim TA and Avraham S (2000) Glutamate-stimulated activation of DNA synthesis via mitogen-activated protein kinase in primary astrocytes: involvement of protein kinase C and related adhesion focal tyrosine kinase. J Neurochem 74(5):1931-1940.

Schuler T, Mesic I, Madry C, Bartholomaus I and Laube B (2008) Formation of NR1/NR2 and NR1/NR3 heterodimers constitutes the initial step in N-methyl-D-aspartate receptor assembly. *J Biol Chem* **283**(1):37-46.

Schumann RR, Leong SR, Flaggs GW, Gray PW, Wright SD, M athison JC, Tobias PS and Ulevitch RJ (1990) Structure and function of lipopolysaccharide binding protein. *Science* **249**(4975):1429-1431.

Seguela P, Haghighi A, Soghomonian JJ and Cooper E (1996) A novel neuronal P2x ATP receptor ion channel with widespread distribution in the brain. *J Neurosci* 16(2):448-455.

Seo DR, Kim KY and Lee YB (2004) Interleukin-10 expression in lipopolysaccharide-activated microglia is mediated by extracellular ATP in an autocrine fashion. *Neuroreport* **15**(7):1157-1161.

Sharp CJ, Reeve AJ, Collins SD, Martindale JC, Summerfield SG, Sargent BS, Bate ST and Chessell IP (2006) Investigation into the role of P2X(3)/P2X(2/3) receptors in neuropathic pain following chronic constriction injury in the rat: an electrophysiological study. *Br J Pharmacol* **148**(6):845-852.

Sheng WS, Hu S, Kravitz FH, Peterson PK and Chao CC (1995) Tumor necrosis factor alpha upregulates human microglial cell production of interleukin-10 in vitro. *Clin Diagn Lab Immunol* **2**(5):604-608.

Shigemoto R, Nakanishi S and Mizuno N (1992) Distribution of the mRNA for a metabotropic glutamate receptor (mGluR1) in the central nervous system: an in situ hybridization study in adult and developing rat. *J Comp Neurol* **322**(1):121-135.

Shimazu R, Akashi S, Ogata H, Nagai Y, Fukudome K, Miyake K and Kimoto M (1999) MD-2, a molecule that confers lipopolysaccharide responsiveness on Toll-like receptor 4. *J Exp Med* **189**(11):1777-1782.

Silva GA, Theriault E, Mills LR, Pennefather PS and Feeney CJ (1999) Group I and II metabotropic glutamate receptor expression in cultured rat spinal cord astrocytes. *Neurosci Lett* **263**(2-3):117-120.

Sim JA, Young-MT, Sung HY, North RA and Surprenant A (2004) Reanalysis of P2X7 receptor expression in rodent brain. *J Neurosci* 24(28):6307-6314.

Sistiaga A and Sanchez-Prieto J (2000) Protein phosphatase 2B inhibitors mimic the action of arachidonic acid and prolong the facilitation of glutamate release by group I mGlu receptors. *Neuropharmacology* **39**(9):1544-1553.

Skeberdis VA, Lan J, Opitz T, Zheng X, Bennett MV and Zukin RS (2001) mGluR1mediated potentiation of NMDA receptors involves a rise in intracellular calcium and activation of protein kinase C. *Neuropharmacology* **40**(7):856-865.

Skidgel RA, Engelbrecht S, Johnson AR and Erdos EG (1984) Hydrolysis of substance p and neurotensin by converting enzyme and neutral endopeptidase. *Peptides* 5(4):769-776.

Somers DL and Clemente FR (2002) Dorsal horn synaptosomal content of aspartate, glutamate, glycine and GABA are differentially altered following chronic constriction injury to the rat sciatic nerve. *Neurosci Lett* **323**(3):171-174.
Chapter 7

Sorkin LS, Westlund KN, Sluka KA, Dougherty PM and Willis WD (1992) Neural changes in acute arthritis in monkeys. IV. Time-course of amino acid release into the lumbar dorsal horn. *Brain Res Brain Res Rev* **17**(1):39-50.

Soto F, Garcia-Guzman M, Gomez-Hernandez JM, Hollmann M, Karschin C and Stuhmer W (1996) P2X4: an ATP-activated ionotropic receptor cloned from rat brain. *Proc Natl Acad Sci U S A* **93**(8):3684-3688.

Souslova V, Cesare P, Ding Y, Akopian AN, Stanfa L, Suzuki R, Carpenter K, Dickenson A, Boyce S, Hill R, Nebenuis-Oosthuizen D, Smith AJ, Kidd EJ and Wood JN (2000) Warm-coding deficits and aberrant inflammatory pain in mice lacking P2X3 receptors. *Nature* **407**(6807):1015-1017.

Sperlagh B, Vizi ES, Wirkner K and Illes P (2006) P2X7 receptors in the nervous system. *Prog Neurobiol* 78(6):327-346.

Stellwagen D, Beattie EC, Seo JY and Malenka RC (2005) Differential regulation of AMPA receptor and GABA receptor trafficking by tumor necrosis factor-alpha. J Neurosci 25(12):3219-3228.

Stevens B (2003) Glia: much more than the neuron's side-kick. Curr Biol 13(12):R469-472.

Stout CE, Costantin JL, Naus CC and Charles AC (2002) Intercellular calcium signaling in astrocytes via ATP release through connexin hemichannels. *J Biol Chem* **277**(12):10482-10488.

Streit W (1995) Microglial cells, in *Neuroglia* (Kettenmann H and Ransom B eds) pp 85-96, Oxford University Press, New York.

Studeny S, Torabi A and Vizzard MA (2005) P2X2 and P2X3 receptor expression in postnatal and adult rat urinary bladder and lumbosacral spinal cord. *Am J Physiol Regul Integr Comp Physiol* **289**(4):R1155-1168.

Suadicani SO, Brosnan CF and Scemes E (2006) P2X7 receptors mediate ATP release and amplification of astrocytic intercellular Ca2+ signaling. *J Neurosci* 26(5):1378-1385.

Sugaya K, Chou S, Xu SJ and McKinney M (1998) Indicators of glial activation and brain oxidative stress after intraventricular infusion of endotoxin. *Brain Res Mol Brain Res* 58(1-2):1-9.

Sung CS, Wen ZH, Chang WK, Ho ST, Tsai SK, Chang YC and Wong CS (2004) Intrathecal interleukin-1beta administration induces thermal hyperalgesia by activating inducible nitric oxide synthase expression in the rat spinal cord. *Brain Res* **1015**(1-2):145-153.

Surprenant A, Rassendren F, Kawashima E, North RA and Buell G (1996) The cytolytic P2Z receptor for extracellular ATP identified as a P2X receptor (P2X7). *Science* **272**(5262):735-738.

Svensson CI, Hua XY, Protter AA, Powell HC and Yaksh TL (2003) Spinal p38 MAP kinase is necessary for NMDA-induced spinal PGE(2) release and thermal hyperalgesia. *Neuroreport* 14(8):1153-1157.

Swarup V, Ghosh J, Duseja R, Ghosh S and Basu A (2007) Japanese encephalitis virus infection decrease endogenous IL-10 production: correlation with microglial activation and neuronal death. *Neurosci Lett* **420**(2):144-149.

Sweitzer SM, Colburn RW, Rutkowski M and DeLeo JA (1999) Acute peripheral inflammation induces moderate glial activation and spinal IL-1beta expression that correlates with pain behavior in the rat. *Brain Res* **829**(1-2):209-221.

Takeuchi H, Jin S, Wang J, Zhang G, Kawanokuchi J, Kuno R, Sonobe Y, Mizuno T and Suzumura A (2006) Tumor necrosis factor-alpha induces neurotoxicity via glutamate release from hemichannels of activated microglia in an autocrine manner. J Biol Chem 281(30):21362-21368.

Takeuchi O, Hoshino K, Kawai T, Sanjo H, Takada H, Ogawa T, Takeda K and Akira S (1999) Differential roles of TLR2 and TLR4 in recognition of gram-negative and gram-positive bacterial cell wall components. *Immunity* **11**(4):443-451.

Tanabe T, Otani H, Mishima K, Ogawa R and Inagaki C (1996) Mechanisms of oxyradical production in substance P stimulated rheumatoid synovial cells. *Rheumatol Int* **16**(4):159-167.

Tanabe Y, Nomura A, Masu M, Shigemoto R, Mizuno N and Nakanishi S (1993) Signal transduction, pharmacological properties, and expression patterns of two rat metabotropic glutamate receptors, mGluR3 and mGluR4. *J Neurosci* **13**(4):1372-1378.

Tanga FY, Nutile-McMenemy N and DeLeo JA (2005) The CNS role of Toll-like receptor 4 in innate neuroimmunity and painful neuropathy. *Proc Natl Acad Sci USA* **102**(16):5856-5861.

Taylor AL, Kudlow BA, Marrs KL, Gruenert DC, Guggino WB and Schwiebert EM (1998) Bioluminescence detection of ATP release mechanisms in epithelia. *Am J Physiol* 275(5 Pt 1):C1391-1406.

Taylor DL, Diemel LT and Pocock JM (2003) Activation of microglial group III metabotropic glutamate receptors protects neurons against microglial neurotoxicity. J Neurosci 23(6):2150-2160.

Taylor DL, Jones F, Kubota ES and Pocock JM (2005) Stimulation of microglial metabotropic glutamate receptor mGlu2 triggers tumor necrosis factor alpha-induced neurotoxicity in concert with microglial-derived Fas ligand. *J Neurosci* **25**(11):2952-2964.

Tikka TM and Koistinaho JE (2001) Minocycline provides neuroprotection against N-methyl-D-aspartate neurotoxicity by inhibiting microglia. *J Immunol* **166**(12):7527-7533.

Tilleux S and Hermans E (2007) Neuroinflammation and regulation of glial glutamate uptake in neurological disorders. *J Neurosci Res* **85**(10):2059-2070.

Tobias PS, Soldau K, Gegner JA, Mintz D and Ulevitch RJ (1995) Lipopolysaccharide binding protein-mediated complexation of lipopolysaccharide with soluble CD14. *J Biol Chem* **270**(18):10482-10488.

Torrens Y, Beaujouan JC, Saffroy M, Daguet de Montety MC, Bergstrom L and Glowinski J (1986) Substance P receptors in primary cultures of cortical astrocytes from the mouse. *Proc Natl Acad Sci US A* **83**(23):9216-9220.

Torrens Y, Beaujouan JC, Saffroy M, Glowinski J and Tence M (1998) Functional coupling of the NK1 tachykinin receptor to phospholipase D in chinese hamster ovary cells and astrocytoma cells. *J Neurochem* **70**(5):2091-2098.

Trapido-Rosenthal HG, Carr WE and Gleeson RA (1990) Ectonucleotidase activities associated with the olfactory organ of the spiny lobster. *J Neurochem* **55**(1):88-96.

Tsai MJ, Chang YF, Schwarcz R and Brookes N (1996) Characterization of L-alphaaminoadipic acid transport in cultured rat astrocytes. *Brain Res* **741**(1-2):166-173.

Tsan MF and Gao B (2004) Endogenous ligands of Toll-like receptors. *J Leukoc Biol* **76**(3):514-519.

Chapter 7

Tsuda M, Shigemoto-Mogami Y, Koizumi S, Mizokoshi A, Kohsaka S, Salter MW and Inoue K (2003) P2X4 receptors induced in spinal microglia gate tactile allodynia after nerve injury. *Nature* **424**(6950):778-783.

Tsuda M, Ueno S and Inoue K (1999) In vivo pathway of thermal hyperalgesia by intrathecal administration of alpha,beta-methylene ATP in mouse spinal cord: involvement of the glutamate-NMDA receptor system. *Br J Pharmacol* **127**(2):449-456.

Uceyler N, Rogausch JP, Toyka KV and Sommer C (2007) Differential expression of cytokines in painful and painless neuropathies. *Neurology* **69**(1):42-49.

Van Bockstaele EJ and Colago EE (1996) Selective distribution of the NMDA-R1 glutamate receptor in astrocytes and presynaptic axon terminals in the nucleus locus coeruleus of the rat brain: an immunoelectron microscopic study. *J Comp Neurol* **369**(4):483-496.

Vaughan DW and Peters A (1974) Neuroglial cells in the cerebral cortex of rats from young adulthood to old age: an electron microscope study. *J Neurocytol* **3**(4):405-429.

Venero JL, Santiago M, Tomas-Camardiel M, Matarredona ER, Cano J and Machado A (2002) DCG-IV but not other group-II metabotropic receptor agonists induces microglial BDNF mRNA expression in the rat striatum. Correlation with neuronal injury. *Neuroscience* **113**(4):857-869.

Ventura R and Harris KM (1999) Three-dimensional relationships between hippocampal synapses and astrocytes. J Neurosci 19(16):6897-6906.

Verdoorn TA, Burnashev N, Monyer H, Seeburg PH and Sakmann B (1991) Structural determinants of ion flow through recombinant glutamate receptor channels. *Science* **252**(5013):1715-1718.

Verma S, Nakaoke R, Dohgu S and Banks WA (2006) Release of cytokines by brain endothelial cells: A polarized response to lipopolysaccharide. *Brain Behav Immun* **20**(5):449-455.

Vincent B, Vincent JP and Checler F (1994) Neurotensin and neuromedin N undergo distinct catabolic processes in murine astrocytes and primary cultured neurons. *Eur J Biochem* **221**(1):297-306.

Viriyakosol S, Tobias PS, Kitchens RL and Kirkland TN (2001) MD-2 binds to bacterial lipopolysaccharide. *J Biol Chem* 276(41):38044-38051.

Chapter 7

Visintin A, Halmen KA, Latz E, Monks BG and Golenbock DT (2005) Pharmacological inhibition of endotoxin responses is achieved by targeting the TLR4 coreceptor, MD-2. *J Immunol* **175**(10):6465-6472.

Visintin A, Mazzoni A, Spitzer JA and Segal DM (2001) Secreted MD-2 is a large polymeric protein that efficiently confers lipopolysaccharide sensitivity to Toll-like receptor 4. *Proc Natl Acad Sci US A* **98**(21):12156-12161.

Viviani B, Bartesaghi S, Gardoni F, Vezzani A, Behrens MM, Bartfai T, Binaglia M, Corsini E, Di Luca M, Galli CL and Marinovich M (2003) Interleukin-1beta enhances NMDA receptor-mediated intracellular calcium increase through activation of the Src family of kinases. *J Neurosci* 23(25):8692-8700.

Vulchanova L, Riedl MS, Shuster SJ, Buell G, Surprenant A, North RA and Elde R (1997) Immunohistochemical study of the P2X2 and P2X3 receptor subunits in rat and monkey sensory neurons and their central terminals. *Neuropharmacology* **36**(9):1229-1242.

Walker KM, Urban L, Medhurst SJ, Patel S, Panesar M, Fox AJ and McIntyre P (2003) The VR1 antagonist capsazepine reverses mechanical hyperalgesia in models of inflammatory and neuropathic pain. *J Pharmacol Exp Ther* **304**(1):56-62.

Wanaka A, Matsumoto K, Kashihara Y, Furuyama T, Tanaka T, Mori T, Tanno Y, Yokoya S, Kitanaka J, Takemura M and Tohyama M (1997) LIM -homeodomain gene family in neural development. *Dev Neurosci* **19**(1):97-100.

Wang Z, Haydon PG and Yeung ES (2000) Direct observation of calciumindependent intercellular ATP signaling in astrocytes. *Anal Chem* 72(9):2001-2007.

Warsame Afrah A, Gustafsson H, Olgart L, Brodin E, Stiller CO and Taylor BK (2004) Capsaicin-evoked substance P release in rat dorsal horn increases after peripheral inflammation: a microdialysis study. *Neurosci Lett* **368**(2):226-230.

Watkins LR, Martin D, Ulrich P, Tracey KJ and Maier SF (1997) Evidence for the involvement of spinal cord glia in subcutaneous formalin induced hyperalgesia in the rat. *Pain* 71(3):225-235.

Whim MD and Lloyd PE (1989) Frequency-dependent release of peptide cotransmitters from identified cholinergic motor neurons in Aplysia. *Proc Natl Acad Sci U S A* 86(22):9034-9038.

White WF, Brown KL and Frank DM (1989) Glycine binding to rat cortex and spinal cord: binding characteristics and pharmacology reveal distinct populations of sites. J Neurochem 53(2):503-512.

Wienrich M and Kettenmann H (1989) Activation of substance P receptors leads to membrane potential responses in cultured astrocytes. *Glia* **2**(3):155-160.

Wieseler-Frank J, Maier SF and Watkins LR (2005) Central proinflammatory cytokines and pain enhancement. *Neurosignals* 14(4):166-174.

Wiesinger H, Schuricht B and Hamprecht B (1991) Replacement of glucose by sorbitol in growth medium causes selection of astroglial cells from heterogeneous primary cultures derived from newborn mouse brain. *Brain Res* **550**(1):69-76.

Williams K, Dooley N, Ulvestad E, Becher B and Antel JP (1996) IL-10 production by adult human derived microglial cells. *Neurochem Int* **29**(1):55-64.

Wilson M, Seymour R and Henderson B (1998) Bacterial perturbation of cytokine networks. *Infect Immun* 66(6):2401-2409.

Wirjatijasa F, Dehghani F, Blaheta RA, Korf HW and Hailer NP (2002) Interleukin-4, interleukin-10, and interleukin-1-receptor antagonist but not transforming growth factor-beta induce ramification and reduce adhesion molecule expression of rat microglial cells. *J Neurosci Res* **68**(5):579-587.

Woiciechowsky C, Schoning B, Stoltenburg-Didinger G, Stockhammer F and Volk HD (2004) Brain-IL-1 beta triggers astrogliosis through induction of IL-6: inhibition by propranolol and IL-10. *Med Sci Monit* **10**(9):BR325-330.

Womack MD, MacDermott AB and Jessell TM (1988) Sensory transmitters regulate intracellular calcium in dorsal horn neurons. *Nature* **334**(6180):351-353.

Woolf CJ and Salter MW (2000) Neuronal plasticity: increasing the gain in pain. *Science* 288(5472):1765-1769.

Wright SD, Ramos RA, Tobias PS, Ulevitch RJ and Mathison JC (1990) CD14, a receptor for complexes of lipopolysaccharide (LPS) and LPS binding protein. *Science* **249**(4975):1431-1433.

Wu G, Whiteside GT, Lee G, Nolan S, Niosi M, Pearson MS and Ilyin VI (2004) A-317491, a selective P2X3/P2X2/3 receptor antagonist, reverses inflammatory mechanical hyperalgesia through action at peripheral receptors in rats. *Eur J Pharmacol* **504**(1-2):45-53.

Eryn Louise Werry

Wu JV, Joo NS, Krouse ME and Wine JJ (2001) Cystic fibrosis transmembrane conductance regulator gating requires cytosolic electrolytes. *J Biol Chem* **276**(9):6473-6478.

Wu YP, Matsuda J, Kubota A, Suzuki K and Suzuki K (2000) Infiltration of hematogenous lineage cells into the demyelinating central nervous system of twitcher mice. *J Neuropathol Exp Neurol* **59**(7):628-639.

Wu Z, Zhang J and Nakanishi H (2005) Leptomeningeal cells activate microglia and astrocytes to induce IL-10 production by releasing pro-inflammatory cytokines during systemic inflammation. *J Neuroimmunol* **167**(1-2):90-98.

Wulff K (1983) Lumionometry, in *Methods of enzymatic analysis* (Bergmeyer HU ed) pp 340-386, Deutsche Bibliothek, Weinheim.

Wurfel MM, Kunitake ST, Lichenstein H, Kane JP and Wright SD (1994) Lipopolysaccharide (LPS)-binding protein is carried on lipoproteins and acts as a cofactor in the neutralization of LPS. *J Exp Med* **180**(3):1025-1035.

Xu J, Weng YI, Simonyi A, Krugh BW, Liao Z, Weisman GA and Sun GY (2002) Role of PKC and MAPK in cytosolic PLA2 phosphorylation and arachadonic acid release in primary murine astrocytes. *J Neurochem* **83**(2):259-270.

Yamada J, Sawada M and Nakanishi H (2006) Cell cycle-dependent regulation of kainate-induced inward currents in microglia. *Biochem Biophys Res Commun* **349**(3):913-919.

Yamagata T and Ichinose M (2006) Agents against cytokine synthesis or receptors. *Eur J Pharmacol* **533**(1-3):289-301.

Yano S, Fukunaga K, Takiguchi M, Ushio Y, Mori M and Miyamoto E (1996) Regulation of CCAAT/enhancer-binding protein family members by stimulation of glutamate receptors in cultured rat cortical astrocytes. *J Biol Chem* **271**(38):23520-23527.

Yashpal K, Dam TV and Quirion R (1990) Quantitative autoradiographic distribution of multiple neurokinin binding sites in rat spinal cord. *Brain Res* **506**(2):259-266.

Yau WM, Bowen DJ and Youther ML (1991) Evidence for an involvement of eicosanoids in neurokinin3-receptor mediated acetylcholine release from myenteric neurons. *Neurosci Lett* **129**(2):259-261.

Zaratin P, Angelici O, Clarke GD, Schmid G, Raiteri M, Carita F and Bonanno G (2000) NK3 receptor blockade prevents hyperalgesia and the associated spinal cord substance P release in monoarthritic rats. *Neuropharmacology* **39**(1):141-149.

Zdanov A (2004) Structural features of the interleukin-10 family of cytokines. *Curr Pharm Des* **10**(31):3873-3884.

Zerari F, Karpitskiy V, Krause J, Descarries L and Couture R (1997) Immunoelectron microscopic localization of NK-3 receptor in the rat spinal cord. *Neuroreport* **8**(12):2661-2664.

Zerari F, Karpitskiy V, Krause J, Descarries L and Couture R (1998) Astroglial distribution of neurokinin-2 receptor immunoreactivity in the rat spinal cord. *Neuroscience* **84**(4):1233-1246.

Zhang J, Shi XQ, Echeverry S, Mogil JS, De Koninck Y and Rivest S (2007a) Expression of CCR2 in both resident and bone marrow-derived microglia plays a critical role in neuropathic pain. *J Neurosci* 27(45):12396-12406.

Zhang JM, Wang HK, Ye CQ, Ge W, Chen Y, Jiang ZL, Wu CP, Poo MM and Duan S (2003) ATP released by astrocytes mediates glutamatergic activity-dependent heterosynaptic suppression. *Neuron* **40**(5):971-982.

Zhang Z, Chen G, Zhou W, Song A, Xu T, Luo Q, Wang W, Gu XS and Duan S (2007b) Regulated ATP release from astrocytes through lysosome exocytosis. *Nat Cell Biol* **9**(8):945-953.

Zhao W, Oskeritzian CA, Pozez AL and Schwartz LB (2005) Cytokine production by skin-derived mast cells: endogenous proteases are responsible for degradation of cytokines. *J Immunol* 175(4):2635-2642.

Zhou F, Yao HH, Wu JY, Yang YJ, Ding JH, Zhang J and Hu G (2006) Activation of Group II/III metabotropic glutamate receptors attenuates LPS-induced astroglial neurotoxicity via promoting glutamate uptake. *J Neurosci Res* 84(2):268-277.

Zhou Z, Peng X, Hao S, Fink DJ and Mata M (2007) HSV-mediated transfer of interleukin-10 reduces inflammatory pain through modulation of membrane tumor necrosis factor alpha in spinal cord microglia. *Gene Ther.*

Zhuang ZY, Wen YR, Zhang DR, Borsello T, Bonny C, Strichartz GR, Decosterd I and Ji RR (2006) A peptide c-Jun N-terminal kinase (JNK) inhibitor blocks mechanical allodynia after spinal nerve ligation: respective roles of JNK activation in

Chapter 7

primary sensory neurons and spinal astrocytes for neuropathic pain development and maintenance. J Neurosci 26(13):3551-3560.

Ziak D, Chvatal A and Sykova E (1998) Glutamate-, kainate- and NMDA-evoked membrane currents in identified glial cells in rat spinal cord slice. *Physiol Res* **47**(5):365-375.

Zocchia C, Spiga G, Rabin SJ, Grekova M, Richert J, Chernyshev O, Colton C and Mocchetti I (1997) Biological activity of interleukin-10 in the central nervous system. *Neurochem Int* **30**(4-5):433-439.

Zorumski CF and Thio LL (1992) Properties of vertebrate glutamate receptors: calcium mobilization and desensitization. *Prog Neurobiol* **39**(3):295-336.

Eryn Louise Werry

CHAPTER 8

APPENDIX

Eryn Werry

Glutamate-stimulated ATP release from spinal cord astrocytes is potentiated by substance P

Eryn L. Werry, Guo Jun Liu and Max R. Bennett

Neurobiology Laboratory, Discipline of Physiology, Bosch Institute, University of Sydney, New South Wales, Australia

Abstract

ATP has recently emerged as a key molecule mediating pathological pain. The aim of this study was to examine whether spinal cord astrocytes could be a source of ATP in response to the nociceptive neurotransmitters glutamate and substance P. Glutamate stimulated ATP release from these astrocytes and this release was greatly potentiated by substance P, even though substance P alone did not elicit ATP release. Substance P also potentiated glutamate-induced inward currents, but did not cause such currents alone. When glutamate was applied alone it acted exclusively through α -amino-3-hydroxy-5-methylisoxazole-4-proprionate receptors

ATP is a key spinal cord neurotransmitter. The ventral horn and intermediolateral column both contain P2X2 and P2X7 receptors (Kanjhan et al. 1999; Deuchars et al. 2001; Deng and Fyffe 2004), and ATP application to phrenic motor neurons in the ventral horn leads to increased firing of these diaphragm-innervating neurons (Miles et al. 2002). In addition, it has recently been discovered that inflammatory and neuropathic pain are completely removed in mice lacking the ATP P2X7 receptor, which is found on microglial cells among others (Chessell et al. 2005; Xiang and Burnstock 2005). Furthermore, formalin-induced hyperalgesia behaviour is significantly reduced in mice lacking the ATP P2X3 receptor, and activation of P2X3 receptors on nociceptor nerve terminals in the spinal cord enhances transmission of chronic inflammatory and neuropathic pain (Kennedy et al. 2003). In addition, nerve injury-induced mechanical allodynia involves up-regulation of ATP P2X4 receptors on microglia as a required molecular mediator (Tsuda et al. 2003). It appears that the level of ATP in the vicinity of nociceptor synapses in the dorsal horn is likely to be an important regulator of inflammatory and neuropathic pain. The question arises as to the source of this ATP in these areas of the spinal cord.

Astrocytes in the spinal cord are a major source of ATP, which they use to transmit propagating waves of intracellular Ca^{2+} increases (Salter and Hicks 1994). Studies suggest that

to stimulate Ca²⁺ influx-dependent ATP release. However, when substance P was co-applied with glutamate, ATP release could be elicited by activation of NMDA and metabotropic glutamate receptors. Activation of neurokinin receptor subtypes, protein kinase C and phospholipases A₂, C and D were needed for substance P to bring about its effects. These results suggest that astrocytes may be a major source of ATP in the spinal cord on activation of nerve fibres that release substance P and glutamate.

Keywords: astrocytes, ATP, glutamate, substance P. J. Neurochem. (2006) 99, 924–936.

astrocytes can release much greater amounts of ATP than neurons (Queiroz et al. 1999). ATP released from astrocytes in response to neurotransmitters can modulate neuronal

E-mail: maxb@physiol.usyd.edu.au

Abbreviations used: A23187, calcimycin A23187; AMPA, a-amino-3-hydroxy-5-methylisoxazole-4-proprionate; Aristol, aristolochic acid; B-Ala, [B-Ala⁸]-neurokinin A fragment 4-10; BSA, bovine serum albumin; CCS, cosmic calf serum; CheCl, chelerythrinechloride; CNQX, 6-cyano-7-nitroquinoxaline-2,3-dione; DDA, 2',5'dideoxyadenosine; DMEM, Dulbecco's modified Eagle's medium; GAMS, y-D-glutamylamino methylsulfonic acid; GFAP, glial fibrillary acidic protein; GYKI, GYKI52466; H 33342, bisBenzimide H 33342 trihydrochloride; IP₃, inositol triphosphate; KA, kainate; L703606, cis-2-(diphenylmethyl)-N-[2-iodophenylmethyl]-1-azabicyclo[2.2.2]octan-3-amine oxalate salt; MCPG, (+)-a-methyl-4-carboxyphenylglycine; mGlu, metabotropic glutamate; MK-801, (+)-MK-801 hydrogen maleate; NBMPR, S-(4-nitrobenzyl)-6-thioinosine; NeuN, Neuron-specific Nuclear protein; PBS, phosphate-buffered saline; PIP2, phosphatidylinositol-4,5-bisphosphate; PKC, protein kinase C; PLA2, phospholipase A2; PLC, phospholipase C; PLD, phospholipase D; Sar, sarcosine; SarMet, [Sar9, Met(O2)11]-substance P; Senk, senktide; SP, substance P; tACPD, trans-(1S,3R)-1-amino-1,3-cyclopentanedicarboxylic acid.

© 2006 The Authors

Journal Compilation © 2006 International Society for Neurochemistry, J. Neurochem. (2006) 99, 924-936

Received May 1, 2006; revised manuscript received July 5, 2006; accepted July 19, 2006.

Address correspondence and reprint requests to Max R. Bennett, Neurobiology Laboratory, Discipline of Physiology, Bosch Institute, University of Sydney, NSW 2006, Australia.

transmission (Zhang *et al.* 2003) and microglial function (Verderio and Matteoli 2001). Astrocytes are in intimate contact with nerve terminals in the spinal cord (Zerari *et al.* 1998) as they are at all central synapses and are exposed to spinal neurotransmitters. Two neurotransmitters released in the dorsal horn, ventral horn and intermediolateral column of the spinal cord are glutamate and substance P (SP) (Helke *et al.* 1982; Kapoor *et al.* 1992; Sasaki *et al.* 1998; Allen *et al.* 1999; Hokfelt *et al.* 2000) and spinal cord astrocytes possess receptors to both transmitters (Wienrich and Kettenmann 1989; Beaujouan *et al.* 1990; Agrawal and Fehlings 1997; Zerari *et al.* 1998).

Given this background, it is natural to examine whether spinal cord astrocytes, excited by the concomitant application of glutamate and SP, release ATP in relatively large amounts which could excite the range of ionotropic purinergic receptors in the spinal cord that have been implicated in events such as respiration and inflammatory pain. We report here that glutamate can stimulate ATP release from spinal cord astrocytes and that this release is greatly enhanced when SP is present. Mechanisms involved in this phenomenon are reported. Preliminary accounts of this work have appeared (Werry *et al.* 2005).

Materials and methods

Spinal cord and cortical astrocyte cultures

Astrocytes were cultured according to Cole and deVellis (1997) with modifications. The University of Sydney Animal Ethics Committee approved the procedures, and procedures were in accordance with the Australian Government National Health and Medical Research Council code of practice for the care and use of animals for scientific purposes. Briefly, whole spinal cords from 0-2-day-old neonatal rats were dissected out, and the meninges and peripheral nerves removed. The remaining spinal cord tissue was incubated in 0.125% porcine trypsin (Sigma-Aldrich, St Louis, MO, USA) dissolved in Hanks balanced salt solution (136.89 mM NaCl, 5.37 mм KCl, 4.17 mм NaHCO3, 0.44 mм KH2PO4, 0.34 mм Na₂HPO₄, 5.55 mM D-glucose, pH 7.4) for 20 min at 37°C and then washed with Dulbecco's modified Eagle's medium (DMEM; Sigma-Aldrich) supplemented with 10% New Zealand cosmic calf serum (CCS; Hyclone Logan, UT, USA) and 1% penicillin/ streptomycin/glutamine (Invitrogen, Carlsbad, CA, USA) and triturated. Dissociated cells were plated on a culture flask coated with poly-D-lysine (20 µg/mL; Sigma) for 4 h in a 5% CO2 incubator at 37°C. Non-attached cells were then removed and cultures were maintained at 5% CO2 and 37°C. The culture medium was replaced with fresh supplemented DMEM every 3-4 days. Twice during in vitro days 14-20 cultures were shaken on an IKA-VIBRAX-VXR shaker (Janke & Kunkel, Staufen, Germany) at 7 g and 37°C for 4 h in a Certomat chamber (B. Braun, Bethlehem, PA, USA). Detached cells were predominantly microglia and oligodendrocytes (Cole and deVellis 1997), and were removed from the culture by replacement of the culture medium. After in vitro day 16 the numbers of remaining microglia and oligodendrocytes were further reduced by replacing DMEM supplemented with CCS and penicillin/streptomycin/glutamine with sorbitol medium [DMEM without glucose (Sigma), 10% horse serum (Trace Bioscientific, Castle Hill, NSW, Australia), 1% penicillin/streptomycin/glutamine and 25 mM sorbitol (Sigma)] (Wiesinger *et al.* 1991). Oligo-dendrocytes and microglia do not have aldose reductase and sorbitol dehydrogenase, and consequently cannot use sorbitol for energy. Thus if sorbitol is substituted for glucose in the culture medium, oligodendrocytes and microglia die off, leaving astrocytes which have the enzymes necessary to use sorbitol for energy (Wiesinger *et al.* 1991). After 21 days, the age at which there becomes only one predominant (> 90%) subtype of spinal cord astrocyte (Black *et al.* 1993), cells were transferred to 13-mm glass coverslips and were used for ATP release, electrophysiology and immunocytochemistry experiments 1–2 days after transfer.

Online bioluminescence

ATP release was measured by online bioluminescence as detailed elsewhere (Taylor *et al.* 1998; Liu *et al.* 2005). A luciferinluciferase enzymatic ATP assay mix (Sigma) was used to measure ATP release; for every molecule of ATP that is evolved in solution, a photon of light is produced (Taylor *et al.* 1998). Consequently, the amount of ATP released by cells can be indirectly calculated by measuring in real time the number of photons produced by cells incubated in the luciferin–luciferase assay with a photomultiplier.

Coverslips were washed in HEPES-buffered salt solution (140 mM NaCl, 5 mM KCl, 10 mM HEPES, 1 mM CaCl₂, 1 mM MgCl₂, pH 7.4) for 5 min. Excess solution was removed from cellfree sides of the coverslip and it was placed on an 18-mm round plastic photomultiplier stage. Eighty microlitres of ATP assay mix (1 mg/mL; Sigma) was immediately added to the coverslip. The photomultiplier stage was placed inside a light-impermeable chamber and photons were counted with a P30CWAD5-45 photodetector package (Electron Tubes, Ruislip, UK) and displayed using an RS232 photon counting program (Electron Tubes). Photons were collected every second and experiments were performed at room temperature (22–24°C).

The initial photon count was high as there was an imbalance between the basal amount of ATP produced and accumulated by cells and the amount of ATP consumed by the ATP assay mix and the ecto-ATPases. Some 60-80 min later the amount of ATP produced by cells and the amount of ATP consumed by the assay and ecto-ATPases reached equilibrium, illustrated by the photon count readings not varying by more than 10%. When this stable photon count extended for 10 min, the average photon count over this period of time was recorded as the baseline level. After the baseline had been reached, recording of photon counts was stopped and agonists of interest were applied at 10% of the volume of liquid on the coverslip. Recording was immediately recommenced (within 20 s) after addition of the chemicals. Chemicals were not washed out during experimentation. At the end of each experiment, 1% Triton X-100 (Bio-Rad, Hercules, CA, USA) was added to lyse cells, allowing an approximation of the number of cells on the coverslip. For each batch of ATP assay mix used for experiments, a standard curve was computed using known concentrations of ATP, allowing conversion of the photon count data to ATP concentration values.

ATP release data were analysed to determine the peak increase in ATP release relative to baseline (referred to as relative peak ATP level). Experiments were repeated at least three times and results were expressed as mean \pm SEM relative peak ATP level. Means were analysed for statistically significant differences with an unpaired Student's two-sample two-tailed *t*-test. p < 0.05 was considered statistically significant.

All inhibitors were applied 1 h before addition of agonists. All agonists and antagonists used in bioluminescence studies were purchased from Sigma-Aldrich, except thapsigargin epoxide from Alomone Laboratories (Jerusalem, Israel) and iodotubercidin from EMD Biosciences (San Diego, CA, USA). All chemicals were dissolved in HEPES-buffered salt solution except the following: thapsigargin, thapsigargin epoxide, aristolochic acid, 2',5'dideoxyadenosine (DDA), A23187, 7-chlorokynurenic acid, S-(4-nitrobenzyl)-6-thioinosine (NBMPR) and iodotubercidin were dissolved in dimethyl sulfoxide (maximum final concentration 0.17%); (+)-a-methyl-4-carboxyphenylglycine (MCPG) was dissolved in 0.1 м NaOH (maximum final concentration 0.7%); GYKI52466 was dissolved in 2.5 mM HCl and [β-Ala8]-neurokinin A fragment 4-10 was dissolved in 20% acetonitrile and 0.01% trifluoroacetic acid. None of the solvents at the concentrations used affected the luciferase activity, induced ATP release from the astrocytes by themselves or affected glutamate-stimulated ATP release. All antagonists used in experiments involving co-application of SP and glutamate were examined for an effect on glutamate-stimulated ATP release, and no significant effects of the antagonists were found, apart from 6-cyano-7-nitroquinoxaline-2,3-dione (CNQX).

Patch-clamp recording

Perforated patch-clamp was used to record whole-cell currents in astrocytes using a Multiclamp 700A amplifier (Axon Instruments, Sunnyvale, CA, USA) at room temperature (22-24°C). Astrocytes were identified according to the morphological features of type I astrocytes, which possess a large flat cell body in comparison to microglia and oligodendrocytes. The morphology of these astrocytes corresponded with the morphology of cells that stained positive for glial fibrillary acidic protein (GFAP) in immunocytochemistry experiments and differed from the smaller CDllb-positive microglia. Resistances of pipettes were 4-8 M Ω when a patch pipette was filled with pipette solution containing 145 mM CsCl, 1 mM MgCl₂, 10 mM EGTA, 10 mm HEPES and 250 µg/mL nystatin (pH 7.3); the external bath solution contained 140 mM NaCl, 5 mM KCl, 1 mM CaCl₂, 1 mм MgCl₂ and 10 mм HEPES (pH 7.4). Mg²⁺ was omitted from the bath solution when NMDA was used as an agonist. Wholecell currents were recorded in voltage-clamp mode at a holding membrane potential of - 60 mV. The currents were sampled on-line using a Digidata 1322A interface and pClamp8 program (Axon Instruments). Glutamate, its analogs and SP were applied by pressure ejection via a picosprizer (General Valve, Fairfield, NJ, USA).

Immunocytochemistry

Cells grown on coverslips from the same culture batches used in bioluminescence experiments were washed twice for 5 min each with the HEPES-buffered salt solution used in bioluminescence studies. Cells on coverslips were then fixed with 2% formaldehyde solution (BDH chemicals, Kilsyth, VA, Australia) in phosphate-buffered saline (PBS; 136.89 mM NaCl, 2.68 mM KCl, 10.14 mM Na₂HPO₄, 1.76 mM KH₂PO₄, pH 7.4) for 10 min. Coverslips were washed twice for 5 min each with PBS and then incubated in PBS

containing 1% bovine serum albumin (BSA; Sigma-Aldrich) and 0.05% saponin (Sigma-Aldrich) for 30 min. Cells on coverslips were then incubated overnight at 4°C with primary antibodies diluted in PBS containing 1% BSA and 0.05% saponin. Primary antibodies were washed out three times for 10 min each with PBS and then cells were incubated in secondary antibodies diluted in PBS containing 1% BSA and 0.05% saponin for 1 h at room temperature (22–24°C).

The coverslips were washed twice for 10 min each with Ca^{2+}/Mg^{2+} -free HEPES-buffered salt solution and incubated in the DNA marker bisBenzimide H 33342 trihydrochloride (H 33342; 5 µg/mL; Sigma) for 15 min. Following this, coverslips were washed twice for 10 min each with Ca^{2+}/Mg^{2+} -free HEPES-buffered salt solution and then mounted on glass slides using 20% glycerol (Sigma-Aldrich) in PBS and sealed with nail polish.

Fluorescence was visualized with an Axioplan2 upright microscope (Zeiss, Jena, Germany) and images were acquired with an AxioCam HR digital monochrome charge-coupled device camera (Zeiss) using Axiovision 4.0 software (Zeiss). Experiments with each antibody were performed on at least three different batches of cultures and were repeated at least twice on each of the batches. For each coverslip at least three fields of view were imaged. No autofluorescence of primary antibodies or non-specific binding of secondary antibodies was observed.

To verify the results obtained using the above method, experiments probing the NMDA receptor 1 and glutamate receptor 4 subunits were repeated with one of the following changes per experiment: omission of incubation of cells in H 33342; substitution of the step in which cells were incubated in saponin and BSA with three steps in which cells were incubated in 1% Triton X-100 which was washed out three times with PBS followed by incubation for 1 h in 1% BSA; use of a different rabbit anti-rat NMDA receptor 1 antibody (1 : 50-1 : 200; Chemicon, Temecula, CA, USA); and incubation of cells in rabbit anti-rat glutamate receptor antibodies before fixation of cells with formaldehyde solution. Results using these different methods were consistent.

Materials

Agonists used in bioluminescence experiments were: neurotransmitters glutamate (30 µм - 10 mм) and SP (3-300 µм); tachykinin receptor agonists [Sar9, Met(O2)11]-SP (100 µм), [β-Ala8]-neurokinin A fragment 4-10 (64 µm) and succinyl-[Asp⁶, N-Me-Phe⁸]-SP fragment 6-11 (senktide; 100 µм); glutamate receptor subtype agonists kainate (5 mm), α-amino-3-hydroxy-5-methylisoxazole-4proprionate (AMPA; 50 µм), NMDA (500 µм to 5 mм), trans-(1S,3R)-1-amino-1,3-cyclopentanedicarboxylic acid (tACPD; 1 mм) and Ca2+ ionophore calcinycin A23187 (10 µM). Inhibitors used in the bioluminescence experiments were: NMDA receptor antagonist (+)-MK-801 hydrogen maleate (MK-801; 40 µM); kainate receptor antagonist y-D-glutamylamino methylsulfonic acid (GAMS; 30 µм); metabotropic glutamate (mGlu) receptor antagonist (+)-α-methyl-4carboxyphenylglycine (MCPG; 500 µм); AMPA receptor antagonist GYKI52466 hydrochloride (50 µм); kainate and AMPA receptor antagonist CNQX (20 µM); phosphatidylinositol-4,5-bisphosphate (PIP₂) synthesis inhibitor LiCl (1 mм); endoplasmic reticulum Ca²⁺dependent ATPase inhibitor thapsigargin (1 µM); the inactive form of thapsigargin, called thapsigargin epoxide (1 µM); protein kinase C (PKC) inhibitor chelerythrine chloride (20 µм); voltage-dependent

© 2006 The Authors Journal Compilation © 2006 International Society for Neurochemistry, J. Neurochem. (2006) 99, 924–936 Ca²⁺ channel inhibitor CdCl₂ (1 mM); Ca²⁺ chelator EGTA (5 mM); phospholipase D (PLD) inhibitor 1-butanol (0.5%); cell-permeable adenylyl cyclase inhibitor DDA (50 μM); neurokinin-1 and neurokinin-2 receptor antagonist spantide I (3 μM); neurokinin-1 antagonist *cis*-2-(diphenylmethyl)-*N*-[2-iodophenylmethyl]-1-azabicyclo[2.2.2] octan-3-amine oxalate salt (L703606; 10 μM); phospholipase A₂ (PLA₂) antagonist aristolochic acid (50 μM); NMDA receptor glycine co-binding-site antagonist 7-chlorokynurenic acid (5 μM); nucleoside transporter 1 inhibitor NBMPR (10 μM); adenosine kinase inhibitor iodotubercidin (10 μM); and ecto-ATPase inhibitor ARL 67156 trisodium salt (100 μM).

Primary monoclonal antibodies used were the astrocyte-specific marker mouse anti-rat GFAP (1:800; Sigma), microglia- and macrophage-specific mouse anti-rat CD11b (1: 100; Chemicon) and neuron-specific antibody anti-Neuron-specific Nuclear protein (NeuN) (1: 150; Chemicon). Primary polyclonal antibodies used were the oligodendrocyte marker rabbit anti-rat galactocerebroside (1:50-1:200; Chemicon), rabbit anti-rat glutamate receptor 2/3 (1:50-1:200; Sigma-Aldrich), rabbit anti-rat glutamate receptor 4 (1:50-1:200; Tocris, Bristol, UK), rabbit anti-rat mGlu receptor 2/3 (1:50-1:200; Sigma-Aldrich), rabbit anti-rat mGlu 10/5 receptor (1: 250; Chemicon) and rabbit anti-rat NMDA receptor 1 (1:50-1:200; Sigma-Aldrich). Secondary antibodies were goat anti-mouse IgG conjugated to Alexa Fluor 488 (1:100), used to detect primary monoclonal antibodies, and goat anti-rabbit IgG conjugated to Alexa Fluor 594 (1:100), used to detect primary polyclonal antibodies. Both secondary antibodies were from Molecular Probes (Eugene, OR, USA).

Results

SP enhances glutamate-stimulated ATP release from spinal cord astrocytes

We initially determined the ability of SP and glutamate to stimulate ATP release from spinal cord astrocytes when applied separately and together using online bioluminescence. Glutamate (1 mm) elicited approximately a 2-fold increase in ATP release levels at the peak of its release compared with the baseline level (2.19 \pm 0.11; n = 67) (Figs 1a and b). This measure is referred to as the relative peak ATP level. Peak ATP release was reached 5-7 min after the application of glutamate and ATP release levels returned to baseline within 30 min after application (Fig. 1b). When applied alone at concentrations of 100 µm or less, SP did not change ATP release levels from baseline level (1 μ M, n = 3; 10 μ M, n = 12; 30 µм, n = 3; 100 µм, n = 15) (Figs 1a and b). However, when 100 µM SP and 1 mM glutamate were co-applied, ATP release levels significantly increased to approximately 7-fold above the baseline level $(7.35 \pm 0.68;$ n = 38; p < 0.001) (Figs 1a and b). This equated to a 336% potentiation of glutamate-stimulated ATP release by SP compared with the effect of glutamate alone. SP did not alter the time course of glutamate-stimulated ATP release (Fig. 1b).

To verify the identity of the cells acted upon by SP and glutamate to bring about the modulation of ATP release,

immunocytochemistry was carried out. Over 90% of cells were positive for the astrocyte-specific marker GFAP, whereas less than 10% were positive for the microglial and macrophage marker CD11b. Anti-galactocerebroside nonspecifically labelled all cells. Consequently the presence of oligodendrocytes was identified on the basis of morphological characteristics and oligodendrocytes were found to comprise < 1% of total cells. There were no neurons present in cultures. To examine a possible contribution of microglia and oligodendrocytes to the modulation of glutamatestimulated ATP release by SP, the effect of SP on glutamate-stimulated ATP release from coverslips with total microglia and oligodendrocyte populations of $\leq 4\%$ were compared with that from coverslips with the same number of astrocytes but with microglia and oligodendrocytes added until the total microglial and oligodendrocyte population reached 30%. In cells with ≤4% microglia and oligodendrocytes, SP (100 µm) increased glutamate (1 mm)-stimulated ATP release by 412%, with an ATP release of 1.64 ± 0.16 (n = 3) times above basal ATP release values being elicited by glutamate only and 6.78 ± 1.32 (n = 4) times by co-application of SP and glutamate. Similarly, on cells with 30% microglia and oligodendrocytes, SP (100 µм) increased glutamate (1 mm)-stimulated ATP release by 379%, with an ATP release of 1.61 ± 0.15 (n = 3) times above basal ATP release values being elicited by glutamate only and 6.10 ± 2.98 (n = 3) times by co-application of SP and glutamate. These results suggest that microglia and oligodendrocytes do not contribute to the interaction of SP with glutamate-stimulated ATP release, and that the potentiation results from SP and glutamate acting on astrocytes.

It was possible that ATP measured on application of agonists could be a consequence of adenosine release through the equilibrative nucleoside transporter 1 and subsequent anabolism by adenosine kinase to ATP. Neither the equilibrative nucleoside transporter 1 antagonist NBMPR ($10 \ \mu M$; 7.06 ± 1.78 ; n = 6) nor the adenosine kinase antagonist iodotubercidin ($10 \ \mu M$; 7.92 ± 1.01 ; n = 3) significantly changed SP and glutamate-stimulated ATP release. Similarly the ecto-ATPase inhibitor ARL 67156 did not have a significant effect on SP and glutamate-stimulated ATP release ($100 \ \mu M$; 6.27 ± 1.10 ; n = 4).

To examine whether SP modulated glutamate-induced currents in spinal cord astrocytes, glutamate-induced wholecell currents were recorded and the effects of SP were determined. Glutamate (1 mM) caused slow inward currents in 17 of 19 cells from which recordings were made, which took approximately 10 s to peak. The average peak amplitude of the inward currents was 49 ± 9 pA. SP (100 μ M) did not cause inward currents in any cells examined (n = 4). However, on co-application of glutamate and SP, inward currents were significantly larger than those recorded from astrocytes exposed to glutamate alone (150 \pm 29 pA; n = 3; p < 0.05) (Fig. 1c).



The potentiation of glutamate-stimulated ATP release by SP is concentration and time dependent

To examine whether the interaction between SP and glutamate is dependent on the concentration of glutamate, ATP release was measured on co-application of increasing concentrations of glutamate with a constant 100 μ M SP, and compared with the level of ATP release stimulated by the corresponding concentration of glutamate alone. In the absence of SP, 30 μ M glutamate elicited ATP release (1.51 ± 0.25; n = 4) and increasing concentrations increased ATP release in a dose-dependent manner (100 μ M, 1.66 ± 0.24, n = 4; 300 μ M, 1.78 ± 0.29, n = 3; 1 mM, 2.19 ± 0.11, n = 67). Concentrations above 1 mM (3 mM,

Fig. 1 SP significantly potentiates glutamate-stimulated ATP release from cultured spinal cord astrocytes. (a) Summary histogram of the effect of application of 100 µM SP, 1 mM glutamate (Glu) and coapplication of 1 mm Glu and 100 µm SP (Glu + SP) on ATP release relative to the basal release level from cultured spinal cord astrocytes. In this figure and histograms in following figures, relative peak ATP level is defined as the peak ATP release divided by the baseline ATP value. p < 0.001 for ATP release stimulated by glutamate plus SP versus that stimulated by glutamate. Values are mean ± SEM. ***p < 0.001 (Student's two-sample two-tailed t-test) (b) Representative time courses of ATP release induced by these agonists from spinal cord astrocytes. The arrow represents time of addition of agonists to the bath. (c) Representative trace of a glutamate (1 mm)-induced inward current from a spinal cord astrocyte that was dramatically enhanced by 100 µM SP, wheras SP itself did not induce currents. The black bars above the current traces represent the duration of agonist application.

 1.99 ± 0.51 , n = 4; 10 mm, 2.15 ± 0.25 , n = 3) did not further increase ATP release. The EC_{50} was 234 μ M and E_{max} was 2.11 when glutamate was applied alone (Fig. 2a). When 100 μM SP was co-administered with glutamate, potentiation of the level of ATP release was seen even at the lowest concentrations of glutamate examined (30 μ M, 2.11 \pm 0.43, n = 3; 100 µm, 2.39 ± 0.66, n = 4). Although ATP release stimulated by co-application of SP with 30 µM and 100 µM glutamate was not significantly greater than that induced by corresponding concentrations of glutamate in the absence of SP, their relative values were comparable with the Emax of glutamate alone. Statistically significantly greater levels of ATP release were seen at concentrations of 300 µM glutamate $(3.41 \pm 0.78, n = 6, p < 0.05)$ and above (1 mm, p < 0.05)7.35 ± 0.68, n = 38, p < 0.001; 3 mm, 7.74 ± 1.40, n = 4, p < 0.01; 10 mm, 7.96 ± 1.51, n = 3, p < 0.05). The EC₅₀ was 347 µm, which was comparable to that obtained with application of glutamate in the absence of SP. However, the Emax was 7.91, which was appreciably higher than that obtained with glutamate alone (Fig. 2a).

To examine whether the interaction between SP and glutamate was dependent on the concentration of SP, ATP release was measured on co-application of various concentrations of SP with a constant 1 mM glutamate, and compared with that elicited by the ATP release stimulated by 1 mM glutamate alone. No significant increases above the glutamate-only level were seen with co-application of 1 μ M (1.65 ± 0.16, n = 3) or 10 μ M (1.63 ± 0.13, n = 3) SP. Significant increases were seen with higher concentrations of SP (30 μ M, 3.47 ± 0.53, n = 4, p < 0.05; 100 μ M, 7.35 ± 0.68, n = 38, p < 0.001; 300 μ M, 7.52 ± 0.92, n = 7, p < 0.001). The EC₅₀ of SP was 34 μ M and the E_{max} was 7.60 (Fig. 2b). Consequently, SP was used at a concentration of 100 μ M and glutamate at 1 mM for the remaining studies.

The time dependence of the potentiation of glutamatestimulated ATP release by SP was investigated by applying glutamate at different intervals after SP application. SP



Fig. 2 Potentiation of glutamate-stimulated ATP release by SP is concentration dependent. (a) Glutamate dose dependently stimulated ATP release (black circles) and the degree of modulation by SP (100 μ M) increased with increasing concentrations of glutamate (white circles). At each concentration of glutamate examined, values obtained with SP were compared with those without SP. (b) SP dose dependently increased glutamate-stimulated ATP release when co-applied with 1 mM glutamate. At each concentration of SP used, values were compared with the ATP release level obtained with application of 1 mM glutamate alone. Values are mean ± SEM. *p < 0.05, **p < 0.01, ***p < 0.001 (unpaired Student's two-sample two-tailed *t*-test).

potentiated glutamate-stimulated ATP release when the two agonists were co-applied (0 min, 7.35 ± 0.68 , n = 38). However when glutamate was added 30 s (2.10 ± 0.21 , n = 3), 2 min (2.23 ± 0.29 , n = 3), 5 min (2.13 ± 0.39 , n = 3), 10 min (2.27 ± 0.35 , n = 3) and 15 min (1.76 ± 0.29 , n = 3) after SP, no potentiation was observed. Therefore SP and glutamate were co-applied in further experiments. An examination of the degree of potentiation when SP was applied at different intervals after glutamate application could not be performed. This was because levels of ATP release were increasing after application of glutamate, and this rendered comparison of potentiation on addition of SP beyond the already increasing levels impossible between experiments. It was not examined whether the time dependency of the potentiation was inherent in the mechanisms leading to the modulation or whether it was attributable to the breakdown of SP by endopeptidases known to be expressed on spinal cord astrocytes (Lentzen and Palenker 1983; Marcel *et al.* 1990).

Neurokinin receptors mediate the potentiation of glutamate-stimulated ATP release by SP

SP is known to bind three subtypes of neurokinin receptors. These are the neurokinin-1, -2 and -3 receptors (Regoli et al. 1994). The selective neurokinin-1 agonist [Sar⁹, Met(O₂)¹¹]-SP (100 μ M; 2.06 \pm 0.45; n = 4), the selective neurokinin-2 agonist [β-Ala⁸]-neurokinin A fragment 4-10 (64 μм; 1.74 ± 0.51 ; n = 3) and the selective neurokinin-3 agonist senktide (100 μ M; 1.83 \pm 0.21; n = 3) did not significantly potentiate glutamate-stimulated ATP release (Fig. 3a). However when [Sar9, Met(O2)11]-SP and [\beta-Ala8]-neurokinin A fragment 4-10 were co-applied with glutamate, a significant potentiation of glutamate-stimulated ATP release was observed $(3.51 \pm 0.38; n = 3; p < 0.05)$ (Fig. 3a). Furthermore, when all three neurokinin agonists were co-applied with glutamate, greater potentiation of glutamate-stimulated ATP release was recorded (4.85 \pm 0.94; n = 6) (Fig. 3a). The neurokinin-1 and -2 preferring antagonist spantide I (3 µm) significantly reduced SP and glutamate-stimulated relative ATP release to levels seen when glutamate was applied alone $(2.03 \pm 0.31; n = 4; p < 0.001)$ (Fig. 3b). Similarly, the neurokinin-1 preferring antagonist L703606 (10 µm) abolished the potentiating effects of SP $(1.90 \pm 0.25; n = 4; p < 0.001)$ (Figs 3b and c).

Glutamate activates AMPA receptors to bring about ATP release

The contribution of NMDA, AMPA, kainate and mGlu receptors to glutamate-stimulated ATP release in the absence of SP was examined. Application of 50 µM AMPA stimulated ATP release $(2.41 \pm 0.34; n = 9)$ whereas 5 mm kainate (n = 4), 1 mM tACPD (n = 4), a mGlu receptor agonist, and 5 mm NMDA (n = 3) did not lead to ATP release (Figs 4a and b). In agreement with these results, glutamate-stimulated ATP release was significantly reduced by 50 µM GYKI52466 hydrochloride, an AMPA receptor antagonist (1.16 \pm 0.16; n = 5; p < 0.001) and 20 μ M CNQX, an AMPA and kainate antagonist $(1.2 \pm 0.07; n =$ 4; p < 0.001 (Figs 4c and d). However, it was not significantly effected by 30 µM GAMS, a kainate receptor antagonist (1.78 ± 0.22; n = 4), 500 µм MCPG, a mGlu 1, 2, 3 and 5 receptor antagonist (1.85 \pm 0.21; n = 4), and 40 μ M MK-801, a NMDA receptor antagonist $(2.39 \pm 0.44; n = 5)$ (Fig. 4c). AMPA (1 mm) induced slow inward currents which were similar to those induced by glutamate in all



Fig. 3 SP acts on neurokinin receptor subtypes to potentiate glutamate-stimulated ATP release. (a) Summary histogram of the effect of neurokinin receptor agonists [Sar9, (SarMet; Met(O2)11]-SP neurokinin-1 receptor agonist; 100 μм), [β-Ala8]-neurokinin A fragment 4-10 (B-Ala; neurokinin-2 receptor agonist; 64 µм) and senktide (Senk; neurokinin-3 receptor agonist: 100 µм) on glutamate-stimulated ATP release. (b) Summary histogram of the effect of spantide I (neurokinin-1 and -2 preferring antagonist; 3 µм) and L703606 (neurokinin-1 preferring antagonist; 10 µм) on ATP release stimulated by co-application of SP and glutamate. (c) Representative time courses of ATP release stimulated by addition of glutamate, glutamate and SP, and glutamate and SP in the presence of L703606. The time courses resulting from glutamate and glutamate plus SP in the presence of L703606 overlap. Values are mean ± SEM. *p < 0.05, ***p < 0.001 (unpaired Student's two-sample two-tailed t-test).

spinal cord astrocytes tested (50–75 pA; n = 3) (Fig. 4e). NMDA (1 mM; n = 4) and tACPD (1 mM; n = 5) did not produce inward currents in spinal cord astrocytes (Fig. 4e).

To examine whether glutamate receptor subtypes were available on the membrane of spinal cord astrocytes to be bound by glutamate, immunocytochemistry was conducted. The AMPA glutamate receptor 2/3 subunit was found in fine clusters throughout the cytoplasm and membrane of all GFAP-positive spinal cord astrocytes imaged (n = 35 astrocytes; Fig. 5a). In contrast, the AMPA glutamate receptor 4 subunit was localized specifically to an area overlapping and immediately surrounding the H 33342-stained nuclei of spinal cord astrocytes (n = 36 astrocytes) (Fig. 5b). The mGlu 1a5 receptor stained positively on 8.8% of imaged GFAP-positive astrocytes (n = 29 of 329 astrocytes) and was distributed in thick clusters throughout the membrane and cytoplasm of spinal cord astrocytes (Fig. 5c). The NMDA receptor glycine co-agonist subunit NR1 was found localized to an area overlapping the H 33342-stained nuclei of spinal cord astrocytes in 64% of cells imaged (n = 74 of 115 astrocytes) (Fig. 5d), whereas the distribution of NR1 also extended throughout the cytoplasm in the remaining cells imaged (n = 41 of 115 astrocytes) (Fig. 5d). The colocalization of glutamate receptor 4 and NR1 with astrocytic nuclei persisted despite the use of a variety of immunocytochemical methods and antibodies.

SP selectively enables glutamate to stimulate ATP release by binding NMDA and mGlu receptors

SP is known to enhance the activity of some neuronal glutamate receptor subtypes, such as the NMDA receptor (Randic *et al.* 1990). It was therefore determined whether SP changes the ability of glutamate receptor agonists to stimulate ATP release. Co-application of SP and 500 μ M NMDA led to ATP release (2.29 \pm 0.26; n = 9) (Figs 6a and b), as did co-application of SP and 1 mm tACPD (2.15 \pm 0.23; n = 7) (Fig. 6a). Application of 500 μ M NMDA, 1 mM tACPD and 100 μ M SP led to further increases in ATP release (3.34 \pm 1.02; n = 6) (Figs 6a and c). No ATP release was seen on co-application of SP and AMPA (50 μ M; n = 3) (Fig. 6a). Co-application of SP and AMPA (50 μ M; 2.46 \pm 0.49; n = 5) led to ATP release, which was not significantly different from that oberseved with application of AMPA alone (Fig. 6a).

The potentiation by SP of glutamate-stimulated ATP release was significantly reduced by 500 μ M MCPG (3.48 ± 0.50; n = 5; p = 0.001) (Fig. 6d) and 40 μ M MK-801 (3.57 ± 0.64; n = 6; p < 0.001) (Fig. 6d). Furthermore,



Fig. 4 Glutamate stimulates ATP release by activating AMPA receptors. (a) Summary histogram of the effects of glutamate receptor agonists glutamate (1 mм), AMPA (50 µм), kainate (KA; 5 mм), tACPD (mGlu receptor agonist; 1 mm) and NMDA (5 mm) on ATP release from spinal cord astrocytes. (b) Time course of the effect of AMPA and kainate on ATP release. (c) Summary histogram of the effects of glutamate receptor antagonists GYKI52466 (GYKI; AMPA receptor antagonist; 50 µm), CNQX (AMPA and kainate receptor antagonist; 20 µм), GAMS (kainate receptor antagonist; 30 µм), MCPG (mGlu receptor antagonist; 500 µм) and MK-801 (NMDA receptor antagonist; 40 µm). (d) Time course of the effect of CNQX on glutamate-stimulated ATP release. (e) Representative traces of patch-clamp recordings from spinal cord astrocytes on application of AMPA (1 mm), NMDA (1 mm) and tACPD (1 mm). The black bars above the current traces indicate the duration of addition of the agonists. Values are mean ± SEM. *** p < 0.001 (Student's unpaired two-sample two-tailed t-test).

SP and glutamate co-stimulated relative ATP release levels were reduced to glutamate-only levels when cells were preincubated with both 500 μ M MCPG and 40 μ M MK-801 (2.13 ± 0.41; n = 4; p < 0.001) (Fig. 6d).

The possibility that SP may have assisted in the modulation of NMDA receptors by increasing extracellular levels of glycine or by potentiating the function of the glycine co-binding site of the NMDA receptor was examined. The potentiation of NMDA (500 μ M)-induced ATP release by SP was not significantly changed by preincubation with the glycine co-binding site antagonist 7-chlorokynurenic acid (5 μ M; 1.76 \pm 0.27; n = 6). Another possible mechanism by which SP may have modulated the NMDA receptor was by depolarizing the cell and removing the Mg²⁺ block. However, because of the Mg²⁺ dependence of the bioluminescence enzymes, it was not possible to conduct experiments in which the concentration of Mg²⁺ was varied.

Multiple intracellular signalling pathways are involved in the potentiation of glutamate-stimulated ATP release by SP

Neurokinin receptors are G protein-coupled receptors linked to four cell signalling pathways: PLC, PLA₂, PLD and adenylyl cyclase (Nakajima *et al.* 1992; Rayner and Van Helden 1997; Torrens *et al.* 1998). Activation of PLC leads to cleavage of PIP₂ to produce inositol triphosphate (IP₃) and diacyl glycerol. IP₃ can liberate intracellular Ca²⁺ stores. Increased intracellular Ca²⁺ and diacyl glycerol as well as PLA₂ activation can activate PKC. Potential roles of each of these cell signalling components were explored.

The PKC inhibitor chelerythrine chloride (20 µм) abolished the effects of SP on glutamate-stimulated ATP release $(2.09 \pm 0.53; n = 6; p < 0.001)$ (Fig. 7a), as did the PLA₂ inhibitor aristolochic acid (50 μ M; 2.40 ± 0.16; n = 4; p < 0.001) (Figs 7a and b). The potentiation of glutamatestimulated ATP release by SP was also significantly reduced by 0.5% 1-butanol, a PLD antagonist (3.44 \pm 0.87; n = 5; p < 0.01), and 1 mM LiCl, an inositol monophosphate as inhibitor $(3.93 \pm 0.73; n = 8; p < 0.01)$. However, these antagonists did not fully abolish the effects of SP (Fig. 7a). Thapsigargin (1 μм), an endoplasmic reticulum Ca²⁺-pump activator inhibitor, also decreased the potentiation $(5.43 \pm 0.56; n = 6; p < 0.05)$. However, thapsigargin reduced the potentiation to the same levels of ATP release as the inactive form of thapsigargin, thapsigargin epoxide (1 μ M; 5.55 ± 1.16; n = 4). The cell-permeable adenylyl cyclase antagonist DDA (50 μ M; 8.63 ± 1.66; n = 5) did not significantly affect SP and glutamate-stimulated ATP release (Fig. 7a). Furthermore, the potentiation of NMDA-stimulated ATP release by SP was significantly decreased by 20 µM chelerythrine chloride $(1.12 \pm 0.33; n = 4; p < 0.05)$ (Fig. 7c), but was not significantly affected by 0.5% 1-butanol (2.54 \pm 0.68; n = 4) (Fig. 7c).



Fig. 5 Representative images of AMPA receptor subunit glutamate receptor 2/3 staining (a), AMPA receptor subunit glutamate receptor 4 staining (b), mGlu 1a/5 receptor staining (c) and NMDA receptor subunit NR1 staining (d) of spinal cord astrocytes. Images on the left show glutamate receptor subtype staining in red, those in the centre show astrocyte intermediate filament marker GFAP staining in green, and those on the right images are merged images with additional H 33342staining of the nucleus in blue. Co-localization of glutamate receptor and GFAP staining appears yellow, whereas co-localization of glutamate receptor and nuclear staining appears violet. Scale bar is 20 µm and applies to all images in the figure.

As NMDA and some AMPA receptors are Ca^{2+} permeable, the role of extracellular Ca^{2+} influx in ATP release was examined. Astrocytes incubated in a Ca^{2+} -free bath solution containing the Ca^{2+} chelator EGTA (5 mM) had significantly reduced glutamate-stimulated ATP release levels (1.30 ± 0.07; n = 5; p < 0.001) compared with glutamate-stimulated ATP release from astrocytes in medium containing 500 μ M Ca²⁺, which was used throughout the studies described previously (Fig. 7d). A possible nonreceptor source of Ca²⁺ influx on application of glutamate may have been voltage-dependent Ca²⁺ channels. However, a voltage-dependent Ca²⁺ channel blocker, CdCl₂, did not significantly affect glutamate-stimulated ATP release (1 mM; 2.40 ± 0.54; n = 3).

The potentiation of NMDA-stimulated ATP release by SP was significantly decreased in Ca²⁺-free medium containing 5 mM EGTA compared with the relative peak ATP release level seen when NMDA and SP were added to cells in medium containing 500 μ M Ca²⁺ (1.10 \pm 0.14; n = 4; p < 0.001) (Fig. 7d). Influx of extracellular Ca²⁺ was shown to be sufficient to induce ATP release as the Ca²⁺ ionophore A23187 triggered ATP release (10 μ M; 2.38 \pm 0.64; n = 3) (Fig. 7d).

Discussion

Receptors involved in glutamate-stimulated ATP release and its potentiation by SP

The glutamate receptors involved in generating glutamateevoked inward currents and ATP release are primarily of the AMPA class as the effects of glutamate on ATP release were blocked by an AMPA receptor antagonist, but were not significantly blocked by antagonists to kainate, NMDA or mGlu receptors. Furthermore, only AMPA evoked inward currents and stimulated ATP release. In agreement with findings in spinal cord astrocytes in situ (Brand-Schieber et al. 2004), immunocytochemistry showed the existence of AMPA glutamate receptor 2/3 subunits on the surface of these cultured spinal cord astrocytes. The glutamate receptor 4 subunit of the AMPA receptor was not found on spinal cord astrocyte membranes. Glutamate receptor 4 subunits have been selectively identified on the endfeet of astrocytes surrounding blood vessels (Brand-Schieber et al. 2004), which were of course absent from our cultures.

Surprisingly ATP release was not elicited by kainate, which is known to activate AMPA receptors. Kainate has been reported to cause small membrane potential changes in



Fig. 6 SP allows glutamate to stimulate ATP release through NMDA and mGlu receptors. (a) Summary histogram of the effect of SP (100 μM) on the amount of ATP release stimulated by glutamate receptor subtype agonists AMPA (50 μM), kainate (5 mM), tACPD (1 mM) and NMDA (500 μM), and co-application of NMDA and tACPD. (b, c) Time course of the effect of SP on (b) NMDA-stimulated and (c) NMDA plus tACPD-stimulated ATP release. (d) Summary histogram demonstrating the effect of application of MCPG (500 μM), MK-801 (40 μM) and co-application of MCPG and MK-801 on SP + glutamatestimulated ATP release. Values are mean ± SEM. ****p* < 0.001 (Student's unpaired two-sample two-tailed *t*-test).

spinal cord astrocytes, although it has not been determined whether these responses are due to the action on kainate or AMPA receptors (Ziak *et al.* 1998). Consequently, we hypothesize that, in the present case, either kainate does not activate spinal cord astrocyte AMPA receptors to the extent needed to produce ATP release or AMPA receptors on spinal cord astrocytes are insensitive to kainate, possibly owing to a primary structure differing at key amino acid sites that regulate the kainate sensitivity of AMPA receptors (Rodriguez *et al.* 2005).

Although NMDA and mGlu receptor agonists did not elicit ATP release when applied alone, when they were each co-applied with SP a release of ATP was observed. SP had no effect on the ATP release responses of spinal cord astrocytes to AMPA and kainate. These results suggest a model whereby glutamate stimulates ATP release through AMPA receptors, but in the presence of SP it can also stimulate ATP release through NMDA and mGlu receptors, leading to the observed potentiation of glutamate-stimulated ATP release.



Fig. 7 Modulation of glutamate-stimulated ATP release by SP requires Ca2+ influx and activation of multiple intracellular signalling pathways. (a) Summary histogram of the effect of intracellular signalling molecule antagonists chelerythrine chloride (CheCl; PKC inhibitor; 20 µм), aristolochic acid (Aristol; PLA2 inhibitor; 50 µм), 1-butanol (PLD inhibitor; 0.5%), LiCl (PIP2 synthesis inhibitor; 1 mм) and DDA (adenylyl cyclase inhibitor; 50 µM) on the potentiation of glutamate-stimulated ATP release by SP. (b) Typical time course of the effect of aristolochic acid on the potentiation of glutamate-stimulated ATP release by SP. The time courses of ATP release stimulated by glutamate and glutamate plus SP in the presence of aristolochic acid overlap. (c) Summary histogram of the effect of chelerythrine chloride and 1-butanol on the potentiation of NMDA-stimulated ATP release by SP. (d) Effect of Ca2+-free medium with 5 mm EGTA (zero Ca2+) on glutamate-stimulated and NMDA plus SP-stimulated ATP release, and the effect of Ca^{2+} ionophore A23187 (10 μ M; n = 3) on ATP release. Values are mean \pm SEM. *p < 0.05, **p < 0.01, ***p < 0.001 (Student's unpaired two-sample two-tailed t-test).

In further support of this conclusion, simultaneous antagonism of NMDA and mGlu receptors completely abolished the potentiation of glutamate-stimulated ATP release by SP. Furthermore, with regard to the concentration studies carried out, the E_{max} value of glutamate-stimulated ATP release in the presence of SP was much higher than that in the absence of SP. This suggests that SP enhanced the number of ATP release-linked glutamate receptors in the membrane.

Immunocytochemistry identified NR1 subunits and mGlu $1\alpha/5$ receptors expressed on the surface of a portion of spinal cord astrocytes, in accordance with previous reports (Aicher

et al. 1997; Silva et al. 1999), yet the fact that glutamate alone does not stimulate ATP release through these receptors suggests that they are not functionally linked to the ATP release pathway. How SP links these receptors to the ATP release pathway remains unclear. In the case of the NMDA receptor, one contribution SP may make is to change the receptor such that glutamate does not rely on glycine cobinding to activate the NMDA receptor, as the potentiation of NMDA-stimulated ATP release by SP occurred even in the presence of a glycine co-binding-site antagonist. In addition, PKC is known to enhance the activity of NMDA receptors (Liao et al. 2001) and arachidonic acid downstream of PLA2 activation is known to enhance the activity of mGlu receptors (Collins et al. 1995). As both of these signalling molecules are known to be activated on binding neurokinin-1 (Rayner and Van Helden 1997; Monastyrskaya et al. 2005) and are central to the effects of SP in this study, SP may possibly potentiate the activity of these receptors to a degree where they can activate enough signalling molecules to release ATP.

Surprisingly, the SP and NMDA co-stimulated response occurred despite the presence of Mg^{2+} at a concentration known to inhibit neuronal NMDA receptor-mediated responses (Ziak *et al.* 1998). There are several examples of NMDA-mediated responses in glial cells that occur at a physiological level of Mg^{2+} , such as NMDA-stimulated ATP release from cortical astrocytes (Queiroz *et al.* 1997) and NMDA-induced inward currents in spinal cord astrocytes from rats younger than 14 days (Ziak *et al.* 1998). As suggested by Ziak *et al.* (1998), perhaps glial NMDA receptors contain Mg^{2+} -resistant subunits such as the $\epsilon 3$ subunit.

To bring about potentiation of glutamate-stimulated ATP release, it is likely that SP acted synergistically on multiple neurokinin receptor subtypes. A neurokinin-1 receptor preferring antagonist abolished the effects of SP, suggesting that an action on the neurokinin-1 subtype was necessary for the effects of SP. However a selective neurokinin-1 receptor agonist was not sufficient to potentiate glutamate-stimulated ATP release. Similarly, selective stimulation of neurokinin-2 or -3 receptors was not sufficient to potentiate glutamatestimulated ATP release. It was only when agonists to the neurokinin receptor subtypes were applied in combination that potentiation was observed. This suggests that the neurokinin-1 receptor is necessary but not sufficient for the actions of SP on glutamate-stimulated ATP release, and that a synergistic action on neurokinin-2 and/or neurokinin-3 receptors is also necessary for the potentiation. This is consistent with reports of SP acting as an agonist at all three neurokinin receptor subtypes (Regoli et al. 1994; Drew et al. 2005) and with reports of functional neurokinin-1, -2 and - 3 receptors on spinal cord astrocytes (Wienrich and Kettenmann 1989; Zerari et al. 1998).

It was unlikely that the ATP found in solution after addition of agonists was due to release of adenosine and subsequent actions of adenosine kinase to produce ATP because antagonists to an adenosine transporter and adenosine kinase did not effect SP and glutamate-stimulated ATP release. Potential mechanisms of direct ATP release were not explored in this study, but previous studies have reported direct ATP release from astrocytes through P2X7 receptors (Suadicani *et al.* 2006), connexin hemichannels (Cotrina *et al.* 1998; Stout *et al.* 2002), exocytosis (Abdipranoto *et al.* 2003; Coco *et al.* 2003) and anion transporters (Abdipranoto *et al.* 2003; Anderson *et al.* 2004).

Ca²⁺ influx and various intracellular signalling pathways are involved in glutamate-stimulated ATP release and its potentiation by SP

Significant decreases in the glutamate-evoked release of ATP in the presence of SP were obtained on blocking PKC, PLA₂, PLD and IP₃, but not adenylyl cyclase. It is known that activation of neurokinin-1 receptors activates PKC in human embryonic kidney 293 cells (Monastyrskaya *et al.* 2005) and PLA₂ is also increased on activation of neurokinin-1 receptors on endothelial cells (Rayner and Van Helden 1997). SP increases PLD, PLC and phosphatidylinositol hydrolysis in Chinese hamster cells and astrocytes (Marriott *et al.* 1991; Torrens *et al.* 1998) but it does not increase (Rougon *et al.* 1983).

ATP release evoked by activation of NMDA receptors in the presence of SP was greatly reduced in Ca2+-free external solution, indicating that Ca2+ influx is necessary for this effect. It is unlikely that this influx of Ca2+ ions is due to activation of neurokinin receptors as we were unable to detect any inward currents in spinal cord astrocytes in response to applied SP. It is known that SP can directly enhance PLC and phophatidylinositol hydrolysis in membrane preparations independently of Ca²⁺ (Nakajima et al. 1992). We suggest therefore that the dependence of the effect of SP on NMDA-stimulated ATP release involves an influx of Ca2+ through the NMDA receptor. In addition, influx of Ca2+ was centrally involved in glutamate-stimulated ATP release as Ca2+ influx was sufficient to elicit ATP release and removal of extracellular Ca2+ greatly attenuated glutamatestimulated ATP release. This Ca2+ influx could be directly through non-glutamate receptor 2-containing AMPA receptors or as a result of Na⁺ influx through AMPA receptors activating the Na⁺/Ca²⁺ exchanger (Smith et al. 2000).

From the present findings it would seem that SP-potentiated glutamate-evoked ATP release is dependent on two pathways. The first is a direct action of glutamate on AMPA receptors resulting in influx of Ca^{2+} . An additional pathway is recruited by SP through its actions on neurokinin receptors, producing activation of PKC, PLA₂, IP₃ and PLD. As a result of this activation glutamate is able to bring about ATP release by binding NMDA and mGlu receptors.

© 2006 The Authors

Possible applications of the potentiation of glutamate-

stimulated ATP release by SP from spinal cord astrocytes SP and glutamate-containing nerve terminals are found in descending bulbospinal neurons in the ventral horn (Hokfelt *et al.* 2000) and in the intermediolateral column (Helke *et al.* 1982; Kapoor *et al.* 1992). The results of the present study suggest that a consequence of SP and glutamate release in these areas may be increased release of ATP from astrocytes. As these areas are rich in purinergic receptors (Kanjhan *et al.* 1999; Deuchars *et al.* 2001; Deng and Fyffe 2004), this may lead to an increase in purinergic transmission in these areas, with possible functional consequences such as increased phrenic motor innervation of the diaphragm (Miles *et al.* 2002).

SP and glutamate are also found in nociceptive nerve terminals in the dorsal horn, from which they are both released by relatively high frequencies of nerve stimulation, and an increase in their release is seen during inflammatory pain (Sasaki et al. 1998; Allen et al. 1999; Honore et al. 2000). Their concomitant release during heightened activity in nociceptor terminals enhances pain transmission in the dorsal horn of the spinal cord (Mjellem-Joly et al. 1992; Zahn and Brennan 1998; Ishizaki et al. 1999). This study suggests that one consequence of increased glutamate and SP release during inflammatory pain may be increased ATP release from astrocytes in close apposition to these terminals. This ATP may then act on P2X3, 4 and 7 receptors in the dorsal horn, the activation of which has been reported to lead to release of pro-inflammatory cytokines, brain-derived neurotrophic factor and neurotransmitters, and consequently chronic pain (Kennedy et al. 2003; Bianco et al. 2005; Coull et al. 2005). Hence this increased ATP release from astrocytes in response to SP and glutamate may be one mechanism that contributes to the pain-enhancing effects of these transmitters.

Acknowledgements

This work was supported by a NSW Spinal Cord Project Grant. We thank T. Knight for assistance with fluorescence microscopy and P. Tan and R. Dampney for donation of anti-NeuN.

References

- Abdipranoto A., Liu G. J., Werry E. L. and Bennett M. R. (2003) Mechanisms of secretion of ATP from cortical astrocytes triggered by uridine triphosphate. *Neuroreport* 14, 2177–2181.
- Agrawal S. K. and Fehlings M. G. (1997) Role of NMDA and non-NMDA ionotropic glutamate receptors in traumatic spinal cord axonal injury. J. Neurosci. 17, 1055–1063.
- Aicher S. A., Sharma S., Cheng P. Y. and Pickel V. M. (1997) The N-methyl-D-aspartate (NMDA) receptor is postsynaptic to substance P-containing axon terminals in the rat superficial dorsal horn. Brain Res. 772, 71–81.

- Allen B. J., Li J., Menning P. M., Rogers S. D., Ghilardi J., Mantyh P. W. and Simone D. A. (1999) Primary afferent fibers that contribute to increased substance P receptor internalization in the spinal cord after injury. J. Neurophysiol. 81, 1379–1390.
- Anderson C. M., Bergher J. P. and Swanson R. A. (2004) ATP-induced ATP release from astrocytes. J. Neurochem. 88, 246–256.
- Beaujouan J. C., Daguet de Montety M. C., Torrens Y., Saffroy M., Dietl M. and Glowinski J. (1990) Marked regional heterogeneity of ¹²⁵I-Bolton Hunter substance P binding and substance P-induced activation of phospholipase C in astrocyte cultures from the embryonic or newborn rat. J. Neurochem. 54, 669–675.
- Bianco F., Pravettoni E., Colombo A., Schenk U., Moller T., Matteoli M. and Verderio C. (2005) Astrocyte-derived ATP induces vesicle shedding and IL-1 beta release from microglia. J. Immunol. 174, 7268–7277.
- Black J. A., Sontheimer H. and Waxman S. G. (1993) Spinal cord astrocytes *in vitro*: phenotypic diversity and sodium channel immunoreactivity. *Glia* 7, 272–285.
- Brand-Schieber E., Lowery S. L. and Werner P. (2004) Select ionotropic glutamate AMPA/kainate receptors are expressed at the astrocyte– vessel interface. *Brain Res.* 1007, 178–182.
- Chessell I. P., Hatcher J. P., Bountra C. et al. (2005) Disruption of the P2X7 purinoceptor gene abolishes chronic inflammatory and neuropathic pain. Pain 114, 386–396.
- Coco S., Calegari F., Pravettoni E., Pozzi D., Taverna E., Rosa P., Matteoli M. and Verderio C. (2003) Storage and release of ATP from astrocytes in culture. J. Biol. Chem. 278, 1354–1362.
- Cole R. and deVellis J. (1997) Astrocyte and oligodendrocyte cultures, in Protocols for Neural Cell Culture, 2nd edn (Fedoroff S. and Richardson A., eds), pp. 117–130. Humana Press Inc., New Jersey.
- Collins D. R., Smith R. C. and Davies S. N. (1995) Interactions between arachidonic acid and metabotropic glutamate receptors in the induction of synaptic potentiation in the rat hippocampal slice. *Eur. J. Pharmacol.* 294, 147–154.
- Cotrina M. L., Lin J. H., Alves-Rodrigues A., Liu S., Li J., Azmi-Ghadimi H., Kang J., Naus C. C. and Nedergaard M. (1998) Connexins regulate calcium signaling by controlling ATP release. *Proc. Natl Acad. Sci. USA* 95, 15 735–15 740.
- Coull J. A., Beggs S., Boudreau D., Boivin D., Tsuda M., Inoue K., Gravel C., Salter M. W. and De Koninck Y. (2005) BDNF from microglia causes the shift in neuronal anion gradient underlying neuropathic pain. *Nature* 438, 1017–1021.
- Deng Z. and Fyffe R. E. (2004) Expression of P2X7 receptor immunoreactivity in distinct subsets of synaptic terminals in the ventral horn of rat lumbar spinal cord. *Brain Res.* 1020, 53–61.
- Deuchars S. A., Atkinson L., Brooke R. E., Musa H., Milligan C. J., Batten T. F., Buckley N. J., Parson S. H. and Deuchars J. (2001) Neuronal P2X7 receptors are targeted to presynaptic terminals in the central and peripheral nervous systems. J. Neurosci. 21, 7143–7152.
- Drew G. M., Mitchell V. A. and Vaughan C. W. (2005) Postsynaptic actions of substance P on rat periaqueductal grey neurons in vitro. *Neuropharmacology* 49, 587–595.
- Helke C. J., Neil J. J., Massari V. J. and Loewy A. D. (1982) Substance P neurons project from the ventral medulla to the intermediolateral cell column and ventral horn in the rat. *Brain Res.* 243, 147–152.
- Hokfelt T., Arvidsson U., Cullheim S., Millhorn D., Nicholas A. P., Pieribone V., Seroogy K. and Ulfhake B. (2000) Multiple messengers in descending serotonin neurons: localization and functional implications. J. Chem. Neuroanat. 18, 75–86.
- Honore P., Rogers S. D., Schwei M. J., Salak-Johnson J. L., Luger N. M., Sabino M. C., Clohisy D. R. and Mantyh P. W. (2000) Murine models of inflammatory, neuropathic and cancer pain each generates a unique set of neurochemical changes in the spinal cord and sensory neurons. *Neuroscience* 98, 585–598.

© 2006 The Authors

Journal Compilation © 2006 International Society for Neurochemistry, J. Neurochem. (2006) 99, 924-936

- Ishizaki K., Sasaki M., Karasawa S., Obata H., Nara T. and Goto F. (1999) Intrathecal co-administration of NMDA antagonist and NK-1 antagonist reduces MAC of isoflurane in rats. *Acta Anaesthesiol. Scand.* 43, 753–759.
- Kanjhan R., Housley G. D., Burton L. D., Christie D. L., Kippenberger A., Thorne P. R., Luo L. and Ryan A. F. (1999) Distribution of the P2X2 receptor subunit of the ATP-gated ion channels in the rat central nervous system. J. Comp. Neurol. 407, 11–32.
- Kapoor V., Minson J. and Chalmers J. (1992) Ventral medulla stimulation increases blood pressure and spinal cord amino acid release. *Neuroreport* 3, 55–58.
- Kennedy C., Assis T. S., Currie A. J. and Rowan E. G. (2003) Crossing the pain barrier: P2 receptors as targets for novel analgesics. *J. Physiol.* 553, 683–694.
- Lentzen H. and Palenker J. (1983) Localization of the thiorphan-sensitive endopeptidase, termed enkephalinase A, on glial cells. *FEBS Lett.* 153, 93–97.
- Liao G. Y., Wagner D. A., Hsu M. H. and Leonard J. P. (2001) Evidence for direct protein kinase-C mediated modulation of N-methyl-Daspartate receptor current. Mol. Pharmacol. 59, 960–964.
- Liu G. J., Werry E. L. and Bennett M. R. (2005) Secretion of ATP from Schwann cells in response to uridine triphosphate. *Eur. J. Neurosci.* 21, 151–160.
- Marcel D., Pollard H., Verroust P., Schwartz J. C. and Beaudet A. (1990) Electron microscopic localization of immunoreactive enkephalinase (EC 3.4.24.11) in the neostriatum of the rat. J. Neurosci. 10, 2804–2817.
- Marriott D. R., Wilkin G. P. and Wood J. N. (1991) Substance P-induced release of prostaglandins from astrocytes: regional specialisation and correlation with phosphoinositol metabolism. J. Neurochem. 56, 259–265.
- Miles G. B., Parkis M. A., Lipski J. and Funk G. D. (2002) Modulation of phrenic motoneuron excitability by ATP: consequences for respiratory-related output *in vitro*. J. Appl. Physiol. 92, 1899–1910.
- Mjellem-Joly N., Lund A., Berge O. G. and Hole K. (1992) Intrathecal co-administration of substance P and NMDA augments nociceptive responses in the formalin test. *Pain* 51, 195–198.
- Monastyrskaya K., Hostettler A., Buergi S. and Draeger A. (2005) The NK1 receptor localizes to the plasma membrane microdomains, and its activation is dependent on lipid raft integrity. J. Biol. Chem. 280, 7135–7146.
- Nakajima Y., Tsuchida K., Negishi M., Ito S. and Nakanishi S. (1992) Direct linkage of three tachykinin receptors to stimulation of both phosphatidylinositol hydrolysis and cyclic AMP cascades in transfected Chinese hamster ovary cells. J. Biol. Chem. 267, 2437– 2442.
- Queiroz G., Gebicke-Haerter P. J., Schobert A., Starke K. and von Kugelgen I. (1997) Release of ATP from cultured rat astrocytes elicited by glutamate receptor activation. *Neuroscience* 78, 1203– 1208.
- Queiroz G., Meyer D. K., Meyer A., Starke K. and von Kugelgen I. (1999) A study of the mechanism of the release of ATP from rat cortical astroglial cells evoked by activation of glutamate receptors. *Neuroscience* 91, 1171–1181.
- Randic M., Hecimovic H. and Ryu P. D. (1990) Substance P modulates glutamate-induced currents in acutely isolated rat spinal dorsal horn neurones. *Neurosci. Lett.* 117, 74–80.
- Rayner S. E. and Van Helden D. F. (1997) Evidence that the substance Pinduced enhancement of pacemaking in lymphatics of the guineapig mesentery occurs through endothelial release of thromboxane A2. Br. J. Pharmacol. 121, 1589–1596.
- Regoli D., Boudon A. and Fauchere J. L. (1994) Receptors and antagonists for substance P and related peptides. *Pharmacol. Rev.* 46, 551–599.

- Rodriguez J., Carcache L. and Rein K. S. (2005) Low-mode docking search in iGluR homology models implicates three residues in the control of ligand selectivity. J. Mol. Recognit. 18, 183–189.
- Rougon G., Noble M. and Mudge A. W. (1983) Neuropeptides modulate the beta-adrenergic response of purified astrocytes *in vitro*. *Nature* 305, 715–717.
- Salter M. W. and Hicks J. L. (1994) ATP-evoked increases in intracellular calcium in neurons and glia from the dorsal spinal cord. *J. Neurosci.* 14, 1563–1575.
- Sasaki M., Tohda C. and Kuraishi Y. (1998) Region-specific increase in glutamate release from dorsal horn of rats with adjuvant inflammation. *Neuroreport* 9, 3219–3222.
- Silva G. A., Theriault E., Mills L. R., Pennefather P. S. and Feeney C. J. (1999) Group I and II metabotropic glutamate receptor expression in cultured rat spinal cord astrocytes. *Neurosci. Lett.* 263, 117–120.
- Smith J. P., Cunningham L. A. and Partridge L. D. (2000) Coupling of AMPA receptors with the Na⁺/Ca²⁺ exchanger in cultured rat astrocytes. *Brain Res.* 887, 98–109.
- Stout C. E., Costantin J. L., Naus C. C. and Charles A. C. (2002) Intercellular calcium signaling in astrocytes via ATP release through connexin hemichannels. J. Biol. Chem. 277, 10 482–10 488.
- Suadicani S. O., Brosnan C. F. and Scemes E. (2006) P2X7 receptors mediate ATP release and amplification of astrocytic intercellular Ca²⁺ signaling. J. Neurosci. 26, 1378–1385.
- Taylor A. L., Kudlow B. A., Marrs K. L., Gruenert D. C., Guggino W. B. and Schwiebert E. M. (1998) Bioluminescence detection of ATP release mechanisms in epithelia. *Am. J. Physiol.* 275, C1391– C1406.
- Torrens Y., Beaujouan J. C., Saffroy M., Glowinski J. and Tence M. (1998) Functional coupling of the NK1 tachykinin receptor to phospholipase D in chinese harnster ovary cells and astrocytoma cells. J. Neurochem. 70, 2091–2098.
- Tsuda M., Shigemoto-Mogami Y., Koizumi S., Mizokoshi A., Kohsaka S., Salter M. W. and Inoue K. (2003) P2X4 receptors induced in spinal microglia gate tactile allodynia after nerve injury. *Nature* 424, 778–783.
- Verderio C. and Matteoli M. (2001) ATP mediates calcium signaling between astrocytes and microglial cells: modulation by IFN-gamma. J. Immunol. 166, 6383–6391.
- Werry E. L., Liu G. J. and Bennett M. R. (2005) Glutamate-stimulated ATP release from spinal cord astrocytes is potentiated by substance P. J. Neurochem. 94, 101.
- Wienrich M. and Kettenmann H. (1989) Activation of substance P receptors leads to membrane potential responses in cultured astrocytes. *Glia* 2, 155–160.
- Wiesinger H., Schuricht B. and Hamprecht B. (1991) Replacement of glucose by sorbitol in growth medium causes selection of astroglial cells from heterogeneous primary cultures derived from newborn mouse brain. *Brain Res.* 550, 69–76.
- Xiang Z. and Burnstock G. (2005) Expression of P2X receptors on rat microglial cells during early development. Glia 52, 119–126.
- Zahn P. K. and Brennan T. J. (1998) Lack of effect of intrathecally administered N-methyl-D-aspartate receptor antagonists in a rat model for postoperative pain. Anesthesiology 88, 143–156.
- Zerari F., Karpitskiy V., Krause J., Descarries L. and Couture R. (1998) Astroglial distribution of neurokinin-2 receptor immunoreactivity in the rat spinal cord. *Neuroscience* 84, 1233–1246.
- Zhang J. M., Wang H. K. YeC. Q., Ge W., Chen Y., Jiang Z. L., Wu C. P., Poo M. M. and Duan S. (2003) ATP released by astrocytes mediates glutamatergic activity-dependent heterosynaptic suppression. *Neuron* 40, 971–982.
- Ziak D., Chvatal A. and Sykova E. (1998) Glutamate-, kainate- and NMDA-evoked membrane currents in identified glial cells in rat spinal cord slice. *Physiol. Res.* 47, 365–375.

© 2006 The Authors

Journal Compilation © 2006 International Society for Neurochemistry, J. Neurochem. (2006) 99, 924-936